#### **REVISED FINAL**

### PHASE II RCRA FACILITY INVESTIGATION REPORT FOR THE BURN PITS (SWMUs 4A - 4F) FORT STEWART, GEORGIA

REGULATORY AUTHORITY RESOURCE CONSERVATION AND RECOVERY ACT 40 CFR 264, TITLE II, SUBPART C, SECTION 3004; 42 USC 6901 ET SEQ.

Prepared for:

U.S. Army Corps of Engineers Savannah District Under Contract DACA21-95-D-0022 Delivery Order Number 0008

Prepared by:

Science Applications International Corporation 800 Oak Ridge Turnpike Oak Ridge, Tennessee 37831

February 1999

I, the undersigned, certify that I am a qualified groundwater scientist who has received a baccalaureate or post-graduate degree in the natural sciences or engineering, and have sufficient training and experience in groundwater hydrology and related fields, as demonstrated by state registration and completion of accredited university courses, that enable me to make sound professional judgments regarding groundwater monitoring and contaminant fate and transport. I further certify that this report was prepared by myself or by a subordinate working under my direction.

Bruce J. Haas. Project Manage SAIC

# SCIENCE APPLICATIONS INTERNATIONAL CORPORATION

contributed to the preparation of this document and should not be considered an eligible contractor for its review.

. /

# CONTENTS

4

EZ	<b>KECU</b>	TIVE SUMMARY	
		IVE SUMMARY ppendices bles	
Li	st of Ta	bles	vii
Lis	st of Fig	gures	vii
Lis	st of Al	gures breviations and Acronyms	ix
		breviations and Acronyms	х
1.0			
	1.1	OBJECTIVES AND SCOPE OF THE DEFENSE	1-1
	1.2		1-1
		REPORT ORGANIZATION	1-2
2.0	SE		
	2.1	TE HISTORY AND CONTAMINANTS	2-1
	2.2		2-1
	2.3		2-1
	2.4		2-1
	2.4	PRELIMINARY CONCEPTUAL SITE MODEL	2-6
3.0	SIT		
2.0	3.1	MMARY OF INVESTIGATION ACTIVITIES	3-1
	5.1		3-1
			3-1
			3-5
	3.2		3-10
	5.2	DATA QUALITY ASSESSMENT.	3-10
4.0	ענו	ZEICAL OTLAD & CTURES	5-10
7.0	4.1	SICAL CHARACTERISTICS OF THE SITE	4-1
	4.2		4-1
	4.2		4-1 4-1
	4.5 4.4		4-1
	4.4 4.5		4-1
			4-1
	4.6		4-2 4-5
	4.7		4-5
	4.8	METEOROLOGY	4-5 4-6
5.0	000		4-0
5.0	CON	TAMINANT NATURE AND EXTENT	5-1
	5,1		
			5-1 5-1
			5-3
	5.2		5-3
	5.3		5-3
	5.4		5-5
	5.5		5-8
<i></i>			5-10
6.0	CONT	AMINANT FATE AND TRANSPORT	
			6-1
	6.2	PHYSICAL AND CHEMICAL PROPERTIES	6-1
			6-1

C. J

C.

		6.2.1 Metals	6-1
		<ul><li>6.2.1 Metals</li><li>6.2.2 Organic Compounds</li></ul>	6-2
		6.2.2 Organic Compounds CONCEPTUAL SITE MODEL	6-4
	6.3	CONCEPTUAL SITE MODEL	6-6
		<ul><li>6.3.1 Water Balance Components</li><li>6.3.2 Contaminant Release Mechanisms and Migration Pathways</li></ul>	6-6
		6.3.2 Contaminant Release Mechanisms and Migration Fail way but	6-'
	6.4	6.3.2 Contaminant Release Mechanisms and Page 19	6-
		6.4.1 Soil Leachability Analysis	6-
		6.4.1 Soil Leachability Analysis	<b>6-</b> 1
	6.5	SUMMARY AND CONCLUSIONS	• -
<b>7</b> 0		AN HEALTH RISK ASSESSMENT	7.
7.0		DATA EVALUATION	7-
	7.1	7.1.1 Date Quality Evaluation	7
		7.1.2 Background Screening	7
		7.1.2 Background Screening	7
	7.2	7.2.1 Receptor Assessment	7
		7.2.1 Receptor Assessment	7
		7.2.2 Migration Pathway Allarysis	7
		7.2.2 Migration 1 autway 7 million attempts and a second s	7
	7.3	SELECTION OF SCREENING VALUES	7
		7.3.1 Screening Values for Soils	7
		7.3.2 Screening Values for Groundwater	-
	7.4		-
	7.5	RISK EVALUATION CONCLUSIONS OF THE HUMAN HEALTH RISK EVALUATION	,
		LOGICAL RISK ASSESSMENT	1
8.0			1
	8.1	8.1.1 Ecological Screening Value Comparison (Step i)	
		8.1.2 Preliminary Problem Formulation (Step ii)	
		8.1.2 Preliminary Effects Evaluation (Step iii)	
		8.1.3 Preliminary Exposure Estimate (Step iv)	
		8.1.4 Preliminary Exposure Estimate (step 1) internation 8.1.5 Preliminary Risk Calculation (Step v)	
		8.1.6 Uncertainties CONCLUSIONS AND RECOMMENDATIONS	
	8.2	CONCLUSIONS AND RECOMMENDATIONS	
9.0	СЧ	ARACTERIZATION OF BURN PIT A (SWMU 4A)	
9.0	9.1	TION OF RURN PULA	ç
	9.1	AND EXTENT OF CONTAMINATION AT BURN FIT A	-
	9,2	o o t G - Contemination	9
		a a a G therefore Soil Contamination	9
		a a g l atta Contomination	9
		CO'	9
	_ ~	THE AND THANGDORT CONSIDERATIONS AT DURING THE AMINING	
	9.3	THE ALL THE AL	
	9.4	9.4.1 Exposure Evaluation	
		9.4.2 Risk Evaluation ECOLOGICAL RISK EVALUATION OF BURN PIT A	
	9.5	ECOLOGICAL RISK EVALUATION OF BURNITH A	
		9.5.1 Ecological Screening Value Comparison (Step i)	
		a to p 1' where Problem Formulation (SICD II)	
		a so put in the Effects Evaluation (Step 11)	
		9.5.3 Preliminary Effects Evaluation (Step 12)	

-

		9.5.5 Preliminary Risk Calculation (Step v)	9-28
			~ ~ ~
	9.6	CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT A	9-29 9-29
10	o		
10.	0 CH	ARACTERIZATION OF BURN PIT B (SWMU 4B)	10-1
		THO TOKE AND DESCRIPTION OF BURN DIT D	
	10.3	2 INATONE AND EXTENT OF CONTAMINIATION AT DUDATION	
		10.2.1 Surface Soli Contamination	
		A View Durban lace Dull Configmination	<b>.</b>
		10.2.5 Croundwater Contamination	10.10
		- Containing of Stic-related L Ontaminanta	
	10.3	AT AND IKANSPORT CONSIDERATIONS AT DIDN DED D	10.00
	10.4	I HOWAN REALTH KISK EVALUATION OF BUDN DUT D	
		10.4.1 Exposure Evaluation	
	10.5	DOCEOUCAL RISK EVALUATION OF BURN PIT B	10.00
		10.5.1 Ecological Screening Value Comparison (Sterni)	
		10.5.2 I Iclininary Problem Formulation (Step ii)	
		Tournand V Effects Evaluation (Step in)	
		Toron Thomandly EXPOSITE Estimate (Step 117)	
		- $        -$	
		10.5.6 Uncertainties	10-28
	10.6	CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT B	
11.0	CHA	RACTERIZATION OF BURN PIT C (SWMU 4C)	
	11.1	HISTORY AND DESCRIPTION OF BURN PIT C	11-1
	11.2	NATURE AND EXTENT OF CONTAMINATION AT BURN PIT C	11-1
		11.2.1 Surface Soil Contamination	11-11
		11.2.2 Subsurface Soil Contamination	
		11.2.3 Groundwater Contamination	11-11
		11.2.4 Summary of Site-related Contaminants	11-19
	11.3	FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT C	11-23
	11.4	HUMAN HEALTH RISK EVALUATION OF BURN PIT C	11-23
		11.4.1 Exposure Evaluation	11-24
			11-24
	11.5	ECOLOGICAL RISK EVALUATION OF BURN PIT C.	11-25
		11.5.1 Ecological Screening Value Comparison (Step i)	11-27
		11.5.2 Preliminary Problem Formulation (Step i)	11-27
		11.5.3 Preliminary Effects Evaluation (Step ii)	11-28
		11.5.4 Preliminary Exposure Estimate (Step in)	11-28
		11.5.5 Preliminary Risk Calculation (Step v)	11-28
		11.5.6 Uncertainties	11-28
	11.6	CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT C	11-30
		ELECTIVITE INDATIONS FOR BURN PIT C	11-30
12.0	CHAR	ACTERIZATION OF BURN PIT D (SWMU 4D)	
	12.1	HISTORY AND DESCRIPTION OF BURN PIT D.	12-1
	12.2	INALURE AND BALLENT OF CONTAINSTATION AT TAXAST	12-1
		14441 OULIACE SOUL OUTSminston	12-11
		A STATE A DEBART RECE STULL OPPOINTS TO STATE	12-11
			12-11

in the second se

		to o a Constant Confermination	12-19
		and a contaminants	12-23
		TARE AND TO ANGDORT CONSIDERATIONS AT BURNETT D	12-23
	12.3	HUMAN HEALTH RISK EVALUATION OF BURN PIT D	12-24
	12.4	12.4.1 Exposure Evaluation	12-24
			12-25
		TOTAL DISK EVALUATION OF BURN PULD	12-27
	12.5	12.5.1 Ecological Screening Value Comparison (Step i)	12-27
		12.5.1 Ecological Screening Value Comparison (Step i) 12.5.2 Preliminary Problem Formulation (Step ii)	12-27
		12.5.2 Preliminary Effects Evaluation (Step iii)	12-28
		12.5.4 Preliminary Exposure Estimate (Step iv)	12-28
		12.5.5 Preliminary Risk Calculation (Step v)	12-28
			12-28
		12.5.6 Uncertainties CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT D	12-28
	12.6		
13.0	CHAR	ACTERIZATION OF BURN PIT E (SWMU 4E)	13-1 13-1
15.0	13.1	THE REPORT AND DESCRIPTION OF BURN PUT F.	13-1
	13.2	NATIOE AND EXTENT OF CONTAMINATION AT BURN PILE	
	12,2	12.0.1 Surface Soil Contamination	13-9
		12.2.2. Subsurface Soil Contamination	13-13
		12.2.2. Company devictor Contamination	13-13
		to a t d of Site related Contaminants	13-19
	13.3	TO AND TO ANODOD'T CONSIDERATIONS AT BUKIN FIT D	13-19
	13.4	TTIMAN HEAT TH RISK CHARACTERIZATION AT BURN FITE	13-21
	15.4	12.4.1 Exposure Evaluation	13-21
			13-22
	13.5	TOOLOGICAL DISK EVALUATION OF BURN PILE	13-22
	15.5	12.5.1 Easlogical Screening Value Comparison (Step 1)	13-22
		12.5.2 Proliminary Problem Formulation (Step 11)	13-24
		12.5.2 Bealiminary Effects Evaluation (Step 111)	13-25
		12.5.4 Dealiminary Exposure Estimate (Step IV)	13-25
		to c & Dustinging Rick Calculation (Step V)	13-25
			10 10
	13.6	13.5.6 Uncertainties CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT E	13-26
	i	RACTERIZATION OF BURN PIT E (SWMU 4F)	14-1
14.0		THE REPORT AND DECORDINATION OF BURN PUT IS A MANAGEMENT AND THE PROPERTY OF T	
	14.1	THE THE PARTY OF CONTAMINATION AT BUKIN 111 F.	7-1-1'Y'
	14.2	14.2.1 Surface Soil Contamination	14-11
		14.2.1 Subsurface Soil Contamination	14-14
		14.2.2 Subsurface Son Contamination	14-19
		the a the summer of Site related Contaminants	. 14-20
	14.3		. 1721
	14.4	HUMAN HEALTH RISK EVALUATION OF BOILTTH Fundamental 14.4.1 Exposure Evaluation	• T-4.770.1
			. 1-1200
		THE REPORT AND A DISTANCE TO A DISTANCE AND A DISTA	14-27
	14.5	5 ECOLOGICAL RISK EVALUATION OF DORUGTER 1 14.5.1 Ecological Screening Value Comparison (Step i)	. 14-27
		14.5.1 Ecological Screening Value Comparison (Step 1)	. 14-27
		14.5.2 Preliminary Problem Formulation (Step II) 14.5.3 Preliminary Effects Evaluation (Step III)	. 14-28
		14.5.3 Preliminary Effects Evaluation (Step II)	14-28
		14.5.4 Preliminary Exposure Estimate (Step W)	

Ì

	14.6	14.5.5 Preliminary Risk Calculation (Step v) 14.5.6 Uncertainties CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT F	14-28 14-29 14-29
15.0	CON	CLUSIONS	
	15.1	SUMMARY OF FINDINGS	15-1
	15.2	CONCLUSIONS	15-1
	15.3	SUPPLEMENTAL PHASE II GROUNDWATER	15-2
	15.4	CHARACTERIZATION	15-3
	19.4	RECOMMENDATIONS	15-4
16.0	REFE	RENCES	16-1

### List of Appendices

.

(\*\*\* )

Sector of the

APPENDIX A APPENDIX B APPENDIX C APPENDIX D APPENDIX E APPENDIX F APPENDIX G	SOIL BORING LOGS MONITORING WELL INSTALLATION DIAGRAMS QUALITY CONTROL SUMMARY REPORT GEOTECHNICAL LABORATORY TEST RESULTS BACKGROUND DATA ANALYTICAL LABORATORY DATA TOXICITY DATA FOR HUMAN HEALTH CHEMICALS/ CONTAMINANTS OF POTENTIAL CONCERN	A-1 B-1 C-1 D-1 E-1 F-1
--	--	--

## List of Tables

3-1	Phase II Monitoring Well Construction Summary, Burn Pits, Fort Stewart	
3-2	Well Development Summary, Burn Pits, Fort Stewart	3-6
3-3	Field Parameter Measurements During Groundwater Sampling Burn Pits	3-8
2 4	Fort Stewart	3-9
3-4	Containing (Soli), Duri Pils, Port Stewart	3-11
3-5	Fintered vs. OnIntered (µg/L) Groundwater Sample Comparison, Burn Pits,	
4-1	Summary of Geotechnical Analyses, Burn Pits, Fort Stewart.	3-12
5-1	Background Media Summary, Burn Dite, Fort Stewart.	4-4
5-2	Background Media Summary, Burn Pits, Fort Stewart	5-2
5-3	Summary of Analytical Results for Surface Soil Samples, Burn Pits, Fort Stewart Summary of Analytical Results for Subsurface Soil Samples, Burn Pits,	5-4
E 1	Fort Stewart	5-6
5-4	Summary of Analytical Results for Groundwater Samples Burn Pite Fort Stowert	5-8
5-5	Summary of Site-Related Constituents, Burn Pits, Fort Stewart	5-12
6-1	Distribution Coefficients (Kd) and Retardation Factors (Rd), Burn Pits, Fort Stewart	
6-2	Physical and Chemical Properties of Organic Site-Related Chemicals, Burn Pits, Fort Stewart	6-2
6-3	Arrival Time Estimates for Contaminant Migration in Groundwater, Burn Pits,	6-3
	Fort Stewart	6.0
6-4	CMCOPUs in Soil and Predicted Maximum Concentrations in Groundwater,	6-8
8-1	Burn Pits, Fort Stewart	6-9
0-1	Derivation of Toxicity Reference Values for Ecological COPCs in Soil	
	Burn Pits A through F, Fort Stewart	8-7

8-2	Exposure Parameters for Surrogate Species Exposed to COPCs in Soil or	8-8
	Groundwater Burn Pits A through F. Fort Stewart	0-0
8-3	Summary of the Ecological COPCs Identified in EPA Region 4 Ecological	
•	Screening Value Comparison for Groundwater at Burn Pits A through P,	0.0
	East Stowner	8-9
8-4	Summary of the Preliminary Risk Evaluation for Ecological COPCs in Surface	0.10
0.4	a 1 + Down Dite & through E Fort Stewart	8-10
9-1	Summary of Analytical Results for Surface Soils, Burn Pit A, Fort Stewart	9-11
9-2	Summary of Analytical Results for Subsurface Soils, Burn Pit A, Port Stewart	9-15
9-3	Summary of Analytical Results for Groundwater, Burn Pit A, Fort Stewart	9-19
9-4	Common of Site Related Constituents, Burn Pit A. Fort Stewart	9-23
9-5	CMCOPCe Paged on Soil Screening for Burn Pit A	9-24
9-6	Comparison to Action Levels-Burn Pit A. Fort Stewart	9-26
9-7	EDA Degion 4 Ecological Screening Value Comparison for Analytes Delected	
J-1	Commentaria et Purn Pit & Fort Stewart	9-27
9-8	Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 1001) at	
7-0	D D' A Dest Clowert	9-29
10-1	Summary of Analytical Results for Surface Soils, Burn Pit B, Fort Stewart	10-11
10-1	Summers of Analytical Results for Subsurface Soils, Burn Pit B, Fort Stewalt	10-15
10.2	Summary of Analytical Results for Groundwater, Burn Pit B, Port Stewart	10-19
10.4	Comments of Site Delated Constituents Burn Pit B. Fort Stewart	10-23
10.5	CMCOPCe Based on Soil Screening for Burn Pit B	10-24
10 6	Comparison to Action Levels—Burn Pit B. Fort Stewart	10-26
10-0	EPA Region 4 Ecological Screening Value Comparison for Analytes Detected	
	- Crowndwater at Burn Pit B Fort Stewart	10-27
10.0	Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 ft) at	
	D	10-29
44.1	Summary of Analytical Results for Surface Soils, Burn Pit C, Fort Stewart	11-11
11.0	Summer of Applytical Regults for Subsurface Soils, Burn Pit C, Fort Siewart.	11-15
11.0	Summer of Analytical Results for Groundwater, Burn Fit C, Fort Stewart	11-19
11.4	Comment of Site Delated Constituents Burn Pit C. Fort Stewart	11-23
<i>-</i>	GMCODCa Deced on Soil Screening for Burn Pit C	11-24
11-0	Comparison to Action Levels—Burn Pit C, Fort Stewart	11-26
11-0	EPA Region 4 Ecological Screening Value Comparison for Analytes Detected	
11-/	in Groundwater at Burn Pit C, Fort Stewart	11 <b>-2</b> 7
	B Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 ft) at	
11-6		11-29
10.1	a contract of the I presented for Surface Soils Burn Pit D FOT SIEWall	12-11
12-1	A polytical Results for Subsurface Soils, Burn Pit D, Fort Stewart	12-15
12-2	Summary of Analytical Results for Groundwater, Burn Pit D, Fort Stewart	12-19
12	<ul> <li>Summary of Site-Related Constituents, Burn Pit D, Fort Stewart</li></ul>	12-23
12-4	5 CMCOPCs Based on Soil Screening for Burn Pit D.	12-24
12-	6 Comparison to Action Levels—Burn Pit D, Fort Stewart	12-26
12-0	7 EPA Region 4 Ecological Screening Value Comparison for Analytes Detected	
12-	Content of Directory Dit D. Fort Stewart	12-27
	CA 1 41-1 Decella for Surface Soils Burn Pit H. FOIL SIGWall	13-9
13-	<ol> <li>Summary of Analytical Results for Subsurface Soils, Burn Pit E, Fort Stewart</li> <li>Summary of Analytical Results for Subsurface Soils, Burn Pit E, Fort Stewart</li> </ol>	13-14
		13-19
13-	<ul> <li>3 Summary of Analytical Results for Groundwatch, Burn Fit E, Fort Stewart.</li> <li>4 Summary of Site-Related Constituents, Burn Pit E, Fort Stewart.</li> </ul>	13-20
13-	<ul> <li>Summary of Site-Related Constituents, Burn Pit E, Fort Stewart, Georgia</li> <li>CMCOPCs Based on Soil Screening for Burn Pit E, Fort Stewart, Georgia</li> </ul>	13-20
13-	<ul> <li>6 Comparison to Action Levels—Burn Pit E, Fort Stewart</li></ul>	13-23
13-	6 Comparison to Action Levels — Built Fit B, Fort Stewart	

	EPA Region 4 Ecological Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit E, Fort Stewart Preliminary Risk Evaluation of Firstley in 1, 2000, in Firstley 1, 2000, in	12.04
13-8	Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 ft) at Burn Pit E, Fort Stewart	13-24
14-1	Summary of Analytical Results for Surface Soils, Burn Pit F, Fort Stewart.	13-26
14-2	Summary of Analytical Results for Subsurface Soils, Burn Pit F, Fort Stewart	14-11
14-3	Summary of Analytical Results for Groundwater, Burn Pit F, Fort Stewart	14-15
14-4	Summary of Site-Related Constituents Due Did D D	14-19
14-5	Summary of Site-Related Constituents, Burn Pit F, Fort Stewart	14-23
14-6	CMCOPCs Based on Soil Screening for Burn Pit F.	14-24
14-7	Comparison to Action Levels—Burn Pit E, Fort Stewart	14-26
	In Orounuwater at Burn Pit F. Fort Stewart	14-28
14-0		
	Burn Pit F, Fort Stewart	14-29
	*****************	1-1-47

10

### List of Figures

2-1	Brownin Douglou May IVI FULL SIEWATT Militany Docomption Commission	
2-2		
2-3	Lood of the Dull Fils (Swive is 4A through AE)	
3-1		_
4-1	Construction of the fort Starsant A.	3-3
6-1	Generalized Conceptual Site Model for Contaminant Fate and Transport at	4-3
	During in Sing unduring report Ntewart	
8-1	Concruit 100003 101 ASSESSING KISK and Selecting Domodical Lange	6-5
	Loord Receptors (adapted from (FPP) 1006)	
9-1	- $        -$	8-3
9-2	Geologic Cross-Section A-A' - Burn Pit A	9-3
9-3	Geologic Cross-Section B-B' - Burn Pit A	
9-4	(40) $(40)$	<b>9-</b> 7
9-5	Accounts of Analyses in Surface Soils - Burn Dit A	9-9
9-6	Results of Analyses in Subsurface Soils - Burn Pit A	9-13
9-7	- Colorido Or / Giaryses III Cholinnwater - Burn Det A	9-17
10-1		9-21
		10-3
		10-5
	The and the contour with rangest rug / - the bit b	10-7
	ACOULD OF THIRDSCS III OUTIACE Solls - Rum Dit R	10-9
	Tresting of Fuldryses HI Subsurface Solic - Hum Dit D	10-13
/	ACOULD OF ANALYSES IN CHAIMAWater - Durn Dit D	10-17
	Evenuous of Damphille Signons - Rum Pit C	10-21
		11-3
	Concerc Cross-Scotion D-B - Billion bit (	11-5
		11-7
A 1 V	incomic of Analyses in Surface Soils - Rum Dit C	11-9
V	ACCOUNT OF ANALYSES IN SUBSULTACE Notic - Rum Dit C	11-13
/	Acould of Analyses in Uroundwater - Rurn Pit C	11-17
		11-21
~~ ~	Could be Cruss-Dection - Birm Pit 1	12-3
12-3	Geologic Cross-Section B-B' - Burn Pit D	12-5
12-4	Water Table Contour Map (August 1997) - Burn Pit D.	12-7
		12-9

		12-13
12-5	Results of Analyses in Surface Soils - Burn Pit D.	12-17
10 1	Devilte of Applyces in Subsurface Soils - Burn Pit D	
12-0	Results of Analyses in Groundwater - Burn Pit D	12-21
12-7	Results of Analyses in Groundwater - Date F	13-3
13-1	Locations of Sampling Stations - Burn Pit E	13-5
10.0	Carlania Cross Sections A-A' and B-B' - Burn Pit E	13-7
10.0	Water Table Contour Man (August 1997) - Burn Pit E	
13-3	Results of Analyses in Surface Soils - Burn Pit E	13-11
13-4	Results of Analyses in Surface Sorias - Jun Pure Dit E	13-15
13-5	Results of Analyses in Subsurface Soils - Burn Pit E	13-17
12 6	Populto of Analyses in Groundwater - Burn Pit E	14-3
1 4 1	Leasting of Sompling Stations - Burn Pit F	
14-1	Geologic Cross-Section A-A' - Burn Pit F	14-5
14-2	Geologic Cross-Section A-A - Duning and the	14-7
14-3	Geologic Cross-Section A-R - Durn Pit F Geologic Cross Section B-B' - Burn Pit F	14-9
111	MI-ton Table Contour Man (August 1997) - Burn Pit F	14-13
A 14	D the of Anolyces in Surface Soils - Burn Pit F	
14-5	Results of Analyses in Subsurface Soils - Burn Pit F	14-17
14-6	Kesuits of Analyses in Subsurface Sons Dam Dit E	14-21
14-7	Results of Analyses in Groundwater - Burn Pit F	

## List of Abbreviations and Acronyms

amsl above mean sea level	
ASTM American Society for Testing and Materials	
AUF area use factor	
BAF bioaccumulation factor	
below ground surface	
BTEX benzene, toluene, ethylbenzene, and xylene	
CAP Corrective Action Plan	19
CERCLA Comprehensive Environmental Response, Co	ompensation, and
Liability Act	
CMCOPC contaminant migration constituent of potentia	al concern
coc chemical/contaminant of concern	
COPC chemical/contaminant of potential concern	
CSM conceptual site model	
DAF dilution attenuation factor	
DPW Directorate of Public Works	
DOCSR Daily Quality Control Summary Report	
DOO Data Quality Objective	
EPA U.S. Environmental Protection Agency	
ERA Ecological Risk Assessment	
ESV ecological screening value	
FSMP Fort Stewart Military Reservation	
GEPD Georgia Environmental Protection Division	
GSSI generic soil screening level	
HHRA human health risk assessment	
HI hazard index	
HO hazard quotient	
IDW investigation-derived wastes	
I OAFL lowest observed adverse effect level	
MCI maximum contaminant level	
NOAEL no observed adverse effect level	
PAH polyaromatic hydrocarbon	
PCE tetrachloroethene	

12

THIS PAGE INTENTIONALLY LEFT BLANK

)

### **EXECUTIVE SUMMARY**

This report summarizes the results of the Phase II Resource Conservation and Recovery Act (RCRA) Facility Investigation (RFI) for the Burn Pits, Solid Waste Management Units (SWMUs) 4A through 4F, at Fort Stewart, Georgia. This report has been prepared by Science Applications International Corporation (SAIC) for the U.S. Army Corps of Engineers (USACE), Savannah District, under Contract DACA21-95-D-0022, Delivery Order No. 0008. The RFI was conducted in accordance with USACE guidance EM200-1-3 and the approved Phase II RFI Work Plan.

SWMU 4 consists of a total of seven separate burn pits located in the southern portion of the Fort Stewart Military Reservation. One of these burn pits, Burn Pit G, has not been used and is, therefore, not included in this Phase II RFI. The remaining six burn pits, Burn Pits A through F, have been used in the past to burn combustible waste such as scrap lumber, timber cuttings, or construction and demolition waste, and for the disposal of ashes, concrete debris, and soil from excavations. Burning operations were not actually done in trenches, or "pits," but in surface piles that were doused with gasoline and burned. The units are, therefore, similar to "burn areas."

Potential contamination at each of the six sites was investigated during a Phase I RFI for 24 SWMUs at Fort Stewart in 1993. Analytical results from soil and groundwater sampling conducted at the former Burn Pits indicated metal and/or volatile organic contamination in both soil and groundwater at each site. Based on these findings, Georgia Environmental Protection Division (GEPD) instructed the Fort Stewart Directorate of Public Works (DPW) to conduct a Phase II RFI.

The objectives of the Phase II RFI for the Burn Pits, as defined in the Work Plan approved by GEPD on June 10, 1997 are to:

- determine the horizontal and vertical extent of contamination;
- determine whether contaminants present a threat to human health or the environment;
- determine the need for future action and/or no further action; and
- gather necessary data to support a Corrective Action Plan, if warranted.

### SUMMARY OF INVESTIGATION ACTIVITIES

The information provided in this report is based upon data collected previously during the Phase I RFI and data collected as part of the Phase II field sampling and analysis. The scope of the Phase II field work included the following activities:

- Redevelopment of 18 existing wells and abandonment of 6 existing wells found to be damaged or unusable.
- Drilling of 3 soil borings at each burn pit site (for a total of 18 borings) with collection of
  associated surface and subsurface soil samples. Surface soil samples were analyzed for
  semivolatile organic compounds (SVOCs) and RCRA metals; subsurface soil samples were
  analyzed for volatile organic compounds (VOCs) and RCRA metals.
- Installation of eight permanent groundwater monitoring wells both upgradient and downgradient of selected sites.

 Collection of groundwater samples from both new and existing wells for a total of 26 samples. Groundwater samples were analyzed for VOCs and RCRA metals

## PHYSICAL CHARACTERISTICS OF THE BURN PIT SITES

The former Burn Pits range between one and four acres in size. The topography at each of the sites is relatively flat, with local relief varying by only 5 to 10 feet. There are no surface water bodies in the vicinity of any of the burn pits. Runoff is collected in ditches that discharge to different receiving streams. Burn Pits A through D are within the Mill Creek and Horse Creek subwatershed areas, Burn Pit E is located within the Canoochee River floodplain, and Burn Pit F is located in the Peacock Creek watershed, with drainage from the site into Goshen Swamp.

The surficial soils at each of the sites are generally a light gray sand or silty sand up to 20 feet thick, the maximum depth explored. Clayey sand and sandy clay layers are interbedded with this sandy material, and wood fragments and charred wood were also encountered in some of the borings. Results of geotechnical analyses indicated that the soils tested are generally non-plastic sands with silt, with only 2 to 12 percent by weight fine-grained particles. Laboratory permeability of the soils tested varied from  $3.11 \times 10^4$  to  $1.73 \times 10^{-3}$  cm/second.

The uppermost hydrologic unit is the surficial aquifer, which ranges from 55 to 150 feet in thickness. Water levels measured during well development and sampling varied from the shallowest (2 to 3 feet) at Burn Pit C, to the deepest (12 to 13 feet) at Burn Pit E. Groundwater flow is generally planar toward the nearest surface stream into which the groundwater discharges. An exception to this is at Burn Pits C and F, where water levels suggest that flow may be radially away from the source area. Horizontal gradients are generally very flat, ranging from 0.001 to 0.009 foot/foot. Calculated groundwater flow velocities vary from a low of 1.2 feet/year at Burn Pit B to a high of 35 feet/year at Burn Pit A. Groundwater discharges to the receptor stream nearest to each of the sites, located at distances that vary from 1000 to 4500 feet from the site.

### CONTAMINANT NATURE AND EXTENT

- SVOCs were found only in a single surface soil sample at low concentrations at Burn Pit E.
- Metals found above background in surface soils include arsenic, barium, cadmium, lead, and mercury in the northern and eastern portions of Burn Pit A; lead and mercury at Burn Pit B; barium, cadmium, lead and mercury in the center of Burn Pit C; barium and mercury at Burn Pit E; and arsenic, chromium, and mercury in the northern and northeastern portion of Burn Pit F.
- Benzene, toluene, ethylbenzene, and xylene (BTEX) compounds were found in subsurface soil at low concentrations sporadically at all six Burn Pit sites. A consistent pattern of distribution is evident only at Burn Pit B and Burn Pit C.
- Metals in subsurface soil were generally found sporadically at all six Burn Pit sites twice background and in no consistent pattern of distribution. Notable presence of metals found in subsurface soils includes lead and mercury in the central and northeastern portion of Burn Pit C, and five metals (predominantly barium, lead, and chromium) throughout the Burn Pit F site.

- In groundwater, the highest levels of BTEX contamination were found in a single well located west (side-gradient) of the Burn Pit F site. Concentrations of BTEX were less than their respective maximum contaminant levels (MCLs) and less than reported Phase I concentrations, suggesting that natural attenuation may be occurring. The source of the BTEX contamination is unknown; BTEX compounds were not found at high concentrations in soils at Burn Pit F.
- Metals in groundwater were found at all six sites at concentrations less than their respective MCLs and much less than reported Phase I concentrations, suggesting that redevelopment of the wells and use of low-flow sampling equipment to reduce turbidity in the samples were successful.
- Metals found in groundwater include barium, mercury, and silver at Burn Pit A; barium, cadmium, chromium, selenium, and silver at Burn Pit B; barium, cadmium, and mercury at Burn Pit C; arsenic, barium, and silver at Burn Pit D; and cadmium at Burn Pits E and F.

### CONTAMINANT FATE AND TRANSPORT

Contaminant fate and transport analysis provided an assessment of the potential migration pathways and transport mechanisms affecting the chemicals at the sites. In particular, the leachability of constituents from soil to groundwater and their natural attenuation in groundwater was evaluated.

Among the metal constituents, only chromium from Burn Pit F is considered a contaminant of potential concern (COPC) for contaminant migration based on contaminant transport analysis for leaching from soil to groundwater. At Burn Pit F, chromium could reach the water table at a predicted concentration (173  $\mu$ g/L) exceeding its MCL of 100  $\mu$ g/L. However, offsite migration of chromium would be very limited due to its high K<sub>d</sub> (i.e., high retardation factor), as well as the slow movement of groundwater toward Goshen Swamp, located approximately 1000 feet west of the site.

Benzene, methylene chloride, and tetrachloroethene could migrate from soils to the water table at one or more of the sites at concentrations exceeding their respective MCLs. However, offsite migration of these constituents would be very limited due to retardation and biodegradation as well as the slow movement of groundwater. Predicted maximum concentrations of these constituents at a distance of 300 feet from the source are not expected to exceed their MCLs even with the most conservative assumption for their biodegradation half-lives. None of the organic compounds that are currently observed in groundwater exceed their respective MCLs in groundwater.

### HUMAN HEALTH RISK ASSESSMENT

The human health risk assessment included a Step 1 risk evaluation to determine potential human health risks associated with the contaminants.

In surface soil, arsenic is a potential human health COPC at Burn Pits A and F, where arsenic exceeds its risk-based screening value for exposure of a residential receptor. It should be noted that the maximum concentration of arsenic at these sites (2.7 mg/kg) is below the average concentration of arsenic for soils regionally (7.4 mg/kg). In addition, a residential screening value was used, since there are no screening values for industrial land use, even though

residential land use at these sites is unlikely. Therefore, arsenic in surface soil is not considered a potential threat to human health at the Burn Pits.

In subsurface soil, no constituent exceeded its respective risk-based screening value for exposure to a residential receptor.

In groundwater, arsenic is a potential human health COPC only at Burn Pit D, where the maximum concentration of arsenic in groundwater exceeded its risk-based screening value for residential use of groundwater as drinking water. However, the maximum concentration of arsenic ( $4.2 \ \mu g/L$ ) was well below its MCL of 50  $\mu g/L$ , and was only slightly above its reference background concentration of  $3.4 \ \mu g/L$ . Arsenic exceeded background in only a single well at Burn Pit D (MW-6). In addition, use of the surficial groundwater at this site for drinking water is unlikely. Therefore, arsenic in groundwater is not considered a potential threat to human health at the Burn Pits.

## ECOLOGICAL RISK ASSESSMENT

The ecological risk assessment included a Phase 1 preliminary risk evaluation (PRE) for potential terrestrial and aquatic receptors at the sites.

Cadmium, chromium, and lead are present in surface soil at one or more of the former Burn Pits at concentrations that exceed the toxicity reference values (TRVs) for the robin, and in some cases, the shrew. Cadmium, chromium. and lead in surface soil at these two sites are, therefore, identified as potential ecological COPCs in the PRE. Based on the results of additional evaluations using published dietary fractions and LOAEL-based TRVs, and the uncertainties in the preliminary risk calculations, cadmium, chromium, and lead in surface soil at the burn pits are unlikely to pose a risk to ecological receptors.

According to GEPD (1996) and U.S. Environmental Protection Agency (EPA) Region 4 (1995) guidance, groundwater was screened as surface water in the ecological PRE. Barium, cadmium, lead, mercury, silver, and ethylbenzene are present in groundwater at one or more of the former burn pits at concentrations that exceed EPA Region 4 ecological screening values for surface water.

Groundwater is not evaluated further in the PRE for the Burn Pits because ecological receptors are not likely to be exposed to groundwater at these sites. There are no surface water bodies at or near the Burn Pits. Also, groundwater contaminants are not expected to migrate from the Burn Pit sites because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in soils at the sites.

# SUPPLEMENTAL GROUNDWATER CHARACTERIZATION

Based upon the results of the original Phase II RFI at the Burn Pit F site, a supplemental groundwater characterization was conducted in November 1998. The purpose of this supplemental characterization is to verify (1) the groundwater contours in the vicinity of Burn Pit F and the ultimate rate and point of discharge of groundwater, (2) the presence or absence of a source of organic contamination, and (3) the natural attenuation of the BTEX compounds. The scope of work included installing eight temporary piezometers within the center of the site, in the northwestern boundary of the site, and to the north and west of the site from MW-4. Grab groundwater samples were taken from these piezometers and were analyzed for BTEX. The four

existing on-site monitoring wells (MW-1 through MW-4) were also sampled and analyzed for VOCs and water quality parameters (methane, ethane, and ethene).

18

VOCs. Two BTEX (ethylbenzene at 289  $\mu$ g/L and total xylenes at 1420  $\mu$ g/L) compounds were detected in groundwater at MW-4. The concentrations are less than the respective MCLs for ethylbenzene (700  $\mu$ g/L) and total xylenes (10,000  $\mu$ g/L). In Phase I sample results, both ethylbenzene and xylenes were reported at much higher concentrations (2,800 and 15,800  $\mu$ g/L, respectively). In the Phase II RFI, both ethylbenzene and total xylenes were also reported at higher concentrations of 637 and 3170  $\mu$ g/L, respectively. These results indicate that the concentrations of BTEX present in groundwater at the site are attenuating naturally, either through dispersion, volatilization, or biodegradation.

Low concentrations of total xylenes (3.8  $\mu$ g/L) were also found in MW-1. Total xylenes were found at a concentration only slightly above its detection limit and significantly less than its MCL of 10,000  $\mu$ g/L. No VOCs were reported above the detection limit in MW-1 during Phase II groundwater sampling.

Chloroform and styrene (2.8 and 12.2  $\mu$ g/L, respectively) were observed in MW-4. Styrene is considered a secondary contaminate in the primary BTEX plume. Chloroform is a common laboratory contaminant, was not detected in the Phase II RFI sampling event, and is substantially less than its MCL of 100  $\mu$ g/L.

The following conclusions and recommendations have been made based on the results of the supplemental groundwater investigation:

- 1. The exact source of the BTEX contamination at Burn Pit F is unknown. MW-4 is located hydraulically higher than the other wells at the Burn Pit F site and is considered side-gradient to the existing groundwater flow. BTEX compounds were not detected at significant concentrations in the soil borings within the potable source area, were not detected in the remaining downgradient wells, and were not detected in the eight additional piezometers installed around MW-4. Therefore, the source and extent of contamination is expected to be localized and confined to the immediate vicinity of MW-4.
- 2. BTEX compounds (ethylbenzene and total xylenes) continue to be present in groundwater at MW-4. However, the concentrations are less than their respective MCLs and less than the concentrations reported during the Phase I and II RFIs, indicating they are attenuating naturally, either through dispersion, volatilization, or biodegradation.
- 3. No further investigation or action is recommended for Burn Pit F. The existing wells should be retained in the event that future site monitoring is required to confirm that concentrations of ethylbenzene and total xylenes continue to decrease.

#### RECOMMENDATIONS

The results of the RFI and supplemental groundwater sampling, conclusions regarding nature and extent of contamination, fate and transport, human health risk, and ecological risk result in the following recommendations:

- 1. No further action is recommended for all six Burn Pits (i.e., SWMUs 4A, 4B, 4C, 4D, 4E, and 4F). If approved by GEPD, Fort Stewart respectfully requests that the installation's Subpart B permit be amended to reflect this change in status.
- Due to the fact that all human health COPCs have been eliminated in site-specific evaluations, none of the Burn Pits will require a baseline risk assessment.
- 3. Due to the fact that all ecological COPCs have been eliminated in site-specific evaluations, none of the Burn Pits will require an ecological risk assessment.
- 4. If GEPD approves Fort Stewart's recommendation for No Further Action at SWMUs 4A, 4B, 4C, 4D, 4E, and 4F, the existing site monitoring wells will be properly abandoned, with the exception of the wells located at SWMU 4F. Fort Stewart proposes to retain the wells at SWMU 4F in the event that future site monitoring is required to confirm that concentrations of ethylbenzene and total xylenes continue to decrease.

### **1.0 INTRODUCTION**

This report summarizes the results of the Phase II Resource Conservation and Recovery Act (RCRA) Facility Investigation (RFI) for the Burn Pits, Solid Waste Management Units (SWMUs) 4A through 4F, at Fort Stewart, Georgia. This report has been prepared by Science Applications International Corporation (SAIC) for the U.S. Army Corps of Engineers (USACE), Savannah District, under Contract DACA21-95-D-0022, Delivery Order No. 0008. This RFI has been conducted in accordance with USACE standards EM 200-1-3 and the approved Phase II RFI Work Plan

SWMU 4 consists of a total of seven separate burn pits located in the southern portion of the Fort Stewart Military Reservation (FSMR). One of these burn pits, Burn Pit G, has not been used and is, therefore, not included in this Phase II RFI. The remaining six burn pits, Burn Pits A through F, have been used in the past to burn combustible waste such as scrap lumber, timber cuttings, or construction and demolition waste, and for the disposal of ashes, concrete debris, and soil from excavations. Burning operations were not actually done in trenches or "pits," but in surface piles that were doused with gasoline and burned. The units are, therefore, similar to

Potential contamination at each of the six sites was investigated during a Phase I RFI for 24 SWMUs at Fort Stewart (Rust 1993). Analytical results from soil and groundwater sampling conducted at the former Burn Pits indicated metal and/or volatile organic contamination in both soil and groundwater at each site. Based on these findings, Georgia Environmental Protection Division (GEPD) instructed the Fort Stewart Directorate of Public Works (DPW) to conduct a

# 1.1 OBJECTIVES AND SCOPE OF THE INVESTIGATION

The specific objectives of the Phase II RFI for the former Burn Pits at Fort Stewart, Georgia, as defined in the Phase II RFI Work Plan (SAIC 1997) (approved by GEPD on June 10, 1997), are

- determine the horizontal and vertical extent of contamination; •
- determine whether contaminants present a threat to human health or the environment; . •
- determine the need for future action and/or no further action; and .
- gather necessary data to support a Corrective Action Plan (CAP), if warranted.

The information provided in this report is based upon data collected previously during the Phase I RFI and data collected as part of the Phase II field sampling and analysis. The Phase II field work was conducted in accordance with the Phase II Work Plan (SAIC 1997), with the exception that the existing wells found to be damaged or unusable were abandoned. The scope of the Phase II field work included the following activities:

- Redevelopment of 18 existing wells.
- Abandonment of six existing wells found to be damaged or unusable.
- Drilling of 3 soil borings at each burn pit site (for a total of 18 borings) with collection of ۰ associated surface and subsurface soil samples.

21

- Installation of eight permanent groundwater monitoring wells, both upgradient and downgradient of selected sites.
- Collection of groundwater samples from both new and existing wells for a total of 26 samples.
- Surveying the position of all sample locations.

### 1.2 REPORT ORGANIZATION

This Phase II RFI Report consists of 16 sections. Section 1.0 describes the purpose of this investigation and summarizes the scope of work performed. Section 2.0 discusses the site history and conceptual site model (CSM) for the former Burn Pits. Section 3.0 summarizes the investigation activities and methodologies used in completing the Phase II RFI field work. Section 4.0 presents the regional setting of the FSMR, including the demographics, topography, regional geology and hydrogeology, surface drainage, soils, and ecology. Section 5.0 describes the results of the investigation, including background data analysis and summary of the nature and extent of contamination. Section 6.0 identifies considerations affecting contaminant fate and transport. Section 7.0 presents the human health risk assessment (HHRA) methodology and summary of the risk characterization. Section 8.0 presents the ecological risk assessment (ERA) methodology and summary of the risk characterization.

Sections 9.0 through 14.0 describe the results of the investigation for each of the six Burn Pit sites, respectively. These sections present the site-specific history, nature and extent of contamination, fate and transport, HHRA, and ERA. Section 15.0 summarizes the conclusions and recommendations for subsequent corrective action. The references cited in this report are listed in Section 16.0.

This revised final report also contains eight appendices. Appendices A through D contain the same information as presented in the final report (March 1998), including boring logs, monitoring well construction diagrams, Quality Control Summary Report (QCSR), and geotechnical laboratory test results. Appendix E, which contains the background data, has been substantially modified to include additional information collected in conjunction with ongoing RFIs at other SWMUs on the FSMR. Appendix F has been modified to include an explanation of acronyms and validation flags. Appendix G has been added to include a summary of the toxicity data for Human Health Chemicals of potential concern. Appendix H is a new appendix that presents the results of the November 1998 supplemental groundwater sampling.

## 2.0 SITE HISTORY AND CONTAMINANTS

#### 2.1 INSTALLATION DESCRIPTION

Fort Stewart (then known as Camp Stewart) was established in June 1940 as an anti-aircraft artillery training center. Between January and September 1945, the installation operated as a prisoner-of-war camp. The installation was deactivated in September 1945. In August 1950, Fort Stewart was reactivated to train anti-aircraft artillery units for the Korean Conflict. The training mission was expanded to include armor training in 1953. Fort Stewart was designated a permanent Army installation in 1956, and became a flight training center in 1966. Aviation training at the Fort Stewart facilities was phased out in 1973. In January 1974, the 1st Battalion, 75th Infantry was activated at Fort Stewart. Fort Stewart then became a training and maneuver area, providing tank, field artillery, helicopter gunnery, and small arms training for regular Army and National Guard units. The 24th Infantry Division, which was reflagged as the 3rd Infantry Division in May 1996, was permanently stationed at Fort Stewart in 1975. These activities comprise the installation's primary mission today.

The FSMR is located in portions of Liberty, Bryan, Long, Tattnall, and Evans Counties, Georgia, approximately 40 miles west-southwest of Savannah, Georgia (Figures 2-1 and 2-2). The cantonment, or garrison area, of the FSMR is located within the Liberty County portion of the FSMR on the southern boundary of the reservation. Hinesville, Georgia, is the nearest city to the garrison area and is located immediately outside of the reservation boundary.

### 2.2 SITE LOCATION AND HISTORY

A total of seven separate burn pits are located around the garrison (or cantonment) area on the FSMR (Figure 2-3), within 10 miles of the garrison area. One of these burn pits, Burn Pit G, was cleared but has not been used (Geraghty and Miller 1992) and is, therefore, not included in this Phase II RFI. The remaining six burn pits, Burn Pits A through F, have been used in the past and are included in this Phase II RFI.

The six burn pits may have been in existence since Fort Stewart operations began, and have been used at various times to burn combustible solid waste such as scrap lumber, timber cuttings, construction and demolition waste, and to dispose of ashes, concrete debris, and soil from excavations. The burn pits were not actually "pits," but surface piles of clearing and grubbing debris that may have been doused with gasoline and burned. The burn pits operations were, therefore, not in trenches, but in surface "burn areas."

Potentially contaminated materials used or generated at the former Burn Pits included flammable liquids, such as gasoline. However, the use of fuels or solvents as ignition sources at the Burn Pits has not been specifically indicated (Geraghty and Miller 1992).

#### 2.3 PREVIOUS INVESTIGATIONS

An RCRA Facility Assessment (RFA) was submitted to the GEPD in June 1990 that listed 24 SWMUs, including the former Burn Pits, as requiring further investigation (Geraghty and Miller 1992). A Phase I RFI, completed in April 1996, was performed in response to that



Figure 2-1. Regional Location Map for Fort Stewart Military Reservation, Georgia

submittal (Rust 1996). The objective of the Phase I RFI was to determine if releases to the environment had occurred from any of the 24 identified SWMUs.

During the Phase I RFI, site characterization activities involved the installation of four monitoring wells at each of the six burn pits, and the sampling and analysis of both soil and groundwater samples collected from these wells. Soil samples were collected during June 1993 (Geraghty and Miller 1993), and groundwater sampling took place during July, August, and November 1993 (Rust 1996). Results of that sampling are presented for each Burn Pit site in their respective site-specific characterizations (Sections 9.0 through 14.0). General trends in those results are described below.

One soil sample was collected by Geraghty and Miller (1993) during drilling of each monitoring well borehole and analyzed for volatile organic compounds (VOCs), RCRA total metals, pH, and specific conductance. Analytical results for the soil samples taken from the monitoring well boreholes were compared to then current GEPD guidelines (1993) or to site-specific background concentrations with the following results:

- VOCs. Benzene, toluene, ethylbenzene, and xylene (BTEX) concentrations were reported in soil samples at levels less than 0.03 mg/kg at each of the sites. The compound tetrachlorethene (PCE) was detected in soil samples from Burn Pits A through D, with the highest value reported at Burn Pit D at 9.7 mg/kg. Methylene chloride, acetone, and carbon disulfide were detected in soil samples at levels above background; however, these compounds are common laboratory artifacts.
- Metals. Five RCRA metals were detected in soil samples from each of the Burn Pit sites at levels exceeding site-specific background concentrations. These metals and their corresponding maximum reported levels included arsenic (6.2 mg/kg), barium (27.0 mg/kg), chromium (15.0 mg/kg), lead (12.0 mg/kg), and mercury (0.063 mg/kg).
- Specific Conductance. Specific conductance in the soil samples ranged from 2.1 to 17.0.
- pH. Values of pH in soil ranged from 3.7 to 6.5, indicating slightly acidic, yet natural, conditions.

Groundwater samples were collected by USACE (1996) and analyzed for VOCs and RCRA total metals; pH and specific conductance were measured in the field. Analytical results for the groundwater samples were compared to Maximum Contaminant Levels (MCLs) and site-specific background concentrations with the following results:

- VOCs. Detection of BTEX reported in groundwater samples at two of the sites: at Burn Pit A, individual BTEX concentrations were less than 1.19 mg/L; at Burn Pit F, BTEX concentrations of 41.1 mg/L for individual constituents were reported. No chlorinated organic compound was reported in groundwater that can be associated with site contamination. Methylene chloride, acetone, and carbon disulfide were detected in groundwater; however, these compounds are common laboratory artifacts.
- Metals. Six RCRA metals were detected in groundwater samples from each of the Burn Pit sites at concentrations exceeding their respective MCLs and site-specific background concentrations. These metals and their corresponding maximum reported concentrations (total metal analysis) included arsenic (0.091 mg/L), chromium (1.4 mg/L), lead (0.625 mg/L), mercury (0.0062 mg/L), selenium (0.110 mg/L), and silver (0.2 mg/L).

- Specific Conductance. Specific conductance in the groundwater samples ranged from 0.02 to 0.5 mhos.
- **pH**. Values of pH in groundwater ranged from 4.6 to 6.5, indicating slightly acidic, yet natural, conditions.

### 2.4 PRELIMINARY CONCEPTUAL SITE MODEL

A preliminary CSM was developed based upon a review of the history of operations at the Burn Pits and the results of the Phase I RFI, which suggest that a release may have occurred at each of these six sites. During the Phase I RFI, soil samples from the monitoring well boreholes surrounding the suspected burn areas contained VOCs and metals at concentrations above site-specific background concentrations. Similarly, groundwater samples from the wells contained VOCs and metals at concentrations above their respective MCLs.

Chemicals of potential concern (COPCs) at this site include gasoline constituents (BTEX) and cleaning solvents (PCE), which may have been used to ignite surface piles of clearing and grubbing debris. Resulting burning residues may include hazardous constituents, primarily semivolatile organic compounds (SVOCs), in the surface soils. Although process knowledge indicates that metal contamination is not anticipated, metal constituents with concentrations greater than background were reported in soil and groundwater in the Phase I RFI and may have been present in the burning residues.

The most likely pathways for contaminant migration at the Burn Pits are via percolation vertically through the subsurface soils to groundwater, then via groundwater flow toward the nearest stream. VOCs and some metals are relatively mobile and may have migrated to subsurface soils or to groundwater. However, because VOCs are not persistent in soils due to their volatile and biodegradable nature, they are not likely to be present in surface soils. SVOCs and most metals, on the other hand, are not very mobile and are relatively persistent; therefore, they are not likely to migrate to groundwater but to remain in the soils on site. Surface water drainages are not present in the vicinity of the Burn Pits; therefore, constituents that reach groundwater are not likely to migrate to surface water. Thus, SVOCs and metals would be expected to be present predominantly in the surface soils within the burn areas; VOCs and metals would be expected in subsurface soil and groundwater localized to the sites.

Potential human receptors include on-site workers or soldiers on maneuvers who may come into contact with contaminated soils, and hypothetical future residents who may ingest groundwater. Because the surficial aquifer is not used as a source of potable water, any ingestion of groundwater by future residents would be accidental. Potential ecological receptors include terrestrial soil-dwelling animals and their predators that may ingest contaminated surface soil at the site, and aquatic biota in the respective receiving stream that may ingest contaminated groundwater discharging to that stream.

# 3.0 SUMMARY OF INVESTIGATION ACTIVITIES

#### 3.1 SAMPLING METHODOLOGIES

This section describes the Phase II RFI field investigations conducted at the Burn Pits from July 8 through August 13, 1997. The sampling methodologies and types of testing for physical and chemical characterization of each site are also described. Locations of the Phase II sampling stations are shown generally in Figure 3-1 and in greater detail in Sections 9.0 through 14.0 for Burn Pits A through F, respectively. The sampling strategy included soil and groundwater sampling at each of the six Burn Pit sites.

#### 3.1.1 Soil Sampling

Three soil borings were drilled at each of the 6 Burn Pit sites, for a total of 18 borings. The soil borings were drilled between July 8 and 14, 1997, and surface and subsurface soil samples were collected for the following reasons:

- to identify whether chemicals occur at concentrations above background in soils at the site;
- to determine the nature of any such contamination, if present;
- to determine the extent of that contamination, together with the peripheral soil borings at monitoring wells;
- to obtain lithographic descriptions of the soil profile at each monitoring well; and
- to evaluate human health and ecological risks due to site contaminants.

Soil samples were also taken during the drilling of boreholes for the installation of monitoring wells at selected sites. A total of eight new monitoring wells were installed, one at Burn Pit A, three at Burn Pit C, two at Burn Pit D, and two at Burn Pit E. The monitoring well boreholes were drilled between July 9 and 24, 1997. The monitoring wells were installed around the periphery of the site, at both upgradient and downgradient locations. Soil samples were collected from the monitoring well boreholes for the following reasons:

- to collect relatively undisturbed samples for geotechnical testing of the screened interval;
- to obtain lithographic descriptions of the soil profile at each monitoring well; and
- to obtain background (upgradient) soil samples for characterization at Burn Pit C.

Auger-drilled soil boreholes were advanced using 4.25-inch inside diameter hollow-stem augers drilling with either a CME-55 or Ingersoll-Rand A-300 drilling rig. Total depth of the undisturbed soil samples ranged from 9.5 to 19.5 feet, and the total depth of the monitoring well boreholes ranged from 14 to 21 feet. During the drilling of each soil borehole, soil samples were collected with a split-barrel sampler continuously over 5-foot intervals from the ground surface to the water table. Two soil samples were collected for off-site chemical laboratory analysis, one from the ground surface and one from the subsurface interval having the highest VOC headspace reading.

The surface soil sample was taken from a depth of 0 to 1 foot below the ground surface and was sent offsite for analysis for SVOCs and RCRA metals. The analysis of surface soils was carried out to provide data suitable for conducting human health and ERAs. SVOCs are target

## THIS PAGE INTENTIONALLY LEFT BLANK



 $R \to 20^{10} - 40^{10} R$ 

1.

\*\*##}:

- See

Figure 3-1. Locations of Sampling Stations at the E

97-160P(DOC)/020899

-

And the

j

3-4

THIS PAGE INTENTIONALLY LEFT BLANK

water and an

f of second

compounds in surface soils because they may occur as residuals from burning operations. RCRA metals are target chemicals because they were found at concentrations above background during the Phase I RFI in peripheral monitoring wells.

For subsurface soil samples, the 5-foot split-barrel core was split into two 2.5-foot sections. A portion of each 2.5-foot section was field tested for VOC headspace gas using a photoionization detector (PID). The borehole sample having the highest detected organic vapor concentration in the headspace gas was then sent offsite for quantitative laboratory analysis for VOCs and RCRA metals. The analysis of subsurface soils was conducted to characterize the extent of subsurface contamination and the potential for contaminants identified in the soil to leach to groundwater. VOCs are target compounds because they may have resulted from gasoline or other flammable liquids used to ignite the clearing and grubbing debris at the Burn Pits. If no VOCs were table was sent for analysis because gasoline-based contaminants of concern (COCs) float on the offsite for laboratory analysis for total organic carbon for use in assessing fate and transport of organic constituents.

Results of the chemical analyses are presented in Sections 9.0 through 14.0. Boring logs for the drilling of both monitoring wells and soil borings are included in Appendix A.

One soil sample from the screened interval in each monitoring well borehole was analyzed for geotechnical parameters to support contaminant transport evaluation. Bulk soil samples were taken directly from the 5-foot split-barrel core and placed into jars. Bulk samples were tested for moisture content, Atterberg limits, and grain size distribution. One relatively undisturbed sample was collected from a boring at each of the six Burn Pit sites using a thin-walled (Shelby) tube sampler. The Shelby tube sampler was inserted into the hollow-stem auger string and hydraulically pushed approximately 2.0 feet. The ends of the Shelby tube sampler were sealed with wax to preserve moisture content in accordance with American Society for Testing and Materials (ASTM) D1587-83, and the tubes were shipped to an off-site laboratory for analysis. The Shelby tube samples were tested for moisture content, Atterberg limits, grain size distribution, specific gravity, soil porosity, and permeability. Geotechnical parameters were obtained to support fate and transport analysis.

Decontamination of drilling and down-hole sampling equipment was accomplished in accordance with the procedures specified in the Phase II RFI Work Plan. These procedures for sampling equipment included washing with water and phosphate-free detergent; rinsing alternately with water and isopropyl alcohol; and placing the equipment on clean plastic, or wrapping in plastic or aluminum foil, to prevent cross-contamination.

### 3.1.2 Groundwater Sampling

## 3.1.2.1 Monitoring Well Installation and Development

A total of eight new monitoring wells were installed between July 9 and 24, 1997-one at Burn Pit A, three at Burn Pit C, two at Burn Pit D, and two at Burn Pit E. The wells were constructed of 2.0-inch-diameter Schedule 40 polyvinyl chloride (PVC) casing with flush-threaded couplings. Well screens were constructed of factory-slotted pipe in 10-foot-long sections. Slot size, determined from the sieve analysis results from the soil boring and field sieve analyses, varied from 0.008 to 0.010 inches (No. 8 to No. 10 slot). Filter pack materials consisted variously of DSI Extra Fine, DSI #1, and DSI #2 Sand. Well construction diagrams are presented in Appendix B. Well construction details are summarized in Table 3-1.

97-160P(DOC)/020899

Table 3-1. Phase II Monitoring Well Construction Summary Burn Pits, Fort Stewart
---

Slot Size         Total         Screen Interval         Lup or future           Slot Size         Coordinates         Total         Screen Interval         Filter Pack         Pack Elevation           Burn Pit A         Burn Pit A         Burn Pit A         Material         (feet)         (feet)           Burn Pit A         Burn Pit A         Burn Pit A         66.47 - 51.97         DSI Extra         62.97           Burn Pit C         61.47 - 51.97         Filter Pack         Pack Elevation         (feet)           0.010         N677164.6         16.0         69.16 - 59.56         DSI #2 Sand         70.79           0.010         E818074.3         14.5         69.90 - 60.40         DSI #2 Sand         71.30           0.010         N677009.4         14.5         67.49 - 57.99         DSI #2 Sand         71.30           0.010         N67788.4         Burn Pit D         67.49 - 57.99         DSI #2 Sand         68.69           0.010         N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           0.010         N675992.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           0.010         N675992.0         13.6         63.78 - 54.08         DSI #2 Sand	┢								Ton of Casing
Incres)         Continues         Burn Pit A         SI = 51.97         DSI Extra         62.97         7           0.008         N686360.3         16.0         61.47 - 51.97         DSI #2 Sand         62.97         7           0.010         N677164.6         16.0         69.16 - 59.56         DSI #2 Sand         70.79           0.010         N677164.6         16.0         69.16 - 59.56         DSI #2 Sand         71.30           0.010         N677009.4         14.5         69.90 - 60.40         DSI #2 Sand         71.30           0.010         N677009.4         14.5         67.49 - 57.99         DSI #2 Sand         71.30           0.010         N676959.2         14.5         67.49 - 57.99         DSI #2 Sand         68.69           0.010         N676959.2         14.5         67.49 - 57.99         DSI #2 Sand         68.69           0.010         N678930.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 55.54         DSI #2 Sand         64.78           0.0010         N675942.0         15.2	Size/		Slot Size	actor for a	Total Depth (feet)	Screen Interval Elevation (feet)	Filter Pack Material	Top of Fliter Pack Elevation (feet)	Lop of Casing Elevation (feet)
0.008         N686360.3         16.0         61.47 - 51.97         DSI Extra         62.97         7           Burn Pit C         Burn Pit C         Burn Pit C $0.010$ N677164.6         16.0         69.16 - 59.56         DSI #2 Sand         70.79         70.79           0.010         N677164.5         16.0         69.16 - 59.56         DSI #2 Sand         71.30           0.010         Barn Pit C         69.90 - 60.40         DSI #2 Sand         71.30           0.010         N676959.2         14.5         67.49 - 57.99         DSI #2 Sand         71.30           0.010         Barn Pit D         68.24 - 58.54         DSI #2 Sand         68.69         68.94           0.010         N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         68.94           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           0	Type		(inches)	COOLUIIIAUCS	Rurn Pit	<b>_</b>			
0.008         N686560.3         16.0         01.47 - 51.57         Fine Sand         70.79           Burn Pit C         Burn Pit C         Burn Pit C         70.79         70.79           0.010         N677009.4         16.0         69.16 - 59.56         DSI #2 Sand         70.79           0.010         Barn Pit C         69.90 - 60.40         DSI #2 Sand         71.30           0.010         Barn Pit D         67.49 - 57.99         DSI #2 Sand         71.30           0.010         B817785.4         Burn Pit D         68.24 - 58.54         DSI #2 Sand         68.69           0.010         N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           0.010         B817785.4         Burn Pit D         63.78 - 54.08         DSI #2 Sand         68.94           0.010         N675942.0         15.2         63.78 - 58.54         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 55.64         DSI #2 Sand         64.78           0.010         N675942.0         15.2         63.78 - 55.64         DSI #1 Sand         64.74						١L	DOI Evtra	62.97	70.38
Burn Pit C         Burn Pit C $16.0$ $69.16 - 59.56$ DSI #2 Sand $70.79$ $70.79$ 0.010         N677009.4 $14.5$ $69.90 - 60.40$ DSI #2 Sand $71.30$ 0.010         N677009.4 $14.5$ $69.90 - 60.40$ DSI #2 Sand $71.30$ 0.010         N676959.2 $14.5$ $67.49 - 57.99$ DSI #2 Sand $71.30$ 0.010         N676959.2 $14.5$ $67.49 - 57.99$ DSI #2 Sand $68.69$ 0.010         N676959.2 $14.5$ $67.49 - 57.99$ DSI #2 Sand $68.69$ 0.010         N675893.0 $13.8$ $68.24 - 58.54$ DSI #2 Sand $68.94$ 0.010         N675942.0 $15.2$ $63.78 - 54.08$ DSI #2 Sand $64.78$ 0.010         N675942.0 $15.2$ $63.78 - 54.08$ DSI #2 Sand $64.78$ 0.010         N675942.0 $15.2$ $63.78 - 54.08$ DSI #2 Sand $64.78$ 0.010         N675942.0 $15.2$ $63.78 - 55.64$ DSI #1 Sand $64.78$ 0.008	2-inch		0.008	N686360.3 F814818 7	16.0	16.10 - 14.10	Fine Sand		
0.010N677164.616.069.16 - 59.56DSI #2 Sand70.79 $Bun010$ E818074.314.569.90 - 60.40DSI #2 Sand71.30 $0.010$ N677009.414.567.49 - 57.99DSI #2 Sand71.30 $0.010$ N676959.214.567.49 - 57.99DSI #2 Sand68.69 $0.010$ N676959.214.567.49 - 57.99DSI #2 Sand68.69 $0.010$ N675893.013.868.24 - 58.54DSI #2 Sand68.94 $0.010$ N675942.015.263.78 - 54.08DSI #2 Sand64.78 $0.010$ N675942.015.263.78 - 54.08DSI #2 Sand68.94 $0.010$ N675942.015.263.78 - 54.08DSI #2 Sand68.04 $0.010$ N675942.015.263.78 - 54.08DSI #2 Sand68.04 $0.010$ N675942.015.263.78 - 54.08DSI #2 Sand68.04 $0.008$ N709229.121.065.84 - 55.64DSI #1 Sand68.04 $0.008$ N709141.521.065.54 - 55.34DSI #1 Sand67.74					Durn Dif	Ţ			
N677164.616.069.16 - 59.50D51 #2 Sand71.30E818074.314.569.90 - 60.40DSI #2 Sand71.30E818206.914.567.49 - 57.99DSI #2 Sand68.69E818206.914.567.49 - 57.99DSI #2 Sand68.69N676959.214.567.49 - 57.99DSI #2 Sand68.69N675893.013.868.24 - 58.54DSI #2 Sand68.94N675942.013.868.24 - 58.54DSI #2 Sand68.94N675942.015.263.78 - 54.08DSI #2 Sand68.94N675942.015.263.78 - 54.08DSI #2 Sand68.04SN709229.121.065.84 - 55.64DSI #1 Sand68.04SN709141.521.065.54 - 55.34DSI #1 Sand67.74Burn Pit EE847932.065.54 - 55.34DSI #1 Sand67.74							There are and	70.79	75.45
E818074.3E818074.314.569.90 - 60.40DSI #2 Sand71.30N677009.414.569.90 - 60.40DSI #2 Sand71.30E818206.914.567.49 - 57.99DSI #2 Sand68.69N676959.214.567.49 - 57.99DSI #2 Sand68.69N675893.013.868.24 - 58.54DSI #2 Sand68.94N675893.013.868.24 - 58.54DSI #2 Sand64.78N675942.015.263.78 - 54.08DSI #2 Sand64.78N675942.015.263.78 - 54.08DSI #2 Sand64.78SN709229.121.065.84 - 55.64DSI #1 Sand68.04SN709229.121.065.54 - 55.34DSI #1 Sand67.74BN709141.521.065.54 - 55.34DSI #1 Sand67.74	2-inch	1	0.010	N677164.6	16.0	00.96 - 00.60	Dilbo 1# IOU		
N677009.414.569.90 - 60.40DSI #2 Sand $1.00$ E818206.914.567.49 - 57.99DSI #2 Sand $68.69$ Burn Pit D $8.749 - 57.99$ DSI #2 Sand $68.94$ N675893.013.8 $68.24 - 58.54$ DSI #2 Sand $68.94$ N675942.013.8 $68.24 - 58.54$ DSI #2 Sand $68.94$ N675942.015.2 $63.78 - 54.08$ DSI #2 Sand $68.94$ N675942.015.2 $63.78 - 54.08$ DSI #2 Sand $68.94$ N075942.015.2 $63.78 - 54.08$ DSI #2 Sand $68.04$ N079229.121.0 $65.84 - 55.64$ DSI #1 Sand $68.04$ N709141.521.0 $65.54 - 55.34$ DSI #1 Sand $67.74$ E847932.0E847941.1 $67.74$ $67.74$	DVC			E818074.3				1 20	74.66
E818206.9       14.5       67,49 - 57.99       DSI #2 Sand       68.69         N676959.2       14.5       67,49 - 57.99       DSI #2 Sand       68.69         Burn Pit D       Burn Pit D       68.24 - 58.54       DSI #2 Sand       68.94         N675893.0       13.8       68.24 - 58.54       DSI #2 Sand       68.94         N675942.0       13.8       68.24 - 58.54       DSI #2 Sand       68.94         N675942.0       15.2       63.78 - 54.08       DSI #2 Sand       68.94         N675942.0       15.2       63.78 - 54.08       DSI #2 Sand       68.04         N0709229.1       21.0       65.84 - 55.64       DSI #1 Sand       68.04         N709141.5       21.0       65.54 - 55.34       DSI #1 Sand       67.74         R847932.0       63.78 - 55.34       DSI #1 Sand       67.74	2-inch	l	0.010	N677009.4	14.5	69.90 - 60.40	DSI #2 Sand	00-17	2011
N676959.2         14.5         67.49 - 57.99         DSI #2 Sand         08.09           E817785.4         Burn Pit D         68.24 - 58.54         DSI #2 Sand         08.04           N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         08.04           N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         68.04           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         68.04           N675942.0         15.2         63.78 - 55.64         DSI #2 Sand         68.04           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74	DVG			E818206.9				02.02	96 VL
E817785.4     Burn Pit D     68.24 - 58.54     DSI #2 Sand     68.94       0.010     N675893.0     13.8     68.24 - 58.54     DSI #2 Sand     68.94       0.010     N675942.0     15.2     63.78 - 54.08     DSI #2 Sand     68.78       0.010     N675942.0     15.2     63.78 - 54.08     DSI #2 Sand     68.78       0.010     N675942.0     15.2     63.78 - 54.08     DSI #2 Sand     68.04       0.010     N675942.0     15.2     63.78 - 55.64     DSI #1 Sand     68.04       0.008     N709229.1     21.0     65.84 - 55.64     DSI #1 Sand     68.04       0.008     N709141.5     21.0     65.54 - 55.34     DSI #1 Sand     67.74	2-inch	1	0.010	N676959.2	14.5	67.49 - 57.99	DSI #2 Sand	00.03	
Burn Pit D         Burn Pit D           N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         68.94           E820512.8         13.8         68.24 - 58.54         DSI #2 Sand         68.94           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         68.04           N67929.1         Burn Pit E         65.84 - 55.64         DSI #1 Sand         68.04           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74	PVC			E817785.4					
N675893.0         13.8         68.24 - 58.54         DSI #2 Sand         08.94           E820512.8         15.2         63.78 - 54.08         DSI #2 Sand         08.94           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74           B847932.0         63.54 - 55.34         DSI #1 Sand         67.74		1			Burn Pit				12 02
E820512.8         15.2         63.78 - 54.08         DSI #2 Sand         64.78           N675942.0         15.2         63.78 - 54.08         DSI #2 Sand         64.78           E820314.1         Burn Pit E         65.84 - 55.64         DSI #1 Sand         68.04           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74	2-inch	1	0.010	N675893.0	13.8	68.24 - 58.54	DSI #2 Sand	08,94	
N675942.0         15.2         63.78 - 54.08         D51 #2 5410         0.1.0           E820314.1         Burn Pit E         0.7.0         65.84 - 55.64         DSI #1 Sand         68.04           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74           B477932.0         21.0         65.54 - 55.34         DSI #1 Sand         67.74	PVC			E820512.8			Dot In Cand	64.78	71.82
E820314.1         Burn Pit E         Burn Pit E         65.84 - 55.64         DSI #1 Sand         68.04           N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         68.04           E847941.1         21.0         65.54 - 55.34         DSI #1 Sand         67.74           B847932.0         51.0         65.54 - 55.34         DSI #1 Sand         67.74	2-inch	1	0.010	N675942.0	15.2	63.78 - 54.08	DIT #2 Dallu	0	
Burn Pit E         Burn Pit E         Burn Pit E         65.84 - 55.64         DSI #1 Sand         68.04           E847941.1         21.0         65.84 - 55.34         DSI #1 Sand         68.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74           E847932.0         67.74         DSI #1 Sand         67.74	PVC			E820314.1					
N709229.1         21.0         65.84 - 55.64         DSI #1 Sand         00.04           E847941.1         21.0         65.84 - 55.34         DSI #1 Sand         05.04           N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74           E847932.0         65.54 - 55.34         DSI #1 Sand         67.74					Burn Pn			70 07	76 53
E847941.1         E347941.1         67.74           0.008         N709141.5         21.0         65.54 - 55.34         DSI #1 Sand         67.74           E847932.0         E847932.0         21.0         65.54 - 55.34         DSI #1 Sand         67.74	2-inch	1	0.008	N709229.1	21.0	65.84 - 55.64	DSI #1 Sand	+0.00	
0.008 N709141.5 21.0 65.54 - 22.34 USL#1 Salla VIII B847932.0 E847932.0	PVC			E847941.1			There at Cand	PL LY	76.24
	2-inch	_	0.008	N709141.5	21.0	65.54 - 55.54			
	PVC			E847932.0					

Note: All elevations in National Geodetic Vertical Datum 1929 datum. PVC = polyvinyl chloride.

÷

Ì

)

Each of the eight new wells was installed in the water table aquifer, to depths ranging from 14 to 21 feet. These wells were installed such that the screened interval bisects the water table, so that any free phase liquid floating on the water table surface could be detected in the well.

The wells were developed between July 13 and 30, 1997. Well development was accomplished using a downhole positive displacement pump. A surge block was used to agitate and mobilize particulates around the well screen by rapidly surging the block up and down. Well development continued until the well water was clear to the eye, sediment within the well was less than 0.1 foot, a minimum of five times the standing water volume in the well had been removed, and five times the volume of any water added during completion had been removed. In addition, water quality parameters [pH, conductivity, temperature, dissolved oxygen, oxidation-reduction potential (Eh), and turbidity] were measured during well development to verify that they had reached equilibrium, and development of the new wells continued until turbidity measured less than 10 nephelometric turbidity units (NTUs) if achievable. Well development is summarized in Table 3-2.

Each of the 24 existing wells at these sites was redeveloped using the same methods as for newly installed wells. The purpose of this redevelopment was to remove any accumulated sediment and to provide well water of comparable turbidity to newly installed wells. The redevelopment of existing wells is also summarized in Table 3-2. Problems were encountered attaining a turbidity reading less than 10 NTUs while redeveloping the four existing wells at Burn Pit A; therefore, discussions between Fort Stewart and GEPD were held, and a decision was made to cease redevelopment once turbidity was less than 25 NTUs. At Burn Pit D, turbidity in MW-2 could not be reduced below 54 NTUs after more than 20 hours of development; a decision was made to cease redevelopment. During sampling, turbidity in MW-2 declined to 16 NTUs.

At Burn Pit C, turbidity could not be reduced below 200 NTUs even after more than 10 hours of development. After consultation with GEPD, these four wells were abandoned by filling the wells with grout. Formal closure of these wells was approved by GEPD in telephone conversations with Ms. Melanie Little of Fort Stewart and followed up with a memorandum to Mr. Rabon of GEPD dated July 30, 1997.

Monitoring well MW-2 at Burn Pit E was discovered to have a bent or separated pipe casing, which prevented the surge block from being inserted into the well. Monitoring well MW-2 at Burn Pit E was, therefore, also abandoned by filling the well with grout, and a new well, MW-6, was constructed to replace it. Monitoring well MW-1 at Burn Pit E could not be developed or sampled, due a depth of water in the well of less than 1 foot and an extremely slow recharge rate, which resulted in the well being purged dry. Therefore, MW-1 was also abandoned by filling the well with grout.

#### 3.1.2.2 Monitoring Well Sampling

A total of 26 monitoring wells were sampled. Groundwater sampling was not conducted until at least 14 days after well development, between July 27 and August 13, 1997. Monitoring wells were sampled using low-flow micropurging techniques to minimize the volume of purge water, minimize disturbance of the aquifer, and thereby minimize turbidity in the sample. Prior to installing the sampling pump, the static water level was recorded in each well. Field parameters (pH, conductivity, temperature, dissolved oxygen, Eh, and turbidity) were monitored during micropurging. The purge rate was adjusted, as necessary, to avoid purging any well to dryness and to equal the recharge of the aquifer. Purging was considered complete when the field

97-160P(DOC)/020899

Well No.	New or Existing	Date Developed	Total Development Time (hours)	Total Volume Removed (gallons)	Final Turbidity Reading (NTUs)
Wen Ito, j			Burn Pit A		
MW-1	Exst	7/22/97	2.5	244	25.4
MW-2	Exst	7/22/97	2.75	180	2.75
MW-3	Exst	7/22/97	4.25	372	25.8
MW-4	Exst	7/16,7/22//97	7	235	17.18
MW-4 MW-5	New	7/30/97	2	156	6.52
IVI W-J	110.00	1120121	Burn Pit B		
MW-1	Exst	7/23/97	1.75	60	7.5
MW-2	Exst	7/23/97	1.25	85	3.3
MW-2 MW-3	Exst	7/23/97	1.25	109	2.1
	Exst	7/23/97	2.5	190	7.67
<u>IVI VV</u>	LASI	112517	Burn Pit C		
N/317 1*	Exst	7/11-7/12/97	21.25	825	very turbid*
<u>MW-1*</u>	Exst	7/11-7/12/97	14.5	135	>200*
MW-2* MW-3*	Exst	7/11-7/12/97	11	65	very turbid*
	Exst	7/12/97	3.75	18	opaque*
MW-4* MW-5	New	7/15-7/16/97	14.75	142	9.74
MW-6	New	7/15/97	2.25	77	4.87
MW-7	New	7/15-7/16/97	11.25	117	5.17
IVI W - /	1464	1120 1120,01	Burn Pit D		
MW-1	Exst	7/9/97	5	184	1.44
	Exst	7/9-7/11/97	21.5	1,595	54
	Exst	7/10/97	2	65	3.95
	Exst	7/10/97	4.25	290	7.2
	New	7/13/97	6.75	555	9.25
C- AN TAF	INCW	7/28/97	1	55	7
MW-6	New	7/13/97	3	110	5.54
IAT AAO		7/28/97	1.5	55	7.25
			Burn Pit E		
MW-1*	Exst	7/14/97	0.5	<1	pumped dry*
MW-2*	Exst	7/14/97	PVC cas	sing bent, well aba	ndoned*
	Exst	7/19/97	1.75	57	2.19
	Exst	7/14/97	3.5	16	2.1
	New	7/27/97	1.25	75	2.04
MW-6	New	7/27/97	1.75	115	1.79
147.11-0			Burn Pit F		
MW-1	Exst	7/14/97	4.75	47	9.96
MW-2	Exst	7/14/97	1	40	1.94
MW-3	Exst	7/28/97	3	25	0.5
MW-4	Exst	7/14/97	2	18	5.75

# Table 3-2. Well Development Summary, Burn Pits, Fort Stewart

\*Monitoring well could not be developed and was abandoned. NTUs = nephelometric turbidity units. PVC = polyvinyl chloride.

)

parameters stabilized within plus or minus 10 percent after a minimum of three readings at 5-minute intervals. Results of field parameter measurements made at the end of purging in each well are listed in Table 3-3.

Monitoring Well	рН (S.U.)	Conductivity (µmho/cm)	Temperature (°C)	Turbidity (NTUs)	Dissolved Oxygen (mg/L)	Eh (mV)	Total Purge Time	
1 4111 4			Burn F			<u> </u>	(hours	
MW-1	4.73	51	21.69	25.0	2.11	444.1	12	
MW-2	4.63	73	22.19	22.4	1.73	417.4	4.5	
MW-3	4.5	53	22.04	86.5	0.88	319.0	4.5	
MW-4	4.37	220	24.9	9.59	1.05	430.4		
MW-5	4.55	114	24.38	8.9	1.36	359.7	2.75	
1 4771 -			Burn P				2.75	
MW-1	4.85	45	22.69	50.9	2.74	442.4	12	
MW-2	4.73	20	24.12	24.1	3.12	448.9	$\frac{12}{12}$	
MW-3	4.77	49	22.8	79.8	1.63	456.2	$\frac{12}{10}$	
MW-4	5.68	376	28.02	6.2	1.31	81.3	4	
N4312 1	<u> </u>		Burn Pi			01.5	4	
MW-1			Monitoring we	ell abandoned	· <b>· · · · · · · · · · · · · · · · · · </b>	·		
MW-2			Monitoring we					
MW-3			Monitoring we	ell abandoned			······	
MW-4			Monitoring we	il abandoned		····· <u>-···</u> _····		
MW-5	6.97	613	23.30	23.9	0.51	-98.3	6.5	
MW-6 MW-7	4.67	296	24.56	35.8	0.99	78.5	11.75	
IVI W-7	5:01	122	23.89	10.0	0.86	358.0	5	
MW-1	4 70 1		Burn Pit				L	
MW-2	4.39	139	24.02	7.59	0.70	113.9	<u> </u>	
MW-3	6.19	199	26.09	16.3	8.73	18.9	3:25	
MW-4	4.49	34	25.37	8.9	2.50	397.5	3.5	
MW-5	4.59	84	22.90	8.4	8.27	78.4	3	
MW-6	4.18	89	24.59	8.6	2.15	434.0	1.5	
101 00-0	4.30	116	23.85	20.6	2.54	427.3	1.5	
MW-1			Burn Pit					
MW-2	<u> </u>		Monitoring wel	l abandoned				
MW-3	4.84 30 23.06 18 2.50 L and a							
MW-4	4.84	the second se	23.06	1.8	2.59	341.6	0.5	
MW-5	5.03	32	24.45	5.5	4.03	395.4	1	
MW-6		46	22.87	10.0	2.07	373.2	0.3	
	4.67	29	21.2	9.9	2.24	404.4	1.5	
MW-1	101		Burn Pit					
MW-2	4.84	23	26.38	8.5	9.34	419.2	2	
MW-3	4.77	35	28.84	1.7	6.23	385.1	0.3	
MW-4	4.77	84	25.44	7.1	3.92	344.0	0.3	
TAT AA and	4.91	48	26.43	4.3	4.67	357.1	0.75	

 Table 3-3. Field Parameter Measurement During Groundwater Sampling

 Burn Pits, Fort Stewart

NTUs = nephelometric turbidity units.

Purging times varied, requiring from 20 minutes to 16 hours per well. A turbidity less than 10 NTUs was attainable in 10 of the 26 wells that were sampled. During the field work, a field change was approved by GEPD (memorandum to Mr. Rabon from Ms. Little, dated August 8, 1997) which allowed Fort Stewart to cease purging and to collect the groundwater sample if turbidity was less than 25 NTUs after at least 3 hours of purging, or if total purging

97-160P(DOC)/020899

time had exceeded 10 hours. The maximum turbidity at the time of sampling was 86.5 NTUs at Burn Pit A, MW-3.

Sampling of each monitoring well began immediately after completion of purging, using the same micropurging pump. Groundwater samples were transferred directly into laboratory sample containers, with the portion designated for volatile organic analysis taken first. Groundwater samples were then sent offsite for laboratory analysis for VOCs and RCRA metals.

A filtered groundwater sample was collected from six of the wells (Burn Pit A wells MW-3 and MW-5, Burn Pit B wells MW-1 and MW-3, Burn Pit C well MW-7, and Burn Pit E well MW-6) and sent offsite for laboratory analysis of RCRA metals. Filtered samples were taken for comparison with total metals analysis to determine whether high metals concentrations in groundwater, which had been observed at a number of SWMUs across the FSMR in the Phase I RFI, were due to turbidity (soil particles) in the samples. Filtered samples were taken from three wells in particular (Burn Pit A well MW-3, and Burn Pit B wells MW-1 and MW-3) because these three wells had shown the highest turbidity readings at the end of purging (between 50.9 and 86.5 NTUs). Filtered samples were taken by attaching a 0.45-micron filter to the end of the low-flow pump sampling line.

### 3.1.3 Investigation-Derived Waste Management

Investigation-derived wastes (IDW) were managed in accordance with the procedures specified in the Phase II RFI Work Plan (SAIC 1997). All IDW were determined to be nonhazardous material. Solid wastes were disposed of by transporting the material to the Fort Stewart Sanitary Landfill for use as daily cover. Liquid wastes were disposed of at the Fort Stewart industrial waste water treatment system.

#### 3.2 DATA QUALITY ASSESSMENT

Multiple activities were performed to achieve the desired data quality in this project. Data quality objectives (DQOs) were established in the Phase II RFI Work Plan to guide the implementation of the field sampling and laboratory analysis. A quality assurance (QA) program was established to standardize procedures and to document activities following the Quality Assurance Project Plan (QAPP) in the Work Plan. Upon receipt by the project team, data were subjected to a verification and validation review that identified and qualified problems related to the analysis. These review steps contributed to a final QCSR, Appendix C, which defines that data used in the investigation met the criteria and are employed appropriately.

The QA program established requirements for both field and laboratory quality control (QC) procedures. In general: field QC duplicates and QA split samples were required for each environmental sample matrix collected at sites being investigated at a frequency of ~10 percent; VOC trip blanks accompanied each cooler containing water samples for VOC determinations; and analytical laboratory QC duplicates, matrix spikes (MSs), laboratory control samples (LCSs), and method blanks were required for every 20 samples or less of each matrix and analyte. The primary goal of the QA program was to ensure that the quality of the results for all environmental measurements were appropriate for their intended use. To this end, a QAPP and standardized field procedures were compiled to guide the investigation and are described in the Work Plan. Through the process of readiness review, training, equipment calibration, QC implementation, and detailed documentation, the project has successfully accomplished the goals set by the QA program.

Project data quality determines its usability. The evaluation is based on the interpretation of laboratory QC measures, field QC measures, and the project DQOs. Daily Quality Control Summary Reports (DQCSRs) and other field-generated documents, such as sampling logs, boring logs, daily health and safety summaries, daily safety inspections, equipment calibration and maintenance logs, and sample management logs, were peer reviewed on site. Analytical data generated for this project have been subjected to a process of data verification, validation, and review. The project implemented the use of data validation checklists to facilitate laboratory data validation. These checklists were completed by the project-designated validation staff and were reviewed by the project laboratory coordinator.

A total of 108 environmental soil, groundwater, and field QC samples were collected with approximately 4700 discrete analyses (i.e., analytes), not including field measurements and field descriptions. The project produced acceptable results for over 97 percent of the sample analyses performed and successfully collected all required investigation samples, with the exception of the wells abandoned at Burn Pits C and E.

The overall quality of the Burn Pit SWMUs 4A through 4F information meets or exceeds the established project objectives listed in the Work Plan. Through proper implementation of the project data verification, validation, and assessment process, project information has been determined to be acceptable for use. This rigorous QA/QC approach demonstrates that the data, as presented, have been qualified as usable but estimated when necessary. Data produced for this study can withstand scientific scrutiny; are appropriate for their intended purpose; are technically defensible; and are of known and acceptable sensitivity, precision, and accuracy. A more detailed data quality assessment (DQA) may be found in Appendix C.

**Common Laboratory Contaminants.** Common laboratory contaminants were detected in one or more soil samples and include acetone, 2-butanone, 2-hexanone, methylene chloride, and toluene. Results for these common laboratory contaminants are questionable when the results are less than 5 times the quantitation limit for these constituents, or when the result is less than the average concentration detected in the background samples. In such cases, the constituent is not considered a site-related contaminant (SRC), but a likely laboratory artifact. Reference values are listed in Table 3-4.

Parameter Acetone	Quantitation Limit	5 × Quantitation Limits	Average Concentration in Background
2-Butanone	10	50	60.21
2-Hexanone	10	50	5.63
Methylene Chloride	10	50	5.63
Toluene	5	25	6.19
	5	25	10.64

 Table 3-4. Common Laboratory Contaminants (Soil)

 Burn Pits, Fort Stewart

Note: All units in µg/kg.

Filtered vs. Unfiltered Analyses. RCRA metals analyses were conducted on both filtered and unfiltered groundwater samples from six Burn Pits wells and two wells at the Former 724th Tanker Purging Station (TPS). Results for detected analytes are listed in Table 3-5.

As discussed in Appendix C, the results indicate no significant relative percent difference (RPD) for any of the analytes. Filtered results were somewhat less than the corresponding unfiltered results in most cases, but the differences were not significant relative to the analytical precision. With few exceptions, the results were less than 5 times the quantitation limit for the analyte. At these low concentrations, the absolute differences between analytical results are not considered significant if the difference is less than 3 times the quantitation limit. Therefore, filtered and unfiltered results were comparable, indicating good correlation in results. These results demonstrate that efforts to reduce effects of turbidity in groundwater samples were successful, and that any residual turbidity (maximum 86.5 in MW-3 at Burn Pit A) did not adversely affect the groundwater sampling results.

<b>A</b>	Station	Sample ID	Parameter	Reporting Limit	Unfiltered Result	Filtered Result
Area		264411	Barium	200	99.20 J	96.90 J
<u>724th TPS</u>	MW-4 MW-5	264511	Barium	200	70.20 J	<u> </u>
724th TPS	MW-3	4A4311	Barium	200	26.20 J	
Burn Pit A		4A4511	Barium	200	66.70 J	62.80 J
Burn Pit A	<u>MW-5</u>	<u>4R4511</u> 4B4111	Barium	200	21.00 J	
Burn Pit B	<u>MW-1</u>	4B4111 4C4711	Barium	200	26.60 J	
Burn Pit C	<u>MW-7</u>		Cadmium	0.5	0.36 J	
Burn Pit A	MW-3	<u>4A4311</u>	Cadmium	0.5	1.20	
Burn Pit E	MW-6	4E4611		10	2.40 J	0.88 J
Burn Pit B	MW-1	<u>4B4111</u>	Chromium	10	3.70J	0.88 J
Burn Pit B	MW-3	4B4311	Chromium	10	1.70J	
Burn Pit E	MW-6	4E4611	Chromium		1.80J	
Burn Pit A	MW-3	4A4311	Lead	1	0,82 J	
Burn Pit A	MW-5	<u>4A4511</u>	Lead	1	2.00	0.84 J
Burn Pit B	MW-1	4B4111	Lead	1	2.00	1.00 J
Burn Pit B	MW-3	<u>4B4311</u>	Lead	1		1.005
724th TPS	MW-4	264411	Mercury	0.05	0.30	0.052=
724th TPS	MW-5	264511	Mercury	0.05	0.58	0.052-
Burn Pit A	MW-3	4A4311	Mercury	0.05	0.22	0.041
Burn Pit B	MW-3	4B4311	Mercury	0.05	<u> </u>	0.04 J
Burn Pit C	MW-7	4C4711	Mercury	0.05	0.28	0.70 1
724th TPS	MW-4	264411	Selenium	5	0.51 J	0.79J
724th TPS	MW-5	264511	Selenium	5	0.78 J	
Burn Pit A	MW-5	4A4511	Selenium	5	0.70 J	<u>1.00 J</u>
724th TPS	MW-4	264411	Silver	0.2	4.10	
Burn Pit A	MW-3	4A4311	Silver	0.2	.028	
Burn Pit B	MW-1	4B4111	Silver	0.2	0.39 J	<u>0.19 J</u>
		4B4311	Silver	0.2	0.14 J	<u>0.11J</u>
Burn Pit B Burn Pit C	MW-7	4C4711	Silver	0.2	0.08 J	

#### Table 3-5. Filtered vs. Unfiltered (µg/L) Groundwater Sample Comparison Burn Pits, Fort Stewart

Note: A blank indicates that analyte was not detected. J indicates estimated value.

An exception is mercury, where results for 4 of the 8 unfiltered samples exceeded 5 times the quantitation limit, but results for the corresponding filtered samples did not exceed the quantitation limit. Another exception is silver in one sample from MW-4 at the Former 724th TPS, where the reported unfiltered result also exceeded 5 times the quantitation limit, but was undetected in the filtered sample. This suggests that mercury and silver may be adhered to soil particles and would be less likely to be transported in the dissolved phase in the groundwater.
## **4.0 PHYSICAL CHARACTERISTICS OF THE SITE**

#### **4.1 DEMOGRAPHICS**

The Burn Pits are located around the cantonment, or garrison, area of the FSMR, within Liberty County, Georgia (Figure 2-2). Liberty County occupies 328,768 acres and had a total population of 52,745 in 1990. Forty-one percent of the county population lives in Hinesville, the largest city in Liberty County. The total population of Fort Stewart in 1990 was 13,774, 55 percent of which were employed by the Armed Forces. Forty-one percent of the Fort Stewart population live in group quarters while the remaining population live in households (U.S. Department of Commerce 1990).

#### **4.2 TOPOGRAPHY**

The FSMR occupies a low-lying, flat region on the coastal plain of Georgia. Surface elevations range from approximately 20 to 100 feet above mean sea level (amsl) within the FSMR and generally decrease from northwest to southeast across the reservation. The topography is dominated by terraces dissected by surface water drainages. The terraces are remnants of sea level fluctuations. The four terraces present within the FSMR are the Wicomico, Penholoway, Talbot, and Pamlico (Metcalf and Eddy 1996). Burn Pits A, B, C, and D are situated west of the cantonment area in the southern portion of the FSMR and are located on the Wicomico Terrace (marsh deposits). Burn Pit E is situated north-northeast of the cantonment area on the Penholoway Terrace (barrier island deposits). Burn Pit F is located northeast of the cantonment area on the Talbot Terrace (barrier island deposits).

The elevation of most of the sites ranges between 65 and 85 feet amsl; the elevation of Burn Pit F ranges between 33 and 40 feet amsl. Each site is relatively flat, with local relief varying by only 5 to 10 feet.

#### 4.3 SURFACE DRAINAGE

The principal surface water body accepting drainage from the FSMR is the Canoochee River, which joins the Ogeechee River (part of the northwestern boundary of the reservation). The individual Burn Pit sites are located within different subwatersheds (Figure 2-3). Burn Pits A through D are located west of the garrison area within the Taylors Creek watershed; drainage from these sites is collected by Mill Creek and Horse Creek tributaries that discharge into Taylors Creek, which in turn discharges into Canoochee Creek. Canoochee Creek is a tributary of the Canoochee River; drainage from the site is collected in small unnamed streams or swampy areas that discharge directly to the Canoochee River. Burn Pit F is located south of Savannah Road in the Peacock Creek watershed; drainage from the site flows into Goshen Swamp, which discharges southward into Peacock Creek and ultimately into the Jerico River.

#### 4.4 REGIONAL GEOLOGY

The FSMR is located within the coastal plain physiographic province. This province is typified by nine southeastward dipping strata that increase in thickness from 0 feet at the fall line (located

approximately 350 miles inland from the Atlantic coast) to approximately 4200 feet at the coast. State geologic records describe a probable petroleum exploration well (the No. 1 Jelks-Rogers) located in the region as encountering crystalline basement rocks at a depth of 4254 feet below land surface. This well provides the most complete record for Cretaceous, Tertiary, and Quaternary sedimentary strata. Figure 4-1 presents a geologic column for the Tertiary and Quaternary section in the Fort Stewart area.

The Cretaceous section is approximately 1970 feet in thickness and dominated by clastics. The Tertiary section is approximately 2170 feet in thickness and dominated by limestone with a 175-foot-thick cap of dark green phosphatic clay. This clay is regionally extensive and is known as the Hawthorn Group. The interval from approximately 110 feet to the surface is Quaternary in age and composed primarily of sand with interbeds of clay or silt. This section is undifferentiated (Metcalf and Eddy 1996).

State geologic records contain information regarding a well drilled in October 1942, 1.8 miles north of Flemington at Liberty Field of Camp Stewart (now known as Fort Stewart). This well is believed to be an artesian well located approximately one-quarter mile north of the runway at Wright Army Airfield within the FSMR. The log for this well describes a 410-foot section, the lowermost 110 feet of which consisted predominantly of limestone above which 245 feet of dark green phosphatic clay, typical of the Hawthorn Group, was encountered. The uppermost 55-foot interval was Quaternary-age interbedded sands and clays. The top 15 feet of these sediments were described as sandy clay (Metcalf and Eddy 1996).

#### 4.5 SOILS

Boring logs showing the types of soils encountered during the Phase II RFI at the Burn Pits in soil borings and monitoring well boreholes are given in Appendix A. Geological cross-sections of the sites are presented in Sections 9.0 through 14.0 for individual Burn Pit sites. In general, the soils are highly variable with abrupt changes in soil types occurring over relatively short distances. The surficial materials are generally a light gray sand or silty sand up to 20 feet thick. Clayey sand and sandy clay layers are interbedded with this sandy material at some boring locations. Wood fragments and charred wood were also encountered in some of the borings. The maximum depth explored in either the Phase I or Phase II RFIs was 21 feet at Burn Pit E.

Geotechnical analyses were conducted on five bulk samples plus six Shelby tube samples taken from the soil borings and monitoring well boreholes. The bulk samples were analyzed for moisture content in accordance with ASTM D2216, grain size distribution (ASTM D422), and Atterberg limits (ASTM D4318). The Shelby tube samples, which included one sample from each of the six Burn Pit sites, were analyzed for the same parameters as bulk samples, plus specific gravity (ASTM D854), porosity (EM1110-2-1906), and permeability (ASTM 5084). Results of the geotechnical analyses are summarized in Table 4-1. The geotechnical laboratory data sheets are included in Appendix D.

These results indicate that the tested soils are generally non-plastic sands with silt. The proportion of fine-grained particles varied over a relatively narrow range in the soils tested, from 2 to 12 percent by weight. The soil from the screened interval in MW-7 at Burn Pit C is a slightly silty sand, and was the only sample tested that showed some low plasticity. The permeability of the soils tested also varied over a relatively narrow range, from  $3.11 \times 10^4$  to  $1.73 \times 10^3$  cm/second, a range that is typical for fine to silty sands.

Lithologic Description	Poorly drain loamv unde		Pale gray to white, well-sorted sands		Argillaceous sands and clays		Buff colored, porous limestone containing foraminifera	Massive, fossilferous limestone	Glauconitic dolomite and limestone	Source Hundlacton Loon.		Jacksonville, FL
Geologic Únit	SURFICIAL SEDIMENTS	UNDIFFERENTIATED	UNDIFFERENTIATED		HAWTHORN GROUP	SUWANNEE LIMESTONE	GLENDON LIMESTONE MARIANNA LIMESTONE	OCALA GROUP	AVON PARK LIMESTONE	Limestone	<u> </u>	
Hydrogeologic Unit		SURFICIAL		CONFINING UNIT		UPPER FLORIDAN	(PRINCIPAL ARTESIAN)		CONFINING UNIT LOWER FLORIDAN AQUIFER	KXXXXXX Limestone	No XX OX OY	
Approximate Age	RECENT	PLEISTOCENE	PLIOCENE	UPPER MIOCENE	UPPER MIOCENE	OLIGOCENE		EOCENE		Surficial Sediments	Sand Sand with Clav	
	,		UD OUN		~~~~	787 1771					] Sand Sand	
										Legend		

Figure 4-1. Geologic and Hydrostratigraphic Column for the Fort Stewart Area

96-053MS(BP)/122297

(

44

4-3

		Permeability	(cm/second)		1.73E-03			7.07E-04				1.24E-03	+			0 455-04	2 7727			3.11E-04			-	0.81E-U4	
		Soil	Porosity		0.4209			03640				03056	22/2-2			0L7L V	0/00'0			0.4641				0.40/8	
		Specific	Gravity		2.662			1671	7/0.7			1 72A	10/-7			1 2 1 1	7.004			2.554				2.663	
ntion		Fines	(percent)		9	4			4		1 12			12			0	80		4	-	7		6	
Crain Size Distribution		Sand	(percent)		94	96		, -	96		87		95	87			94	91		90	2	98		06	
aio-C	1915	Gravel	(percent)		c		5		0			     	0				0				>	0			
	its	Plasticity Ladex	(percent)	Rurn Pit A			NP	Burn Pit B	đN	Rurn Pit C		N	đ	15		Burn Pit D	AN	dN	Durn Dir F		NP	đŊ	Rurn Pit F		INC
	Atterburg Limits	Plastic	(nercent)	<u>, , , , , , , , , , , , , , , , , , , </u>		dN N	M		AN			NP	đŻ				đŻ		IN .		đ	diz			NL NL
	At	Liquid	Limit (nercent)	(maxind)		đ	NP		đ			đz	đ		30		aiv I		N		ďZ	dN	11		dN
		Moisture	Content	(percent)		19	19.9		15.4			13.3	16.4	10.4	16.8		000	7.02	16.9		0.00	21.6	0.10		23
			Sample	No.		4A1213*	4A1513		******	451515	:	101513		4C16137	4CI713			4D1513*	4D1613		401513*	451717	<u>  4E1613</u>		4F1213*
			Depth	(feet)		9.5-11.5	45-65			13 - 15		76 100	//	12 - 14	10-12.5			10.7 - 12.5	12.8 - 13.8		11	11-01	15 - 17.5		19.5 - 21.5
				Station		SR-7	AVW 5	C- M M		SB-3			- MM	9-MW	NIW_7			MW-5	9-MW			NW-5	MW-6		SB-2

Table 4-1. Summary of Geotechnical Analyses, Burn Pits, Fort Stewart

NOTE: NP = Non-plastic. A blank indicates the sample was not tested for that parameter. \*indicates Shelby tube sample.

· )

## 97-160P(DOC)/020899

4-4

#### 4.6 HYDROGEOLOGY

The hydrogeology in the vicinity of the FSMR is dominated by two aquifers, referred to as the Principal Artesian and the surficial, that are separated by a confining unit (Figure 4-1).

The Principal Artesian aquifer is the lowermost hydrologic unit which is regionally extensive from South Carolina through Georgia, Alabama, and most of Florida, and is regionally known as the Floridan Aquifer. This aquifer is subdivided into upper and lower hydrogeologic units. The upper hydrogeologic unit is composed primarily of Miocene-age argillaceous sands and clays and Oligocene- to Eocene-age limestones (including the Ocala Group and the Suwannee Limestone, where present) at the top. The upper hydrogeologic unit ranges in thickness from 200 to 260 feet and is most productive where it is thickest and where secondary permeability is most developed. The lower hydrologic unit is comprised of the Eocene-age Avon Park Limestone at the base. The transmissivity of the aquifer in the Savannah area ranges from about 28,000 to 33,000 square feet/day (Krause and Randolph 1989). Groundwater from this aquifer is primarily used for drinking water (Arora 1984). Thirteen groundwater production wells screened in this aquifer are used for potable water supply on the FSMR, and one additional production well is available for

The confining layer for the Principal Artesian aquifer is the phosphatic clays of the upper Hawthorn Group. These sediments are regionally extensive and range from 60 to 80 feet in thickness at the FSMR. There are minor occurrences of aquifer material within the Hawthorn Group; however, they have limited utilization (Miller 1990).

The uppermost hydrologic unit is the surficial aquifer, which consists of widely varying amounts of sand, silt, and clay ranging from 55 to 150 feet in thickness. This aquifer is primarily utilized for domestic lawn and agricultural irrigation with wells typically yielding 2 to 180 gallons per minute. The top of the water table ranges from 2 to 10 feet below ground level (Geraghty and Miller 1992) but was found as deep as 13 feet in the current investigation.

Water levels were measured during well development and groundwater sampling in the monitoring wells at each of the six Burn Pits. Depth to water varied from 2 to 3 feet at Burn Pit C, to 12 to 13 feet at Burn Pit E. Water table contour maps, showing directions and rates of groundwater flow, are presented for each of the Burn Pit sites in Sections 9.0 through 14.0. Groundwater flow at each site is generally planar toward the nearest surface stream into which the groundwater discharges. An exception to this is at Burn Pits C and F, where water levels suggest that flow may be radially away from the burn area, prior to joining the direction of the general water table flow in the surrounding area.

#### 4.7 ECOLOGY

Approximately 7.8 square miles of the 436.8 square miles at FSMR comprise the garrison area. The remainder is used for ranges and training areas (approximately 11 percent) or held as non-use areas.

Eighty-four percent of the land is forested (approximately 367.2 square miles). Sixty-six percent of the forest area is pine with the major species including the slash pine, loblolly pine, and longleaf pine. Thirty-four percent of the forest is composed of river bottom lands and swamps whose major species include the tupelo, other gum trees, water oak, and bald cypress trees. The open range and training areas comprise 11 percent of the base and consist of grasses, shrubs, and scrub tree (oak) growth.

97-160P(DOC)/020899

Aquatic habitats on FSMR include a number of natural or man-made ponds and lakes, the Canoochee River, Canoochee Creek and tributaries, and a number of bottom land swamps and pools. The Ogeechee River borders the installation along its northeast boundary. Organic detritus content is high, and dark coloring of the water is not unusual. Dense growths of aquatic vegetation are also typical, especially during the summer months.

Both terrestrial and aquatic fauna are abundant in the unimproved areas of FSMR. Major game species found on the installation include white-tailed deer, feral hog, wild turkey, rabbit, squirrel, and bobwhite in addition to numerous other mammal, bird, reptile, and amphibian species (Environmental Science and Engineering 1982). Dominant fish include bluegill, largemouth bass, crappie, sunfish, channel catfish, minnows, and shiners. Three federally listed threatened or endangered species reside at FSMR: the American bald eagle, Eastern indigo snake, and the red-cockaded woodpecker.

#### 4.8 METEOROLOGY

Fort Stewart has a humid, subtropical climate with long, hot summers. Average temperatures range from 50°F in the winter to 80°F in the summer. Average annual precipitation is 48 inches, with slightly over half falling from June through September. Prolonged drought is rare in the area, but severe local storms (tornadoes and hurricanes) do occur. Under normal conditions, wind speeds rarely exceed 5 knots, but gusty winds of over 25 knots may occur during summer thunderstorms (Geraghty and Miller 1992).

## 5.0 CONTAMINANT NATURE AND EXTENT

This section summarizes the results of the chemical laboratory analyses of the soil and groundwater samples collected at the former Burn Pit sites. An assessment of contaminant concentrations, patterns of distribution at the six sites, and general conclusions as to the nature and extent of site-related contamination is also presented. This section provides an overview of the six sites; Sections 9.0 through 14.0 provide details specific to each of the six sites individually, including maps of contaminant distribution. Complete analytical results for the Phase II chemical data are included in Appendix F of this report. Analytical results for the Phase I chemical data were presented in Appendix U of the Phase I RFI Report (Rust 1994).

## 5.1 BACKGROUND DATA ANALYSIS AND SCREENING

The reference background criteria for the Burn Pits have been developed based on data from background samples collected from SWMUs across the FSMR. In general, reference background samples were collected in each medium at locations upgradient or upslope of each site so as to be representative of naturally occurring conditions at SWMUs under Phase II investigation. In addition, soil samples collected during the Phase I investigation [i.e., Burn Pits (SMWUs 4A, 4B, 4D, 4E, and 4F), Active Explosive Ordnance Disposal (EOD) Area (SWMU 12A), etc.], were included as reference background samples if they were upgradient of the site and if the data were of sufficient quality to be representative of natural background conditions at the FSMR. A summary of the sample stations, SWMUs, and the source of the data (Phase I or II RFI) is presented in Table 5-1 for each medium.

EPA Region IV methodology (EPA 1996b) was used as guidance for the development of the background data set for screening metals data. In cases where enough samples (e.g., more than 20) are collected to define background, a background upper tolerance level can be calculated. In cases where fewer samples (e.g., less than 20) are collected to define background, background concentration (EPA 1996b). Given that fewer than 20 background samples were collected for the FSMR, the latter method was used for calculating reference background concentrations for metals.

Appendix E presents the summary of background data and presents the two-times-mean background concentrations for metals. Given the limited number of reference background samples, the mean concentration of metals for soils in the eastern United States is also presented, for comparative purposes only. The locations of all reference background samples are also shown in Appendix E, on Figures E-1 and E-2.

The detected concentrations of organics in background samples were not used to calculate reference background criteria because all organic compounds are considered to be potentially man-made. Organic compounds were not screened against background. All detected organic compounds are considered SRCs. The following sections discuss the background data analysis for each medium.

#### 5.1.1 Surface Soil

Surface soil samples were taken from the ground surface to a depth of 1 or 2 feet below ground surface (bgs) depending on the amount of recovery from the sampling device. Thirteen surface

		······································		Station		
SWMU	SWMU Name on Hazardous Waste Permit HW-045	Surface Soil	Subsurface Soil	Groundwater	Surface Water	Sediment
Number		SC-M17 <sup>a</sup>	SC-M17 <sup>a</sup>	SC-M10 <sup>a</sup>	NA	NA
1	South Central Landfill	MW5 <sup>c</sup>	MW5 <sup>c</sup>	MW5 <sup>c</sup>	NA	NA
2	Camp Oliver Landfill	MW5 <sup>c</sup>	MW5°	MW5 <sup>c</sup>	NA	NÁ
3	TAC-X Landfill	141 44 5	MW1 <sup>b</sup>	MW1 <sup>d</sup>	NA	NA
4A	Burn Pit A		(Phase I)			<u> </u>
			MW3 <sup>b</sup>	MW3 <sup>d</sup>	NA	NA
4B	Burn Pit B		(Phase I)		]	
<u> </u>		MW7 <sup>d</sup>	MW7 <sup>d</sup>	MW7 <sup>d</sup>	NA	NA
<u>4C</u>	Burn Pit C	178.97.7	MW2 <sup>b</sup>	MW2 <sup>d</sup>	NA	NA
4D	Burn Pit D		(Phase I)			
			MW3 <sup>b</sup>	MW3 <sup>d</sup>	NA	NÅ
4E	Burn Pit E		(Phase I)			
			MW1 <sup>b</sup>	MW1 <sup>d</sup>	NA	NA
4F	Burn Pit F		(Phase I)			
	DOD senteining Open	MW1 <sup>e</sup>	MW1 <sup>e</sup>	MW1 <sup>c</sup>	NA	NA
12A	Active EOD containing Open Detonation Unit and Open Burn Pit		(Phase I)			
			1	MW8 <sup>c</sup>	NA	NA
14	Old Fire Training Area DRMO Hazardous Waste Storage	MW1 <sup>c</sup>	MW1 <sup>c</sup>	MW1 <sup>c</sup>	NA	NA
17						
	Area Industrial Wastewater Treatment	MW1 <sup>c</sup>	MW1 <sup>c</sup>	MW1 <sup>c</sup>	NA	NA
18	Plant	•••••		•		
- 26	Former 724th Tanker Purging Station	MW1 <sup>c</sup>	MW1 <sup>c</sup>	MW1 <sup>c</sup>	SWS-1	
26 29	Evans Army Heliport POL Storage	MW5°	MW5 <sup>c</sup>	MW5 <sup>c</sup>	NA	NA
	Facility	N GINY C	MW1 <sup>c</sup>	MW1 <sup>c</sup>	NA	NA
31	DEH Asphalt Tanks	MW1 <sup>c</sup>	MW1 <sup>c</sup>	MW1 <sup>c</sup>	NA	NA
32	Supply Diesel Tank	MW1 <sup>c</sup>	MW1 MW1	MW1 <sup>c</sup>	NA	NA
34	DEH Equipment Wash Rack	MW1 <sup>c</sup>		MW9	NA	NA
35	Wright Army Airfield Bulk Fuel	HA-05		(Phase I)		
1	System	(Phase I)	(Phase I)	(1110301)		

Table 5-1. Background Media Summary, Burn Pits, Fort Stewart

DEH = Directorate of Engineering and Housing.

DRMO = Defense Reutilization and Marketing Office.

EOD = Explosive Ordnance Disposal.

NA = Not applicable; surface water and sediment background are site specific.

POL = Petroleum Oil and Lubricant.

Bold indicates background groundwater sample collected from the same borehole as sample for soil (i.e., monitoring well was constructed in the borehole).

"Science Applications International Corporation (SAIC), September 1998. Phase II RCRA Facility Investigation Report for the South Central Landfill (SWMU 1), Fort Stewart, Georgia (Final Report), U.S. Army Corps of Engineers, Savannah District,

Contract No. DACA21-95-D-0022, Delivery Order 0012. <sup>b</sup>Rust Environment and Infrastructure, May 1996. Phase I RCRA Facility Investigation Report for 24 Solid Waste Management Units at Fort Stewart, Georgia, Volume I of III (Corrected Final Report), U.S. Army Corps of Engineers, Savannah District,

Contract No. DACA21-93-D-0029, Delivery Order 0005. Science Applications International Corporation (SAIC), September 1998. Phase II RCRA Facility Investigation Report for 16 Solid Waste Management Units at Fort Stewart, Georgia, Volume 1 (Draft Report), U.S. Army Corps of Engineers,

Savannah District, Contract No. DACA21-95-D-0022, Delivery Order 0009. <sup>d</sup>Science Applications International Corporation (SAIC), March 1998. Phase II RCRA Facility Investigation Report for the Burn Pits (SWMUs 4A - 4F) at Fort Stewart, Georgia (Final Report), U.S. Army Corps of Engineers, Savannah District, Contract No.

DACA21-95-D-0022, Delivery Order 0008. Radian International, LLC, January 1997. Site Characterization Report, Open Burn/Open Detonation Units, Fort Stewart, Georgia

(Draft Report), U.S. Army Corps of Engineers, Mobile District. Metcalf & Eddy, Inc., December 1996. Final Phase I RFI Report for Bulk Fuel Storage System at Wright Army Airfield, Fort Stewart, Georgia, U.S. Army Corps of Engineers, Contract No. DACA21-93-D-0049, Delivery Order 0018.

2

soil samples were used in the development of the surface soil background data set (Table 5-1). The reference background concentration for metals in surface soils was calculated as 2 times the average concentration of these 13 locations. Phase I data from SWMU 12A and SWMU 35 were determined to be of sufficient quality to include in the background data set. If a chemical was not detected at a site, then one-half the detection limit was used as the concentration in calculating the mean background concentration. The sample results included in the data set are presented in Table E-2, Appendix E. Metals are considered SRCs if their concentrations exceed the calculated reference background concentration. Organics were not screened against background; all organic compounds are considered SRCs if they were detected.

#### 5.1.2 Subsurface Soil

Subsurface soil samples were taken from the interval between a depth of 2 feet bgs and the water table. Eighteen subsurface soil samples were used in the development of the subsurface soil background data set (Table 5-1). Phase I data from SWMUs 4A, 4B, 4D, 4E, 4F, 12A, and 35 were determined to be of sufficient quality to include in the subsurface soil background set. The reference background concentration for metals in subsurface soil was calculated as 2 times the mean of the chemical detected at the 19 locations. If a chemical was not detected in a sample, then one-half the detection limit was used in calculating the mean background concentration. The sample results included in the background data set are presented in Table E-3, Appendix E. Metals are considered SRCs if their concentrations exceed the calculated reference background concentration. Organics were not screened against background; all organic compounds are considered SRCs if they were detected.

#### 5.1.3 Groundwater

Only groundwater samples collected using low-flow techniques (Phase II RFI) were used in the development of the groundwater background data set. Groundwater samples from 19 SWMUs were used (Table 5-1). The reference background concentration groundwater was calculated as two times the average of these 19 samples. If a chemical was not detected at a site, then one-half the detection limit was used in calculating the mean background concentration. The sample results included in the background data set are presented in Table E-4, Appendix E. Metals in groundwater are considered SRCs if their concentrations exceed the calculated reference background concentration. Organics were not screened against background; all organic compounds are considered SRCs if they were detected.

## 5.2 SURFACE SOIL CONTAMINATION

The nature and extent of surface soil contamination (0 to 1 foot bgs) was evaluated using the results from surface soil samples taken from three soil borings at each of the six Burn Pit sites and a total of eight monitoring well boreholes drilled during the Phase II investigation. This assessment presents Phase II contaminant data only, because no surface soil samples were collected during Phase I. The surface soil samples were analyzed for SVOCs and RCRA metals. Table 5-2 summarizes the analytical results for surface soil samples. Detailed discussions of the contaminants and their distribution in surface soils at each Burn Pit site are presented in Sections 9.0 through 14.0.

		M	avimum De	tected Conc	entrations in	Surface So	ils
Parameter	Reference Background	Burn Pit A	Burn Pit B	Burn Pit C	Burn Pit D	Burn Pit E	Burn Pit F
	Sem	ivolatile Org	zanic Compo	unds (µg/kg,	)		
Benzo(a) pyrene	0.0					<u>21</u> 54.8	
Benzo(b)fluoranthene	0.0		<u> </u>	<u> </u>		63	
Fluoranthene	0.0		<u> </u>	<u> </u>			,
			Metals (mg/	<u>(kg)</u>	0.34	0.17	2.7
Arsenic	2.10	2.5	14.4	20.2	12.6	18.3	13.6
Barium		0.36	-	0.35			<u> </u>
Cadmium	0.18	3.7	1.5	3.7	2.1	2.3	13.5
Chromium	6.21		10.5	33.2	7.7	8.5	7.3
Lead	8.81	10.4		0.06	0.03	0.04	0.04
Mercury	0.03	0.04	0.04	1 0.00		0.39	
Selenium	0.41		_ <b> </b>	_ <u>_</u>	+		
Silver	0.15				<u> </u>		<u>_ I</u>

#### Table 5-2. Summary of Analytical Results for Surface Soil Samples **Burn Pits, Fort Stewart**

Note: A blank indicates analyte not detected at that Burn Pit site.

Note: BOLD font indicates value exceeds reference background criteria for Resource Conservation and Recovery Act (RCRA) metals.

SVOCs. SVOCs were detected in surface soils at Burn Pit E, but not at any of the other five Burn Pit sites. Benzo(a)pyrene (21  $\mu$ g/kg), benzo(b)flouranthene (54.8  $\mu$ g/kg), and flouranthene (63  $\mu$ g/kg), were detected at Burn Pit E in a single surface soil sample from MW-6 at the southeastern edge of the former burn area. SVOCs were not detected in any of the other surface soil samples at Burn Pit E.

RCRA metals. Arsenic, barium, cadmium, chromium, lead, and mercury were reported at concentrations above their reference background concentrations in surface soils at one or more Burn Pit sites. Selenium and silver were not detected above background at any of the six Burn Pit sites and are, therefore, not COPCs in surface soils.

It should be pointed out that the metal detections in surface soils are not overly high for natural soil conditions. With the exception of lead in surface soils at Burn Pit C, most detections are within reasonable ranges for inorganics in soils in the eastern United States. Published reference values (USGS 1984), as listed in Appendix E, are average concentrations of metals in regional soils. Lead (33.2 mg/kg) at Burn Pit C exceeds the USGS reference value (17 mg/kg) for lead in soils in the eastern United States.

- At Burn Pit A, arsenic (2.5 mg/kg), barium (18.8 mg/kg), cadmium (0.36 mg/kg), lead (10.4 mg/kg), and mercury (0.04 mg/kg) were detected above background in surface soils in the northern and eastern portions of the site.
- At Burn Pit B, lead (maximum 10.5 mg/kg) and mercury (0.04 mg/kg) were detected in one of the three surface soil samples taken from the site (SB3). Both lead and mercury were ٠ detected at concentrations only slightly greater than their respective reference background value.
- At Burn Pit C, four metals were reported above reference background, predominantly at two boring locations in the middle of the source area. These metals include barium (20.2 mg/kg), cadmium (0.35 mg/kg), lead (33.2 mg/kg), and mercury (0.06 mg/kg). Only mercury was found above background in two other borings at the site.

- At Burn Pit D, no constituent was detected above background. Therefore, no site-related constituents above the referenced background were located at Burn Pit D.
- At Burn Pit E, barium was detected in four of the five surface soil samples at a maximum of 18.3 mg/kg. Mercury was detected in a single sample at a concentration (0.04 mg/kg) that is only slightly greater than background and is not likely to be site related.
- At Burn Pit F, maximum concentrations of arsenic (2.7 mg/kg) and chromium (13.5 mg/kg) were reported above background at boring SB-1 in the northeastern portion of the source area. Chromium also exceeded reference background concentrations at the other two boring locations (SB-2 and SB-3). Mercury was detected in a single sample at a concentration (0.04 mg/kg) that is only slightly greater than background and is not likely to be site related.

## 5.3 SUBSURFACE SOIL CONTAMINATION

The nature and extent of subsurface soil contamination was evaluated using the results of VOC and RCRA metals analyses on soil samples taken from both soil borings and monitoring well boreholes. Subsurface soil samples were taken during both Phase I and Phase II investigations, including 4 samples at each of the 6 sites from the Phase I well boreholes, 3 samples at each of the 6 sites from the Phase I well boreholes, 3 samples at each of total of 50 subsurface soil samples. Table 5-3 summarizes the analytical results for subsurface soil samples. Detailed discussions of the contaminants and their distribution in subsurface soils at each Burn Pit site are presented in Sections 9.0 through 14.0.

VOCs. BTEX compounds are target analytes because gasoline may have been used to ignite the debris at the Burn Pits. BTEX compounds were detected in both Phase I and Phase II subsurface soil samples from all six of the Burn Pit sites, but at relatively low concentrations. Maximum values were reported for benzene (30  $\mu$ g/kg at Burn Pit B), toluene (69.6  $\mu$ g/kg at Burn Pit C), and total xylenes (22  $\mu$ g/kg at Burn Pit D). Ethylbenzene was not detected at any of the six sites. BTEX contamination in subsurface soils shows a consistent pattern of distribution only at Burn Pits B and C, although at relatively low concentrations.

- At Burn Pit A, BTEX compounds were identified in 4 of the 8 subsurface soil samples and from both Phase I and Phase II borings, but at low concentrations (28 µg/kg benzene, 12 µg/kg toluene, and 8.5 µg/kg total xylenes) that showed no consistent pattern of distribution.
- At Burn Pit B, BTEX compounds were detected in 5 of the 7 samples. The BTEX contamination, although present at relatively low concentrations (30 µg/kg benzene, 60.8 µg/kg toluene, and 9.9 µg/kg total xylenes), extends over most of the 1-acre Burn Pit B site.
- At Burn Pit C, BTEX compounds were found in 3 of the 10 samples (69.9 µg/kg toluene and 14 µg/kg total xylenes) at locations in the northern to northwestern portion of the Burn Pit C site.

5-5

			Maximum Do	etected Concen	tration in Sub	SUFIACE SOILS	
Parameter	Reference Background	Burn Pit A	Burn Pit B	Burn Pit C	Burn Pit D	Burn Pit E	Burn Pit F
		Volatila	Organic Comp	ounds (µg/kg)			<del></del>
		63	110	39.4	110		L
Acetone	0.0	28	30		12	7.9	<u> </u>
Benzene	0.0		33.1	17.1			
2-Butanone	0.0	2.4	55.1		1.4		
2-Hexanone	0.0		5	4.3	1		7.4
Methylene Chloride	0.0			48	9.7		
Tetrachloroethene	0.0	8.8	8.9	69.9	21	8.8	5.5
Toluene	0.0	12	60.8				
1,1,1 Trichloroethane	0.0			10	22		
Xylenes, Total	0.0	8.5	9.9	14			
76/10/102, 10/10			RCRA Metals		2 21	0.2	3.3
Arsenic	8.04	0.5	1.4	0.39	3.1	27	125
Barium	17.0	6.5	11.6	13.9	15.6		
A REAL PROPERTY AND A REAL	0.24			0.12	0.54	12	65.8
Cadmium	11.6	9	13	9.6	11.1		26.9
Chromium	11.1	12	8.5	21.9	11	5.9	0.051
Lead	0.05	0.04	0.06	0.07	0.063	0.08	0.051
Mercury	1.12	+					
Selenium	0.46	+				0.19	2.6
Silver		3210	546	9770	2400	824	279
Total Organic Carbon	n <u>2200</u>	3410					

## Table 5-3. Summary of Analytical Results for Subsurface Soil Samples Burn Pits, Fort Stewart

NOTE: A blank indicates analyte not detected at that Burn Pit site. NOTE: BOLD font indicates value exceeds reference background criteria for Resource Conservation and Recovery Act (RCRA) metals.

- At Burn Pit D, BTEX compounds were reported in 3 of the 4 Phase I boreholes around the perimeter of the site, but were not detected in any of the 5 Phase II samples, including the 3 soil borings in the middle of the source area. Therefore, BTEX contamination is not considered to be present at Burn Pit D.
- At Burn Pit E, toluene was detected in 4 of the 5 Phase II samples, but at concentrations lower than the average detection limit for toluene and may, therefore, not be indicative of site-related contamination. Benzene, the only BTEX compound found in any of the Phase I subsurface soil samples at Burn Pit E, was not found in any Phase II sample. Therefore, BTEX contamination is not considered to be present at Burn Pit E.
- At Burn Pit F, toluene was detected in 2 of the 7 subsurface soil samples, but at concentrations lower than the average detection limit for toluene. As discussed in Section 5.4, BTEX contamination was found in groundwater at Burn Pit F, even though it was not found at significant concentrations in subsurface soils.

Two chlorinated organic compounds (PCE and 1,1,1-trichloroethane) were reported in subsurface soil at four of the Burn Pit sites. PCE was detected at Burn Pits A, B, and D at concentrations less than 10  $\mu$ g/kg in Phase I samples only. At Burn Pit C, both PCE (maximum 48  $\mu$ g/kg) and 1,1,1-trichloroethane (maximum 10  $\mu$ g/kg) were detected in Phase I subsurface soil samples. However, neither compound was detected in any Phase II sample at any of the Burn Pit sites, including any sample taken from borings in the middle of the burn areas. In addition, as discussed below, neither compound was detected above levels of concern in groundwater at any of the six sites. Therefore, PCE and 1,1,1-trichloroethane are not considered related to releases which may have occurred from activities conducted at the Burn Pit sites.

Acetone, 2-butanone, 2-hexanone, and methylene chloride were detected in subsurface soil samples from most of the six Burn Pit sites. However, these constituents were found sporadically and at concentrations near those found in reference background samples and are not considered related to contaminant releases at the Burn Pit sites. These constituents are common laboratory contaminants.

**RCRA metals.** Metals are present in subsurface soils at each of the Burn Pit sites at concentrations exceeding their respective reference background concentrations. However, at most of the Burn Pit sites, the metal contamination does not appear to be site related. The maximum concentrations at most sites are generally less than twice background, and the metal contamination is not present in a consistent distribution in subsurface soils across the various sites. The most significant metal contamination was found at Burn Pits C and F, predominantly barium, lead, and chromium.

It should be pointed out that barium detections in subsurface soils are not overly high for natural soil conditions. Barium (maximum 125 mg/kg) in subsurface soils is within the reasonable range for barium (420 mg/kg) in soils in the eastern United States (USGS 1984), as listed in Appendix E. Lead in subsurface soils at Burn Pit C (maximum 21.9 mg/kg) and at Burn Pit F (maximum 26.9 mg/kg) exceeds the U.S. Geological Service (USGS) reference value (17 mg/kg) for lead in soils in the eastern United States. Similarly, chromium (65.8 mg/kg) in subsurface soils at Burn Pit F exceeds the USGS reference value (52 mg/kg).

- At Burn Pit A, lead (12 mg/kg) was reported in a single sample at a concentration only slightly above background. Lead is not present in a consistent pattern and is not considered site related.
- At Burn Pit B, chromium (13 mg/kg) was reported in one sample and mercury (0.06 mg/kg) in another sample at concentrations only slightly above background. These metals are not present in a consistent pattern and are not considered site related.
- At Burn Pit C, lead and mercury were found at concentrations above background in the soil borings made in the middle of the site and extend to the northeast. Maximum values of lead (21.9 mg/kg) and mercury (0.07 mg/kg) were found at Burn Pit C. Lead and mercury were also found at concentrations above background in surface soils at Burn Pit C.
- At Burn Pit D, although two metals were reported at concentrations above their reference background value, their distribution is not present in a consistent pattern. Maximum values of cadmium (0.54 mg/kg) and mercury (0.063) were reported at three different sample locations around the perimeter of the site only and, therefore, are not considered site related.
- At Burn Pit E, barium was detected at concentrations above background in one of the 9 subsurface soil samples; the maximum value of barium (27 mg/kg) was reported at MW-1, which is located more than 300 feet downgradient from the source area. Chromium and mercury, which were detected at concentrations only slightly above background, do not show a consistent pattern of distribution and are not considered site related.
- At Burn Pit F, five metals were reported in subsurface soils at concentrations above their reference background value. Barium was detected at concentrations above background in 3 of the 7 subsurface soil samples, with a maximum value (125 mg/kg) at SB-3 in the western portion of the source area. Chromium was also found in 3 of the 7 samples, with a maximum concentration of 65.8 mg/kg, at SB-3. Chromium was also found at

concentrations above background in surface soils at Burn Pit F. Other metals detected above background in subsurface soils at Burn Pit F, but with less consistency, include lead, mercury, and silver.

## 5.4 GROUNDWATER CONTAMINATION

A total of 26 groundwater samples were collected from both existing Phase I and new Phase II monitoring wells. The samples were analyzed for VOCs and RCRA metals. Both filtered and unfiltered water samples were collected; only the total metal analysis on unfiltered water samples is presented in this section. Table 5-4 summarizes the analytical results for groundwater samples. This assessment presents Phase II contaminant data only, because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

			Maximut	n Detected	Concentrat	ions in Gr	oundwate	r
Parameter	Reference Background	Federal MCL	Burn Pit A	Burn Pit B	Burn Pit C	Burn Pit D	Burn Pit E	Burn Pit
J Al Allictel		Volatile	Organic Con	ipounds (µg	/L)			
Acetone	0.0						53.4	(27
Ethylbenzene	0.0	700					1.1	<u>637</u> 33.5
Styrene	0.0	100					1.0	33.5
Tetrachloroethene	0.0	5					1.0	3170
Xylenes, Total	0.0	10000					6.3	5170
Jtyteneo, ret_	1		RCRA Metals	<u>(μg/L)</u>			<u></u>	r
Arsenic	3.02	50	1.5		2.5	4.2		40.1
	71.72	2000	97.5	73.4	134	99.6	17.7	49.1
Barium	0.43	5	0.36	1.2	3	0.43	4.3	1.1
Cadmium		100	2.4	5.9		2.9	1.7_	2.8
Chromium	3.56		3.3	2.2	1.1	1.2		0.08
Lead	4.69	15		2.2	0.28			
Mercury	0.14	2	0.22	<u></u>		1.6	0.43	
Selenium	1.90	50	0.7	3.5	1.6			0.1
Silver	1.12		1.6	4.9	0.77	3.6	0.16	

#### Table 5-4. Summary of Analytical Results for Groundwater Samples **Burn Pits, Fort Stewart**

NOTE: A blank indicates analyte not detected in that medium.

NOTE: BOLD font indicates value exceeds reference background criteria for Resource Conservation and Recovery Act (RCRA) metals.

MCL = Maximum Contaminant Level.

VOCs. No VOC contamination was found in groundwater at Burn Pits A, B, C, or D. BTEX compounds were detected at both Burn Pits E and F, PCE and acetone were detected at Burn Pit E, and styrene was detected at Burn Pit F.

- At Burn Pit E, low concentrations of BTEX compounds were found in a single well (MW-4). Ethylbenzene (1.1  $\mu$ g/L) and xylenes (6.3  $\mu$ g/L) were reported at concentrations only slightly above their detection limits. PCE was detected at 1.0  $\mu$ g/L, which is less than its average detection limit in background samples. Acetone was detected in 2 wells at a maximum of 53.4  $\mu$ g/L. No VOCs were reported in Phase I groundwater.
- At Burn Pit F, two BTEX compounds (ethylbenzene at 637  $\mu$ g/L and xylenes at 3170  $\mu$ g/L) were found in groundwater at MW-4. These concentrations are less than the respective MCLs for ethylbenzene (700  $\mu$ g/L) and xylenes (10,000  $\mu$ g/L). In the Phase I sample

results, both ethylbenzene (2,800  $\mu$ g/L) and xylenes (15,800  $\mu$ g/L) were reported at much higher concentrations, and both benzene (22,700  $\mu$ g/L) and toluene (41,100  $\mu$ g/L) were reported above their MCLs. These results suggest that the high concentrations of BTEX present in groundwater at the site may be attenuating naturally, either through dispersion, volatilization, or biodegradation. Styrene was also observed in MW-4 at 33.5  $\mu$ g/L, and is considered a secondary contaminant in the primary BTEX plume.

The source and extent of the BTEX contamination is unknown. MW-4 is located hydraulically higher than other wells at the Burn Pit F site and is considered side-gradient to groundwater flow at the site. BTEX compounds were not detected at significant concentrations in the soil borings within the probable source area and were not detected in the remaining (downgradient) wells.

**RCRA metals.** Metals are present in groundwater at each of the Burn Pit sites at concentrations exceeding their respective reference background concentrations. However, none of the metal concentrations exceed their respective MCLs at any of the Burn Pit sites. In addition, metals in groundwater were found at significantly lower concentrations than reported in the Phase I RFI Report, suggesting that redevelopment of the wells and use of low-flow sampling equipment to reduce turbidity were successful.

Barium and cadmium are notable in that they exceeded background at more than half the sites. The broad presence of barium and cadmium may indicate that they are of natural origin, rather than site-related contamination. Given the limited background data, the reference background concentrations of these metals may not be representative of the range of naturally occurring background conditions.

- At Burn Pit A, barium was reported in a single downgradient well at a maximum concentration of 97.5  $\mu$ g/L. Mercury (0.22  $\mu$ g/L) and silver (1.6  $\mu$ g/L) were detected in separate wells at concentrations only slightly above background and, therefore, do not indicate a pattern of distribution indicating a potential release of these metals from the former burn pit.
- At Burn Pit B, five metals were reported at concentrations above background in groundwater. Barium (73.4 µg/L), cadmium (1.2 µg/L), chromium (5.9 µg/L), selenium (3.5 µg/L), and silver (4.9 µg/L) were detected at concentrations only slightly exceeding background and in separate wells. These metals do not indicate a pattern of distribution indicating a potential release from the former burn pit.
- At Burn Pit C, three metals were reported at concentrations above background in groundwater. Barium (134 µg/L) and cadmium (3 µg/L) were detected in one well. Mercury was detected in a second, MW-7, but at a concentration (0.28 µg/L), that is only slightly above background. Well MW-7 is considered upgradient of the site; however, as discussed in Section 11.0, water table contours suggest that the water table may potentially be mounded beneath the Burn Pit C site.
- At Burn Pit D, three metals were reported at concentrations above background in different wells. Arsenic (4.2 µg/L) was reported in one well, and barium (99.6 µg/L) and silver (3.6 µg/L) were reported in another well. Because these metals were detected in separate wells at concentrations only slightly above background and do not indicate a pattern of distribution indicating a potential release of these metals from the former burn pit, they are not considered site related.

- At Burn Pit E, cadmium was the only RCRA metal detected above background in groundwater. Cadmium was found in two of the three downgradient wells at a maximum of 4.3 µg/L.
- At Burn Pit F, cadmium was again the only RCRA metal detected above its reference background concentration in groundwater. Cadmium was detected at a maximum of  $1.1 \mu g/L$  at well MW-1. Well MW-1 is considered upgradient of the site; however, as discussed in Section 11.0, water table contours suggest that the water table may potentially be mounded beneath the Burn Pit F site.

## 5.5 SUMMARY OF NATURE AND EXTENT OF CONTAMINATION

The following summarizes the major findings of the Phase II sampling and analysis:

- SVOCs were found only in a single surface soil sample at low concentrations at Burn Pit E.
- Metals found above background in surface soils include arsenic, barium, cadmium, lead, and mercury in the northern and eastern portions of Burn Pit A; lead and mercury at Burn Pit B; barium, cadmium, lead, and mercury in the center of Burn Pit C; barium and mercury at Burn Pit E; and arsenic, chromium, and mercury in the northern and northeastern portion of Burn Pit F. With the exception of lead at Burn Pit C, these metals are well within the regional background range for metals in soils and may be indicative of natural soil conditions.
- BTEX compounds were found in subsurface soils at low concentrations at all six Burn Pit sites. A consistent pattern of distribution is evident only at Burn Pit B (whole site) and Burn Pit C (northern and northwestern portion of the site), although at low concentrations.
- Metals in subsurface soils were generally found at concentrations less than twice background and in no consistent pattern of distribution. Notable presence of metals in subsurface soils is characterized by lead and mercury in the central and northeastern portion of Burn Pit C and five metals (predominantly barium, lead, and chromium) throughout the Burn Pit F site.
- In groundwater, the highest levels of BTEX contamination were found in a single well located west (side-gradient) of the Burn Pit F site. Concentrations of BTEX were less than their respective MCLs and less than reported Phase I concentrations, suggesting that natural attenuation may be occurring. The source of the BTEX contamination is unknown; BTEX was not found at high concentrations in soils at Burn Pit F.
- Metals in groundwater were found at all six sites at concentrations less than their respective MCLs and much less than reported Phase I concentrations, suggesting that redevelopment of the wells and use of low-flow sampling equipment to reduce turbidity were successful.
- Metals found in groundwater are characterized by barium, mercury, and silver at Burn Pit A; barium, cadmium, chromium, selenium, and silver at Burn Pit B; barium, cadmium, and mercury at Burn Pit C; arsenic, barium, and silver at Burn Pit D; and cadmium at Burn Pits E and F.

A summary of the SRCs by medium is presented in Table 5-5. SRCs include all organics that are detected and inorganics detected above reference background criteria. These SRCs are carried through for evaluation under fate and transport, human health preliminary risk evaluation (PRE), and ecological PRE.

C

Ć

							1			Ē	0.410 mm		Build	Rurn Pit F.	6	Bu	<b>Burn Pit</b>	Ē
		Down Dit A	<	ď	Rurn Pit B	~~	Bu	Burn Pit C	5	ā	ILU EW							
	ă U U U		m <sup>2</sup>	SS	SB	<u>G</u> W	SS	SB	GW	SS	SB	M	SS	B	3	20	3	ؤ
	8	<b>,</b>		3	}			×			×				×			
Acetone		×			< ;						×			×				
Renzene		×			×	Ī					\$							
2 Duratorie		×		-	×			×					-		>			×
Z-Dullatonic															<			
Ethylbenzene											Х			"				
2-Hexanone					;	T		>									Х	
Methylene chloride					×			<										х
Styrene								÷			^				×			
Tetrachloroethene		×			×			<b>-</b>						×			Х	
Tolnene		×			×			X			<							
1 0 1 1 T-1-L1-conthone								Х							>			>
1,1,1-1 Include liame	-	>			×			X			X				<			ł
Xylenes, total													×					
Benzo(a)pyrene													×					
Benzo(a) flouranthene													×					
Flouranthene	-											×				×		
Arsenic	×						; ;		\$			: >	×	×			×	
Barium	×		×			×	× ;		< >		<b> </b> >	{	(	1	×			×
Cadmium	x					×	×		<		<	_		×		×	×	
Chrominm					×	×											×	
T and	×	×		×			×	×			;		>	>		×	×	  _
Manuel	×		×	×	X		×	×	×		×		<	<			   	
Mercury		-				×											>	
Selenum	$\left  \right $	-	>			×						×					<	
Silver			< -			<								1				

X = Site-related constituent (SRC)
 SS = Surface soil
 SB = Subsurface soil
 GW = Groundwater

## 6.0 CONTAMINANT FATE AND TRANSPORT

#### **6.1 INTRODUCTION**

This section describes the potential migration pathways and mechanisms for transport of chemical substances found in surface and subsurface soils and groundwater at the Burn Pits (SWMUs 4A-4F). Based on the information presented in previous sections, the refined site conceptual model is developed in this section. Simple analytical methods were used to define contaminant movement from source areas to receptor locations. The overall objectives of these analyses are to evaluate potential future impact to human health and the environment.

Section 6.2 discusses the persistence, mobility, and other physical and chemical properties of the organics and metals found at the Burn Pits. Section 6.3 presents a conceptual model for potential contaminant migration pathways and describes contaminant release mechanisms through primary transport media (groundwater). Section 6.4 discusses the fate and transport of the contaminants at the Burn Pits with respect to their leachability and natural attenuation in the groundwater. Section 6.5 summarizes the conclusions drawn from the results of the analyses and discusses the uncertainties associated with the analyses.

## 6.2 PHYSICAL AND CHEMICAL PROPERTIES

The fate and transport of organic compounds and metals are functions of both site characteristics and the physical/chemical properties of the contaminants. Such properties include solubility in water, tendency to transform or degrade (usually described by a half-life or an environmental half-life in a given media), and chemical affinity for solids or organic matter (usually described by a partitioning coefficient K<sub>d</sub>, K<sub>oc</sub>, or K<sub>ow</sub>). These properties and how they affect inorganic and organic contaminant behavior are described below.

#### 6.2.1 Metals

Inorganic SRCs for soils at the Burn Pits site include arsenic, barium, cadmium, chromium, lead, mercury, and silver. These metals are subject to movement with soil moisture, and may be transported through the vadose zone to groundwater. Metals do not degrade, although some metals can transform to other oxidation states in soil, reducing their mobility and toxicity. Metals also react with soils or other solid surfaces by ion exchange, adsorption, precipitation, or complexation. Such reactions are affected by pH, oxidation-reduction conditions, and the type and amount of organic matter, clay, and hydrous oxides present. In general, these reactions are reversible and cause an element's mobility to be retarded. The retardation factor (R<sub>d</sub>) describes, numerically, the extent to which the velocity of the contaminant, relative to water, is slowed. The  $R_d$  is largely derived from the partitioning coefficient ( $K_d$ ), expressed by the following relation:

$$R_d = 1 + K_d \cdot \rho_h / \theta$$

where:  $\rho_b$  = the soil bulk density (g/cm<sup>3</sup>),  $\theta$  = soil moisture content.

 $K_d$  for the metals at the Burn Pits may vary by large ranges. It has been found that  $K_d$  can even vary by orders of magnitude between samples from the same site. The range of measured  $K_d$ 

97-160P(DOC)/020899

values [obtained from EPA (1996)] and the corresponding range of calculated  $R_d$  for the SRCs are presented in Table 6-1.

Site-related Analytes	K <sub>d</sub> Range <sup>a</sup> (mL/kg)	R <sub>d</sub> Range <sup>b</sup>		
	25 to 31	60 to 75		
Arsenic	11 to 52	27 to 126		
Barium	15 to 4300	37 to 10,320		
Cadmium	14 to 31	35 to 75		
Chromium	19 to 1405	25 to 3370		
Lead		1.1 to 480		
Mercury	0.04 to 200	6.3 to 44		
Selenium	2.2 to 18	1.2 to 265		
Silver	0.1 to 110	1.2 10 205		

## Table 6-1. Distribution Coefficients (Kd) and Retardation Factors (Rd) Burn Pits, Fort Stewart

"The K<sub>d</sub>-ranges represent the pH-dependent values for metals developed for soil screening level application (EPA 1996).

(EFA 1220). <sup>b</sup> The R<sub>d</sub>-ranges represent calculated values using the K<sub>d</sub>-range and site-specific parameters.

#### 6.2.2 Organic Compounds

The organic compounds detected in soils at the Burn Pits are primarily VOCs, with the exception in Burn Pit E, where three SVOCs were detected in a single Phase II surface soil sample. These compounds may be degraded in the environment by various processes, including hydrolysis, oxidation/reduction, photolysis, or biodegradation. Half-lives of organic compounds in various media can vary from minutes to years, depending on the chemical and environmental conditions. Degradation may either enhance or reduce the toxicity of contaminants if daughter products have different toxicological effects than the parents of a chemical. The biodegradation rates for relevant organic compounds are presented in Table 6-2. These values are based on the biodegradation half-lives taken from the *Handbook of Environmental Degradation Rates* (Howard et al. 1991). Although a range of values is presented in this book, only the lowest biodegradation rates corresponding to the highest half-lives are presented here to ensure conservatism in discussing contaminant loss through degradation/decay.

The mobility of an organic compound is affected by its volatility and its partitioning behavior between solids and water, water solubility, and concentration. The Henry's Law constant value  $(K_H)$  for a compound is a measure of the ratio of the compound's vapor pressure to its aqueous solubility. The  $K_H$  value can be used to make general predictions about the compound's tendency to volatilize from water. Substances with  $K_H$  values less than 10<sup>-7</sup> atm/m<sup>3</sup>/mol will generally volatilize slowly, while compounds with  $K_H$  greater than 10<sup>-3</sup> atm/m<sup>3</sup>/mol will volatilize rapidly. Vapor pressure is a measure of the pressure at which a compound and its vapor are in equilibrium. The value can be used to determine the extent to which a compound would travel in air, as well as the rate of volatilization from soils and solution. In general, compounds with vapor pressures lower than 10<sup>-7</sup> mm Hg will not be present in the atmosphere or soil air in significant amounts, while compounds with vapor pressures higher than 10<sup>-2</sup> mm Hg will exist primarily in the air. Unless the soil is saturated, VOCs will exist primarily in the atmosphere and soil air. Polyaromatic hydrocarbons (PAHs) and other SVOCs will exist in both the air and the soil. The air diffusion coefficient is a measure of the rate of spontaneous mixing, presented in units of cm<sup>2</sup>/second, of one substance with another when in contact or separated by a permeable Table 6-2. Physical and Chemical Properties of Organic Site-Related Chemicals at Burn Pits, Fort Stewart

Constituents	Mol. Wt	Solubility S <sub>w</sub> (mg/L)	S* @ Temp.	K <sub>ow</sub> (mL/mL)	Vapor Pressure (torr @ °C)	Vapor     Henry's     Kh       Pressure     Constant (Kh)     @Temp.       (torr @ °C)     (atmem³/mol)     (°C)	Kh @Temp. (°C)	Alr Diff. Coeff. (cm/s)	K <sub>oc</sub> ('mL/g)	Calculated K <sub>d</sub> (mL/g)	Biodegradation Rate Constant A (1/day)	Log (Kow)
Acetone						omme organic compounds	SL					
	1.80	1.00E+06		5.75E-01	270 @ 30	5.14F-07	1 23 C	19110				<u></u>
Benzene	78.1	1.78E+03	20	1 35ETU2	05 @ 26		3	11.0	5/5.0	1.61E-03	2.48E-02	-0.24
2-Butanone	72.1	2 75E+05		201.0001	(7 m) Cc	3.35E-03	25	$0.0932^{b}$	62 <sup>c</sup>	1.74E-01	9.63E-04	212
Ethylbenzene	106.2			1.625+00	100@25	6.61E-07	25 <sup>a</sup>	$0.092^{b}$	1.15	3.21F-03	10 201 6	
2. Uovor 2.		1.325702	70	1.41E+03	10 @ 25.9	6.44E-03	25	0.075	3046		70-704-7	07.0
2-ricxanone	100.2	3.50E+04	20	2.40E+01	2@20	7 520 06		20.0	<b>1</b> 0	2./1E-UI	3.04E-03	3,15
Methylene chloride	84.9	1.67E+04	25	1.78E+01	170 0 15	00-3001	70	0.078	15.1	4.23E-02		138
Styrene	104.1	3.00E+02	20	1 ASELO2		0.19E-03	52	$0.1037^{b}$	10	2.80E-02	6.19E-03	175
Tetrachloroethene	165.8		2	CUTUCE.	2 @ 20	2.28E-03		0.0071	9126	2.55E+00	3 30F 02	
Toluce		1-306+02	3	3.39E+02	19 @ 25	2.87E-02	35	94400	2220		ch-ancir	3.10
1 oluene	92.1	5.15E+02	20	4.90E+02	28 @ 25	5 070 03		1100	67	7.42E-01	4.19E-04	2.53
I, I, I - Trichloroethane	133.4	4.40E+03	00	2 05 2 02	100 0 10	J.92E-U3	្ន	0.087	140°	3.92E-01	3.30E-03	2 69
Xylene	106.2	2 00E+02		201710212	57 m ni	4.08E-03	25	0.019	135	3.78E-01	6.35E_04	<u>- 4 - 6</u>
		1017001		201240.0	5 @ 20	5.25E-03	25	0.073	196	5.49E-01	1 015 01	
					Semivolatile	Semivolatile Organic Compounds	spund				C0-366-1	1
Benzo(a)pyrene	252.3	3.80E-03	25	9 558405	55.0 @ 21	1 001 00						
Benzo(b)fluoranthene	252.3	1.00E-03		- 1		4.90E-07	25	0.043 968,774	68,774°	2.71E+03	3.27E-04	5 08
Fluoranthene	202.3	2 65E-01	36		1 @_/-ac	2.94E-07	25ª	0.044" 2,340,000	340,000	6.55E+03	2 84F.04	2 2 4
				4014C	0.0-35	6.50E-06	25	0.069 4	49,100 <sup>c</sup>	1.37E+02	3 946-04	
Solubilities, Henry's Constant, and I on (V ) house the	int and I o	a (K. ) have been	6 								10.71.7	

Solubilities, Henry's Constant, and Log (K<sub>w</sub>) have been taken from RREL Treatability Data Base (EPA 1994) except as otherwise indicated. Biodegrdation half-lives are based on biodegradation half-lives taken from <u>Hand Book of Environmental Degradation Rates</u> (Howard et. al. 1991) except as otherwise indicated. Air diffusion coefficients are obtained from EPA 1987, except as otherwise indicated. "STF Data Base (EPA 1991). <sup>a</sup>Indicates Shen et. al. 1993 as the source. "Measured K<sub>w</sub> values (EPA 1996). "Source from EPA 1995. "EPA (1996). "Measured K<sub>w</sub> values (EPA 1996). "Source from EPA 1995. "EPA (1996). K<sub>d</sub> = K<sub>w</sub> \* f<sub>w</sub>, where f<sub>w</sub> is fraction of organic carbon content with a site average value of 0.028 obtained based on measurements from the six Burn pits.

97-160P(DOC)/020899

64

membrane. The rate of diffusion is proportional to the concentration gradient of a substance, increases with temperature, and is inversely related to density and pressure. In soil systems, the principal type of diffusion is from a region of high concentration to a region of low concentration. Diffusion occurs most readily in gases, to a lesser extent in liquids, and least in solids.

Water solubility and the tendency to adsorb to particles or organic matter can correlate with retardation in groundwater transport. The adsorption coefficient/partition coefficient ( $K_d$ ) of an organic compound is related to the organic carbon/water partition coefficient ( $K_{oc}$ ) by

$$K_d = f_{oc} \times K_{oc}$$

where:

 $f_{oc}$  = fraction of soil organic carbon content.

Chemical-specific  $K_{oc}$  values may be obtained from literature or may be calculated using empirical formulas relating the octanol-water partitioning coefficient ( $K_{ow}$ ) to the  $K_{oc}$ . The  $K_{ow}$ (mL/mL) is the ratio of a contaminant's concentrations in a system containing water and octanol. The most commonly used formula to relate  $K_{ow}$  to  $K_{oc}$  is given by (Mills et al. 1985):

$$K_{oc} = 0.63 \times K_{ow}$$

Chemicals with relatively high water solubilities and low adsorption coefficients (e.g., acetone, methylene chloride, etc.) are expected to remain primarily as dissolved phases and be transported at the same rate as the groundwater flow. Chemicals with lower water solubilities and higher adsorption coefficients (e.g., PAHs) are expected to remain primarily adsorbed to the surface of the soils; their transportation with the groundwater would be very limited and at a much slower rate. Table 6-2 presents the solubility, Henry's Law constant ( $K_H$ ), vapor pressure, air diffusion coefficients, and biodegradation rate constants for the organic compounds detected in soils and groundwater at the Burn Pits. Log  $K_{ow}$ ,  $K_{ow}$ ,  $K_{oc}$ , and  $K_d$  for these compounds are also presented in this table.

#### 6.3 CONCEPTUAL SITE MODEL

The CSM is a statement of expected site conditions that serves as a paradigm against which observations can be compared and within which predictions can be made. The predictive function of the CSM, of primary importance to contaminant fate and transport analysis, relies on known information and informed assumptions about the site. The better the information and the greater the accuracy of the assumptions, the more accurately the CSM describes the site.

The CSM presented in this section summarizes the hydrogeologic components (presented in Section 4.0) and the distribution of contaminants in the subsurface soils and groundwater (presented in Section 5.0). Contaminant migration pathways and release mechanisms are also based on the information presented in Section 5.0. The CSM for contaminant fate and transport at the Burn Pits is diagrammatically illustrated in Figure 6-1. The summary of the model elements follows.



66

(

## 6.3.1 Water Balance Components

The potential for contaminant transport begins with precipitation. The actual amount of rainwater available for flow is highly variable and dependent upon soil type and climatic conditions. A water balance calculation can be used as a tool to quantitatively account for all the components of the hydrologic cycle at the Burn Pits. The components of a simple steady-state water balance model include: precipitation (P), evapotranspiration (ET), surface runoff (Sr), and groundwater recharge or percolation (Gr), and is defined as follows:

$$P = ET + Sr + Gr$$

or

Rain water available for flow = Sr + Gr = P - ET

The annual average water balance estimates for the Fort Stewart area indicate an evapotranspiration of 65.5 percent (31.4 inches) of total precipitation (48 inches) as compared to 34.5 percent (16.4 inches) for rain water available for flow. Of this 34.5 percent (14.7 inches), groundwater recharge (percolation) accounts for 30.7 percent (14.7 inches) and surface runoff accounts for the remaining 3.8 percent (1.8 inches). The water balance estimations were based on Hydrologic Evaluation of Landfill Performance model (EPA 1994) calculations using precipitation and temperature data for the years 1974 through 1978 at Savannah, Georgia.

## 6.3.2 Contaminant Release Mechanisms and Migration Pathways

Past pathways were:

- The burn pits were not actually "pits," but surface piles of clearing and grubbing debris that could have been doused with gasoline and burned. Some of these liquids might have infiltrated the soil and contaminated soil and groundwater below the individual Burn Pits.
- Organic contamination in groundwater has spread and/or been degraded to reach its current extent.
- Residues from burning have left some metals at concentrations higher than reference background criteria in surface and subsurface soils.
- If past operations at Burn Pit F have contaminated soils with BTEX compounds, then those contaminants have degraded to negligible concentrations in the mean time.

Current pathways are:

- Percolating rainwater through contaminated soil below the Burn Pits leaches contaminants and transports contaminants to the water table.
- Contaminants migrate in groundwater along the flow path by advection toward the downgradient receiving streams.
- Organic compounds in groundwater and probably in soil are being biologically degraded.
- Organic compounds in soil and probably in groundwater are being volatilized.

97-160P(DOC)/020899

**Release Mechanisms.** The potential release mechanism at the Burn Pit sites is infiltration of rain water with leaching to groundwater. Precipitation that does not leave the site as surface runoff infiltrates into the subsurface. The rate of percolation is controlled by soil cover, ground slope, saturated conductivity of the soil, and meteorological conditions. As discussed previously, the rate of percolation at these sites is quite high (14.7 inches/year).

Factors that affect the leaching rate include a contaminant's solubility and partitioning coefficient  $(K_d)$  and the amount of percolation. Insoluble compounds will precipitate out of solution in the subsurface or remain in their insoluble form with little leaching. Those contaminants with a small  $K_d$  (lower sorption) will be leached more effectively than those with a larger  $K_d$  (higher sorption).

A factor that affects the persistence of a contaminant is the contaminant's rate of decay. Most of the organic compounds decay or break down at characteristic rates that are described by the substance's half-life. For a given percolation rate, those contaminants with long half-lives have a greater potential for contaminants groundwater than do those contaminants with shorter half-lives. Organic contaminants with shorter half-lives and higher  $K_{oc}$  values will be completely degraded before reaching the water table.

Release by gaseous emission and airborne particulates is not significant at the Burn Pits. Concentrations of organic contaminants in the surface soil are relatively insignificant so gaseous emissions to the atmosphere would be minor. Wind erosion is not likely to be significant at the Burn Pits because of the vegetative cover.

Migration Pathways. The most likely pathway of contaminant migration from the Burn Pits is via groundwater discharge to downgradient streams. Groundwater from Burn Pits A through D flows toward Mill Creek and Horse Creek tributaries that, in turn, discharge into Taylors Creek. Groundwater from Burn Pit E discharges to a small unnamed stream or swampy area that discharges directly to the Canoochee River. Groundwater from Burn Pit F discharges into Goshen Swamp, which in turn discharges southward into Peacock Creek.

In the saturated zone, the contaminants are carried either in solution or adsorbed to fine particulates (colloids) laterally to the hypothetical receptor locations. The horizontal hydraulic conductivity, which controls the flow rate, is a function of soil grain size and the pressure gradient. Saturated hydraulic conductivities for the Burn Pit sites range from  $3.1 \times 10^{-4}$  to  $1.7 \times 10^{-3}$  cm/second with an overall average of  $9.2 \times 10^{-4}$  cm/second (Section 4.0). The horizontal hydraulic gradient varies from 0.0006 foot/foot across the Burn Pit B site to 0.009 foot/foot across the Burn Pit E with groundwater flow occurring in different directions from the individual sites. The calculated groundwater seepage velocity with these gradients and hydraulic conductivities range from 1.2 feet/year (Burn Pit B) to 35 feet/year (Burn Pit A) using the effective porosity measurements in Table 4-1. Therefore, the arrival time for the site groundwater to reach the nearest receptor location is expected to range from 43 years (for Burn Pit A) to 3750 years (for Burn Pit B). These ranges are summarized in Table 6-3.

## 6.4 FATE AND TRANSPORT ANALYSIS

#### 6.4.1 Soil Leachability Analysis

Contaminant fate and transport analysis at this site involves a series of screening steps to define the contaminant migration constituents of potential concern (CMCOPCs). The CMCOPCs are defined as the constituents that may pose potential problems in groundwater if they were to

97-160P(DOC)/020899

migrate from the site source. The CMCOPCs were identified through a screening process, using a simple analytical approach for vertical migration from source soils to groundwater and lateral migration in groundwater from the source site to the nearest receptor stream. The screening steps are discussed in the following sections.

Cit-	Hydraulic Conductivity, k (cm/second)	Horizontal Hydraulic Gradient, i (feet/feet)	Effective Porosity, n	Groundwater Flow Velocity, v (feet/year)	Distance to Receptor Stream (feet)	Receptor Stream	Arrival Time (years)
Site			0.42	35	1500	Mill Creek	43
Burn Pit A	1.73E-03	0.0083		1.2	4500	Horse Creek	3750
Burn Pit B	7.07E-04	0.0006	0.365			Mill Creek	284
Burn Pit C	1.24E-03	0.0043	0.396	14.1	4000		
	9.45E-04	0.0015	0.368	4.0	2500	Mill Creek	625
Bum Pit D			0.464	6.2	1000	Canoochee R	161
Burn Pit E	3.11E-04	0.009			1000	Goshen	294
Burn Pit F	5.81E-04	0.0023	0.408	3.4	1000	Swamp	

## Table 6-3. Arrival Time Estimates for Contaminant Migration in Groundwater Burn Pits, Fort Stewart

The first step of the screening process represents the development of the SRCs. The SRCs were selected by comparing the maximum detected concentrations of all the analytes measured in surface and subsurface soils with their respective reference background concentration. The reference background concentrations represent the average site background concentration multiplied by a factor of two. If the maximum concentration of an analyte in the soil exceeds its reference background concentration, then that analyte is selected as an SRC.

The second step of the screening process involves comparing the maximum concentrations of all the SRCs, developed in the previous step, with EPA generic soil screening levels (GSSLs). The GSSLs are set for Superfund Sites for the migration to the groundwater pathway (EPA 1996). For conservatism, a default dilution attenuation factor (DAF) of 1, as applicable based on source area and depth to water table, was used to select the GSSLs for the organic chemicals. A DAF of 1 is appropriate for organic chemicals because organic constituents are not easily adsorbed to the sandy inorganic soils present at the Burn Pits and because the depth to the water table is less than 13 feet. However, for the metals, because of their higher retardation factor, a DAF of 20 was used to select GSSLs. The GSSL is defined as the concentration of a contaminant in soil that represents a level of contamination below which there is no concern under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), provided conditions associated with soil screening levels (SSLs) are met. Generally, if contaminant concentrations in soil fall below the GSSL, and there are no significant receptors of concern, then no further study or action is warranted for that area. However, it should be noted here that the purpose of this screen is not to identify the contaminants that may pose risk at a downgradient location, but to target those contaminants that may pose a potential problem if they were to migrate from the site. The results of this screening for individual Burn Pits are presented in Sections 9.0 through 14.0. The resulting CMCOPCs are summarized in Table 6-4. The predicted maximum concentration of a contaminant in groundwater is calculated as follows (EPA 1996):

$$C_{\text{GW}(P)} = \frac{C_{\star}}{\text{GSSL}} \times MCL$$

where:

$C_{GW(P)}$	.=	predicted maximum concentration in groundwater,
C,	=	observed maximum concentration in groundwater,
GSSL	=	generic soil screening level,
MCL	=	maximum contaminant level.

# Table 6-4. CMCOPCs in Soil and Predicted Maximum Concentrations in Groundwater Burn Pits, Fort Stewart

Site	Parameter	Maximum Observed Concentration in Soil <sup>a</sup>	GSSL⁴	MCL (µg/L)	Predicted Maximum Concentration in Groundwater (µg/L)
Burn Pit A	Benzene	28	2	5	
	Tetrachloroethene	8.8	3		70
Burn Pit B	Benzene	30	2		15
	Tetrachloroethene	8.9	3	5	75
	Methylene Chloride	5		5	15
Burn Pit C	Tetrachloroethene	48		5	25
	Methylene Chloride		3	5	80
Burn Pit D	Benzene	4.3	1	5	22
	the second se	12	2	5	30
Burn Pit E	Tetrachloroethene	9.7	3	5	16
	Benzene	7.9	2	5	20
Burn Pit F	Chromium	65.8	38	100	
	Methylene Chloride	7.4	<u> </u>	5	<u> </u>

"Concentrations for metals in mg/kg, for organics in µg/kg.

CMCOPCs = Contaminant Migration Constituents of Potential Concern. GSSL = Generic Soil Screening Level.

MCL = Maximum Contaminant Level.

## 6.4.2 Natural Attenuation of the CMCOPCs

The organic CMCOPCs in soils at the Burn Pits include benzene, methylene chloride, and PCE. These contaminants may be degraded in the environment by various processes, including hydrolysis, oxidation/reduction, photolysis, or biodegradation. As already discussed in Section 6.2, environmental half-lives of organic compounds in various media can vary from minutes to years, depending on the chemical and environmental conditions. Organic chemicals with differing chemical structures will biodegrade at different rates. Primary biodegradation consists of any biologically induced structural change in an organic chemical, while complete biodegradation is the biologically mediated degradation of an organic compound into carbon dioxide, water, oxygen, and other metabolic inorganic products. The biodegradation rate of an organic compound is proportional to the concentration:

$$-dC/dt = kC^n$$

where:

C = concentration,

k = biodegradation rate constant = 1/t Ln (a/[a-x]),

t = time,

97-160P(DOC)/020899

- a = initial concentration,
- x = change in concentration with time,
- n = reaction order, n=1 for first order kinetics.

The half-life  $(t_{1/2} = Ln2/k)$  is the time necessary for half of the chemical concentration to react. The biodegradation rate of an organic chemical is generally dependent on the presence and population size of soil micro-organisms, capable of degrading the chemical.

Based on the above equation and the predicted maximum concentrations of these constituents if they were to migrate to groundwater, a simple first-order correlation can be obtained between the constituents' half-lives and the time required to degrade the contaminant to the concentration equal to its MCL. These correlations indicate that benzene, with its predicted maximum concentration of 75  $\mu$ g/L (using linear equilibrium relationship) in groundwater, will degrade to its MCL value in less than 4 times its half-life. Using the most conservative half-life reported in the literature, it appears that the maximum expected concentration of benzene will degrade to its MCL value in less than 10 years. This time frame is much less than the 43 years that is the shortest time expected for the site groundwater to reach the receptor location among all of the individual Burn Pit sites. Similarly, PCE with its predicted maximum concentration of 42 parts per billion (ppb) will degrade to its MCL in less than 14 years, and methylene chloride with its should be further noted here that this analysis does not account for attenuation due to adsorption and dispersion, making the analysis highly conservative.

## 6.5 SUMMARY AND CONCLUSIONS

Based on site characterization and monitoring data, some metals and VOCs were detected in soils at the Burn Pit sites. Among the metals, only chromium from Burn Pit F is considered a CMCOPC based on the contaminant transport analysis. At Burn Pit F, chromium could reach the water table at a predicted concentration exceeding its MCL. However, off-site migration of chromium would be very limited due to its high  $K_d$  (i.e., high retardation factor) as well as the slow movement of groundwater flow to the receptor location (unnamed tributary of Goshen Swamp located approximately 1000 feet west of the site).

None of the organic compounds currently observed in the site groundwater exceeds its respective MCL in groundwater. Based on contaminant transport analysis, benzene, methylene chloride, and PCE could migrate from soils to the water table at concentrations exceeding their respective MCLs; however, off-site migration of these contaminants will be very limited due to retardation and biodegradation as well as the slow movement of groundwater flow. Predicted maximum concentration of these constituents at a distance 500 feet from the source is not expected to exceed 1 ppb (i.e., less than the MCL of 5 ppb) even with the most conservative assumption for their biodegradation half-lives.

## 7.0 HUMAN HEALTH RISK ASSESSMENT

This HHRA uses a Step 1 risk evaluation approach that is based on guidance from the GEDP. This is done to determine if there are potential risks to human health associated with contamination detected at the Burn Pits (SWMUs 4A through 4F). This process involves the following steps:

- for inorganics, compare detected concentrations to naturally occurring background levels to determine if detected inorganics are naturally occurring or are associated with past activities at the site;
- identify potential migration and exposure pathways associated with the site, and identify potential exposure scenarios in order to identify appropriate action levels;
- identify available risk-based action levels for each contaminant detected above background levels or develop levels if they do not exist; and
- compare sample concentrations to action levels to determine if site conditions warrant further evaluation.

Chemicals that exceed action levels will be identified as COPCs and will be evaluated in a baseline risk assessment (BRA).

#### 7.1 DATA EVALUATION

The objective of this evaluation is to develop a set of chemical data which is suitable for use in the HHRA. The data for the Burn Pits were evaluated to establish (1) which data are of sufficient quality for use in the quantitative risk assessment and (2) which detected chemicals are believed to be site related. This is discussed in site-specific chapters.

### 7.1.1 Data Quality Evaluation

The data used in the risk assessment were verified and validated using the methodology described in the QAPP. Data qualified during the validation as rejected data ("R") were not used in the risk assessment.

Detection limits achieved during sample analysis were reviewed to ensure that the required detection limits were met. Typically, detection limit requirements are established to ensure that characterization has occurred to levels that are low enough to determine if chemicals are present at hazardous levels. These levels are chemical specific and related to each chemical's toxicity. Required detection limits are presented in the QAPP. In some cases recommended detection limits cannot be achieved by a laboratory, e.g., if matrix or chemical interference requires that a sample be diluted. Samples with elevated detection limits that exceed 10 times the required detection limit were excluded from the risk assessment data set.

An organic chemical was removed from further consideration if it was a common laboratory contaminant and the reported sample concentration was less than 10 times the concentration in an associated quality control sample (i.e., trip blank, field blank, equipment rinsate, or laboratory blank). Common laboratory contaminants include acetone, 2-butanone, methylene chloride,

toluene, or phthalate esters. Other organic chemicals were not included if results were less than five times the highest concentration detected in an associated quality control sample. Where this has occurred, it is discussed in the site-specific chapter.

#### 7.1.2 Background Screening

EPA Region 4 methods for comparison to background (EPA 1995) were used to screen metals data. If enough samples (e.g., >20) are collected across the entire site to develop a more robust background level, background upper tolerance level (UTL) can be calculated and compared to site data. In cases where there are a low number of samples to define background, as is the case for FSMR, a second method can be used to define background: 2 times the mean background concentration (EPA 1995). Given that less than 20 background samples were collected for the site, the latter method was used for estimating background concentrations in this study.

Metal concentrations in surface soils, subsurface soils, and groundwater were compared with respective background concentrations to determine what chemicals are site related. Appendix E presents the summary of background data and presents the 2 times mean background concentrations. The mean concentrations for soils in the eastern United States are also presented for comparative purposes (USGS 1981). Given the limited number of background samples at the FSMR, the screening value for background may not be representative of the true upper range of background levels. The mean concentrations for soils in the eastern United States provide some perspective on how representative the background screening values are relative to regional background concentration. The results of the site-specific background screening for various environmental media are discussed in the respective chapters for each SWMU.

Organic chemicals are generally not ubiquitous in the environment. Therefore, it is assumed that any organic chemicals detected are the result of site-specific activities. No background screening was performed for organic chemicals, and all detected organics are considered SRCs.

### 7.2 EXPOSURE EVALUATION

The objective of this exposure evaluation is to identify potential human populations that may be exposed to site-related chemicals under current and future land-use conditions. A complete exposure pathway consists of five elements: (1) a potential receptor population, (2) a source of contamination, (3) a transport or retention medium, (4) a point of contact for a receptor, and (5) a route of exposure (ingestion dermal absorption or inhalation) at the point of contact through which the chemical may be taken into the body. When all of these elements of an exposure pathway are present, an exposure of a receptor population can take place. The assessment considers both on-site and off-site receptors and their relationship to the potential migration pathways, and exposure pathways and points of exposure for site-related chemicals.

### 7.2.1 Receptor Assessment

This section identifies those populations that may be exposed to site-related chemicals. Although a receptor population may be identified under current conditions, potential changes in land use may result in the presence of more sensitive receptor populations in the future.

Generally, receptor populations are divided into two groups: on-site and off-site receptors. On-site receptors are those individuals who may be present within the site boundaries and come into direct contact with contaminants present. The exposure to an off-site receptor requires that the contaminant be transported to an off-site exposure point.

The on-site receptors will vary from site to site depending on the current land use. The on-site receptors for those sites which are currently in active use would include occupational receptors. On-site receptors for inactive sites would include juvenile trespassers and hunters. None of the sites are secured; therefore, a juvenile trespasser may come onto the site to explore or play. In addition, hunting takes place near many of the Burn Pits. Hunters may come in contact with site-related contaminants while hunting in the area.

Land use may change in the future. The burn pits are all located in areas that have been designated for use as maneuver areas or similar purposes. Military operations may increase in the areas of sites which are not currently used. This would result in the presence of military personnel on site. Heavy industrial development of these areas is highly unlikely.

The future on-site receptor populations identified will be representative of those populations that are likely to be present given the potential changes in current land use. The current land use at a site may remain unchanged. Given this scenario, the future receptor populations would remain the same as those identified as current receptor populations. However, to be sure that the first step of the risk assessment process does not exclude any potential receptors, this assessment assumes that residential use of the site could occur in the future. Residential use of the site is highly unlikely, but is presented as a scenario in accordance with Risk-Based Corrective Action (RBCA) guidance.

The current off-site receptor populations are individuals working or hunting in the area. The off-site receptor populations would remain unchanged for those sites where the land use at the site and in the surrounding area remain unchanged. The off-site receptor population would change from occupational receptors to residential receptors if the area surrounding the site were to be developed into base housing or a residential area, which is highly unlikely.

### 7.2.2 Migration Pathway Analysis

This section describes the potential pathways related to chemical transport that may result in potential exposure points for humans. In general, the major routes of migration from these sites are volatilization into air, wind erosion resulting in fugitive dust, and leaching of contaminants into groundwater. All of the sites are relatively flat, and most are completely vegetated; therefore, migration via surface water runoff is not a viable migration pathway.

Air Pathway. Site-related compounds in soils may be released via volatilization. This could result in potential exposures to human receptors via inhalation. This migration pathway is generally limited to VOCs found in the surface soils.

Particulate-bound chemicals may also be transported through soil erosion or generation of fugitive dust. This pathway is limited to compounds that have a high affinity for soils and a low vapor pressure, thus reducing the possibility of volatilization. This migration pathway is limited to chemicals found in surface soils at sites that lack complete vegetative cover (i.e., Burn Pit A).

**Groundwater**. Migration of soil contaminants to groundwater could occur from infiltration and percolation of rainwater through the soil. The extent of contaminant migration depends primarily on the amount of rainfall, evaporation, solubility of the chemical in water, the absorption coefficient, and distance to the groundwater. In general, VOCs (such as BTEX) travel more easily through soils than SVOCs. Solubility of metals is dependent on the metal species and is difficult to generalize. Groundwater occurs approximately 2 to 13 feet bgs at the Burn Pit sites. Calculation of the likelihood for contaminants to migrate to groundwater at the Burn Pits is discussed in Section 6.0.

97-160P(DOC)/020899

There are no surface water bodies in the vicinity of any of the Burn Pits. The sites are relatively flat and most are completely vegetated. Therefore, migration of contaminants to surface waters via the discharge of groundwater is not a viable migration pathway.

## 7.2.3 Identification of Exposure Pathways

Potential human exposure may occur by primary pathways (e.g., dermal contact, inhalation, or direct consumption of soil or water) or through secondary pathways involving the transfer of site-related chemicals into food sources (i.e., crops, livestock, and game).

**Primary Pathways.** Potential primary pathways for exposure of receptor populations include ingestion of soils, inhalation of volatile organics and air-borne particulates, and dermal contact with soils.

Current on-site receptor populations may be exposed to contaminants in surface soil via inadvertent ingestion of soils, inhalation of volatile organics from soils, inhalation of fugitive dust and dermal contact with soils. Military personnel are at greatest potential risk for exposure because training may require the soilders to lie on the ground, resulting in dermal contact of soils, inadvertant ingestion of soils adhering to the hands, and inhalation of volatiles and fugitive dust.

The hunter is not expected to come in direct contact with the soils. The hunter's activities include primarily walking and standing, with minimal kneeling or sitting. The hunter is expected to wear long pants and boots; therefore, dermal contact with soils is expected to be minimal. The hunter's activities are expected to be in areas that are well vegetated, providing habitat for game animals. Therefore, exposure via inhalation of fugitive dust is expected to minimal because the vegetation would prevent generation of fugitive dust.

Off-site occupational receptors and hunters may be exposed via inhalation of fugitive dust. It is unlikely the volatile compounds will migrate off site in any significant concentrations, therefore, off-site receptor populations would not be exposed to volatilized compounds.

The surficial groundwater aquifer underlying Fort Stewart is 2 to 13 feet bgs. Below this aquifer is the Principal Artesian aquifer, which is hydrogeologically isolated from the surficial aquifer. The surficial groundwater aquifer is too shallow to be used as a drinking water source. If groundwater were to be used as a source of drinking water, the Principal Artesian aquifer would be used. The shallow aquifer is currently used in some areas of Fort Stewart for irrigation or watering lawns. However, the groundwater underlying the Burn Pits is not currently in use; therefore, there are no current exposure pathways for contaminants in groundwater.

Potential future on-site and off-site occupational receptor populations would have the same exposure pathways as current occupational receptors. The on-site resident may be exposed to contaminants in soils via inadvertent ingestion, inhalation of volatile organics and fugitive dust, and dermal contact. The construction worker may be exposed to contaminants in surface and subsurface soils via inadvertent ingestion, inhalation of volatiles, and dermal exposure. Surficial groundwater may be used for watering lawns and grounds. Potential exposure to contaminants in groundwater includes inhalation of volatile organics and dermal exposure.

Indirect pathways. Hunting is allowed on the FSMR. Game species may bioaccumulate contaminants as a result of ingesting soils and contaminated vegetation. Current off-site receptors may be exposed as a result of consuming contaminated game. Fort Stewart does not currently

lease agricultural lands and is unlikely to allow agricultural practices within the base in the future. Therefore, exposure via uptake into food crops is not a viable exposure pathway.

The risk-based screening values are based on a residential receptor population. The potential exposure pathways addressed in deriving the screening values for soils include soil ingestion, inhalation of fugitive dust and volatiles, and ingestion of contaminants leaching into groundwater from soils. Exposure pathways for groundwater include ingestion and inhalation of volatiles during showering. The residential exposure scenario does address all of the potential exposure pathways, although none of the potential receptor populations would be exposed via <u>all</u> of the pathways addressed under a residential exposure scenario. The derivation of the risk-based screening values is discussed further in the following section.

## 7.3 SELECTION OF SCREENING VALUES

Screening values represent risk-based concentrations that are publicly available. Due to their conservative nature, screening values can be used with a high degree of confidence to indicate sites for which no further action is required. Screening values inherently incorporate assumptions about land use. In identifying COPCs, it is generally accepted that screening levels will reflect any potential future land uses and, thus, usually reflect a conservative residential use scenario (EPA 1991, 1996b; ASTM 1995).

If risk-based screening values are not available, it generally reflects (1) that the chemical is not considered to be toxic except perhaps at extremely high concentrations (e.g., aluminum, sodium, etc.); (2) no dose-response data indicate a toxic effect; or (3) EPA is currently reviewing toxicity information, and no reference dose or cancer slope factor currently is available.

### 7.3.1 Screening Values for Soils

The EPA Region III risk-based screening values for ingestion of soils were taken from EPA (1996c). The risk-based values were adjusted to reflect a potential incremental lifetime cancer risk of  $1 \times 10^{-6}$  or a hazard index of 0.1. The risk-based values are given for residential and industrial land use. Residential land use is unlikely to occur at this site. However, as a conservative measure, residential land use values were used to screen surface soil samples. Exposure of subsurface soils would be limited to a person working with an excavation, i.e., an industrial exposure scenario. Therefore, the industrial land use values were used to screen the subsurface soil samples.

Step 1 screening levels generally reflect residential land uses; use of these levels in the first step of the risk process ensures that no chemical will be screened from consideration prematurely. The EPA does provide guidance and default parameter values for developing screening levels that reflect industrial land-use assumptions. These levels are developed using equations and default values from EPA (1991). Residential land use is unlikely at any of the Burn Pits.

The default residential exposure assumptions for soil are as follows:

- Soil ingestion—For noncarcinogens, the receptor is a child (age 1 to 6) who ingests 200 mg of soil/day for 6 years; for carcinogens, the soil ingestion rate is age-adjusted over a time period of birth until age 30, assuming an adult ingests 114 mg/day (EPA 1996a, 1996c).
- Inhalation of volatiles or fugitive dust, where a resident is exposed to airborne contaminants for 30 years (EPA 1996a).

97-160P(DOC)/020899

• Leaching of contaminants to groundwater, with subsequent ingestion of groundwater (EPA 1996a).

The potential exposure pathways for soils present at the Burn Pits include ingestion of surface soils, ingestion of subsurface soils (construction worker), inhalation of volatiles, and inhalation of fugitive dust for future land-use scenarios. For those chemicals detected, the screening values for soil ingestion are lower (i.e., more conservative) than the risk-based inhalation values; therefore, the soil ingestion values were selected.

The EPA Region III risk-based screening values give two values for arsenic. One value is based on the carcinogenic risks for arsenic, and the other value is based on systemic toxicity or noncarcinogenic risk. As a conservative measure, the lower value (carcinogenic value) will be used for the screening value,

Chromium may exist in two valence states, trivalent ( $Cr^{+3}$ ) and hexavalent ( $Cr^{+6}$ ) chromium. The hexavalent chromium is significantly more toxic than  $Cr^{+3}$  and  $Cr^{+6}$  is more mobile in the environment. However,  $Cr^{+6}$  is not naturally occurring and is unstable in the environment, oxidizing to the trivalent state. The risk-based screening values for residential soils include both trivalent and hexavalent chromium. It is unlikely that the chromium present is hexavalent chromium, given there is no likely source for  $Cr^{+6}$ . In addition, the value given represents the total chromium present, which includes the naturally occurring trivalent chromium. As a conservative assumption, the hexavalent chromium value will be used for the screening value.

## 7.3.2 Screening Values for Groundwater

The groundwater screening values reflect the use of groundwater as a source of drinking water (EPA, 1996b, 1996d). These values include the Region III screening values for tap water based on a cancer risk of  $10^{-6}$  and a hazard index (HI) of 0.1. As previously discussed, groundwater at this site is unlikely to be used as a drinking water source, but may be used as a source of water for irrigation or watering in the future. The drinking water screening values are considered to be health protective values given the conservative assumptions used.

The default residential exposure assumptions for groundwater are as follows:

- Groundwater ingestion—For noncarcinogens, the receptor is an adult who ingests 2 L of groundwater/day; for carcinogens, the water ingestion rate is age-adjusted over a time period of birth until age 30, assuming a child age 1-6 ingests 1 L/day (EPA 1996a, 1996c).
- Inhalation of volatiles during showering.

There is no reference dose for lead; however, the EPA has derived an Action Level for acceptable lead levels at the tap (EPA 1991). The 15  $\mu$ g/L action level will be used for the lead screening value.

Region III risk-based screening values for arsenic include values for carcinogenic and noncarcinogenic effects. As a conservative measure, the carcinogenic value for arsenic will be used for exposure via drinking water. However, it should be noted that the drinking water scenario is not applicable at this site, and is being used in the absence of a more appropriate screening value.

As previously discussed, chromium may exist in two valence states, trivalent ( $Cr^{+3}$ ) and hexavalent ( $Cr^{+6}$ ) chromium. Hexavalent chromium is more toxic but is also unlikely to be

79

present at this site because of the absence of a potential source and the instability of the chemical, which oxidizes readily into  $Cr^{+3}$ . As a conservative assumption the hexavalent chromium value will be used for the screening value.

#### 7.4 RISK EVALUATION

The risk evaluation compares the maximum value detected in each medium with its respective screening value. If chemicals exceed a screening value, then a risk may exist and those chemicals should be evaluated more carefully. Contaminants identified as COPCs will be evaluated further in a BRA.

The selection of COPCs for each environmental media (surface soil, subsurface soil, groundwater, sediment and surface water) at each of the Burn Pits is addressed in the respective chapters (Sections 9.0 through 14.0). The selection process involves two steps. The initial step is the comparison of the maximum concentrations to the appropriate screening values. Given the conservative nature of the screening values, a weight-of-evidence analysis of those chemicals passing the screen is done to determine if those chemicals selected should be analyzed further in a BRA.

The potential risks associated with exposure to chemicals are not quantified. However, toxicity values and associated data (reference doses, target organs, cancer slope factors, etc.) are presented in Appendix G for informational purposes.

## 7.5 CONCLUSIONS OF THE HUMAN HEALTH RISK EVALUATION

In surface soil, arsenic is a potential COPC at Burn Pits A and F. At these two sites, arsenic exceeds its risk-based screening value for exposure of a residential receptor. Maximum concentrations of arsenic of 2.5 mg/kg and 2.7 mg/kg were found at Burn Pits A and F, respectively, which exceed the risk-based screening value of 0.43 mg/kg. It should be noted that these maximum concentrations of arsenic are below the average concentration of arsenic for soils regionally, which is 7.4 mg/kg. In addition, a residential screening value was used, although residential land use at this site is unlikely. No other constituent in surface soil exceeded its risk-based screening value at any of the Burn Pits.

In subsurface soil, no constituent exceeded its respective risk-based screening value for exposure to a residential receptor. As discussed in Section 6.0, benzene, PCE, methylene chloride, and chromium are potential COPCs in subsurface soils due to potential for leaching to groundwater resulting in groundwater concentrations exceeding their respective MCLs. However, off-site migration of these contaminants would be very limited due to retardation and biodegradation as well as the slow movement of groundwater. In addition, none of these four constituents has been detected in groundwater at levels exceeding its risk-based screening value or its MCL.

In groundwater, arsenic is a potential COPC at Burn Pit D, where the maximum concentration of arsenic in groundwater (4.2  $\mu$ g/L) exceeded its risk-based screening value of 0.045  $\mu$ g/L for residential use of groundwater as drinking water. However, the maximum concentration of arsenic was well below its MCL of 50  $\mu$ g/L. Arsenic was detected at a concentration only slightly above its reference background concentration of 3.4  $\mu$ g/L, and in only a single well at Burn Pit D (MW-6). In addition, use of the surficial groundwater at this site for drinking water is unlikely. Therefore, arsenic is not considered a potential threat to human health at the Burn Pits.

Similarly, cadmium is a potential COPC in groundwater at Burn Pits C and E, where the maximum concentrations of cadmium in groundwater (3  $\mu$ g/L and 4.3  $\mu$ g/L, respectively) exceed their risk-based screening value of 1.8  $\mu$ g/L for residential use of groundwater as drinking water. The maximum concentration of cadmium was less that its MCL of 5  $\mu$ g/L at each site. In addition, use of the surficial groundwater at these sites for drinking water is unlikely. Therefore, cadmium is not considered a potential threat to human health at the Burn Pits.

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at the Burn Pits. A human health BRA is not warranted. No further action is required for protection of human health.
# 8.0 ECOLOGICAL RISK ASSESSMENT

42

The State of Georgia requires that all RCRA facilities choosing to set remediation levels based on an assessment of risk to human health and the environment prepare risk assessment documentation and propose remediation levels according to the *Guidance for Selecting Media Remediation Levels at RCRA Solid Waste Management Units* (GEPD 1996). GEPD (1996) guidance is based on the guidance contained in EPA Region 4 Bulletins, *Supplemental Guidance to RAGS, Ecological Risk Assessment* (EPA 1996a) and a 1994 draft of *Ecological Risk Assessment for Superfund, Process for Designing and Conducting Ecological Risk Assessments* (EPA 1997). The EPA has also proposed guidelines for conducting ecological risk assessments (EPA 1996b). Where there are differences with EPA guidance (EPA 1996a, 1997), the GEPD 1996 guidance document takes precedence.

Risk is the likelihood of experiencing adverse effects. ERAs identify and evaluate the risk to biota exposed to chemical contaminants and physical and biological hazards. The ERA for the Burn Pits A through F focuses on evaluating the potential for harmful effects on ecological receptors as a result of exposure to chemicals.

The assessment of risk for ecological receptors at the Burn Pits A through F is being conducted in a phased approach according to GEPD guidance (GEPD 1996). As shown in the flowchart of the GEPD ecological risk assessment process, the two phases are:

- PRE and
- ERA.

The PRE compares measured concentrations of SRCs to conservative ecological screening values (ESVs) for one or more ecological receptors. Only those facility-related contaminants that are indicated to be potential hazards in the PRE are evaluated as ecological COPCs in an ERA. The basic approach to ERAs is similar to that of the PRE, but site-specific data are used to quantify exposure and evaluate effects in the ERA (GEPD 1996). Appropriate site-specific data include concentrations of contaminants in animals and plants (tissue residues) and toxicity tests (EPA 1996a). Remediation levels for protection of ecological resources are developed and proposed only for those COPCs identified as ecological COCs in the ERA, if one is required.

The medium of primary concern to ecological receptors at the Burn Pits A through F is surface soil. Chemicals in surface soil can be contacted directly by ecological receptors, or their presence in soil can result in the accumulation of contaminants in plants and animals, which can cause ecological receptors to be exposed by ingesting those biota. Groundwater at the Burn Pits A through F is also evaluated because EPA Region 4 guidance (EPA 1996a) directs that, in the PRE, groundwater be screened as surface water. However, there are no surface water bodies in the vicinity of any of the Burn Pits. Therefore, migration of contaminants to surface waters via the discharge of groundwater is not a complete exposure pathway at the Burn Pits.

### 8.1 PRELIMINARY RISK EVALUATION

The purpose of the PRE is to identify substances detected at the Burn Pits that pose a potential hazard to ecological receptors. Ecological COPCs are those substances that are detected at the Burn Pits A through F at concentrations exceeding ESVs.

According to GEPD (1996), the PRE consists of five steps:

- i. ESV comparison,
- ii. Preliminary problem formulation,
- iii. Preliminary ecological effects evaluation,
- iv. Preliminary exposure estimate, and
- v. Preliminary risk calculation.

As shown in the flowchart of the GEPD ERA process (Figure 8-1), all substances detected are screened as ecological COPCs by comparing the maximum detected concentration to the ESVs. This approach assumes that the most sensitive receptors are those that live in direct contact with the medium and are exposed by multiple pathways to contaminants. If no ecological COPCs are identified based on the screening (Step i), then no further evaluation is required. If ecological COPCs are identified based on the screening, then ecological COPCs are evaluated further (Steps ii through v). Because there are no ESVs for soil, all analytes detected in surface soil at the Burn Pits A through F are evaluated further in PRE Steps ii through v.

# 8.1.1 Ecological Screening Value Comparison (Step i)

The ESVs used to identify ecological COPCs at the Burn Pits A through F are EPA Region 4 screening values for hazardous waste sites. For analytes without Region IV ESVs, screening values are proposed based on other methods and data obtained from published sources (e.g., Clayton and Clayton 1981) and toxicological data bases, e.g., Hazardous Substances Data Bank, Integrated Risk Information System. Screening values are conservative to prevent elimination of any contaminant that may pose ecological risk (EPA 1997). If no data are available to support the development of an ESV for an analyte, the analyte is an ecological COPC by default (GEPD 1997a).

For groundwater, EPA Region 4 ESVs are chronic ambient water quality criteria (AWQC) for the protection of aquatic life, such as aquatic plants, invertebrates, and fish, or similarly derived values (EPA 1996a). There are no EPA Region 4 ESVs for barium, acetone, and xylenes, so proposed ESVs for these analytes are identified from published data sources (Suter and Tsao 1996, Clayton and Clayton 1981).

The results of the screening value comparisons for groundwater are presented in the individual sections for each burn pit. The maximum detected concentration in groundwater sampled from the monitoring well locations at each of the former burn pits (Burn Pits A through F) is used to screen for ecological COPCs.

A preliminary problem formulation (Step ii), preliminary ecological effects evaluation (Step iii), preliminary exposure estimate (Step iv), and preliminary risk calculation (Step v) are conducted for SRCs in surface soil because there are no ESVs for soil. These four steps evaluate the potential for risk from COPCs to categories of receptors potentially occurring at the Burn Pits.

### 8.1.2 Preliminary Problem Formulation (Step ii)

The preliminary problem formulation (Step ii) qualitatively identifies categories of potential ecological receptors that occur at the Burn Pits A through F and contaminants that may pose a risk to those receptors in the environmental setting of the Burn Pits A through F. Ecological





COPCs identified in the PRE screen for groundwater are considered, as well as all analytes detected in soil. Preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors are selected for evaluation in the preliminary risk calculation.

GEPD (1996) specifies that the PRE develop "risk characterization for a model ecological receptor." Developing risk characterization for multiple ecological receptors, e.g., mammals and birds, is allowable for sites where more than one type of potentially hazardous chemical is detected (GEPD 1997b). Characterizing the risk to multiple receptors, where each may be more sensitive to one or more chemical contaminant, can make the PRE more protective of ecological resources. At the Burn Pits A through F, the risk characterization for surface soil considers both mammals and birds as ecological receptors.

#### **Environmental Setting**

The Burn Pits A through F are located in the area surrounding the cantonment, or garrison, area of the FSMR (Figure 2-3). This area of the FSMR is used for ranges and training areas or held as non-use areas. The open range and training areas comprise 11 percent of the base and consist of grasses, shrubs, and scrub oak.

Sixty-six percent of the approximately 367.2 square miles of forest in the FSMR is pine, with the major species being slash pine, loblolly pine (*Pinus taeda*), and longleaf pine (*P. palustris*). The remainder of FSMR forests is composed of river bottomlands and swamps, whose major species include tupelo, other gum trees, water oak, and bald cypress trees. The understory of the pine forest is saw-palmetto (*Serenoa repens*) and is managed by controlled burning, as evidenced by the presence of burn marks on the mature trees.

The terrestrial habitat in the vicinity of the Burn Pits A through F consists primarily of palmetto-pine forest. The individual burn pits are in various phases of succession from deforested open grassy areas, through shrub and scrub oak to the surrounding pine forest community.

The principal surface water body receiving drainage from the FSMR is the Canoochee River, which joins the Ogeechee River (part of the northwestern boundary of the reservation). The individual Burn Pit sites are located within different subwatersheds (Figure 2-3). Burn Pits A through D are located west of the garrison area within the Taylors Creek watershed; drainage from these sites is collected by Mill Creek and Horse Creek tributaries that discharge into Taylors Creek, which in turn discharges into Canoochee Creek. Canoochee Creek is a tributary of the Canoochee River that drains much of the western portion of the FSMR. Burn Pit E is located near the Canoochee River; drainage from the site is collected in small unnamed streams that discharge directly to the Canoochee River. Burn Pit F is located south of Savannah Road in the Peacock Creek watershed. Drainage from the Burn Pit F site flows into Goshen Swamp, which discharges southward into Peacock Creek and, ultimately, into the Jerico River.

Numerous mammals and birds were noted by SAIC field personnel while conducting the field investigations at the former burn pits. The red-headed woodpecker (Melanerpes eryhrocephalus) and the turkey vulture (Cathartes aura) were observed by SAIC personnel on numerous occasions. Scat or tracks of white-tailed deer (Odocoileus virginianus), coyote (Canis latrans), armadillo (Dasypus novemcinctus), and raccoon (Procyon lotor) were noted by SAIC field personnel.

### Surface Soil at Burn Pits A through F

The PRE for surface soil (0 to 1 foot) at the Burn Pits A through F evaluates the potential for risk to ecological receptors from ecological COPCs detected at surface soil sampling locations at each of the Burn Pits. All analytes detected in surface soil are ecological COPCs because there are no EPA Region 4 ESVs for soil.

The categories of ecological receptors that are potentially directly exposed to substances in surface soil at the Burn Pits A through F are soil bacteria and fungi, vegetation, and animals that come in direct contact with or ingest soil, e.g., soil-dwelling invertebrates. Other categories of receptors are potentially exposed indirectly to soil contaminants that are taken up and stored in the cells or tissues of those organisms that are directly exposed. Herbivorous invertebrates (e.g., insects) and vertebrates (e.g., birds and mammals) are potentially indirectly exposed when they ingest animals exposed directly or indirectly to contaminate soil.

Nine substances from two broad categories of chemicals were detected above background criteria in surface soil samples from one or more of the Burn Pits A through F: 15 RCRA metals and 3 SVOCs. Based on the greater amount of published data on the effects of these nine substances on vertebrate wildlife, mammals and birds in particular, and the potential for three of the metals to biomagnify in soil-dwelling invertebrates (cadmium, lead and mercury), the proposed ecological receptors for soil at the Burn Pits A through F are carnivorous small mammals and birds that prey upon soil-dwelling invertebrates.

The preliminary assessment endpoint for soil at the Burn Pits A through F is protection of small mammals and bird populations from adverse effects. The surrogate species to represent the generic small mammal and bird receptors are the short-tailed shrew (*Blarina brevicauda*) and the American robin (*Turdus migratorius*). The home range of the shrew is small, and robins are territorial during the spring mating season. Earthworms and other soil-dwelling invertebrates represent a large percentage of both species' diets. The life history and behavior of these two surrogate species ensure a conservative estimate of risk.

### Groundwater at Burn Pits A through F

The PRE for the Burn Pits A through F screens groundwater for exposure to ecological COPCs per GEPD (1996) and EPA Region 4 (1995) guidance. However, groundwater is not evaluated further in the PRE for the Burn Pits because ecological receptors are not likely to be exposed to groundwater at these sites. There are no surface water bodies in the vicinity of any of the Burn Pits. Also, groundwater contaminants are not expected to migrate from the Burn Pits because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit soils. Therefore, migration of contaminants to surface waters via groundwater discharge is not a complete exposure pathway, and PRE steps ii through v are not conducted for groundwater.

### 8.1.3 Preliminary Effects Evaluation (Step iii)

The preliminary ecological effects evaluation (Step iii) identifies toxicity reference values (TRVs) for use in the preliminary risk calculation. TRVs are derived from no observed adverse effect levels (NOAELs) from laboratory toxicity studies on test species from groups closely related to the surrogate species (EPA 1996a).

In the PRE for the Burn Pits A through F, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the Burn Pits. For these receptors, TRVs are expressed as threshold concentrations of the contaminant in the soil. The TRVs for ecological receptors exposed by ingestion of ecological COPCs in soil are calculated from dietary limit concentrations corresponding to the NOAEL doses. Dietary limit concentrations (mg/kg) are calculated from the NOAEL (mg/kg/d) by multiplying by the body weight (kg) and dividing by the ingestion rate (kg/d). That is,

dietary limit concentration = NOAEL × body weight/ingestion rate.

NOAELs and dietary limits for surrogate species are reported in Sample et al. (1996).

For shrews and robins, which are exposed indirectly by ingestion of biota, the maximum detected soil concentration is compared to the threshold soil concentration (i.e., the TRV), which is calculated as the dietary limit concentration associated with the NOAEL dose divided by the unitless bioaccumulation factor (BAF) for the contaminant in the tissue of the ingested soil-dwelling biota. That is,

 $HQ = maximum soil concentration (mg/kg) \times BAF/dietary limit concentration (mg/kg).$ HQ = maximum soil concentration (mg/kg)/dietary limit concentration (mg/kg)/BAF.

This approach to calculating and presenting TRVs allows direct comparison of measured concentrations of COPCs in the abiotic media against the abiotic media concentration assumed to be protective of the ecological receptor by virtue of its equivalence to the NOAEL.

If toxicity data are not available for the surrogate species, data for a test species of the same taxonomic class may be used and adjusted for the body weight of the surrogate species; i.e., mammal test species data are used for mammal surrogate species, bird test species data are used for bird surrogate species. NOAELs for test species based on daily dose (mg/kg body weight/day) are adjusted to surrogate species based on body weight, according to the following equation:

surrogate species NOAEL = test species NOAEL x  $(bw_{ts}/bw_{ss})^2$ ,

where  $bw_{ss}$  and  $bw_{ss}$  are the body weights (kg) of the test species (ts) and surrogate species (ss), respectively, and z = 0.25 for mammals (EPA 1997) and z = 0 for birds (Sample et al. 1996). For example, the published NOAEL for a chemical might be based on data for a 0.35-kg rat. The NOAEL for a 0.022-kg field mouse would be nearly 2 times larger than the rat NOAEL. If a NOAEL is not available for a contaminant, a concentration associated with the lowest observed adverse effect level (LOAEL) divided by a conservative uncertainty factor of 10 (LOAEL/10) will be used as the screening value (EPA 1996a). If neither a NOAEL nor LOAEL is available for a contaminant divided by an uncertainty or 'safety' factor of 10 may be used as a surrogate TRV (GEPD 19976a).

The TRVs derived for shrews and robins for ecological COPCs detected in soil at Burn Pits A through F are presented in Table 8-1. There are mammal and bird TRVs for seven of eight RCRA metals, silver being the exception. For organic ecological COPCs, there is only one TRV, and that is for shrews for benzo(a)pyrene. Surrogate TRVs are derived for benzo(b)fluoranthene and fluoranthene by dividing the benzo(a)pyrene TRV by an uncertainty factor of 10. There are no TRVs for birds exposed to organic contaminants.

Analyte	(mg/kg	AEL <sup>a</sup> 3 BW/d)		y Limit <sup>b</sup> z/kg)	Earthworm	Tł	
	Shrew	Robin	Shrew	Robin	BAF	(mg Shrew	Robin
			Inorganic	5			
Arsenic	0.15	5.1	0.25	4.3	0.0000		<u> </u>
Barium	11.8	20.8	19.7		0.0066	37.9	651.5
Cadmium	2.12			17.2	0.0075	2627	2293
Chromium		1.45	3.5	1.2		0.321	0.109
Lead	6015	1.0	10026	0.83	0.16	62663	
	17.6	1.13	29.3	0.94	0.4	the second s	5.2
Mercury <sup>d</sup>	2.9	0.45	4.8	0.37		73.3	2.35
	·····				0.34	14.1	1.1
Benzo(a)pyrene	1.19		tile Organic	Compounds			······································
	1.13		1.98	-	0.05	39.6	
Benzo(b)fluoranthene	-						
Fluoranthene					0.05	3.96	-
		• /	-	-	0.05	3.96*	······································

#### Table 8-1. Derivation of Toxicity Reference Values for Ecological COPCs in Soil, Burn Pits A through F, Fort Stewart

"Sample et al. (1996).

<sup>b</sup>Dietary Limit = NOAEL x BW (kg)/Food Ingestion Rate (kg/d); see Table 8-2 for BW and Food Ingestion Rate.

Earthworm BAFs from HAZWRAP (1994); lead BAF assumes calcium concentration >500 mg/kg and lead <sup>d</sup>Assumed to be mercuric chloride.

Surrogate value = benzo(a)pyrene TRV × 0.1.

COPCs = Chemicals/Contaminants of Potential Concern. NOAEL = No Observed Adverse Effect Level.

TRV = Toxicity Reference Value = Dietary Limit/BAF<sub>earthworm</sub>. - = No data to derive TRV.

### 8.1.4 Preliminary Exposure Estimate (Step iv)

The preliminary exposure estimate (Step iv) evaluates the potential pathways of exposure appropriate to the preliminary assessment endpoints and ecological receptors at Fort Stewart SWMUs. Exposure factors are selected for receptors likely to be exposed to ecological COPCs in groundwater or surface water by ingestion of contaminated groundwater or soil and biota exposed to contaminated soil.

The exposures of the surrogate species for the Burn Pits A through F are estimated using conservative assumptions. It is assumed that the surrogate species at each Burn Pit spend their entire lives and obtain 100 percent of their diet or drinking water from that area, i.e., the area use factor (AUF) equals 1. Shrews and robins are assumed to eat only soil-dwelling invertebrates, such as worms, that bioaccumulate contaminants from soil, in accordance with guidance that the screen be based on exposure through two trophic transfers (EPA 1997, GEPD 1997b). Contaminants are assumed to bioaccumulate in the soil-dwelling invertebrate prey of ecological receptors at levels equal to published BAFs for earthworms and other invertebrates (e.g., HAZWRAP 1994). The exposure parameters for shrews and robins exposed to COPCs in soil are presented in Table 8-2.

The concentration of ecological COPCs to which endpoint receptors at the Burn Pits are directly or indirectly exposed is estimated by the maximum detected concentration at each former Burn Pit.

	Surrogat	e Species
Parameter	Shrew	Robin
	0.015*	0.077*
Body Weight (kg) Food Ingestion Rate (kg/d)	0.009*	0.093*
AUF	1	1
Relative Bioavailability	1	11
Diet	100 percent earthworm	100 percent earthworm
Source Medium	Surface soil	Surface soil

Table 8-2. Exposure Parameters for Surrogate Species Exposed to COPCs in Soil or Groundwater, Burn Pits A through F, Fort Stewart

\*Sample et al. (1996); Table B-1.

AUF = Area Use Factor.

COPCs = Chemicals/Contaminants of Potential Concern.

### 8.1.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses hazard quotients (HQs), the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to produce an HI. An HI greater than 1 for a category of COPCs is a useful indicator of potential risk when no individual COPC in that category has an HQ greater than 1. An HI assumes that the effects of the individual COPCs in the category are additive. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1 and when HQs are calculated for more than one chemical.

Because of uncertainties in quantifying exposure and effects, the exposure and effects assessments for the Burn Pits A through F are designed to produce HQs that minimize the probability of falsely concluding that there is no risk when, in fact, there is. Therefore, ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. To minimize the probability of falsely concluding there is risk when there is none, an ERA for those ecological COPCs with calculated HQs or HIs exceeding 1 is performed using site-specific data (GEPD 1996).

#### 8.1.6 Uncertainties

To evaluate the uncertainty associated with preliminary HQs that exceed 1, supplemental risk calculations are presented for selected ecological COPCs identified in the PRE. Supplemental risk calculations are made using published dietary fractions of plant tissue, animal tissue, and incidental soil for shrews and robins (EPA 1993b), instead of 100 percent earthworms, and an estimated fractional absorption from ingested soil of 10 percent, instead of 100 percent absorption. Evaluating the exposure from ingested plant tissues requires uptake factors, which are similar to BAFs for animal tissue, for both vegetative parts and reproductive parts of plants. Plant uptake factors are those reported in Baes et al. (1986). These adjustments to the exposure estimates are used to derive TRVs based on both published NOAELs and LOAELs. The LOAELs used to derive TRVs for the supplemental evaluation are those reported in Sample et al. (1996).

Supplemental risk calculations are presented for shrews and robins exposed to cadmium, chromium, or lead in surface soil at one or more of the six Burn Pits. The shrew dietary fractions used in the supplemental risk calculations are 13.8 percent plant vegetative parts, 86.2 percent earthworms, and an incidental soil ingestion of 13 percent of total food ingestion. The robin dietary fractions are 62.5 percent plant reproductive parts, 37.5 percent earthworms, and an incidental soil ingestion of 3.9 percent of total food ingestion. The NOAELs used to derive TRVs are those used and reported in the PRE. The LOAEL used to derive the cadmium TRV for the shrew (1.74 mg/kg) is the 10 mg/kg/d chronic cadmium chloride dose causing a 50 percent reduction in fetal rat survival. This level of reproductive effect may not be ecologically significant for small mammals with high fecundity, such as shrews and mice. The LOAEL used to derive the cadmium TRV for the robin (3.9 mg/kg) is the 20 mg/kg/d chronic cadmium chloride dose causing a significant reduction in egg production in ducks. The LOAEL used to derive the chromium TRV for the robin (62.1 mg/kg) is the 5 mg/kg/d chronic dose causing a reduction in duckling survival. The LOAEL used to derive the lead TRV for the shrew (366.3 mg/kg) is the 80 mg/kg/d chronic lead acetate dose causing reduced offspring weight and kidney damage in young rats. The LOAEL used to derive the lead TRV for the robin (48.1 mg/kg) is the 11.3 mg/kg/d chronic lead acetate dose causing a reduction in successful quail egg hatching. These diets and LOAELs provide a lower bound on the preliminary risk estimate to compliment the conservative risk estimate reported in the PRE.

### 8.2 CONCLUSIONS AND RECOMMENDATIONS

A PRE is conducted for surface soil and groundwater at each individual Burn Pit site, according to GEPD (1996) and EPA (1996a) guidance, and is presented in Sections 9 through 14.

Barium, cadmium, lead, mercury, silver, and ethylbenzene are present in groundwater at one or more of the former Burn Pits at concentrations that exceed EPA Region 4 ESVs for surface water (Table 8-3). However, treating groundwater as surface water is unrealistic because there are no surface water bodies in the vicinity of any of the Burn Pits. As indicated in Section 6.0, off-site migration of these contaminants would be limited due to retardation as well as the slow movement of groundwater. Therefore, migration of contaminants to surface waters via groundwater discharge is not a complete exposure pathway for ecological receptors.

Cadmium, lead, and chromium are present in surface soil at one or more of the former Burn Pits at concentrations that exceed the TRVs for the robin and, in some cases, the shrew (Table 8-4).

	<u> </u>		Burn	Pits		······
and a second	<u>A</u>	B	C	D	E	F
Concentration Exceeds ESV	Barium Mercury Silver	Barium Cadmium Silver	Barium Cadmium Mercury	Barium Silver	Cadmium	Ethylbenzene

 Table 8-3. Summary of the Ecological COPCs Identified in EPA Region 4 Ecological Screening

 Value Comparison for Groundwater at Burn Pits A through F, Fort Stewart

COPCs = Chemicals/Contaminants of Potential Concern.

EPA = U.S. Environmental Protection Agency.

ESV = Ecological Screening Value.

# Table 8-4. Summary of the Preliminary Risk Evaluation for Ecological COPCs in Surface Soil at Burn Pits A through F, Fort Stewart

		·····	Burn Pit	S		<del></del>
	A	В	С	D	E	<u>  F</u>
HQ > 1	Cadmium (shrew, robin) Lead (robin)	Lead (robin)	Cadmium (shrew, robin) Lead (robin)	(none)	(none)	Chromium (robin)

COPCs = Chemicals/Contaminants of Potential Concern.

HQ = Hazard Quotient.

TRV = Toxicity Reference Value.

Supplemental site-specific ecological risk evaluations are presented for Burn Pits A, B, C, and F in their individual sections. For these sites, the PRE conducted according to GEPD and EPA Region 4 guidance (GEPD 1996, EPA 1996a) identified site-related ecological COPCs in surface soil, indicating that there is a possibility of adverse effects to ecological receptors from exposure to contaminated media. Further evaluation in the uncertainty sections considers the potential for soil concentrations to exceed TRVs associated with more realistic exposure estimates and effects levels for cadmium, chromium, and lead that are associated with low-level effects. There are no ecological COPCs for Burn Pits D and E, and no further action is recommended for those sites. Based on the results of the supplemental evaluation, it is concluded that the ecological COPCs identified in the PRE do not pose a risk to ecological receptors, and no further action is recommended for Burn Pits A, B, and F. These recommendations are made in the individual sections for each Burn Pit and are summarized in Section 15.0.

# 9.0 CHARACTERIZATION OF BURN PIT A (SWMU 4A)

## 9.1 HISTORY AND DESCRIPTION OF BURN PIT A

Burn Pit A (SWMU 4A) comprises approximately 2.3 acres and is located on Fort Stewart Route 38, approximately 1200 feet southwest from its junction with Fort Stewart Route 40. The site is essentially flat and surrounded by woods. Sparse grass currently covers the area. Burn Pit A is currently being used as a staging area for (1) the Rail Marshaling Area and (2) the Tank Commander's Course.

The elevation of the site varies between approximately 69 and 73 feet above mean seal level (amsl) (Figure 9-1). Surface (overland) drainage occurs through overland flow radially to the west, north, and east. A roadside ditch south of the site captures runoff from the road, but does not receive runoff from the site. An unnamed tributary of Taylors Creek lies approximately 1500 feet west of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 9-1). VOCs, including acetone and PCE, and RCRA metals, including barium and lead, were detected above site-specific background levels in soils. VOCs, including carbon disulfide, benzene, and acetone, and RCRA metals, including arsenic, selenium, chromium, lead, and mercury, were detected above site-specific background levels in groundwater.

During the Phase II RFI activities, one additional downgradient monitoring well (MW-5) was nstalled, and three soil borings (SB-1 through SB-3) were drilled at the site (Figure 9-1). These wells and borings were used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of slightly silty sands. Geologic cross-sections of the Burn Pit A site are presented in Figures 9.2 and 9.3. Geotechnical parameters were measured in a bulk soil sample taken from MW-5 at a depth of 4.5 to 6.5 feet and in a Shelby tube sample taken from boring SB-2 at a depth of 9.5 to 11.5 feet. The results of the geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic sands with less than 6 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from SB-2, is  $1.73 \times 10^{-3}$  cm/second, typical for a fine sand.

Water levels in the monitoring wells were measured during well development between July 16 and 30, 1997, and again during well sampling between August 6 and 13, 1997. The water table is present at a depth between 5 and 12 feet bgs. A water table contour map is provided as Figure 9-4. Groundwater contours measured during the Phase I RFI activities in August 1993 indicate similar groundwater elevations and flow directions. The shallow groundwater flow across the site is to the east, discharging ultimately to Mill Creek, which is located approximately 1500 feet east of the Burn Pit A site. The calculated horizontal hydraulic gradient across the site was 0.0083 foot/foot. Based on a hydraulic conductivity of  $1.73 \times 10^{-3}$  cm/second and a porosity of 0.42, the groundwater flow rate is approximately 35 feet/year towards Mill Creek.

9-1



(

[.

Figure 9-1. Locations of Sampling

2

)

Ì

97-160P(DOC)/020899

~~~~

j.

1. Alexandre and the second se

|    |     | <u>ر</u>                                             | 60 <u>1</u>                           | 120 1 | 80 2                                                                                                            | 240         |
|----|-----|------------------------------------------------------|---------------------------------------|-------|-----------------------------------------------------------------------------------------------------------------|-------------|
| 75 |     |                                                      |                                       |       |                                                                                                                 |             |
| 70 |     |                                                      |                                       |       |                                                                                                                 |             |
| 65 | MM  | SAND<br>SILTY SAND                                   |                                       | SILT  | in the second |             |
| 60 |     |                                                      |                                       | CLAYE | Y SAND                                                                                                          | WATER TABLE |
| 55 |     | MEDIUM TO<br>FINE SAND                               |                                       |       |                                                                                                                 |             |
| 50 |     |                                                      |                                       |       |                                                                                                                 |             |
| 45 |     |                                                      |                                       |       | n v sen<br>Rohme , n <u>i</u>                                                                                   |             |
| 40 | 1 1 | ۲ میں ایک میں کا | a a a a a a a a a a a a a a a a a a a |       |                                                                                                                 |             |

Figure 9-2. Geologic Cross-5

1

};

31

97-160P(DOC)/020899

1.000

C.

ſ

| Γ  |    | <u> </u>                   | 60 1:       | 20 18         | 80 24         | 40 30 |
|----|----|----------------------------|-------------|---------------|---------------|-------|
| 75 | В  |                            |             |               |               |       |
| 70 |    |                            |             |               |               |       |
| 65 |    | MM<br>SAND                 | ILTY<br>AND |               | SILTY<br>SAND |       |
| 60 | ¥- | SILTY<br>- SAND CL<br>SAND | AYEY SAND   |               | CLAYEY SAND   |       |
| 55 |    | SAND                       |             |               |               |       |
| 50 |    | SILTY<br>SAND              |             |               |               |       |
| 45 |    |                            |             | e e e e e e e |               |       |
| 40 |    |                            |             |               |               |       |

Figure 9-3. Geologic Cross-Se

97-160P(DOC)/020899

-----



Figure 9-4. Water Table Contour Map (.

and the second

÷

1

### 9.2.1 Surface Soil Contamination

. . . .

The nature and extent of surface soil contamination was evaluated using the results from four surface soil samples taken from the monitoring well MW-5 borehole and the three soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 9-1 summarizes the analytical results for surface soil samples, and Figure 9-5 shows their distribution. This assessment presents Phase II contaminant data only, because no surface soil samples were collected during Phase I.

|           |                         |                                          | Phase II S                               | amples                                   |                                          |
|-----------|-------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|
| Parameter | Reference<br>Background | SB-1<br>4A1111<br>07/11/97<br>(0-1 foot) | SB-2<br>4A1211<br>07/12/97<br>(0-1 foot) | SB-3<br>4A1311<br>07/11/97<br>(0-1 foot) | MW-5<br>4A1511<br>07/11/97<br>(0-1 foot) |
| A         | <i>R</i>                | CRA Metals (mg/                          |                                          |                                          |                                          |
| Arsenic   | 2.1                     |                                          | 0.99                                     | <u></u>                                  |                                          |
| Barium    | 14.7                    | 4.5                                      | 18.8                                     |                                          | 2.5                                      |
| Cadmium   | 0.18                    |                                          |                                          | 8                                        | 8.6                                      |
| Chromium  | 6.21                    |                                          | 0.36                                     |                                          | 0.15                                     |
| Lead      |                         | 2                                        | 3.5                                      | 2.5                                      | 3.7                                      |
| Мегсигу   | 8.81                    | 2.6                                      | 10.4                                     | 5.3                                      | 8.8                                      |
|           | 0.03                    | 0.02                                     | 0.04                                     | 0.04                                     | 0.0                                      |

| Table 9-1. | Summary of Analytical Results for Surface Soils, |
|------------|--------------------------------------------------|
|            | Burn Pit A, Fort Stewart                         |

Note: A blank indicates analyte not detected.

Note: BOLD font indicates value exceeds reference background criterion.

RCRA = Resource Conservation and Recovery Act.

SVOCs. No SVOCs were reported above the detection limit in surface soil samples.

RCRA metals. Analytical results were compared to the soil reference background concentrations presented in Section 5.0. Only those metals that exceeded their respective background concentrations are considered SRCs. Three metals have been identified as SRCs in surface soils, and they include arsenic (maximum 2.5 mg/kg at MW-5), barium (maximum 18.8 mg/kg at SB-2), cadmium (maximum 0.36 mg/kg at SB-2), lead (maximum 10.4 mg/kg at SB-2), and mercury (maximum 0.04 mg/kg at SB-2, SB-3, and MW-5).

### 9.2.2 Subsurface Soil Contamination

Subsurface soil samples were collected from all five monitoring well boreholes and the three soil borings. The samples were analyzed for VOCs and RCRA metals. Table 9-2 summarizes analytical results for the subsurface soil samples, and Figure 9-6 shows their distribution. Both Phase I and Phase II data are shown.

VOCs. BTEX compounds were identified in both Phase I and Phase II subsurface soil samples. Maximum concentrations of benzene (28  $\mu$ g/kg), toluene (12  $\mu$ g/kg), and xylenes (8.5  $\mu$ g/kg) were reported at MW-4 during Phase I. Lower concentrations of BTEX were reported at MW-1, MW-2, and SB-2.

.



Figure 9-5. Results of Analyses in Surf

je E Table 9-2. Summary of Analytical Results for Subsurface Soils Burn Pit A, Fort Stewart

(

(

|                                               |                   |             | Fnase I     | <b>Fnase I Samples</b>   |                                 |               | Dhafe II           | -                  |              |
|-----------------------------------------------|-------------------|-------------|-------------|--------------------------|---------------------------------|---------------|--------------------|--------------------|--------------|
|                                               |                   | IWM         | MW2         | MW3                      | MWA                             | ATTUP         | ruase 11 Samples   | Samples            |              |
|                                               |                   | SL1-6       | SL2-4       | SL34                     | SI.4-8                          | CINIMS 441512 | SB-1               | SB-2               | SB-3         |
| Parameter                                     | Reference         | 06/25/93    | 06/25/93    | 06/25/93                 | 06/22/93                        | 76/11/20      | 4A1112<br>07/11/97 | 4A1212<br>07/12/07 | 4A1312       |
|                                               | I nino 19vvar T   | (1391 0-+-) | (2-4  feet) | (2-4 feet)               | (6-8 feet)                      | (7-9.5 feet)  | (2.5-4.5 feet)     | (4.5-7 feet)       | (5-7 5 feet) |
|                                               |                   |             | Volatila O. |                          |                                 |               |                    |                    | 1231         |
| 2-Butanone                                    | 0.0               |             |             | Ranic comp               | viune of game compounds (ug/kg) |               |                    |                    |              |
| Acetone                                       | 00                |             | 5           |                          |                                 |               |                    | 2.4                |              |
| Benzene                                       |                   |             | 3           |                          |                                 |               |                    | A R                |              |
| Tetrachloroethene                             |                   |             |             |                          | 28                              |               |                    | 2                  |              |
| Toluene                                       |                   | 0.1         | 8.8         |                          |                                 |               |                    |                    |              |
| Vylence Tot-1                                 | 0.0               | 8.1         |             |                          | 2                               |               |                    |                    |              |
| Ayrenes, 10tal                                | 0.0               | 6.1         | 63          |                          |                                 |               |                    | 4                  |              |
|                                               |                   |             | 3.0         |                          | 8.5                             |               |                    |                    |              |
|                                               |                   | ĺ           | RCI         | RCRA Metals (malla)      | alba)                           |               |                    |                    |              |
| Arsenic                                       | 8.04              |             |             | 111 0101211              | E/nE/                           |               |                    |                    |              |
| Barium                                        | 17 00             |             |             |                          |                                 |               | <b> </b>           | 1.4                | T            |
| Chromium                                      | 0.11              |             | 6.5         | 1.4                      | 6.4                             | 14            | 10                 |                    |              |
| T and                                         | 11.60             | 9           | 2.5         | c                        | ЧY                              |               | 4.7                | 0.8                | 3.9          |
| Trau                                          | 11.10             | 9.4         | 4           | 1 5                      |                                 |               | 1.3                | 2.2                | 1.5          |
| IMERCULY                                      | 0.05              |             | 0.018       | 100                      | 7                               | 1.3           | 2.4                | 1.9                | 3            |
|                                               |                   |             | 0.42        | 1/10-0                   | 01010                           |               | 0.04               | 0.01               | 0.04         |
|                                               |                   |             | Other       | Other Inorganics (mg/kg) | ne/ke)                          |               |                    |                    |              |
| Total Organic Carbon                          | 2,200.00          |             |             |                          | à                               |               |                    |                    |              |
| Notes A LII.                                  |                   |             |             |                          |                                 |               |                    | 3210               |              |
| Note: A blank indicates analyte not detected. | nalyte not detect | ted.        |             |                          |                                 |               |                    |                    |              |

Note: BOLD font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

97-160P(EXCEL)/011499



Ć

Figure 9-6. Results of Analyses

97-160P(DOC)/020899

a modern

Ì

PCE was reported in two Phase I samples at a concentration of 8.8  $\mu$ g/kg at MW-2 and 6.1  $\mu$ g/kg at MW-1. PCE was not detected in the Phase II soil samples. Acetone and 2-butanone were detected at MW-2 and SB-2 at concentrations at or below their average concentration in background samples and are, therefore, not considered site related.

RCRA metals. Only one metal was reported at a concentration above its reference background value. The maximum value of lead (12 mg/kg at MW-4) was reported at a concentration above background in only a single Phase I soil sample. Therefore, lead does not appear to be due to releases that may have occurred from site-related activities.

#### 9.2.3 Groundwater Contamination

Groundwater contamination was evaluated using the results from water samples taken from the four monitoring wells installed during Phase I (MW-1 through MW-4) and one well (MW-5) installed during the Phase II field work at the site. These samples were analyzed for VOCs and RCRA metals. Both filtered and unfiltered water samples were collected; only the total metal analysis on unfiltered water samples is presented in this section. Table 9-3 summarizes the analytical results for groundwater samples, and Figure 9-7 shows their distribution. This assessment focuses on the Phase II contaminant data, because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

|               |                         |      | Pha                        | se II Groun                | dwater Sam                 | ples                                   |                             |
|---------------|-------------------------|------|----------------------------|----------------------------|----------------------------|----------------------------------------|-----------------------------|
| Parameter     | Reference<br>Background | MCL  | MW-1<br>4A4111<br>08/08/97 | MW-2<br>4A4211<br>08/07/97 | MW-3<br>4A4311<br>08/07/97 | MW-4<br>4A4411<br>08/08/97             | MW-5<br>54A4511<br>08/13/97 |
| Arsenic       |                         |      | RCRA Metals                | (µg/L)                     |                            |                                        |                             |
|               | 3.02                    | 50   |                            |                            |                            | ······                                 |                             |
| <u>Barium</u> | 71.72                   | 2000 | 18.5                       | (2, 5                      |                            | 1.5                                    |                             |
| Cadmium       | 0.43                    | 5    |                            | 63.5                       | 26.2                       | <u> </u>                               | 66.7                        |
| Chromium      | 3.56                    |      | 0.34                       |                            | 0.36                       |                                        |                             |
| Lead          | 4.69                    |      | 2.4                        |                            |                            | 1.5                                    | •• <u> </u>                 |
| Mercury       |                         | 15   | 1.1                        | 3.3                        | 1.8                        |                                        | 0.82                        |
|               | 0.14                    | 2    |                            | 0.2                        | 0.22                       | —————————————————————————————————————— | 0.82                        |
| Selenium      | 1.9                     | 50   |                            |                            | 0.42                       |                                        |                             |
| Silver        | 1.12                    |      | 0.14                       | 0.44                       | 0.20                       |                                        | 0.7                         |
| Note: A blant | · · ·                   |      |                            | 0.44                       | 0.28                       | 1.6                                    |                             |

#### Table 9-3. Summary of Analytical Results for Groundwater, Burn Pit A, Fort Stewart

Note: A blank indicates analyte not detected.

Note: BOLD font indicates value exceeds reference background criterion. MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act. -- = No MCL available.

VOCs. No VOCs were reported above the detection limit in Phase II groundwater samples. During Phase I, benzene was reported in MW-4 at a concentration of 51.8  $\mu$ g/L; this benzene contamination was not confirmed by the Phase II data. Although carbon disulfide and acetone were also detected in Phase I, these analytes are common laboratory artifacts and were not found in Phase II samples.

RCRA metals. Analytical results were compared to the reference groundwater background concentrations presented in Section 5.0. Only those metals that exceed their respective background concentrations are considered SRCs. Barium was reported at a maximum

 $HL^{1}$ 



ł

ĺ

Figure 9-7. Results of Analyse

97-160P(DOC)/020899

and the second s

concentration in a single groundwater sample from MW-4 (97.5  $\mu$ g/L), which is downgradient from the site. Barium was not present at concentrations above background in any other wells.

115

Mercury was reported in two wells (MW-2 and MW-3) at concentrations of 0.2 and 0.22  $\mu$ g/L, respectively. Silver exceeds its reference background concentration in a single well (MW-4 at 1.6  $\mu$ g/L).

### 9.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 9-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

| Analyte           | Surface Soil | Subsurface Soil | Groundwater     |
|-------------------|--------------|-----------------|-----------------|
| Acetone           | (µg/kg)      | (µg/kg)         | (µg/L)          |
| Benzene           | na           | 63              | nd              |
| 2-Butanone        | na           | 28              | nd              |
|                   | па           | 2.4             | nd              |
| Tetrachloroethene | na           | 8.8             |                 |
| Toluene           | na           | 12              | nd              |
| Xylenes, total    | na           | 8.5             | nd              |
|                   |              |                 |                 |
| Arsenic           | (mg/kg)      | (mg/kg)         | (µg/l)          |
| Barium            | 2.5          | brc             | brc             |
| Cadmium           | 18.8         | brc             | 97.5            |
|                   | 0.36         | nd              | <u>97.5</u> brc |
| Chromium          | brc          | brc             | brc             |
| Lead              | 10.4         | 12              |                 |
| Mercury           | 0.04         | brc             | brc             |
| Selenium          | nd           | nd              | 0.22            |
| Silver            |              | 10              | brc             |

# Table 9-4. Summary of Site-Related Constituents Burn Pit A, Fort Stewart

brc = Below background reference criteria.

na = Not analyzed in any sample in that medium.

nd = Not detected in any sample in that medium.

BOLD font indicates the analyte is an SRC in that medium.

# 9.3 FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT A

The results of contaminant migration soil screening (discussed in Section 6.0) for Burn Pit A are presented in Table 9-5. Only benzene and PCE are identified as CMCOPCs based on leaching to groundwater. However, these constituents were not currently observed in groundwater indicating that they may be degraded before reaching the water table.

| SRCs              | Maximum<br>Concentration       | GSSL                                                                                                            | Is Maximum<br>Concentration > GSSL? |
|-------------------|--------------------------------|-----------------------------------------------------------------------------------------------------------------|-------------------------------------|
| SRCS              | RCRA Metals (mg                | /kg)                                                                                                            |                                     |
| A                 | 2.5                            | 29                                                                                                              | No                                  |
| Arsenic           | 18.8                           | 1600                                                                                                            | No                                  |
| Barium            | 0.36                           | 8                                                                                                               | No                                  |
| Cadmium           | 12.0                           | 400                                                                                                             | No                                  |
| Lead              |                                | 0.40                                                                                                            | No                                  |
| Mercury           | 0.04<br>Volatile Organic Compo | the second se |                                     |
|                   | 2.4                            | 38.4                                                                                                            | No                                  |
| 2-Butanone        |                                | 800                                                                                                             | No                                  |
| Acetone           | 63.0                           | 2                                                                                                               | Yes                                 |
| Benzene           | 28.0                           | 2                                                                                                               | Yes                                 |
| Tetrachloroethene | 8.8                            |                                                                                                                 | <u>No</u>                           |
| Toluene           | 12.0                           | 600                                                                                                             |                                     |
| Xylenes           | 8.5                            | 10000                                                                                                           | No                                  |

### Table 9-5. CMCOPCs Based on Soil Screening for Burn Pit A

CMCOPCs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

RCRA = Resource Conservation and Recovery Act.

The predicted maximum concentration of benzene in groundwater, based on a maximum of 28  $\mu$ g/kg in soil and a DAF of 1, is 70  $\mu$ g/L. Through biodegradation, even if the benzene were to reach groundwater at such a concentration, it would degrade to its MCL value (5  $\mu$ g/L) in less than 3.8 times its half-life. Assuming a conservative half-life of 2.0 years, the benzene would degrade to its MCL value in 7.6 years. Similarly, PCE with its predicted maximum concentration of 15  $\mu$ g/L in groundwater, would degrade to its MCL in less than 1.6 times its half-life, or 7.2 years.

The maximum concentrations of benzene and PCE were reported in Phase I soil samples only and were not detected in Phase II samples. The most likely pathway of contaminant migration from Burn Pit A is via groundwater discharge to Mill Creek, located approximately 1500 feet east of the site. Based on the calculated groundwater flow velocity of 35 feet/year, the estimated arrival time for the site groundwater to reach Mill Creek is expected to be 43 years. Therefore, both benzene and PCE would degrade to below their MCLs within 300 feet of the Burn Pit A site, well before reaching Mill Creek.

Metals detected in the groundwater at this site include barium, lead, mercury, and selenium. However, the maximum concentrations of these metals do not exceed their respective MCLs and, based on fate and transport analysis, they are not expected to increase in the future. Therefore, none of the constituents is expected to be of concern for contaminant migration from Burn Pit A.

# 9.4 HUMAN HEALTH RISK CHARACTERIZATION OF BURN PIT A

SRCs were identified in Section 9.2 for surface soil, subsurface soil, and groundwater. The identification of risk-based screening values for all of the Burn Pits has been previously discussed in Section 7.3 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit A is given below.

#### 9.4.1 Exposure Evaluation

This site is partially covered by vegetation. In areas where active maneuvers are taking place, the grass and other vegetation have been killed as a result of physical stress (i.e., repeated wear from tanks, vehicles, etc.). Current potential receptors would include soldiers involved in training operations at the site. Hunting is not allowed in this area; however, hunting occurs in nearby areas. Therefore, a hunter is considered to be an off-site receptor. A child may visit the site, but given the types of activity occurring in this area, frequent visits are unlikely.

Future land use at this site is unlikely to change from its current use as a training area; however, the type of training taking place may change. The current receptor populations are likely to be present in the future. Construction activities may take place at the site in the future. A construction worker would be an additional potential receptor population in the future.

A summary of the potential exposure pathways for each of the receptor populations is given below.

Current and future juvenile trespasser. The juvenile trespasser may be exposed to contaminants in surface soil. Inhalation of fugitive dust, incidental ingestion, and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Off-site hunter. The hunter may be exposed to fugitive dust and contaminants that have bioaccumulated in game.

**On-site military personnel.** The on-site worker may be exposed to surface soil. Inhalation of fugitive dust, incidental ingestion, and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

**Construction Worker.** The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

#### 9.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals, identified as possible COPCs, are evaluated further below.

Arsenic in surface soil is the only contaminant that exceeds its risk-based screening concentration for direct exposure of a receptor population (Table 9-6). Arsenic in surface soil is a potential COPC at Burn Pit A, because it exceeds the risk-based screening value for exposure of a residential receptor. It should be noted that the maximum concentration of arsenic (2.5 mg/kg) is below the average concentration for soils in the eastern United States and only slightly exceeds background criteria (2.1 mg/kg) at one location, MW-5. Arsenic was not found above was used although residential land use at this site is unlikely. Therefore, arsenic is not considered a potential threat to human health at Burn Pit A.

| in to Action Levels-Burn Pit A, Fort Stewart |
|----------------------------------------------|
| Comparison (                                 |
| Table 9-6.                                   |

1

|                   |             |            |         |                                                           |             | EPA III   |       |                                        |
|-------------------|-------------|------------|---------|-----------------------------------------------------------|-------------|-----------|-------|----------------------------------------|
|                   |             | Frequency  | Minimum | Maximum                                                   | Background  | Screening | COPC? | Justification                          |
| Analyte           | Units       | of Detects | Detect  | Detect                                                    |             | CITICITA  |       |                                        |
|                   |             |            |         | Burn Fut A Durlace June June June June June June June Jun | 101 C       | 0.43      | Yes   | Max Detect >                           |
| Arsenic           | mg/kg       | 2/4        | 66'N    |                                                           |             |           |       | Screening Criteria                     |
| Barium            | mg/kg       | 4/4        | 4.5     | 18.8                                                      | 14.70       | 550       | No    | Max Detect <<br>Screening Criteria     |
| Cadmium           | mg/kg       | 2/4        | 0.15    | 0.36                                                      | 0.18        | 3.9       | No    | Max Detect <<br>Screening Criteria     |
| Lead              | mg/kg       | 4/4        | 2.60    | 10.40                                                     | 8.81        | 400       | No    | Max Detect <<br>Screening Criteria     |
| Mercury           | mg/kg       | 4/4        | 0.02    | 0.04                                                      | 0.03        | 2.30      | No    | Max Detect <<br>Screening Criteria     |
|                   |             |            | 1<br>W  | Rurn Pit A Subsurface Soil                                | urface Soil |           |       |                                        |
| Lead              | mg/kg       | LIL        | 1.3     | 12                                                        | 11.1        | 400       | No    | Max Detect <<br>Screening Criteria     |
| 2-Butanone        | mg/kg       | 1/4        | 0.0024  | 0.0024                                                    |             | 100,000   | No    | Max Detect <<br>Screening Criteria     |
| Acetone           | mg/kg       | 2/7        | 0.0045  | 0.063                                                     |             | 20,000    | No    | Max Detect <<br>Screening Criteria     |
| Benzene           | mg/kg       | 11         | 0.028   | 3 0.028                                                   |             | 200       | οN    | Max Detect <<br>Screening Criteria     |
| Tetrachloroethene | mg/kg       | 1/6        | 0.0088  | 3 0.0088                                                  |             | 110       | Ŷ     | Max Detect <<br>Screening Criteria     |
| Toluene           | mg/kg       | 2/6        | 0.004   | 4 0.012                                                   |             | 41,000    | No    | Max Detect <<br>Screening Criteria     |
| Xylenes, Total    | mg/kg       | 2/6        | 0.0082  | 2 0.0085                                                  |             | 100,000   | No    | Max Detect <<br>Screening Criteria     |
|                   |             |            |         | Burn Pit A Groundwater                                    | 1.1         |           | No    | May Detect <                           |
| Barium            | hg/L        | 4/4        | 26.2    | 2 97.5                                                    | 5 71.72     | 007       |       | Screening Criteria                     |
| Mercury           | <u>1/8п</u> | 2/4        | 0.2     | 2 0.22                                                    | 2 0.14      | 1.1       | No    | Max Detect <<br>Screening Criteria     |
| Silver            | hg/L        | 3/4        | 0.28    | 8 1.6                                                     | 6 1.12      | 18        | No    | Max Detect < All<br>Screening Criteria |
|                   |             |            |         |                                                           |             |           |       |                                        |

COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency.

97-160P(DOC)/021099

)

)
None of the other contaminants in surface soils, subsurface soils, or groundwater exceeded its respective risk-based screening value for direct exposure of a potential receptor population (Table 9-6).

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit A. A human health BRA is not warranted. No further action is required for protection of human health at Burn Pit A.

## 9.5 ECOLOGICAL RISK EVALUATION OF BURN PIT A

The ecological risk evaluation of Burn Pit A is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are considered further (PRE Steps ii through v). The results of the five steps of the PRE are reported in the following sections.

## 9.5.1 Ecological Screening Value Comparison (Step i)

Three RCRA metals were detected in groundwater at Burn Pit A at concentrations above reference background criteria. The results of the ecological screening value comparison for groundwater at Burn Pit A are presented in Table 9-7. Where detected in the four downgradient monitoring wells (MW-2 through MW-5), barium, mercury, and silver exceed the ESVs for those analytes. The ecological COPCs identified by the ESV comparison for groundwater at Burn Pit A are barium, mercury, and silver.

| Analyte  | ESV          | MW-1*                                  | <u>MW-2</u> | MW-3       | MW-4   | MW-5   | Maximum         |
|----------|--------------|----------------------------------------|-------------|------------|--------|--------|-----------------|
| <u> </u> |              | ······································ | RCRA Met    | als (µg/L) |        |        |                 |
| Barium   | <u>3.9</u> ª | <u>    18.5   J    </u>                | 63.5 J      | 26.2 J     | 97.5 J | 66.7 J | 07.5 7          |
| Mercury  | 0.0123       |                                        | 0.2         | 0.22       |        | 00.7 J | <u>97.5 J</u>   |
| Silver   | 0.012        | 0.14 J                                 | 0.44        | 0.28       | 1.6 J  |        | 0.22 J<br>1.6 J |

#### Table 9-7, EPA Region 4 Ecological Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit A, Fort Stewart

\*Upgradient monitoring well.

J = Estimated concentration.

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1996) and, where indicated, alternative values for analytes without Region 4 ESVs. RCRA = Resource Conservation and Recovery Act.

"Office of Solid Waste and Emergency Response Ambient Water Quality Criteria or Tier-II value (Suter and Tsao

Bold font indicates detected concentration exceeds ESV.

Although ecological COPCs are identified in groundwater at Burn Pit A when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit A. There are no surface water bodies in the vicinity of Burn Pit A. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit A soils.

Because there are no ESVs for soil, all analytes detected in soil at the Burn Pit A are evaluated further in PRE Steps ii through v.

## 9.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit A comprises approximately 2.3 acres (Figure 9-1). The site is essentially flat, sparsely covered with grass, and surrounded by woods. An unnamed tributary of Taylors Creek lies approximately 500 feet west of Burn Pit A, and Mill Creek, also a tributary of Taylors Creek, lies approximately 1500 feet east of the site, which is the flow direction of the surficial groundwater.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit A are described in Section 8.1.2.

## 9.5.3 Preliminary Effects Evaluation (Step iii)

In the PRE for Burn Pit A, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the facility. The derivations of TRVs are discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8.1.

## 9.5.4 Preliminary Exposure Estimate (Step iv)

Ecological receptors at Burn Pit A are likely exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews, and robins exposed to COPCs in soil are presented in Table 8-2.

## 9.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate an HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0 and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

Surface Soil. The preliminary risk calculation for shrews and robins exposed to ecological COPCs detected in soil at the Burn Pit A is presented in Table 9-8. This table shows the maximum detected concentrations and the TRVs for shrews and robins. Concentrations resulting in HQs exceeding 1.0 are shown in boldface font. The ecological COPCs present in Burn Pit A surface soil at concentrations exceeding the TRVs for the surrogate species are cadmium and

lead. Both the shrew and robin are potentially at risk from cadmium in surface soil. However, cadmium was detected in a single soil sample at a concentration (0.36 mg/kg) above background (0.18 mg/kg). Therefore, cadmium may not be site related. Lead was detected in all four surface soil samples at concentrations exceeding the TRV for robins (2.35 mg/kg). However, lead was detected in a single soil sample at a concentration (10.4 mg/kg) above background (8.81 mg/kg). Therefore, lead may not be site related. An HI is not calculated for the surrogate species exposed to RCRA metals detected in surface soil, because they are assumed to have dissimilar mechanisms of toxicity.

97-160P(DOC)/020899

# Table 9-8. Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 foot) at Burn Pit A, Fort Stewart

|         | TI    | RV    |               |               |        |              | 1                                     |
|---------|-------|-------|---------------|---------------|--------|--------------|---------------------------------------|
| Analyte | Shrew | Robin | MW-5          | SB-1          | SB-2   | SB-3         | Monimum                               |
|         |       |       | RCRA Meta     | ls (mo/ko)    |        |              | Maximum                               |
| Arsenic | 37.8  | 651   | 2.5 J         | <u></u>       | 0.00 7 |              | · · · · · · · · · · · · · · · · · · · |
| Barium  | 2627  | 2293  | 8.6 J         |               | 0.99 J |              | 2.5 J                                 |
| Cadmium | 0.32  |       |               | <u>4.5</u> J  | 18.8 J | 8 J          | 18.8 J                                |
| Lead    |       | 0.11  | <u>0.15</u> J |               | 0.36   |              | 0.36                                  |
|         | 73.3  | 2.35  | 8.8 J         | 2.6 J         | 10.4   | 52 T         |                                       |
| Mercury | 14.1  | 1.1   | 0.04          | 0.02 J        |        | <u>5.3 J</u> | 10.4                                  |
|         |       |       |               | <u>0.02</u> J | 0.04   | 0.04         | 0.04                                  |

COPCs = Chemicals/Contaminants of Potential Concern.

J = estimated concentration.

RCRA = Resource Conservation and Recovery Act.

TRV = Toxicity Reference Value = (NOAEL x BW/Food Ingestion)/BAF<sub>earthworm</sub> (see Table 8-1). Blank cells indicate analyte not detected at location.

Bold font indicates detected concentration exceeds TRV (HQ > 1).

#### 9.5.6 Uncertainties

The risk to ecological receptors from cadmium and lead in surface soil at Burn Pit A is overestimated by the preliminary risk calculations. The single cadmium detect in surface soil (0.36 mg/kg in SB-2) does not exceed the NOAEL-based TRV for shrews (0.365 mg/kg) calculated using published dietary fractions, and only barely exceeds that for the robin (0.284 mg/kg). Cadmium in surface soil at Burn Pit A is many times less than the LOAEL-based TRVs for shrews and robins calculated using published dietary fractions, 1.74 and 3.92 mg/kg, respectively. Likewise, lead was detected in two surface soil samples at Burn Pit A at concentrations exceeding the NOAEL-based TRV for the robin (5.89 mg/kg), but lead in Burn Pit A surface soil does not exceed the LOAEL-based TRV for the robin calculated using published dietary fractions (48 mg/kg). Therefore, cadmium and lead in surface soil at Burn Pit A are unlikely to pose a risk to ecological receptors.

# 9.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT A

The following are the conclusions of the Phase II RFI for Burn Pit A:

- In surface soils, arsenic, barium, cadmium, lead, and mercury were detected above background in the northern and eastern portions of the site.
- Arsenic is a potential human health COPC in surface soil at Burn Pit A because it exceeds the risk-based screening value for exposure of a residential receptor. However, the maximum concentration of arsenic is below the average concentration for regional soils, and residential land use is unlikely. Therefore, arsenic is not considered a potential threat to human health, and there are no human health COPCs in surface soil.
- Cadmium and lead are potential ecological COPCs in surface soils because they exceed the TRVs for the surrogate species (shrew and robin). However, both cadmium and lead were detected in only a single sample at a concentration above background and may, therefore, not be site related. The maximum concentration for both cadmium and lead is well within the USGS range of background concentrations for soils in the Eastern United States (2 mg/kg for

cadmium and up to 300 mg/kg for lead). The risk to ecological receptors from cadmium and lead in surface soil at Burn Pit A is overestimated by the preliminary risk calculations. As discussed in the uncertainties section (9.5.6), the maximum detected concentrations of cadmium and lead in surface soil are expected to result in exposures to the surrogate ecological receptors (shrews or robins) that are less than the lowest exposures associated with adverse effects on them (LOAEL-based TRV). Therefore, neither cadmium nor lead is considered to be an ecological COPC for Burn Pit A, and further investigation and/or evaluation of these constituents is not required.

- In subsurface soil, lead was reported in a single sample at a concentration only slightly above background. Lead is not present in a consistent pattern and is not considered site related. ٠ Similarly, BTEX compounds, PCE, 2-butanone, and acetone were identified in four of the eight subsurface soil samples, but at low concentrations that showed no consistent pattern of distribution.
- There are no human health COPCs in subsurface soil because no constituent exceeded its risk-based screening value.
- PCE and benzene are potential contaminant migration COPCs in subsurface soil because these chemicals may leach to groundwater, resulting in groundwater concentrations exceeding their respective MCLs. However, they were not detected in groundwater and were detected in only one soil sample of the seven collected. Off-site migration of these contaminants would be very limited due to retardation and biodegradation and the slow movement of groundwater. Even if these constituents were to reach groundwater at their predicted maximum concentrations, they would degrade to concentrations less than their MCLs in less than 7.6 years, and would have traveled less than 300 feet from the Burn Pit A site. Therefore, migration of these constituents is considered unlikely, and there are no contaminant migration COPCs in subsurface soil.
- In groundwater, barium, mercury, and silver were detected in separate wells at concentrations only slightly above background and, therefore, do not indicate a pattern of • distribution indicating a potential release from the former Burn Pit.
- There are no human health COPCs in groundwater because no constituent exceeded its risk-based screening value or its MCL in groundwater.
- Barium, mercury, and silver are potential ecological COPCs in groundwater because they exceed the ESVs for surface water and may present a potential risk to amphibians. However, there are no surface water bodies in the vicinity of Burn Pit A, and off-site migration would be limited due to retardation and the slow movement of groundwater. Therefore, exposure of ecological receptors to these metals in surface water bodies downgradient from Burn Pit A is not a complete pathway. Thus, no constituent in groundwater is considered a threat to ecological receptors and there are no ecological COPCs in groundwater.

The following are the recommendations for Burn Pit A:

- 1. No further action is recommended for Burn Pit A.
- 2. All potential human health COPCs were eliminated for Burn Pit A (Section 9.4), and a BRA is not recommended for the site.
- 3. All potential ecological COPCs were eliminated for Burn Pit A (Section 9.5), and an ERA is not recommended for the site.

## 10.0 CHARACTERIZATION OF BURN PIT B (SWMU 4B)

124

## **10.1 HISTORY AND DESCRIPTION OF BURN PIT B**

Burn Pit B (SWMU 4B) comprises approximately 1 acre and is located on Fort Stewart Route 90, approximately 200 feet northeast from its junction with the cut-off to State Route 196. The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees.

The site is generally flat, with elevations varying between approximately 75 and 78 feet amsl (Figure 10-1). A large pile of wood stumps and logs about 4 feet high is present in the southern portion of the site, which prevented drilling of borings beneath the pile. Drainage occurs through overland flow radically in all directions. The closest surface stream to the site is Horse Creek, a tributary of Taylors Creek, which lies approximately 4500 feet east of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 10-1). VOCs, including acetone and PCE, and RCRA metals, including arsenic, barium, chromium, lead, and mercury were detected above site-specific background levels in soils. VOCs were not detected in groundwater samples above detection limits. RCRA metals detected above site-specific background levels in groundwater included chromium and lead.

During the Phase II RFI activities, three soil borings (SB-1 through SB-3) were drilled at the site (Figure 10-1). These borings, together with the Phase I wells, were used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of slightly silty sands. Geologic cross-sections of the Burn Pit B site are presented in Figures 10-2 and 10-3. Geotechnical parameters were measured in a Shelby tube sample taken from boring SB-3 at a depth of 13 to 15 feet. The results of the Geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic sands with less than 4 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from SB-3, is  $7.07 \times 10^4$  cm/second, typical for a fine sand.

Water levels in the monitoring wells were measured during well development on July 23, 1997, and again during well sampling between August 8 and 10, 1997. The water table is present at a depth between 3 and 6 feet bgs. A water table contour map is provided as Figure 10-4. Groundwater contours measured during the Phase I RFI activities in August 1993 indicate similar groundwater flow directions and gradients, although water levels were generally 5 feet lower in August 1993 than measured in August 1997. The shallow groundwater flow across the site is to the east, discharging ultimately to Horse Creek, which is located approximately 4500 feet east of the Burn Pit B site. The water table is very flat, with a calculated horizontal hydraulic gradient across the site of 0.0006 foot/foot. Based on a hydraulic conductivity of  $7.07 \times 10^{-4}$  cm/second and a porosity of 0.365, the groundwater flow rate is approximately 1.2 feet/year toward Horse Creek.



1

(

Ć

|    | · · · · · · · · · · · · · · · · · · · | 6                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 0                                      | 120 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 80 2                                  | 40 3           | 00      |
|----|---------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|----------------|---------|
| 75 | and the set of the second second      | 1. (1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1.1)<br>1 |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       | SAND           |         |
|    | SAND<br>CLAYEY<br>SAND                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | SILTY<br>SAND                          | SIL-T<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                       | SILTY<br>SAND  |         |
| 70 |                                       | SANDY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | SAND                                   | FRAGMENTS-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                       | CLAYEY<br>SAND | -<br> - |
| 65 | CLAYEY<br>SAND<br>SILTY<br>SAND       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | CLAYEY<br>SAND                         | CL AYEY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       | SAND           |         |
| 60 | SAND<br>CLAYEY<br>SILTY<br>SAND       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | SAND W/<br>CLAY                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                | · · ·   |
|    |                                       | a a anti a a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | SAND                                   | <ul> <li>All with a weight of the second second</li></ul> | e e see e                             |                |         |
| 55 |                                       | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                |         |
| 50 |                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                |         |
| 50 |                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | · · · · · · · · · · · · · · · · · · · |                | · -     |
| 45 |                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                |         |
|    |                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1999 1999 1999 1999 1999 1999 1999 199 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                |         |
|    |                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                        | 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                       |                |         |

Figure 10-2. Geologic Cross

.

. )

ĺ

or same of board on the

ŝ

( 

Ĺ

ſ

|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 120                                                                                                                                                                                                                                | 1                                                                                                              | 80                                                                                                              | 240                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 3   |
|----------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| 75       | SILTY SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                | SB-02                                                                                                           | SB-01                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | MW4 |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | St  |
|          | SANDY<br>CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                | SILTY<br>SAND                                                                                                   | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |     |
| 70       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    | WOOD<br>PRESE                                                                                                  | FRAGMENT                                                                                                        | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |     |
|          | SILTY CI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | AYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | CLAYEY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1   |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 19. – Lenne |                                                                                                                                                                                                                                    | l I                                                                                                            |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          | SANDY -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                              | CLAYEY<br>SAND                                                                                                 | VERY                                                                                                            | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1   |
| 65       | - SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | х <sub>н</sub> к                                                                                                                                                                                                                   | · · · ·                                                                                                        | SAND                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    | <u>م</u> روند می دادد.<br>برومیرو در مد ما داده                                                                |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          | м <sup>1</sup>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | e de la seconda de la second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                    |                                                                                                                | 6 - 6                                                                                                           | ¥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 1   |
| <u> </u> | این پر در باره کار میرد و می<br>روی این از در آنسان میروند های<br>در ویسید در در آنسان میروند های                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |     |
| 60       | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    | and a second | n an                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | and a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | en de la companya de<br>la companya de la comp |                                                                                                                |                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          | nen ann an San San San San San San San San                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | n an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                    |                                                                                                                | n na ser                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
| 55       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                | سويت کې<br>د د م                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                | na ana ana ana ana ana ana ana ana ana                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | -   |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                | · · · · · · · · · · · · ·                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          | <ul> <li>A manufacture of the second sec</li></ul> |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | in an                                                                                                                                                                                          |                                                                                                                |                                                                                                                 | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |     |
| 50       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | n de la companya de l<br>La companya de la comp                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | n name a standard                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | se e opi de colo                                                                                                                                                                                                                   | · · · · · · · · ·                                                                                              | a a san a san a san a san a                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | -   |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | en e                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                    |                                                                                                                | 1                                                                                                               | e e superior<br>a superior de la superi |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | a balanti a sa ang sa                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | en e                                                                                                                                                                                           | and and an and an and an                                                   | an a                                                                        | n na serie and and a<br>Na serie a serie afo<br>Fili                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                              |                                                                                                                | ينين معرف<br>ماني ماني<br>موجود ماني م                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | ·<br>·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | -   |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | n an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |     |
| y.       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                    |                                                                                                                |                                                                                                                 | a a caracteria.<br>La caracteria de la caracter<br>La caracteria de la caracter                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |     |
| 40       | server a ser                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | · · · · · ·                                                                                                                                                                                                                        | and the second second second                                                                                   | ester de la desta de la des | and the second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 1   |

Figure 10-3. Geologic Cross-

-

 $\langle \, \rangle$ 

1

 $\langle \rangle$ 

97-160P(DOC)/020999

- ----



K

Figure 10-4. Water Table Conto

Comparison Comparison (Comparison)

"INVOL.

## 10.2 NATURE AND EXTENT OF CONTAMINATION AT BURN PIT B

#### 10.2.1 Surface Soil Contamination

The nature and extent of surface soil contamination was evaluated using the results of analyses on surface soil samples taken from the three soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 10-1 summarizes the analytical results for surface soil samples, and Figure 10-5 shows their distribution. This assessment presents Phase II contaminant data only because no surface soil samples were collected during Phase I.

|           |                         |                                          | Phase II Samples                         |                                          |
|-----------|-------------------------|------------------------------------------|------------------------------------------|------------------------------------------|
| Parameter | Reference<br>Background | SB-1<br>4B1111<br>07/12/97<br>(0-1 foot) | SB-2<br>4B1211<br>07/17/97<br>(0-1 foot) | SB-3<br>4B1311<br>07/12/97<br>(0-1 foot) |
|           | R                       | CRA Metals (mg/kg)                       |                                          |                                          |
| Barium    | 14.7                    | 11.5                                     | 14.4                                     | 10.9                                     |
| Chromium  | 6.21                    | 1.4                                      | 1.4                                      |                                          |
| Lead      | 8.81                    | 7.8                                      | 5.4                                      | 1.5                                      |
| Mercury   | 0.03                    | 0.03                                     | 0.03                                     | <u> </u>                                 |

# Table 10-1. Summary of Analytical Results for Surface Soils Burn Pit B, Fort Stewart

Note: **BOLD** font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

SVOCs. No SVOCs were reported above the detection limit in surface soil samples.

RCRA metals. Lead (maximum 10.5 mg/kg at SB-3) and mercury (0.04 mg/kg at SB-3) were reported above reference in a single boring (SB-3). No other metals were detected above background.

### 10.2.2 Subsurface Soil Contamination

The subsurface soil samples were collected from the four Phase I monitoring well boreholes and the three Phase II soil borings. The samples were analyzed for VOCs and RCRA metals, Table 10-2 summarizes analytical results for the subsurface soil samples, and Figure 10-6 shows their distribution. Both Phase I and Phase II data are shown.

**VOCs.** BTEX compounds were identified in both Phase I and Phase II subsurface soil samples. Maximum concentrations of benzene ( $30 \mu g/kg$  at MW-4), toluene ( $60.8 \mu g/kg$  at SB-2), and xylenes ( $9.9 \mu g/kg$  at MW-4) were reported. Lower concentrations of BTEX, notably toluene, were reported also at MW-1, SB-1, SB-2, and SB-3. The extent of BTEX contamination, although present at relatively low concentrations, extends over most of the 1-acre site.



ſ

ſ

C

Figure 10-5. Results of Analyse

)

Table 10-2. Summary of Analytical Results for Subsurface Soils Burn Pit B, Fort Stewart

ſ

ſ

(

|                                     |                 |                                               | Phase I                  | <b>Phase I Samples</b>           |            | đ              | Phase II Samulae |              |
|-------------------------------------|-----------------|-----------------------------------------------|--------------------------|----------------------------------|------------|----------------|------------------|--------------|
|                                     |                 | IWM                                           | MW2                      | EWM3                             | MW4        | SB-1           | SR-2             | 583          |
|                                     |                 | SL1-8                                         | SL2-8                    | SL3-8                            | SL4-4      | 4B1112         | <b>4R1717</b>    | 401317       |
| -                                   | Reference       | 06/25/93                                      | 06/25/93                 | 06/25/93                         | 06/25/93   | 07/11/07       | 212121           | 710101-0     |
| rarameter                           | Background      | (4-6 feet)                                    | (2-4 feet)               | (2-4 feet)                       | (6-8 feet) | (2.5-4.5 feet) | -                | (5-7.5 feet) |
|                                     |                 | Volat                                         | le Oronnie (             | Volatile Organic Communds ( ~ 4) |            |                |                  | Inni cu al   |
| 2-Butanone                          |                 |                                               |                          | minodulo                         | TR/KR)     |                |                  |              |
| Acetone                             |                 |                                               |                          |                                  |            |                | 33.1             |              |
| Benzaug                             | 0.0             |                                               |                          |                                  | 110        | 7.6            | 42.8             |              |
| DOINCOLO                            | 0.0             |                                               |                          |                                  | 30         |                |                  |              |
| Methylene Chloride                  | 0.0             |                                               |                          |                                  | 2          |                |                  |              |
| Tetrachloroethene                   | 00              | 0                                             |                          |                                  |            |                | 5                |              |
| Tolnana                             |                 | <u>, , , , , , , , , , , , , , , , , , , </u> |                          |                                  |            |                |                  |              |
|                                     | 0.0             | 8.3                                           |                          |                                  |            | 5 2            | 0.03             |              |
| <b>Aylenes</b> , <b>Total</b>       | 0.0             |                                               |                          |                                  | d          | 1              | 0.00             | 0.0          |
|                                     |                 |                                               |                          |                                  |            |                |                  |              |
|                                     |                 |                                               | RCRA Metals (mo/ko)      | als (mo/ko)                      |            |                |                  |              |
| Arsenic                             | 8.04            | 14                                            |                          | 1 00                             |            |                |                  |              |
| Barium                              | 17 M            |                                               |                          |                                  |            | 0.31           | 0.5              | 0.76         |
| Chromium                            | 11 60           | v č                                           |                          | 3.7                              | 9.8        | 6.7            | 11.6             | 5.7          |
| ead                                 | 00.11           | 3                                             | 7.7                      | 4.5                              | 7.1        | 1.7            | 1.5              | 5            |
|                                     |                 | 7.3                                           | 2.2                      | 2.7                              | 4.2        | 1.0            | 2.5              | 36           |
| wickcury                            | 0.05            | 0.013                                         |                          |                                  | 0.021      | 0.06           |                  | 0.0          |
|                                     |                 |                                               |                          |                                  |            | 200            | +0.0             |              |
|                                     |                 | -                                             | Other Inorganics (mg/kg) | nics (mg/kg)                     |            |                |                  |              |
| Total Organic Carbon                | 2,200           |                                               |                          |                                  |            |                |                  |              |
|                                     |                 |                                               |                          |                                  |            |                |                  | 546          |
| Note: A blank indicates analyte not | te not detected |                                               |                          |                                  |            |                |                  |              |

Note: A blank indicates analyte not detected. Note: BOLD font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

97-160P(EXCEL)/011499

.



Ţ

ĺ

K.

Figure 10-6. Results of Analyse

);

1

97-160P(DOC)/020999

PCE was reported in a single Phase I sample at a concentration of 8.9  $\mu$ g/kg at MW-1. PCE was not detected in the Phase II soil samples. Acetone (110 µg/kg), 2-butanone (33.1 µg/kg), and methylene chloride (5  $\mu$ g/kg) were detected at SB-1, SB-2, and MW-4.

RCRA metals. Two metals were reported at concentrations above their reference background value. Maximum values of chromium (13 mg/kg at MW-1) and mercury (0.06 mg/kg at SB-1) were reported at concentrations only slightly above background.

#### **10.2.3 Groundwater Contamination**

T-12 40 0

Groundwater contamination was evaluated using the results from water samples obtained during the resampling of the four monitoring wells installed during Phase I (MW-1 through MW-4). These samples were analyzed for VOCs and RCRA metals. Both filtered and unfiltered water samples were collected; only the total metal analysis on unfiltered water samples is presented in this section. Table 10-3 summarizes the analytical results for groundwater samples, and Figure 10-7 shows their distribution. This assessment focuses on the Phase II contaminant data, because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

|           |                                       |      | P                         | hase II Sample             | 5                          |                            |
|-----------|---------------------------------------|------|---------------------------|----------------------------|----------------------------|----------------------------|
| Parameter | Reference<br>Background               | MCL  | MW-1<br>4B111<br>08/10/97 | MW-2<br>4B4311<br>08/09/97 | MW-3<br>4B4411<br>08/09/97 | MW-4<br>4B4411<br>08/09/97 |
|           | · · · · · · · · · · · · · · · · · · · | RC   | CRA Metals (ug/L          | .)                         |                            |                            |
| Barium    | 71.72                                 | 2000 | 21                        | 23.9                       |                            |                            |
| Cadmium   | 0.43                                  | 5    |                           |                            |                            | 73.4                       |
| Chromium  | 3.56                                  | 100  | 24                        | 1.2                        | ·                          |                            |
| Lead      | 4.69                                  |      | 2.4                       | 5.9                        | 3.7                        | 0.61                       |
| Selenium  |                                       | 15   | 2                         |                            | 2.2                        |                            |
|           | 1.90                                  | 50   |                           | 3.2                        |                            | 3.5                        |
| Silver    | 1.12                                  |      | 0.39                      | 0.53                       | 0.14                       | 4.9                        |

| 1 able 10-3. | Summary of Analytical Results for Groundwater |
|--------------|-----------------------------------------------|
|              | Burn Pit B, Fort Stewart                      |

Note: A blank indicates analyte not detected.

Note: BOLD font indicates value exceeds reference background criterion.

MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act.

VOCs. No VOCs were reported above the detection limit in either Phase I or Phase II groundwater samples. These results suggest that, although low concentrations of BTEX are present in subsurface soils at the site, the soil contamination has not impacted groundwater.

RCRA metals. Five metals were reported at concentrations above background in groundwater. Barium was reported at a maximum concentration in MW-4 (73.4 µg/L); cadmium in MW-2 (1.2 µg/L); chromium in MW-2 and MW-3 (5.9 and 3.7 µg/L, respectively); selenium in both MW-2 and MW-4 (3.2 and 3.5 µg/L, respectively); and silver in MW-4 (4.9 µg/L). These metals are present at concentrations only slightly exceeding background.

97-160P(DOC)/020999



Figure 10-7. Results of Analy

)

( )

97-160P(DOC)/020999

ł

### 10.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 10-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

| Analyte            | Surface Soil | Subsurface Soil | Groundwater |
|--------------------|--------------|-----------------|-------------|
|                    | (µg/kg)      | (µg/kg)         | (µg/L)      |
| Acetone            | па           | 110             | nd          |
| Benzene            | na           | 30              | nd          |
| 2-Butanone         | па           | 33.1            | nd          |
| Methylene Chloride | na           | 5               | nd          |
| Tetrachloroethene  | na           | 8.9             | nd          |
| Toluene            | na           | 60.8            | nd          |
| Xylenes, total     | na           | 9.9             | nd          |
|                    | (mg/kg)      | (mg/kg)         | (µg/L)      |
| Arsenic            | nd           | brc             | nd          |
| Barium             | brc          | brc             | 73.4        |
| Cadmium            | nd           | nd              | 1.2         |
| Chromium           | brc          | 13              | 5.9         |
| Lead               | 10.5         | brc             | brc         |
| Mercury            | 0.04         | 0.06            | nd          |
| Selenium           | nd           | nd              | 3.5         |
| Silver             | nd           | nd              | 4.9         |

# Table 10-4. Summary of Site-Related Constituents Burn Pit B, Fort Stewart

brc = Below background reference criteria

na = Not analyzed in any sample in that medium

nd = Not detected in any sample in that medium

BOLD font indicates the analyte is an SRC in that medium

## **10.3 FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT B**

The results of contaminant migration soil screening (discussed in Section 6.0) for Burn Pit B are presented in Table 10-5. Only benzene and PCE are identified as CMCOPCs based on leaching to groundwater. However, these constituents were not currently observed in groundwater, indicating that they may be degraded before reaching the water table. The predicted maximum concentration of benzene in groundwater, based on a maximum of 30  $\mu$ g/kg in soil and a DAF of 1, is 75  $\mu$ g/L. Through biodegradation, even if the benzene were to reach groundwater at such a concentration, it would degrade to its MCL value (5  $\mu$ g/L) in less than 3.9 times its half-life. Assuming a conservative half-life of 2.0 years, the benzene would degrade to its MCL value in 7.8 years. Similarly, PCE with its predicted maximum concentration of 15  $\mu$ g/L in groundwater would degrade to its MCL in less than 1.6 times its half-life, or 7.2 years, and methylene chloride with its predicted maximum concentration of 25  $\mu$ g/L in groundwater would degrade to its MCL in less than 2.3 times its half-life, or 0.7 years.

146

The maximum concentrations of benzene and PCE were reported in Phase I soil samples only and were not detected in Phase II samples. The most likely pathway of contaminant migration from Burn Pit B is via groundwater discharge to Horse Creek, located approximately 4500 feet east of the site. Based on the calculated groundwater flow velocity of 1.2 feet/year, the estimated arrival time for the site groundwater to reach Horse Creek is expected to be more than 3750 years. Therefore, benzene, PCE, and methylene chloride would degrade to less than their MCLs within 10 feet of the Burn Pit B site, well before reaching Horse Creek.

| SRCs               | Maximum<br>Concentration | GSSL             | Is Maximum<br>Concentration > GSSL? |
|--------------------|--------------------------|------------------|-------------------------------------|
| UNCO               | RCRA Meta                | ıl (mg/kg)       |                                     |
| Chromium           | 13.00                    | 38               | No                                  |
| Lead               | 10.5                     | 400              | No                                  |
| Mercury            | 0.06                     | 0.4              | No                                  |
|                    | Volatile Organic C       | ompounds (µg/kg) | ·····                               |
| 2-Butanone         | 33.10                    | 38.4             | No                                  |
| Acetone            | 110.00                   | 800              | No                                  |
| Methylene Chloride | 5                        | 1                | Yes                                 |
| Benzene            | 30.00                    | 2                | Yes                                 |
| Tetrachloroethene  | 8.90                     | 3                | Yes                                 |
| Toluene            | 60.80                    | 600              | No                                  |
| Xylenes            | 9.90                     | 10000            | No                                  |

| Table 10-5. CMCOPCs Based on Soi | il Screening for | Burn Pit B |
|----------------------------------|------------------|------------|
|----------------------------------|------------------|------------|

CMCOPs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

RCRA = Resource Conservation and Recovery Act.

Metals detected in the groundwater at this site include barium, cadmium, chromium, selenium, and silver. However, the maximum concentrations of these metals do not exceed their respective MCLs and, based on fate and transport analysis, they are not expected to increase in the future. Therefore, none of the constituents are expected to be of concern for contaminant migration from Burn Pit B.

### 10.4 HUMAN HEALTH RISK EVALUATION OF BURN PIT B

SRCs were identified in Section 10.2 for surface soil, subsurface soil, and groundwater. The identification of risk-based action levels for all of the Burn Pits has been previously discussed in Section 4.7 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit B is given below.

#### **10.4.1 Exposure Evaluation**

This site is completely covered by vegetation, and no VOCs were detected in surface soils. Therefore, exposure via inhalation based on current land use is not a viable exposure pathway. The site is not currently used by the U.S. Army. Therefore, current on-site receptors include a juvenile trespasser and a hunter. Current off-site receptors would be represented by an off-site hunter. Land use at this site is unlikely to change in the future. However, the U.S. Army may use the site as a training area in the future. Construction may occur on the site or the vegetative cover may be gone as a result of on-site activities. Future on-site receptors include military personnel and a construction worker. Future off-site receptors include hunters in the area.

A summary of the potential exposure pathways for each of the receptor populations is given below.

**Current juvenile trespasser.** The juvenile trespasser may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Hunter. The hunter is representative of a current on-site receptor and a current and future off-site receptor. The current on-site receptor is not likely to be exposed to chemicals given the absence of an air migration pathway, and clothing the hunter would be wearing would eliminate potential dermal exposure. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

The current off-site hunter may be exposed via ingestion of contaminants bioaccumulated in game animals. The future off-site hunter may be exposed via ingestion of game and inhalation of fugitive dust if the vegetative cover was removed.

Future on-site military personnel. The on-site military personnel may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Future construction worker. The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

#### 10.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals will be evaluated further in a BRA.

None of the contaminants in surface soils or subsurface soils exceeded its respective risk-based screening value for direct exposure of a potential receptor population (Table 10-6).

None of the contaminants in groundwater exceeded its screening value (Table 10-6).

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit B. A human health BRA is not warranted. No further action is required for protection of human health.

Table 10-6. Comparison to Action Levels-Burn Pit B, Fort Stewart

|                                       |                    |                |            |                           |             | EPA III               |          |                                     |
|---------------------------------------|--------------------|----------------|------------|---------------------------|-------------|-----------------------|----------|-------------------------------------|
|                                       |                    | Frequency of   | Minimum    | Maximum                   | Background  | Screening<br>Criteria | COPC?    | Justification                       |
| Analyte                               | Units              | Detects        | Detect     | Rurn Pit R Surface Soil   | ace Soil    |                       |          |                                     |
|                                       |                    |                | 10         |                           | ł           |                       | No       | Max Detect < Screening Criteria     |
|                                       | mø/kø              | 3/3            | 5.4        | 10.5                      | 8.81        | 004                   | 221      |                                     |
| Lead                                  |                    | 5/2            | 0.03       | 0.04                      | 0.03        | 2.30                  | No       | Max Detect < Screening Uniteria     |
| Mercury                               | mg/kg              |                | Ruy<br>Ruy | Rum Pit R Subsurface Soil | irface Soil |                       |          |                                     |
|                                       |                    |                |            | 12                        |             | 1.000                 | No       | Max Detect < Screening Criteria     |
| Chromium                              | mg/kg              | 6/6            | <u>ci</u>  | <u>-</u>                  |             |                       | N.S.     | Max Detect < Screening Criteria     |
| Marcury                               | mg/kg              | 4/6            | 0.013      | 0.06                      | 0.05        |                       |          | Max Detection Commission Criteria   |
| INTELCULY                             |                    |                | 0.0331     | 0.0331                    |             | 100,000               | No<br>No | Max Detect < Screening Cincila      |
| 2-Butanone                            | mg/Kg              | C/I            | 72000      |                           |             | 20,000                | 0N<br>N  | Max Detect < Screening Criteria     |
| Acetone                               | mg/kg              | 3/6            | 0/00/0     |                           |             | 200                   | °N<br>N  | Max Detect < Screening Criteria     |
| Danrene                               | mg/kg              | 1/5            | 0.03       | 0.03                      |             |                       |          | Consenting Criteria                 |
| DCIIZCIIC                             |                    | 1/6            | 0 0080     | 0.0089                    | 6           | 110                   | °Z       | Max Detect < Screening Chicking     |
| T etrachloroethene                    | mg/kg              | C/I            |            |                           |             | 41.000                | 0N<br>No | Max Detect < Screening Criteria     |
| Toluene                               | mg/kg              | 4/6            | 0.0052     | 0.0000                    | 0           |                       |          | May Detect < Screening Criteria     |
|                                       | mo/ko              | 1/5            | 6600.0     | 6600.0                    | 6           | 100,000               | ov,      | Max Depen                           |
| Xylenes, 1 0tat                       | 9. A.              |                |            | Rurn Pit B Groundwater    | oundwater   |                       |          |                                     |
|                                       |                    |                |            |                           | CL 12       | 51 260                | 0N (     | Max Detect < Screening Criteria     |
| Banium                                | Π/gμ               | 3/3            | 71         | 1.01                      |             |                       |          | May Detect < Screening Criteria     |
|                                       |                    | 1/3            | 1.2        |                           | 1.2 0.43    | 3                     |          |                                     |
| Caomun                                |                    | 10             | 0.61       |                           | 5.9 3.56    | 6 18                  | N.       | Max Detect < All Screening Cliferia |
| Chromium                              | л/дн               |                | C.E.       |                           | 3.5 1.90    | 0 18                  | No<br>No | Max Detect < Screening Criteria     |
| Selenium                              | μg/L               | C17            |            |                           | 1 12        |                       | 18 No    | Max Detect < Screening Criteria     |
| Silver                                | µg/L               | 3/3            | 66.0       |                           |             |                       |          |                                     |
| 222 Charlenting of Potential Concern. | antominant of Pote | ntial Concern. |            |                           |             |                       |          |                                     |

COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency.

97-160P(DOC)/020999

## **10.5 ECOLOGICAL RISK EVALUATION OF BURN PIT B**

The ecological risk evaluation of Burn Pit A is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are evaluated further (Steps ii through v). The results of the five steps of the PRE are reported below.

## 10.5.1 Ecological Screening Value Comparison (Step i)

Five RCRA metals were detected in groundwater at Burn Pit B at concentrations exceeding reference background criteria. The results of the ecological screening value comparison for groundwater at Burn Pit B are presented in Table 10-7. Where detected in the three downgradient monitoring wells (MW-1, MW-2, and MW-4), barium and silver exceed the ESVs for those analytes. Cadmium was only detected in MW-2, and it exceeds the ESV. The ecological COPCs identified by the ESV comparison for groundwater at Burn Pit B are barium, cadmium, and silver.

|          | ESV                |          |               | 1            |          | T             |
|----------|--------------------|----------|---------------|--------------|----------|---------------|
| Analyte  |                    | MW-3*    | MW-1          | MW-2         | MW-4     | Maximum       |
|          |                    | RCRA M   | letals (µg/L) |              |          |               |
| Barium   | 3.94               |          | 21 J          | 23.9 J       | 73.4 J   |               |
| Cadmium  | 0.66*              | <u> </u> |               |              | <u> </u> | <u>73.4 J</u> |
| Chromium | 117 <sup>b.c</sup> | 3.7 J    | 24.7          | 1.2          | <u> </u> | 1.2           |
| Selenium |                    | 5.7 3    | 2.4 J         | <u>5.9 J</u> | 0.61 J   | 5.9 J         |
|          |                    |          |               | 3.2 J        | 3.5 J    | 3.5 J         |
| Silver   | 0.012              | 0.14 J   | 0.39 J        | 0.53         | 4.9      | 4.9           |

\*Upgradient monitoring well.

J = Estimated concentration.

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1996) and, where indicated, alternative values for analytes without Region 4 ESVs.

RCRA = Resource Conservation and Recovery Act.

<sup>a</sup>Office of Solid Waste and Emergency Response Ambient Water Quality Criteria or Tier-II value (Suter and Tsao 1996).

<sup>b</sup>Hardness dependent, assumes 50 mg/L CaCO<sub>3</sub>

<sup>c</sup>Chromium (III); hardness dependent, assumes 50 mg/L CaCO<sub>3</sub>; ESV for Chromium (VI) = 11.

Bold font indicates detected concentration exceeds ESV.

Although ecological COPCs are identified in groundwater at Burn Pit B when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit B. There are no surface water bodies in the vicinity of Burn Pit B. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit B soils.

Because there are no ESVs for soil, all analytes detected in soil at the Burn Pit B are evaluated further in PRE Steps ii through v.

150

### 10.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit B comprises approximately 1 acre (Figure 10-1). The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees. A large pile of wood stumps and logs about 4 feet high is present in the southern portion of the site. The closest surface stream is Horse Creek, a tributary of Taylors Creek, which lies approximately 4500 feet east of the site.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit B are described in Section 8.1.2.

### 10.5.3 Preliminary Effects Evaluation (Step iii)

In the PRE for Burn Pit B, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the site. The derivation of TRVs is discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8-1.

## 10.5.4 Preliminary Exposure Estimate (Step iv)

Ecological receptors at the Burn Pit B are likely to be exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews, and robins exposed to COPCs in soil are presented in Table 8-2.

#### 10.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate a HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0, and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

**Surface Soil.** The preliminary risk calculation for shrews and robins exposed to ecological COPCs detected in soil at Burn Pit B is presented in Table 10-8. This table shows the maximum detected concentrations and the TRVs for shrews and robins. Concentrations resulting in HQs exceeding 1.0 are shown in boldface font. The ecological COPC present in Burn Pit B surface soil at concentrations exceeding the TRV for the surrogate species is lead. Lead was detected in all three samples at concentrations exceeding the TRV for the robin. However, lead was detected in a single soil sample at a concentration (10.5 mg/kg) above background criteria (8.81 mg/kg). Therefore, lead may not be site related. An HI is not calculated for the surrogate species exposed to RCRA metals detected in surface soil because they are assumed to have dissimilar mechanisms of toxicity.

#### Table 10-8. Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 foot) at Burn Pit B, Fort Stewart

| TRV (mg/kg) |       | mg/kg) |                 |      | <u> </u> | - <u></u> |
|-------------|-------|--------|-----------------|------|----------|-----------|
| Analyte     | Shrew | Robin  | SB-1            | SB-2 | SB-3     | Maximum   |
|             |       |        | RA Metals (mg/k | (g)  |          |           |
| Lead        | 73.3  | 2.35   | 7.8             | 5.4  | 10.5     | 10.5      |
| Мегсигу     | 14.1  | 1.1    | 0.03 J          | 0.03 | 0.04     | 0.04      |

COPCs = Chemicals/Contaminants of Potential Concern.

J = Estimated concentration.

RCRA = Resource Conservation and Recovery Act.

 $TRV = Toxicity Reference Value = NOAEL \times BW/Food Ingestion)/BAF_{earthworm}$  (see Table 8-1). Blank cells indicate analyte not detected at site

Bold font indicates detected concentration exceeds TRV (HQ > 1).

#### **10.5.6 Uncertainties**

The risk to ecological receptors from lead in surface soil at Burn Pit B is overestimated by the preliminary risk calculations. Lead was detected in the three surface soil samples at concentrations ranging from 5.4 to 10.5 mg/kg. Lead was detected in two surface soil samples at Burn Pit B at concentrations exceeding the NOAEL-based TRV for the robin with a realistic diet (5.89 mg/kg). However, lead concentrations in Burn Pit A surface soil are many times less than the LOAEL-based TRV for the robin calculated using published dietary fractions (48 mg/kg). Therefore, lead in surface soil at Burn Pit A is unlikely to pose a risk to ecological receptors.

## 10.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT B

The following are the conclusions of the Phase II RFI for Burn Pit B:

- In surface soils, lead and mercury were detected in one of the three surface soil samples • taken from the site. Neither lead nor mercury poses a human health risk.
- Lead is a potential ecological COPC in surface soil because it exceeds the TRV for the ۰ surrogate species (robin). However, lead was detected in only a single sample at a concentration above background and may not be site related. The maximum concentration of lead was well within the USGS range of background concentrations for soils in the Eastern United States (up to 300 mg/kg) and many times less than the LOAEL-based TRV. Therefore, lead is estimated as an ecological COPC in surface soil at Burn Pit B, and further investigation and/or evaluation of this constituent is not required.
- In subsurface soils, PCE and BTEX compounds were detected in five of the seven samples, • extending over most of the site. 2-Butanone, acetone, and methylene chloride were also detected. Chromium and mercury were reported in one sample at concentrations only slightly above background. These metals are not present in a consistent pattern and are not considered site related.
- There are no human health COPCs in surface soil or subsurface soil because no constituent exceeded its risk-based screening value.

- PCE, benzene, and methylene chloride are potential contaminant migration COPCs in subsurface soil because these chemicals may leach to groundwater, resulting in groundwater concentrations exceeding their respective MCLs. However, they were not detected in groundwater and were detected in only one soil sample of the nine collected. Off-site migration of these contaminants would be very limited due to retardation and biodegradation and the slow movement of groundwater. Even if these constituents were to reach groundwater at their predicted maximum concentrations, they would degrade to concentrations less than their MCLs in less than 7.8 years, and would have traveled less than 10 feet from the Burn Pit B site. Therefore, migration of these constituents is considered unlikely, and there are no contaminant migration COPCs in subsurface soils.
- In groundwater, five metals were detected at concentrations only slightly exceeding background and in separate wells. These metals (barium, cadmium, chromium, selenium, and silver) do not indicate a pattern of distribution indicating a potential release from the former Burn Pit.
- There are no human health COPCs in groundwater because no constituent exceeded its risk-based screening value.
- Barium, cadmium, and silver are ecological COPCs in groundwater because they exceed the ESVs for surface water. However, there are no surface water bodies in the vicinity of Burn Pit B, and off-site migration would be limited due to retardation and the slow movement of groundwater. Therefore, exposure of ecological receptors to these metals in surface water bodies downgradient from Burn Pit B is not a complete pathway. Thus, no constituent in groundwater is considered a threat to ecological receptors, and there are no ecological COPCs in groundwater.

The following are the recommendations for Burn Pit B:

- 1. No further action is recommended for Burn Pit B.
- 2. Because there are no human health COPCs, a BRA is not recommended for Burn Pit B.
- 3. All potential ecological COPCs were eliminated for Burn Pit B (Section 10.5), and an ERA is not recommended for the site.

# 11.0 CHARACTERIZATION OF BURN PIT C (SWMU 4C)

IN

## 11.1 HISTORY AND DESCRIPTION OF BURN PIT C

Burn Pit C (SWMU 4C) comprises approximately 2 acres and is located on Fort Stewart Route 90, approximately 300 feet west from its junction with 15th Street. The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees.

The site is slightly mounded, with elevations varying between approximately 71 feet amsl around the perimeter of the site and 76 feet amsl in the center of the site (Figure 11-1). Drainage occurs through overland flow radically in all directions and is collected in a series of man-made ditches that convey runoff to Mill Creek. The closest surface stream to the site is Mill Creek, a tributary of Taylors Creek, which lies approximately 4000 feet northeast of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 11-1). RCRA metals, including barium, chromium, and lead were detected above site-specific background levels in soils. RCRA metals were also detected above site-specific background levels in groundwater and included arsenic, chromium, lead, and silver. No VOCs were detected in either soil or groundwater samples in Phase I.

During the Phase II RFI activities, three new monitoring wells (MW-5 through MW-7) were installed, and three soil borings (SB-1 through SB-3) were drilled at the site (Figure 11-1). Redevelopment of the existing Phase I wells was attempted, but the wells remained very turbid even after several hours of purging. The four Phase I wells were, therefore, abandoned by filling the well casing with grout. The Phase II wells and borings were therefore used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of silty to clayey sands. Geologic cross-sections of the Burn Pit C site are presented in Figures 11-2 and 11-3. Geotechnical parameters were measured in bulk soil samples taken from MW-5 at a depth of 7.5 to 10 feet and MW-7 at a depth of 10 to 12.5 feet, and in a Shelby tube sample taken from MW-6 at a depth of 12 to 14 feet. The results of the Geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic or low-plasticity silty to clayey sands with less than 12 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from MW-6, is  $1.24 \times 10^{-3}$  cm/second, typical for a fine sand.

Water levels in the monitoring wells were measured during well development between July 11 and 15, 1997, and again during well sampling on August 6 and 7, 1997. The water table is present at a depth between 2 and 3 feet bgs around the perimeter of the site, and is likely mounded beneath the center of the site. A water table contour map is provided as Figure 11-4, showing an interpreted water table mound beneath the site with radial flow in all directions. Groundwater contours measured during the Phase I RFI activities in August 1993 indicate similar water levels, although the groundwater flow direction was interpreted to be planar flow to the east. The current interpretation is that groundwater may be slightly perched within the clayey soils beneath the site, forming a mound beneath the site; groundwater flow is likely radial, flowing in all directions before joining the regional shallow groundwater flow. Regional groundwater flow is expected to be to the northeast, discharging ultimately to Mill Creek, which is located approximately 4000 feet northeast of the Burn Pit C site.

parente anno 11

wiw^~/


Figure 11-1. Locations of S:

.

A REPORT OF A R

|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 120                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 180                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 240                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 300                | 360              |
|----|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|------------------|
| 75 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 88                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Sector of the se | AND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    | SAND             |
| 70 | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | e manana ang ang ang ang ang ang ang ang an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | an hair a she a ga |                  |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | W(                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ILTY SAND &<br>DOD DEBRIS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SILT<br>WOOD       | SAND &<br>DEBR1S |
| 65 | SILTY-<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
| 60 | CLAYEY SILTY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | Y                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
| 55 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | <ul> <li>And Andrew State State State State State</li> <li>And Andrew State State State State State</li> <li>Andrew State State State State State State</li> <li>Andrew State State State State State</li> <li>Andrew State State State State</li> <li>Andrew State State State</li> <li>Andrew State State State</li> <li>Andrew State State State</li> <li>Andrew State State</li> <li>Andrew State State</li> <li>Andrew Stat</li></ul> | n han san fan 'e ser ser ser ser ser<br>New Yester ser ser ser ser ser ser ser ser ser s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and the second sec | en al construction de la |                    |                  |
| 50 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
| 45 | and the second sec                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |
| 40 | Provide and the second seco | n an an an ann an an an an ann an an an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                    |                  |

Figure 11-2. Geologic Cros

The second

~

Ĉ

٠,

|    |   |                                                                                                                                                                                                                                                                                                              | 50                                                                                                        | 1:   | 20 1      | 80 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 240 | 3                                                                             | 00                                           |
|----|---|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|------|-----------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|-------------------------------------------------------------------------------|----------------------------------------------|
| 75 | B |                                                                                                                                                                                                                                                                                                              |                                                                                                           | 2    | SB-02     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     | یون میں<br>دیک در ایک<br>ایک در ایک<br>ایک در ایک<br>ایک در ایک<br>ایک در ایک |                                              |
| 70 |   |                                                                                                                                                                                                                                                                                                              | 1                                                                                                         | SANE | SILTY     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     |                                                                               |                                              |
| 65 |   | SAND 7                                                                                                                                                                                                                                                                                                       |                                                                                                           | SIL  | TY SAND & | WOOD<br>DEBRIS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |     | SAN                                                                           | , <u>7</u>                                   |
| 60 |   |                                                                                                                                                                                                                                                                                                              |                                                                                                           | -    |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     | NO                                                                            | REC                                          |
| 55 |   | SILTY<br>SAND                                                                                                                                                                                                                                                                                                |                                                                                                           |      |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     |                                                                               | н (р. с. |
| 50 |   |                                                                                                                                                                                                                                                                                                              |                                                                                                           |      |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     | · · · · · · · · · · · · · · · · · · ·                                         |                                              |
| 45 |   | enementar que en estas de una por o origan<br>en presentar una encontra de una porte en any<br>en presentar en encontra de una porte<br>en presentar en any encontra de una porte<br>en presentar en any encontra de una<br>encontra encontra de la presentar en<br>encontra encontra de la presentar en any | n na San ay<br>Mara Nang<br>San San San San San<br>Mara San San San San<br>San San San San San<br>San San |      |           | In the second se<br>Second second secon<br>second second sec |     | na se                                                                         |                                              |
| 40 |   |                                                                                                                                                                                                                                                                                                              | n an                                                                  |      |           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |     |                                                                               |                                              |

Figure 11-3. Geologic Cr

4

k.,

)

( )



Figure 11-4. Water Table Contour

a contractor or

THIS PAGE INTENTIONALLY LEFT BLANK

# 11.2 NATURE AND EXTENT OF CONTAMINATION AT BURN PIT C

### **11.2.1 Surface Soil Contamination**

The nature and extent of surface soil contamination was evaluated using the results of analyses on surface soil samples taken from the three new monitoring wells (MW-5 through MW-7) and the three soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 11-1 summarizes the analytical results for surface soil samples, and Figure 11-5 shows their distribution. This assessment presents Phase II contaminant data only, because no surface soil samples were collected during Phase I.

| r         | - <u></u>               |                                          |                                          | Phase II                                 | Samples                                  | ,                                        |                                          |
|-----------|-------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|
| Parameter | Reference<br>Background | SB-1<br>4C1111<br>07/10/97<br>(0-1 foot) | SB-2<br>4C1211<br>07/13/97<br>(0-1 foot) | SB-3<br>4C1311<br>07/10/97<br>(0-1 foot) | MW-5<br>4C1511<br>07/10/97<br>(0-1 foot) | MW-6<br>4C1611<br>07/10/97<br>(0-1 foot) | MW-7<br>4C1711<br>07/11/97<br>(0-1 foot) |
|           |                         |                                          | RCRA Meta                                | ls (mg/kg)                               |                                          | · · · · · · · · · · · · · · · · · · ·    |                                          |
| Barium    | 14.7                    | 7                                        | 20.2                                     | 13.3                                     | 6.2                                      | 6.9                                      | 5.2                                      |
| Cadmium   | 0.18                    |                                          | 0.35                                     |                                          |                                          | 0.9                                      | 2                                        |
| Chromium  | 6.21                    | 3.7                                      | 3.5                                      | 2.8                                      | 0.78                                     | 2                                        | 0.77                                     |
| Lead      | 8.81                    | 6.5                                      | 33.2                                     | 29.9                                     | 3.5                                      |                                          | 0.67                                     |
| Mercury   | 0.03                    | 0.05                                     | 0.06                                     | 0.06                                     | 0.03                                     | 5.3<br><b>0.06</b>                       | 3.1                                      |

# Table 11-1. Summary of Analytical Results for Surface Soils Burn Pit C, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

RCRA = Resource Conservation and Recovery Act.

SVOCs. No SVOCs were reported above the detection limit in surface soil samples.

RCRA metals. Four RCRA metals were reported above reference background values. Maximum concentrations were reported at SB-2 for all four metals, including barium (20.2 mg/kg), cadmium (0.35 mg/kg), lead (33.2 mg/kg), and mercury (0.06 mg/kg). Lead and mercury were also above background at SB-3, and mercury was above background at SB-1 and MW-6. The extent of these metals in surface soil is limited to the area surrounding these borings in the center of the Burn Pit C site, approximately one acre in extent.

### 11.2.2 Subsurface Soil Contamination

The subsurface soil samples were collected from the four Phase I monitoring well boreholes, the three Phase II well boreholes, and the three Phase II soil borings. The samples were analyzed for VOCs and RCRA metals. Table 11-2 summarizes analytical results for the subsurface soil samples, and Figure 11-6 shows their distribution. Both Phase I and Phase II data are shown.

VOCs. BTEX compounds were identified in both Phase I and Phase II subsurface soil samples. Maximum concentrations of toluene (69.9  $\mu$ g/kg at SB-2) and xylenes (14  $\mu$ g/kg at MW-3) were reported. Lower concentrations of BTEX, notably toluene at 5  $\mu$ g/kg, were reported also at SB-1. The extent of BTEX contamination, although present at relatively low concentrations, extends from the area of SB-2 to the northwest at MW-3.

-1



)

)

Table 11-2. Summary of Analytical Results for Subsurface Soils Burn Pit C, Fort Stewart

Ć

(

|                        |            |          | Phase I  | Phace I Samples |                         |                                  |           | ļ         |                         |          |           |
|------------------------|------------|----------|----------|-----------------|-------------------------|----------------------------------|-----------|-----------|-------------------------|----------|-----------|
|                        |            | NAX74    | T.       |                 |                         |                                  |           | Phase II  | <b>Phase II Samples</b> |          |           |
|                        |            | TAT      | 7 M M    | MW3             | MW4                     | MW-5                             | MW-6      | MW-7      | SB-1                    | SR-2     | SR.3      |
|                        |            | SL14     | SL2-4    | SL3-4           | SL4-4                   | 4C1512                           | 4C1612    | 401712    | AC1112                  | 101111   |           |
|                        |            | 06/23/93 | 06/23/93 | 06/23/93        | 06/23/03                | 07/10/07                         | 07/10/07  |           |                         | 717174   | 401312    |
|                        |            | 0-4      | 0.4      | S               |                         |                                  | 1.6/01/10 | 16/TT// 0 | 1.6/01//0               | 113/97   | 07/10/97  |
|                        |            |          | י<br>י   | ţ               | +<br>7                  | C-C77)                           | (2.5-5)   | (2.5-5    | (2.5-5                  | (2.5-4.5 | (10-10.5] |
|                        |            | (leet)   |          | feet)           | feet)                   | feet)                            | feet)     | feet)     | feet)                   | feet)    | feet)     |
| Douronter              | Keference  | 0002.0   |          | 0002.0          | 0002.0                  | 0002.5                           | 0002.5    | 0002.5    | 0002.5                  | 0002.5   | 0010.0    |
|                        | Background | 0004.0   | 0004.0   | 0004.0          | 0004.0                  | 0005.0                           | 0005.0    | 0005.0    | 0005.0                  | 0004.5   | 0010.5    |
|                        |            |          | Νo       | latile Oron     | nic Como                | Volatile Oroanic Compande (malle |           |           |                         |          |           |
| 1,1,1 Trichloroethane  | 00         |          |          |                 |                         | 21/21/1 cm/m                     |           |           |                         |          |           |
| 2-Butanone             |            |          |          |                 |                         |                                  |           |           |                         |          |           |
| Acetone                |            |          |          |                 |                         |                                  | _         |           |                         | 17.1     |           |
| A future of the second | n'n        |          |          |                 |                         |                                  |           |           |                         | 105      |           |
| Internytene Untoride   | 0.0        |          |          |                 |                         |                                  |           |           |                         |          |           |
| Tetrachloroethene      | 0.0        |          |          | 48              | 4                       |                                  |           |           |                         | 4.3      |           |
| Toluene                | 0.0        |          |          | 21              | 2                       |                                  |           |           |                         |          |           |
| Xylene                 | 0.0        |          |          |                 |                         |                                  |           |           | S                       | 6.69     |           |
|                        |            |          |          |                 |                         | -                                |           |           |                         |          |           |
|                        |            |          |          | RCRA            | RCRA Metals (mo/ko)     | o/ko)                            |           |           |                         |          |           |
| Arsenic                | 8.04       |          |          |                 |                         |                                  |           |           |                         |          |           |
| Barium                 | 17.0       | 13       | 1        | 3 5             | 1 2                     | 0 0                              | -         |           | 0.37                    | 0.39     |           |
| Cadmium                | 0.24       |          |          | ;               | *                       |                                  | 4         | 1.1       | 10.2                    | 13.4     | 5.4       |
| Chromium               |            |          |          |                 |                         |                                  |           |           | 0.12                    |          |           |
| Lead                   |            |          |          | 4               | 4.8                     | 9.6                              | 2.7       | 1.2       | 2.9                     | 1.2      | 46        |
|                        |            | /0.0     | 0.73     | 2               | 2.7                     | 9.1                              | £         | 1.6       | 21.9                    | 14       | 2.2       |
| INICICALLY             | 0.05       |          |          |                 |                         | 0.06                             | 0.02      | 0.03      | 0.07                    | 0.03     | 2.00      |
|                        |            |          |          | Other In        | Other Inorpanics (mo/ko | na/ka)                           |           |           |                         | 1000     | c0-0      |
| Total Organic Carbon   | 000 0      |          |          |                 |                         |                                  |           |           |                         |          |           |
|                        | 7,200      |          |          |                 |                         | _                                | 9770      |           |                         |          |           |
| Note: A blant indiani- |            |          |          |                 |                         |                                  |           |           |                         |          | ]         |

Note: A blank indicates analyte not detected. Note: **Bold** font indicates value exceeds reference background criterion.

RCRA = Resource Conservation and Recovery Act

97-160P(EXCEL)/011499

11-16

THIS PAGE INTENTIONALLY LEFT BLANK



Figure 2-2. Location Map for Fort Stewart Military Reservation, Georgia

171

2-3

C



Figure 2-3. Location of the Burn Pits (SWMUs 4A through 4F)



1 al

(

....

Figure 11-6. Results of Analys

(

PCE was reported in two Phase I samples at a concentration of 48  $\mu$ g/kg at MW-3 and 15  $\mu$ g/kg at MW-4. 1,1,1-Trichloroethane was also reported in a Phase I sample (MW-3 at 10  $\mu$ g/kg). Neither PCE nor 1,1,1-trichloroethane was detected in the Phase II soil samples. Acetone, 2-butanone, and methylene chloride were detected only at SB-2 at concentrations near their average concentration in background samples and are, therefore, not considered site related.

**RCRA metals.** Two metals were reported at concentrations above their reference background value. Maximum values of lead (21.9 mg/kg at SB-1) and mercury (0.07 mg/kg at MW-1) were reported. The extent of these metals above background is characterized by borings SB-1, SB-2, and MW-5, extending from the center of the site to the northeast, approximately 1.5 acres in size.

#### **11.2.3 Groundwater Contamination**

Groundwater contamination was evaluated using the results from water samples taken from the three monitoring wells installed during Phase II (MW-5 through MW-7). Waters removed from the Phase I monitoring wells were very turbid, and the wells could not be adequately redeveloped; after consultation with GEPD, the wells were abandoned. The groundwater samples from MW-5 through MW-7 were analyzed for VOCs and RCRA metals. Both filtered and unfiltered water samples were collected; only the total metal analysis on unfiltered water samples is presented in this section. Table 11-3 summarizes the analytical results for groundwater samples, and Figure 11-7 shows their distribution. This assessment focuses on the Phase II contaminant data because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

|           |                         |               | Phase I                   | Samples                    | ************************************** |
|-----------|-------------------------|---------------|---------------------------|----------------------------|----------------------------------------|
| Parameter | Reference<br>Background | MCL           | MW-5<br>8/07/97<br>4C4511 | MW-6<br>08/06/97<br>4C4611 | MW-7<br>08/07/97<br>4C4711             |
|           |                         | RCRA Metals ( | ug/L)                     |                            |                                        |
| Arsenic   | 3.02                    | 50            | 1                         | 2.5                        |                                        |
| Barium    | 71.72                   | 2000          | 70.6                      | 134                        | 26.6                                   |
| Cadmium   | 0.43                    | 5             |                           | 3                          |                                        |
| Lead      | 4.69                    | 15            | 0.3                       | 1.1                        | ·····                                  |
| Mercury   | 0.14                    | 2             | 1                         | 0.14                       | 0.28                                   |
| Selenium  | 1.90                    | 50            | 0.88                      | 1.6                        | 0.20                                   |
| Silver    | 1.12                    |               | 0.24                      | 0.77                       | 0.08                                   |

 Table 11-3. Summary of Analytical Results for Groundwater,

 Burn Pit C, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act.

**VOCs.** No VOCs were reported above the detection limit in either Phase I or Phase II groundwater samples. These results suggest that although low concentrations of BTEX are present in subsurface soils at the site, the soil contamination has not impacted groundwater.

**RCRA metals.** Three metals were reported at concentrations above background in groundwater. Barium was reported at a maximum concentration in MW-6 (134  $\mu$ g/L), cadmium in MW-6 (3  $\mu$ g/L), and mercury in MW-7 (0.28  $\mu$ g/L). Because MW-6 is downgradient on the site, barium

)



Figure 11-7. Results of Analys

)

97-160P(DOC)/020999

and cadmium are considered potentially site related. MW-7 is considered upgradient of the site; however, as discussed in Section 11.1, the water table may be mounded beneath the Burn Pit C site.

### 11.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 11-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

| Analyte               | Surface Soil | Subsurface Soil | Groundwater |
|-----------------------|--------------|-----------------|-------------|
|                       | (µg/kg)      | (µg/kg)         | (µg/L)      |
| Acetone               | na           | 39.4            | nd          |
| 2-Butanone            | na           | 17.1            | nd          |
| Methylene Chloride    | na           | 4.3             | nd          |
| Tetrachloroethene     | na           | 48              | nd          |
| Toluene               | na           | 69.9            | nd          |
| 1,1,1-Trichloroethane | na           | 10              | nd          |
| Xylenes, total        | na           | 14              | nd          |
|                       | (mg/kg)      | (mg/kg)         | (μg/L)      |
| Arsenic               | nd           | brc             | brc         |
| Barium                | 20.2         | brc             | 134         |
| Cadmium               | 0.35         | brc             | 3           |
| Chromium              | brc          | brc             | nd          |
| Lead                  | 33.2         | 21.9            | brc         |
| Mercury               | 0.06         | 0.07            | 0.28        |
| Selenium              | nd           | nd              | brc         |
| Silver                | nd           | nd              | brc         |

#### Table 11-4. Summary of Site-Related Constituents Burn Pit C, Fort Stewart

brc = Below background reference criteria

na = Not analyzed in any sample in that medium

nd = Not detected in any sample in that medium

BOLD font indicates the analyte is an SRC in that medium

# 11.3 FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT C

The results of contaminant migration soil screening (discussed in Section 6) for Burn Pit C are presented in Table 11-5. Only PCE is identified as a CMCOPC based on leaching to groundwater. However, PCE is not currently observed in groundwater, indicating that it may be degraded before reaching the water table. The predicted maximum concentration of PCE in groundwater, based on a maximum of 48  $\mu$ g/kg in soil and a DAF of 1, is 80  $\mu$ g/L. Through biodegradation, even if the PCE were to reach groundwater at such a concentration, it would degrade to its MCL value (5  $\mu$ g/L) in less than 4.0 times its half-life. Assuming a conservative half-life of 4.5 years, the PCE would degrade to its MCL value in 18 years. Similarly, methylene chloride with its predicted maximum concentration of 22  $\mu$ g/L in groundwater would degrade to its MCL in less than 2.1 times its half-life, or 0.6 years.

The maximum concentrations of PCE were reported in Phase I samples only and were not detected in Phase II samples.

| SRCs                  | Maximum<br>Concentration   | GSSL    | Is Maximum<br>Concentration > GSSL? |
|-----------------------|----------------------------|---------|-------------------------------------|
|                       | RCRA Metals (mg/kg)        |         |                                     |
| Barium                | 20.20                      | 1600    | No                                  |
| Cadmium               | 0.35                       | 8       | No                                  |
| Lead                  | 33.20                      | 400     | No                                  |
| Mercury               | 0.07                       | 0.4     | No                                  |
|                       | Volatile Organic Compounds | (µg/kg) |                                     |
| 1,1,1-Trichloroethane | 10.00                      | 100.0   | No                                  |
| 2-Butanone            | 17.10                      | 38.4    | No                                  |
| Acetone               | 39.40                      | 800     | No                                  |
| Methylene Chloride    | 4.3                        | 1       | Yes                                 |
| Tetrachloroethene     | 48.00                      | 3       | Yes                                 |
| Toluene               | 69.90                      | 600     | No                                  |
| Xylenes               | 14.00                      | 10000   | No                                  |

### Table 11-5. CMCOPCs Based on Soil Screening for Burn Pit C

CMCOPCs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

RCRA = Resource Conservation and Recovery Act.

SRCs = Site-related Contaminants.

The most likely pathway of contaminant migration from Burn Pit C is via groundwater discharge to Mill Creek, located approximately 4000 feet northeast of the site. Based on an average groundwater flow velocity of 14.1 feet/year, the estimated arrival time for the site groundwater to reach Mill Creek is expected to be more than 284 years. Therefore, PCE and methylene chloride would degrade to less than their MCLs within 300 feet of the Burn Pit C site, well before reaching Mill Creek.

Metals detected in the groundwater at this site include barium, cadmium, and mercury. However, the maximum concentrations of these metals do not exceed their respective MCLs and, based on fate and transport analysis, they are not expected to increase in the future. Therefore, none of the constituents are expected to be of concern for contaminant migration from Burn Pit C.

# 11.4 HUMAN HEALTH RISK EVALUATION OF BURN PIT C

SRCs were identified in Section 11.2 for surface soil, subsurface soil, and groundwater. The identification of risk-based action levels for all of the Burn Pits has been previously discussed in Section 4.7 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit C is given below.

#### **11.4.1 Exposure Evaluation**

This site is completely covered by vegetation, and no VOCs were detected in surface soils. Therefore, exposure via inhalation based on current land use is not a viable exposure pathway. The site is currently used by the Army to conduct outdoor classroom type training. Therefore, current on-site receptors include a juvenile trespasser and a hunter. Current off-site receptors would be represented by an off-site hunter. Land use at this site is unlikely to change in the future. However, the Army may use the site as a training area in the future. Construction may occur on the site or the vegetative cover may be gone as a result of on-site activities. Future on-site receptors include military personnel and a construction worker. Future off-site receptors include hunters in the area.

A summary of the potential exposure pathways for each of the receptor populations is given below.

**Current juvenile trespasser.** The juvenile trespasser may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Hunter. The hunter is representative of a current on-site and a current and future off-site receptor. The current on-site receptor is not likely to be exposed to chemicals given the absence of an air migration pathway, and clothing the hunter would be wearing would eliminate potential dermal exposure. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

The current off-site hunter may be exposed via ingestion of contaminants bioaccumulated in game animals. The future off-site hunter may be exposed via ingestion of game and inhalation of fugitive dust if the vegetative cover were to be removed.

Future on-site military personnel. The on-site military personnel may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Future construction worker. The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

### 11.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals will be evaluated further in a BRA.

None of the contaminants in surface soils or subsurface soils exceeded its respective risk-based screening value for direct exposure of a potential receptor population (Table 11-6).

The maximum concentration of cadmium in groundwater  $(3 \mu g/L)$  only slightly exceeded the screening value of 1.8  $\mu g/L$  (Table 11-6). However, the maximum concentration of cadmium was below its MCL of 5  $\mu g/L$ . The remaining chemicals were below their respective screening values. In addition, the use of surface ground water at the sites for drinking water is unlikely. Therefore, cadmium is not considered a potential threat to human health.

|                       |       |                         |                   | ·                 |                            |                       |         |                                 |
|-----------------------|-------|-------------------------|-------------------|-------------------|----------------------------|-----------------------|---------|---------------------------------|
|                       |       |                         |                   | -                 | •                          | EPA III               |         |                                 |
| Analyte               | Units | Frequency<br>of Detects | Minimum<br>Detect | Maximum<br>Detect | Background<br>Criteria     | Screening<br>Criteria | COPC?   | Justification                   |
|                       |       |                         |                   | Burn Pit          | Burn Pit C.Surface Soil    |                       |         |                                 |
| Barium                | mg/kg | 9/9                     | 6.2               | 20.2              | 14.70                      | 550                   | No      | Max Detect < Screening Criteria |
| Cadmium               | mg/kg | 1/6                     | 0.35              | 0.35              | 0.18                       | 3.90                  | °N.     | Max Detect < Screening Criteria |
| Lead                  | mg/kg | 6/6                     | 3.5               | 33.2              | 8.81                       | 400                   | No      | Max Detect < Screening Criteria |
| Mercury               | mg/kg | 6/6                     | 0.03              | 0.06              | 0.03                       | 2.3                   | No      | Max Detect < Screening Criteria |
|                       |       |                         |                   | Burn Pit (        | Burn Pit C Subsurface Soil | !!                    |         |                                 |
| Lead                  | mg/kg | 10/10                   | 0.67              | 21.9              | 11.1                       | 400                   | No      | Max Detect < Screening Criteria |
| Mercury               | mg/kg | 6/10                    | 0.02              | 0.07              | 0.05                       | 61.0                  | No      | Max Detect < Screening Criteria |
| 1,1,1-Trichloroethane | mg/kg | 1/9                     | 0.01              | 0.01              |                            | 4,100                 | No      | Max Detect < Screening Criteria |
| 2-Butanone            | mg/kg | 1/6                     | 0.0171            | 0.0171            |                            | 100,000               | No      | Max Detect < Screening Criteria |
| Acetone               | mg/kg | 1/10                    | 0.0394            | 0.0394            |                            | 20,000                | No      | Max Detect < Screening Criteria |
| Tetrachloroethene     | mg/kg | 2/8                     | 0.015             | 0.048             |                            | 110                   | No      | Max Detect < Screening Criteria |
| Toluene               | mg/kg | 4/10                    | 0.005             | 0.0699            |                            | 41,000                | No      | Max Detect < Screening Criteria |
| Xylenes, Total        | mg/kg | 1/8                     | 0.014             | 0.014             |                            | 100,000               | No<br>N | Max Detect < Screening Criteria |
|                       |       |                         |                   | Burn Pit          | Burn Pit C Groundwater     | <b>F</b>              |         |                                 |
| Barium                | µg/L  | 2/2                     | 20.6              | 134               | 71.72                      | 260                   | No      | Max Detect < Screening Criteria |
| Cadmium               | μg/L  | 1/2                     | 3                 | m                 | 0.43                       | 1.8                   | Yes     | Max Detect > Screening Criteria |
| Mercury               | μg/L  | 2/2                     | 0.14              | 0.14 <sup>a</sup> | 0.14                       | 1.1                   | No      | Max Detect < Screening Criteria |
|                       | _     |                         |                   |                   |                            |                       |         |                                 |

Table 11-6. Comparison to Action Levels-Burn Pit C, Fort Stewart

<sup>1</sup> <sup>4</sup>Maximum detected value of mercury in MW-7 (upgradient well) was 0.28 μg/L. COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency.

Ì

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit C. A human health BRA is not warranted. No further action is required for protection of human health.

### 11.5 ECOLOGICAL RISK EVALUATION OF BURN PIT C

The ecological risk evaluation of Burn Pit C is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are considered further in PRE Steps ii through v. The results of the five steps of the PRE are reported below.

### 11.5.1 Ecological Screening Value Comparison (Step i)

Three RCRA metals were detected in groundwater at Burn Pit C at concentrations exceeding reference background criteria. The results of the ecological screening value comparison for groundwater at Burn Pit C are presented in Table 11-7. At MW-7, barium and mercury concentrations exceed the surface water ESVs. Where detected in the remaining two monitoring wells (MW-5 and MW-6), barium, cadmium, and mercury exceed the ESVs for those analytes. The ecological COPCs identified by the ESV comparison for groundwater at Burn Pit C are barium, cadmium, and mercury.

| Analyte | ESV              | MW-7            | MW-5   | MW-6                                   | Maximum                                |
|---------|------------------|-----------------|--------|----------------------------------------|----------------------------------------|
|         | R                | CRA Metals (µg/ | L)     | ······································ | ······································ |
| Barium  | 3.9 <sup>4</sup> | 26.6 J          | 70.6 J | 134 J                                  | 134 J                                  |
| Cadmium | 0.66             |                 |        | 3                                      | 3                                      |
| Mercury | 0.0123           | 0.28            |        | 0.14                                   | 0.28                                   |

# Table 11-7. EPA Region 4 Ecological Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit C, Fort Stewart

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1995) and, where indicated, alternative values for analytes without Region 4 ESVs.

RCRA = Resource Conservation and Recovery Act.

<sup>4</sup>Office of Solid Waste and Emergency Response Ambient Water Quality Criteria or Tier-II value (Suter and Tsao 1996).

<sup>b</sup>Hardness dependent, assumes 50 mg/L CaCO<sub>3</sub>.

Bold font indicates detected concentration exceeds ESV.

Although ecological COPCs are identified in groundwater at Burn Pit C when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit C. There are no surface water bodies in the vicinity of Burn Pit C. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit C soils.

Because there are no ESVs for soil, all analytes detected in soil at Burn Pit C are evaluated further in PRE Steps ii through v.

#### 11.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit C comprises approximately 2 acres (Figure 11-1). The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees. Drainage occurs through overland flow radically in all directions and is collected in a series of man-made ditches that convey runoff to Mill Creek. The closest surface stream to the site is Mill Creek, a tributary of Taylors Creek, which lies approximately 4000 feet northeast of the site.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit C are described in Section 8.1.2.

#### 11.5.3 Preliminary Effects (Step iii)

In the PRE for Burn Pit C, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the site. The derivation of TRVs is discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8-1.

#### 11.5.4 Preliminary Exposure (Step iv)

Ecological receptors at the Burn Pit C are likely exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews and robins exposed to COPCs in soil are presented in Table 8-2.

#### 11.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate a HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0 and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

Surface Soil. The preliminary risk calculation for shrews and robins exposed to ecological COPCs detected in soil at Burn Pit C is presented in Table 11-8. This table shows the maximum detected concentrations and the TRVs for shrews and robins. Concentrations resulting in HQs exceeding 1.0 are shown in boldface.

The ecological COPCs present in Burn Pit C surface soil at concentrations exceeding the TRVs for the surrogate species are cadmium and lead. Both the shrew and robin are potentially at risk from cadmium in surface soil. However, cadmium was detected in only a single sample at a concentration above background criteria and may not be site related. The robin is potentially at risk from lead in surface soil. The concentrations of lead in all six surface soil samples exceed the robin TRV. An HI is not calculated for the surrogate species exposed to RCRA metals in surface soil because they are assumed to have dissimilar mechanisms of toxicity.

Table 11-8. Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 foot) at Burn Pit C, Fort Stewart

(

|          | TRV   | Δ     |             |                     |       |            |                                            |         |           |
|----------|-------|-------|-------------|---------------------|-------|------------|--------------------------------------------|---------|-----------|
| Analyte  | Shrew | Robin | <b>MW-5</b> | 9-WW                | MW-7  | SR-1       | SR-3                                       | CB 2    | Marine    |
|          |       |       |             |                     |       |            | 7-770                                      | C-00    | INTAXIMUM |
|          |       |       | RCRA M      | RCRA Metals (mg/kg) |       |            |                                            |         |           |
| Barium   | 2627  | 2203  | 627         |                     | F     |            | ) ( ) ( )<br>( ) ( ) ( ) ( ) ( ) ( ) ( ) ( |         |           |
|          |       |       | 7.7.7       | r c•n               | 1 2.0 | <b>(</b> ) | 70.7 J                                     | 13.3 J  | 20.2 J    |
| Caumum   | 0.321 | 0.109 |             |                     |       |            | 750                                        |         |           |
| I and    |       |       |             |                     |       |            | cc.n                                       |         | 0.35      |
| L-Vau    | <br>  | 2.30  | 5           | 1.6.2               |       | 1 2 9      | 12.7                                       | 1000    | C CC      |
| Maroning |       |       |             |                     |       | 2          | 100                                        | 5 C C 7 | 23.6      |
| 1 manate | 14.1  | 1.09  | 0.03        | 0.06                |       | 0.05       | 0.06                                       | 0.06    | 200       |
|          |       |       |             |                     |       | 2212       | 2010                                       | 0.00    | 00.00     |
|          |       |       |             |                     |       |            |                                            |         |           |

COPCs = Chemicals/Contaminants of Potential Concern.

J = Estimated concentration.

= No data to derive TRV.

RCRA = Resource Conservation and Recovery Act. TRV = Toxicity Reference Value = (NOAEL x BW/Food Ingestion)/BAF<sub>earthworm</sub> (see Table 8-1).

Blank ceils indicate analyte not detected at site.

Bold font indicates detected concentration exceeds TRV (HQ > 1).

#### 11.5.6 Uncertainties

The risk to ecological receptors from cadmium and lead in surface soil at Burn Pit C is overestimated by the preliminary risk calculations. The single cadmium detect in surface soil (0.35 mg/kg in SB-2) does not exceed the NOAEL-based TRV for shrews (0.365 mg/kg) calculated using published dietary fractions, and only barely exceeds that for the robin (0.284 mg/kg). Cadmium in surface soil at Burn Pit C is many times less than the LOAEL-based TRVs for shrews and robins calculated using published dietary fractions, 1.74 and 3.92 mg/kg, respectively. Likewise, lead was detected in three of six surface soil samples at Burn Pit C at concentrations exceeding the NOAEL-based TRV for the robin (5.89 mg/kg), but in no sample did lead exceed the LOAEL-based TRV for the robin calculated using published dietary fractions (48 mg/kg). Therefore, cadmium and lead in surface soil at Burn Pit C are unlikely to pose a risk to ecological receptors.

# 11.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT C

The following are the conclusions of the Phase II RFI for Burn Pit C:

- In surface soils, four metals (barium, cadmium, lead, and mercury) were reported above reference background at boring locations in the middle of the source area. The extent of these metals in surface soil is, therefore, limited to the area surrounding these borings (SB-1, SB-2, SB-3, and MW-6) in the center of the Burn Pit C site, approximately one acre in size.
- There are no human health COPCs in surface soil or subsurface soil because no constituent exceeded its risk-based screening value.
- Cadmium and lead are ecological COPCs in surface soils because they exceed the TRVs for • the surrogate species (shrew and robin). The maximum concentration of cadmium (0.35 mg/kg) is much less than the average USGS background concentration for soils in the Eastern United States (2 mg/kg). Cadmium was detected above background in only one sample and may not be site related. Lead is possibly site related and may pose a risk to ecological receptors (mammals and birds ingesting earthworms and other soil-dwelling invertebrates). However, the maximum concentration of lead (33.2 mg/kg) is well within the USGS range of background concentration for soils in the Eastern United States (up to 300 mg/kg). In addition, lead concentrations in surface soil are many times less than the LOAEL-based TRV and are unlikely to pose a risk to ecological receptors. The risk to ecological receptors from cadmium and lead in surface soil at Burn Pit C is overestimated by the preliminary risk calculations. As discussed in the uncertainties section (11.5.6), the maximum detected concentrations of cadmium and lead in surface soil are expected to result in exposures to the surrogate ecological receptors (shrews or robins) that are less than the lowest exposures associated with adverse effects on them (LOAEL-based TRV). Therefore, cadmium and lead are eliminated as ecological COPCs for Burn Pit C, and further investigation and/or evaluation of these constituents is not required.
- In subsurface soil, two metals (lead and mercury) occur in the middle of the site and extend to the northeast, defining the extent of potential metal contamination. Similarly, BTEX compounds and PCE were identified in four of the eight subsurface soil samples, extending from the center of the site to the northwest.

- There are no human health COPCs in subsurface soil because no constituent exceeded its risk-based screening value.
- PCE is a potential contaminant migration COPC in subsurface soil because it may leach to groundwater, resulting in groundwater concentrations exceeding its respective MCL. Off-site migration of PCE would be very limited due to retardation and biodegradation and the slow movement of groundwater. Even if it were to reach groundwater at its predicted maximum concentration, it would degrade to concentrations less than its MCL in less than 18 years, and having traveled less than 300 feet from the Burn Pit C site. Therefore, migration of PCE is considered unlikely, and there are no contaminant migration COPCs in subsurface soil.
- In groundwater, three metals (barium, cadmium, and mercury) were reported above background and are considered SRCs. Mercury was detected in well MW-7 at concentrations only slightly above background. Well MW-7 is considered upgradient of the site; however, water table contours suggest that the water table may potentially be mounded beneath the Burn Pit C site.
- Cadmium is a potential human health COPC at Burn Pit C where the maximum concentration
  of cadmium in groundwater exceeded its risk-based screening value for residential use of
  groundwater. The maximum concentration of cadmium was less than its MCL of 5 µg/L at
  the site. In addition, use of the surficial groundwater at the site for drinking water is
  unlikely. Therefore, cadmium is not considered a potential threat to human health at Burn
  Pit C and does not require further investigation and/or evaluation.
- Barium, cadmium, and mercury, are ecological COPCs in groundwater because they exceed the ESVs for surface water. However, there are no surface water bodies in the vicinity of Burn Pit C, and off-site migration would be limited due to retardation and the slow movement of groundwater. Therefore, exposure of ecological receptors to these metals in surface water bodies downgradient from Burn Pit C is not a complete exposure pathway. Therefore, barium, cadmium, and mercury are eliminated as ecological COPCs for Burn Pit C, and further investigation and/or evaluation of these constituents is not required.

The following are recommendations for Burn Pit C:

- 1. No further action is recommended for Burn Pit C.
- 2. All potential human health COPCs were eliminated for Burn Pit C, and a BRA is not recommended for the site.
- 3. All potential ecological COPCs were eliminated for Burn Pit C, and an ERA is not recommended for the site.

ŝ

# 12.0 CHARACTERIZATION OF BURN PIT D (SWMU 4D)

## 12.1 HISTORY AND DESCRIPTION OF BURN PIT D

Burn Pit D (SWMU 4D) comprises approximately 4 acres and is located on Fort Stewart Route 90, approximately 500 feet west from its junction with 6th Street. The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees. Trees in the center of the site have become large enough that access to truck-mounted drilling equipment was restricted.

The site is slightly mounded, with elevations varying between approximately 71 to 72 feet amsl around the perimeter of the site and 77 feet amsl in the center of the site (Figure 12-1). Drainage occurs through overland flow radially in all directions and is collected in a series of man-made ditches that convey runoff to Mill Creek. The closest surface stream to the site is Mill Creek, a tributary of Taylors Creek, which lies approximately 2500 feet east-northeast of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 12-1). PCE and four RCRA metals, including arsenic, chromium, lead, and mercury were detected above site-specific background levels in soils. RCRA metals were also detected above site-specific background levels in groundwater and included chromium, lead, mercury, and selenium. No VOCs were detected in groundwater samples in Phase I.

During the Phase II RFI activities, two new monitoring wells (MW-5 and MW-6) were installed, and three soil borings (SB-1 through SB-3) were drilled at the site (Figure 12-1). These wells and borings, together with the existing Phase I wells, were used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of slightly silty sands. Geologic cross-sections of the Burn Pit D site are presented in Figures 12-2 and 12-3. Geotechnical parameters were measured in a bulk soil sample taken from MW-6 at a depth of 12.8 to 13.8 feet and in a Shelby tube sample taken from MW-5 at a depth of 10.7 to 12.5 feet. The results of the geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic sands with less than 8 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from MW-5, is  $9.45 \times 10^{-4}$  cm/second, typical for a fine sand.

Water levels in the monitoring wells were measured during well development between July 9 and 13, 1997, and again during well sampling between July 28 and August 10, 1997. The water table is present at a depth between 7 and 8 feet bgs around the perimeter of the site and up to 13 feet bgs beneath the center of the site. A water table contour map is provided as Figure 12-4. Groundwater contours measured during the Phase I RFI activities in August 1993 indicate similar water levels and groundwater flow direction. Groundwater flow is generally to the east-northeast, discharging ultimately to Mill Creek, which is located approximately 2500 feet east-northeast of the Burn Pit D site. The water table is relatively flat, with a calculated horizontal hydraulic gradient across the site of 0.0015 foot/foot. Based on a hydraulic conductivity of  $9.45 \times 10^4$  cm/second and a porosity of 0.368, the groundwater flow rate is approximately 4.0 feet/year toward Mill Creek.

۰.



Figure 12-1. Locations of S:

÷
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 120                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 180                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 240                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 300                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|----------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | alar Alar Sarahan ang ang ang ang ang ang ang ang ang a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | -frankrast-radionalise.ch.m.j                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┥┥┥┥┾                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ┼┼┼┼                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | and a second secon                                                                                                                                                                                                                                            | al and an a second and a second a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and the second s | www.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | and and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┉┉┉┉┼╌┼╌┠╌┟╌┥╌┥╌                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and from the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┉┉┉┝┉┟┉┟┉┟┉┼┉┾╺                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ┉┉┉┉┉┉┉┉┉╻                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | <u>LIP</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | tit titt                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | <u>╺</u> ╋╼╋╼╋╼╋ <mark>╼</mark> ╋╍╋╍╋╼╋╼                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 75       | A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | for send at a fact second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | have a property of the second se                                                                                                                                                                                                                                             | ┍┿┟┾┧┝┥┼┾                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ┽ <del>╏╻┥╻┥╸┥</del>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ╅╬┽╋╋┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and the second sec                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | <del>┊╶┥┈╪╶</del> ╁┯ <mark>┠╌</mark> ╆╌ <u></u> ┠╌ <mark>┟╌</mark> ┿╸ <mark>┽</mark> ╶┿                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | - had a find a first start of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | historia - to the total total and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 11111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | itit titit                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | ┥┥┥┥┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | m                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ┉┿╍╴┫╼┑┿╼╍┿╼╍┿╺╌┥╼╍┥╍╍┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 8-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | <b>X</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SB-02                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┿┿┿┿┿                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | a dentri de la contra de la con | and the second second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 711111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a in an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 70       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a star a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ala da                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <del>┥┥┥┥</del>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ₹.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|          | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | an fan brander an fan ster ster ster ster ster ster ster ster                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ╺┽┾┼┥╸                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ┿┉┪┉┪╍┿╍┶┿╍┝╍┝╺                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a de la companya de l                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| -        | The second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and a fair a second and a second and a second a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ┝╾┝╼┝╼┝╼┝╼┝╼┝╼┝                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | -SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | and and and an and a special and and a set of the set of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | SILT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Y                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|          | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | and and and and and a straight of the straight |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┽╇┲                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | har far far far far far far far far far f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 65       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a jeneraljene je se je na j<br>Na na je na na jeneraljene je na j                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and in free free free free free free free fre                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Z                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|          | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | mber frequencies for frequencies                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | + + + + + + + + + + + + + + + + + + +                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| -        | and the second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | andren and providence for an for and a second for.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ┉┼╌┼╍┨╍┥╺┽╍╎╍╎╌┼                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| -        | The                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | n na                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | and and and and an end and an end and an end and and and and and and an end and an end and an end and an end an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | FY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|          | and a set of the set o | alan mining ang ang ang ang ang ang ang ang ang a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | SL HH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | <mark>╞╍<mark>╞╍╞╍╞</mark>╍╋┉╞╍┢╍╞╍╞╸╉</mark>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 60       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ng wang o dan panang sa basa ba                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 11111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ┽┥╋╼┢                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| <u> </u> | And a start of a start | an sea an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | un ser algebra confine de sur fragada a africa fraga                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 141                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ander for an and a stand a<br>And a stand and a stand                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | the first of the f |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ┥┥┥┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ┿╫┿                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|          | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | NO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| İ        | and a state of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | to an dout of a link for                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | from the forther of the state o |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ╼┿╍╋╼┶╍╏╍╍╬╍╋╍╼╬╍╗                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | RECOVE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 7.Y                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | for a container for an an an an farme for a farme for  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | an general and an a general and a second and a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 55       | and a second a second s | 1999 per og en stjerne former frem stjerne og en se stjerne og en stjerne og en stjerne og en stjerne og en st<br>For sterne store for andere store                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | nine for a subscription for a first subscription of the subscription of the subscription of the subscription of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and rate of a second conference in the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | and man and a second and a second a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | man and a fear for the former                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | miter free and a star free for the star                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | - for for the second se | and a standard and a standard and a standard and a standard a standard a standard a standard a standard a stand                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | - Jan San Jan Andrew                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|          | ราง กลักษณฑิตระกร่า <b>งส</b> ะสุขาย สีนกระสุขารณ์ <sub>การส</sub> าร 3 (กระ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | an managera a san ana ang ang ang ang ang ang ang ang a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | A new construction of the state | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and the second sec                                                                                                                                                                                                                                             | from a soft of an and an interview of a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | And a second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | har her showed and an grant production of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | n numero di manie al manie al medare                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 6.10 <sup>-2</sup>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | a service at more                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and the second contraction of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | an ang mangkan ng mang<br>Na mangkan ng                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | radione some of the state of th | **************************************                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | an also a sin a succession and a success | fring - name from the only a from the original                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | m - + - + - + - +                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 50       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a no a series de series contra e la serie de series de series de series de series de series de series de series<br>No antima de series d                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | and the second s | and the second sec | a den en ser de la companya de la co<br>Companya de la companya de la company                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | Trademond and a start of the st |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | and an in some and and some from the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | million sampling of human and the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | need and the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | a second and the second se                                                                                                                                                                                                                                                                                                             | الى بى بىلىدىكى بى ئىيە ئەن مەمەر مىمەردى<br>بەر بى بىڭ بىڭ بىڭ بى ئومىلى، مەيمەر مەر مىچە                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | والمؤالية والمستجوم فيتوكينا والمتحا والمحاصين                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - 14-240 (a 15-24) (a 14-24)<br>- 14-240 (a 15-24) (a 16-24)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|          | and a standard and a standard a s<br>Standard a standard a st                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | أيهم والمرافعية بالمراجع المرجع متحاصر والع                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | here we we also a wanter of a sec                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | animation of a fraction of the same fraction of the same                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | درمانه میشود. میشود بیشمانه میکومان میکومانی میکومانی میکود.<br>درمانه میشود میکود میکومانی میکومانی میکومانی می                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | many months and an include a sub-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | - transformer in formation of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | water where a second mean through the second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | andreas dan serang serang perintahan di kerangkan dan serang serang perintahan di kerangkan dan perintahan di<br>Kerangkan di kerangkan di kerang perintahan di kerang serangkan di kerangkan di kerangkan di kerangkan di kerang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | en ment die streefte van die soegee en gebruik van die en gebruik van die en<br>weren en weer die meerste were were weren were die soege was en die en die soege were die soege was die soege                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 4 - I    | 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999<br>1999 - 1997 - 1998 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n and a sub-and and a sub-angles of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and when a second or a second se | يبد جي وسيد والمحمد المحمد الم                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | a ganded the set of a set of the |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 45       | ma                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and the second s | and the production of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | na na mana ana ana ana ana ana ana ana a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | n han a star and and a star barren a star br>A star a star                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | and and a start of the second s |
|          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| -        | e de la competencia de la competencia                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | وليستو المتحمة المحمد المحاد                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | بيمارير الممكر والمراجع مح مراجع المراجع                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | - Low a standard of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|          | nan yan yan baran yan ngana ngan<br>Nangan ngana ng                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | en professionade en en en en en                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ويقونهم ورويتهم والجار بالمراو يقاربهم بالمراجع والاستار ال                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ىرىمى بېرىسىمورىدۇ دىرىد ، بىرىس                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | a manager in the same sugar is the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | and a fair to share so a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Second second and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|          | and the second sec                                                                                                                                                                                                                                             | an ange again and a second a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ang mang meng meng meng mengen sigan sigan si                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | nan antar antar an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| a.       | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | en el Garage de Sources                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | gene of new continuous and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | and the second states are second as a second s                                                                                                                                                                                                                                             | and reaction is a set of the second set of the second second second second second second second second second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ite receipt na maan barang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
| ·•       | n Nen ger men normal minneg serå avgir<br>Senerativ Sport den som avgir avgir Sport                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | the construction of the company was as                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | The second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | والماجا وأواجي والصلام فالطيم لأسلأ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | and a surprise in a straight of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 40       | Construction of New Western 2001 By                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | And a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and the second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | وليرك والأمر المستحد ومحاصر متحاوية المتحاوي                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | for the second states in the s |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |

Γ

1

Ć

Figure 12-2. Geologic Cros

Ì

Ĵ

on the second second

\*\*\*\*\*\*

10.000

|    | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 120                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 180                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 240                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 300                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 36                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|----|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|    | en en fan de fan de fan it en fan de fan<br>en en fan de f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | landa ang sa Sana kana kana kana di na sa                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ederada a factor a formada a formada a factor da seria d<br>Antimada a seria da se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | egenegan fas por ministrasfranjansjan jandarska<br>- Jangan fas por ministrasfranjan fas jandarska                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | - Antonia and a statement and a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | afrantes frantes en frantes de la contes de la                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and a star a<br>The star and a star and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 75 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and an international statement and an advantation of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and the second state of th | a ang mang mang mang mang mang mang mang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 1111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | franker og sen som en som                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | -sil1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|    | X                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | in the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 70 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | to a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 70 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ╺┝╍╼┝╍╼┿╼╼┝╍╼┢<br>┿╍╍┝╌╌┝╼╸┿╍╍╶╍                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┉┼┉┼┉┠╍╉┈╞┈┟╍╁╍╁╍╁╼╎                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | lanan farin samad nan samad nan samad nan samad na samad na sa<br>man samad nan samad nan samad nan samad na samad                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ****                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | Jacquer La Constant                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | WZ Z                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 65 | SILTY SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | and a set of the former and an end and an end and an end and a set of the set | en en de sand avec br>avec de sand avec de sand de sand avec de sand                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | ar valence and a subject for the subject for a subject for a subject for the s |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ¥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | , and an a start when a start                                                                                                                                                                                                                                             | ╾╍╡╾╍╡╸╍┧╸╴┠╴╍╞╼╼┟╌╴╞╼╍╪╼╼╞<br>╾┧┝╌╼┽╴╘╶┿┍╼╺┨┙┈┇╴╌┿╍╌╽╌╴┙╺┲┨╾╸┝                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | • • • • • • • • • • • • • • • • • • • •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and a second and a second and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 60 | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | หระสินการสุขภาพสินการสี่ เพราะสี่ (การสุขมองนี้) การสุขิมสรรมีการสำนัก<br>1999 ที่ - สำนักประการสี่งการสี่ง เพราะส์ (การสุขมองนี้) การสุขิมสรรมีการกับ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - Marine and Anna and Anna and Anna Anna Anna An                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ما معاصف المحمد المحمد المراجع معاملة من المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد<br>موال معارضة المحمد ا                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ······································                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 55 | and the second sec                                                                                                                                                                                                                                             | a series and an entry of the series of the s | and a second sec |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and an analysis of the second s |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an fan referin yn arste ar ar gener fan sy'n ar fan yn                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | ╺╋╼╌╬╍╊╍╊╼╞╍╞╼ <u>╞</u> ╼╞╼┝╸╞╍<br>╈━╬╍╬╍┲╴┢╍┟╍┟╺╻┪┙╏╍┪╸                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | andresi Seconda e a seconda                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | and and and an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|    | and a second and a second and a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and and a second and a second se                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1999 1999 1999 1997 200 and 1991 20 and 200                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ╼न्द्रेः द्वाराष्ट्रव्यक्रियोः<br>चन्द्रेः द्वाराष्ट्रव्यक्रियाः                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 50 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | n fa renne an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | alan di seri d<br>Na seri di seri                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | nderselsen iser in en men den sige sige sonder sonder.<br>Net er selsen sig sonderse sonderse sige sonderse sonder                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | · ways when the proved the part of the part of the proved of the part of the p | an al harden a gan a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and and a second s                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and an analysis and and an and an and a second and a second                                                                                                                                                                                                                                              | and the a strength of the state of the strength of the strengt | alin minan marine marine marine and a same                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second on the same of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | . alter and the second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|    | a contraction of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a dia mandri da a su a dia mandri da a su a dia mandri da ana di ana<br>Na dia mandri da a su a dia mandri da a su a dia mandri da ana di ana                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | and a state of the | rijev in graat merendenigen ander de nader al.<br>19 mij - 19                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | และการรู้และหรือสาวการรู้และการรู้และสาวการรู้และสาวการรู้และสาวการรู้และการรู้และการรู้และการรู้และการสาวการร                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n and a start of the start of t |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 45 | and any other than the set of the second and the se | n na serie a serie de la completa de                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | high weighter by the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 45 | and grant a suffer product of the same second and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | ۳۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۰۰۰ ۲۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹<br>۲۰۰۹ ۲۰۰۹ ۲۵٬۰۰۹ ۲۰۰۹ ۲۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹ ۲۵٬۰۰۹                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | a fan her fan de ser f<br>Geografie de ser fan de                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|    | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | a-1-12-10 http://www.analastania.com                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | in sum and a second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | and a second sec | ามมร้องสารเหตุการจะเพราะจะสารเรา (การการการการการการการการการการการการการก                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | for an and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a and a second and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|    | - month in the production of an interest of the second sec | the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | tan - marine and marine                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 10. مارسان به معامل المراقع الم                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ng transformations<br>ng transforma |
|    | and A. S. Salahan and a start of the second st | a series of the  | er 10,900 inner der der eine eine der eine der eine eine eine eine eine eine eine ei                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ter miner daman sandra di manda manda mangan sa pangan sa pangan sa pangan sa pangan sa pangan sa pangan sa pan<br>Ina sa pangan sa pang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | is when part could a particular factor is a<br>second could a start a start for the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 40 | When the information of compare any effective spectra that years a first state year of the state ye    | n in dees a finite provide provide a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | kannin nijeran najeratika na kanala angana<br>na nye najeran anjan kanalan kanala kanala                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | har han a har a har har har har har har har har                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| 40 | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | na lan sebasa terma termi (s.<br>mgan basan termi penantan terjar)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |

I

Ć

(

Ć

()

1



Ć

Figure 12-4. Water Table Contou

()

## 12.2 NATURE AND EXTENT OF CONTAMINATION AT BURN PIT D

## 12.2.1 Surface Soil Contamination

The nature and extent of surface soil contamination was evaluated using the results of analyses on surface soil samples taken from the two new monitoring wells (MW-5 and MW-6) and the three soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 12-1 summarizes the analytical results for surface soil samples and Figure 12-5 shows their distribution. This assessment presents Phase II contaminant data only, because no surface soil samples were collected during Phase I.

|           |                         |                                          | Ph                                       | ase II Sampl                             | es                                       |                                          |
|-----------|-------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|
| Parameter | Reference<br>Background | SB-1<br>4D1111<br>07/08/97<br>(0-1 foot) | SB-2<br>4D1211<br>07/08/97<br>(0-1 foot) | SB-3<br>4D1311<br>07/08/97<br>(0-1 foot) | MW-5<br>4D1511<br>07/09/97<br>(0-1 foot) | MW-6<br>4D1611<br>07/09/97<br>(0-1 foot) |
|           |                         | RCRA Me                                  | tals (mg/kg)                             |                                          |                                          |                                          |
| Arsenic   | 2.10                    | 0.34                                     |                                          | 0.27                                     |                                          | ·····                                    |
| Barium    | 14.7                    | 8.6                                      | 0.71                                     | 12.6                                     | 2.8                                      | 2                                        |
| Chromium  | 6.21                    | 1.1                                      |                                          | 2                                        | 2.1                                      | 0.74                                     |
| Lead      | 8.81                    | 6.4                                      |                                          | 7.7                                      | 2.6                                      | 1.2                                      |
| Mercury   | 0.03                    | 0.02                                     |                                          | 0.02                                     | 0.03                                     |                                          |

# Table 12-1. Summary of Analytical Results for Surface Soils Burn Pit D, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

RCRA = Resource Conservation and Recovery Act.

SVOCs. No SVOCs were reported above the detection limit in surface soil samples.

**RCRA metals**. None of the RCRA metals was reported above reference background values; therefore, metals in surface soil are not considered site related.

## 12.2.2 Subsurface Soil Contamination

The subsurface soil samples were collected from the four Phase I monitoring well boreholes, the two Phase II well boreholes, and the three Phase II soil borings. The samples were analyzed for VOCs and RCRA metals. Table 12-2 summarizes analytical results for the subsurface soil samples, and Figure 12-6 shows their distribution. Both Phase I and Phase II data are shown.

VOCs. BTEX compounds were identified only in the Phase I subsurface soil samples at MW-1, MW-2, and MW-3. Maximum concentrations of benzene ( $12 \mu g/kg$  at MW-1), toluene ( $21 \mu g/kg$  at MW-2), and xylenes ( $22 \mu g/kg$  at MW-2) were reported. No BTEX compounds were found in any of the Phase II subsurface soil samples.

PCE was reported in a single Phase I sample at a concentration of 9.7  $\mu$ g/kg at MW-4. PCE was not detected in the Phase II soil samples. Acetone was detected at SB-1, SB-3, and MW-2 at a maximum concentration of 110  $\mu$ g/kg, and 2-hexanone was detected in SB-1 at a concentration less than its average detection limit in background samples and is, therefore, not considered site related.

1



Ţ

Ć

Figure 12-5. Results of Analy:

Table 12-2. Summary of Analytical Results for Subsurface Soils Burn Pit D, Fort Stewart

(

(

(

|                      |                    |            | Phase I    | Phase I Samples |                                    |              |                 | Phase II Samulos                                                                                                           | loc          |                |
|----------------------|--------------------|------------|------------|-----------------|------------------------------------|--------------|-----------------|----------------------------------------------------------------------------------------------------------------------------|--------------|----------------|
|                      |                    | IWM        | MW2        | MW3             | MW4                                | MW-5         | MW-6            | SB-1                                                                                                                       | SR-7         | CR.3           |
|                      | -                  | SL1-4      | SL2-6      | SL3-6           | SL4-4                              | 4D1512       | 4D1612          | 4D1112                                                                                                                     | 401212       | 4D1317         |
|                      | Reference          | 06/24/93   | 06/24/93   | 06/24/93        | 06/24/93                           | 76/60/20     | <i>L6/60/L0</i> | 7/08/97                                                                                                                    | 76/80/20     | 07/08/97       |
| rarameter            | Background (2-4 fe | (2-4 feet) | (4-6 feet) | (4-6 feet)      | (2-4 feet)                         | (2.5-5 feet) | (5-7.5 feet)    | eet)   (4-6 feet)   (4-6 feet)   (2-4 feet)   (2.5-5 feet)   (5-7.5 feet)   (10-12.1 feet)   (5-7.5 feet)   (10-12.5 feet) | (5-7.5 feet) | (10-12.5 feet) |
|                      |                    |            | 0A         | latile Orga     | Volatile Organic Compounds (ue/ke) | inds (ue/ko) |                 |                                                                                                                            |              |                |
| 2-Hexanone           | 0.0                |            |            |                 |                                    | 0            |                 | 1                                                                                                                          |              |                |
| Acetone              | 0.0                |            | 110        |                 |                                    |              |                 | 76.7                                                                                                                       |              | 0.00           |
| Benzene              | 0.0                | 12         |            |                 |                                    |              |                 | 101                                                                                                                        |              | 2.26           |
| Tetrachloroethene    | 0.0                |            |            |                 | 9.7                                |              |                 |                                                                                                                            |              |                |
| Toluene              | 0.0                | 5.8        | 21         | 5.8             |                                    |              |                 |                                                                                                                            |              |                |
| Xylene               | 0.0                | 5.5        | 22         |                 |                                    |              |                 |                                                                                                                            |              |                |
|                      |                    |            |            | RCRA            | RCRA Metals (mo/ko)                | 140)         |                 |                                                                                                                            |              |                |
| Arsenic              | 8.04               |            |            | 1 2             |                                    | 1 1          |                 |                                                                                                                            |              | ,              |
| Barium               | 17.0               | 2.9        | 5 9        | AA              | C V                                | 0.2          | 15 6            | 67.7                                                                                                                       | 0.4/         | 3.1            |
| Cadmium              | 0.24               | 0.54       |            | 2               | 2                                  |              | 1.0             | c.c                                                                                                                        | 10.4         | 11.8           |
| Chromium             | 11.6               | 3.4        | 3.3        | 6.4             | 3.7                                | 0            | 22              | 5 0                                                                                                                        | -            | c u            |
| Lead                 | 11.1               | 2.7        | 3.7        |                 | 2.5                                | 64           | 5.3             | 2.2                                                                                                                        | 3 2          |                |
| Mercury              | 0.05               | 0.016      | 0.016      | 0.063           | 0.017                              | 0.06         | 0.05            | 60.0                                                                                                                       | 200          | <u></u>        |
|                      |                    |            |            | Other In        | Other Inorganics (me/kg            |              |                 | 1010                                                                                                                       | co.o         | C0.0           |
| Total Organic Carbon | 2,200              |            |            |                 |                                    | 2400         |                 |                                                                                                                            |              |                |
|                      |                    |            |            |                 |                                    |              |                 |                                                                                                                            |              |                |

Note: A blank indicates analyte not detected. Note: **Bold f**ont indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

97-160P(EXCEL)/011499

204

1

)



C

ſ

Acres 10400

1.000

**RCRA metals.** Two metals were reported at concentrations above their reference background value. Maximum values of cadmium (0.54 mg/kg at MW-1) and mercury (0.063 mg/kg at MW-3) were reported. The distribution of these metals at concentrations above background does not indicate a distinctive pattern; the metals are sporadically within the area defined by borings MW-1, MW-3, and MW-5.

#### 12.2.3 Groundwater Contamination

Groundwater contamination was evaluated using the results from water samples taken from the four monitoring wells installed during Phase I (MW-1 through MW-4) and two monitoring wells installed during Phase II (MW-5 and MW-6). The groundwater samples were analyzed for VOCs and RCRA metals (total metal analysis on unfiltered water samples). Table 12-3 summarizes the analytical results for groundwater samples, and Figure 12-7 shows their distribution. This assessment focuses on the Phase II contaminant data because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

| r         |                         |      |                            | P                          | hase II San                | ples                       |                            |                            |
|-----------|-------------------------|------|----------------------------|----------------------------|----------------------------|----------------------------|----------------------------|----------------------------|
| Parameter | Reference<br>Background | MCL  | MW-1<br>07/27/97<br>4D4111 | MW-2<br>07/29/97<br>4D4211 | MW-3<br>07/28/97<br>4D4311 | MW-4<br>07/28/97<br>4D4411 | MW-5<br>08/10/97<br>4D4511 | MW-6<br>08/10/97<br>4D4611 |
|           |                         |      | RCRA                       | Metals (µg                 | /L)                        |                            |                            |                            |
| Arsenic   | 3.02                    | 50   |                            |                            |                            |                            |                            | 4.2                        |
| Barium    | 71.72                   | 2000 | 99.6                       | 58                         | 12                         | 63.5                       | 51.7                       | 50.1                       |
| Cadmium   | 0.43                    | 5    |                            | 0.43                       |                            |                            | 0.22                       |                            |
| Chromium  | 3.56                    | 100  | 1.8                        |                            | 0.67                       | 2.9                        | 2.7                        | 2.5                        |
| Lead      | 4.69                    | 15   |                            | 0.2                        |                            |                            | 2.7                        | 1.2                        |
| Selenium  | 1.90                    | 50   |                            |                            |                            |                            | 1.1                        | 1.2                        |
| Silver    | 1.12                    | ·    | 3.6                        | 0.37                       | 0.09                       | 0.53                       | 0.23                       | 0.85                       |

| Table 12-3. | Summary of Analytical Results for Groundwater, |
|-------------|------------------------------------------------|
|             | Burn Pit D, Fort Stewart                       |

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act.

**VOCs.** No VOCs were reported above the detection limit in either Phase I or Phase II groundwater samples. These results suggest that although low concentrations of BTEX are present in subsurface soils at the site, the soil contamination has not impacted groundwater.

**RCRA metals.** Three metals were reported at concentrations above background in groundwater. Arsenic was reported at a maximum concentration in MW-6 (4.2  $\mu$ g/L), barium in MW-1 (99.6  $\mu$ g/L), and silver in MW-1 (3.6  $\mu$ g/L).

)



and the second sec

ſ

ĺ

Figure 12-7. Results of Analy

ŝ

-----

#### 12.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 12-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

| Analyte           | Surface Soil | Subsurface Soil | Groundwater |
|-------------------|--------------|-----------------|-------------|
|                   | (µg/kg)      | (µg/kg)         | (µg/L)      |
| Acetone           | na           | 110             | nd          |
| Benzene           | na           | 12              | nd          |
| 2-Hexanone        | na           | 1.4             | nd          |
| Tetrachloroethene | na           | 9.7             | nd          |
| Toluene           | па           | 21              | nd          |
| Xylenes, total    | na           | 22              | nd          |
|                   | (mg/kg)      | (mg/kg)         | (μg/L)      |
| Arsenic           | brc          | brc             | 4.2         |
| Barium            | brc          | brc             | 99.6        |
| Cadmium           | nd           | 0.54            | brc         |
| Chromium          | brc          | brc             | brc         |
| Lead              | brc          | brc             | brc         |
| Mercury           | brc          | 0.063           | nd          |
| Selenium          | nd           | nd              | brc         |
| Silver            | nd           | nd              | 3.6         |

# Table 12-4. Summary of Site-Related Constituents Burn Pit D, Fort Stewart

brc = Below background reference criteria.

na = Not analyzed in any sample in that medium.

nd = Not detected in any sample in that medium.

BOLD font indicates the analyte is an SRC in that medium.

## 12.3 FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT D

The results of contaminant migration soil screening (discussed in Section 6) for Burn Pit D are presented in Table 12-5. Benzene and PCE are identified as CMCOPCs based on leaching to groundwater. However, these constituents were not currently observed in groundwater, indicating that they may be degraded before reaching the water table. The predicted maximum concentration of benzene in groundwater, based on a maximum of 12  $\mu$ g/kg in soil and a DAF of 1, is 30  $\mu$ g/L. Through biodegradation, even if the benzene were to reach groundwater at such a concentration, it would degrade to its MCL value (5  $\mu$ g/L) in less than 2.6 times its half-life. Assuming a conservative half-life of 2.0 years, the benzene would degrade to its MCL value in 5.2 years. Similarly, PCE with its predicted maximum concentration of 16  $\mu$ g/L in groundwater would degrade to its MCL in less than 1.7 times its half-life, or 7.7 years.

The maximum concentrations of benzene and PCE were reported in Phase I samples only and were not detected in Phase II samples.

| SRCs              | Maximum<br>Concentration | GSSL            | Is Maximum<br>Concentration > GSSL? |
|-------------------|--------------------------|-----------------|-------------------------------------|
|                   | RCRA Meta                | l (mg/kg)       |                                     |
| Cadmium           | 0.54                     | 8               | No                                  |
| Mercury           | 0.063                    | 0.4             | No                                  |
|                   | Volatile Organic Co      | mpounds (µg/kg) |                                     |
| 2-Hexanone        | 1.40                     |                 | No                                  |
| Acetone           | 76.20                    | 800             | No                                  |
| Benzene           | 12.00                    | 2               | Yes                                 |
| Tetrachloroethene | 9.70                     | 3               | Yes                                 |
| Toluene           | 5.80                     | 600             | No                                  |
| Xylenes           | 5.50                     | 1000            | No                                  |

#### Table 12-5. CMCOPCs Based on Soil Screening for Burn Pit D

CMCOPCs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

RCRA = Resource Conservation and Recovery Act.

The most likely pathway of contaminant migration from Burn Pit D is via groundwater discharge to Mill Creek, located approximately 2500 feet east-northwest of the site. Based on the calculated groundwater flow velocity of 4.0 feet/year, the estimated arrival time for the site groundwater to reach Mill Creek is expected to be over 625 years. Therefore, benzene and PCE would degrade to less than their MCLs within 40 feet of the Burn Pit D site, well before reaching Mill Creek.

Metals detected in the groundwater at this site include arsenic, barium, and silver. However, the maximum concentrations of these metals do not exceed their respective MCLs and, based on fate and transport analysis, they are not expected to increase in the future. Therefore, none of the constituents is expected to be of concern for contaminant migration from Burn Pit D.

## 12.4 HUMAN HEALTH RISK EVALUATION OF BURN PIT D

SRCs were identified in Section 12.2 for subsurface soil and groundwater; there were no SRCs identified for surface soil. The identification of risk-based action levels for all of the Burn Pits has been previously discussed in Section 4.7 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit D is given below.

#### 12.4.1 Exposure Evaluation

This site is completely covered by vegetation, including thick stands of pine trees and other trees, and no VOCs were detected in surface soils. Therefore, exposure via inhalation based on current land use is not a viable exposure pathway. The site is not currently used by the Army. Therefore, current on-site receptors include a juvenile trespasser and a hunter. Current off-site receptors would be represented by an off-site hunter.

Land use at this site is unlikely to change in the future. However, the Army may use the site as a training area in the future. Construction may occur on the site or the vegetative cover may be gone as a result of on-site activities. Future on-site receptors include military personnel and a construction worker. Future off-site receptors include hunters in the area.

A summary of the potential exposure pathways for each of the receptor populations is given below.

214

**Current juvenile trespasser.** The juvenile trespasser may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Hunter. The hunter is representative of a current on-site and a current and future off-site receptor. The current on-site receptor is not likely to be exposed to chemicals given the absence of a air migration pathway, and clothing the hunter would be wearing would eliminate potential dermal exposure. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

The current off-site hunter may be exposed via ingestion of contaminants bioaccumulated in game animals. The future off-site hunter may be exposed via ingestion of game and inhalation of fugitive dust if the vegetative cover were to be removed.

Future on-site military personnel. The on-site military personnel may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

**Future construction worker.** The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

#### 12.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals will be evaluated further in a BRA.

None of the contaminants in surface soils or subsurface soils exceeded its respective risk-based screening value for direct exposure of a potential receptor population (Table 12-6).

In groundwater, arsenic is a potential COPC because the maximum concentration of arsenic  $(4.2 \ \mu g/L)$  exceeds its risk-based screening value of  $0.045 \ \mu g/L$  for residential use of groundwater as drinking water. However, the maximum concentration of arsenic is well below its MCL of 50  $\mu g/L$ . Arsenic was detected at a concentration only slightly above its reference background concentration of  $3.02 \ \mu g/L$ , and in only a single well at Burn Pit D (MW-6). Arsenic was not found above reference background criteria in either surface or subsurface soil of Burn Pit D. In addition, use of the surficial groundwater at this site for drinking water is unlikely. Therefore, arsenic is not considered a potential threat to human health at Burn Pit D. No other constituent exceeded its risk-based screening value or its MCL in groundwater at Burn Pit D.

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit D. A human health BRA is not warranted. No further action is required for protection of human health.

Table 12-6. Comparison to Action Levels-Burn Pit D, Fort Stewart

|                   |               |            |             |                            |             | EPA III   | -     |                    |
|-------------------|---------------|------------|-------------|----------------------------|-------------|-----------|-------|--------------------|
|                   |               | Frequency  | Minimum     | Maximum                    | Background  | Screening |       |                    |
| Analyte           | Units         | of Detects | Detect      | Detect                     | Criteria    | Criteria  | COPC? | Justification      |
|                   |               |            | Bu          | Burn Pit D Subsurface Soil | urface Soil |           |       |                    |
| Cadmium           | ma/ko         | 1/8        | 0.54        | 0.54                       | 0.24        | 1,000     | No    | Max Detect <       |
|                   | 0. A.         | )          |             |                            |             |           |       | Screening Criteria |
| Mercury           | mo/ko         | 8/8        | 0.016       | 0.063                      | 0.05        | 61.0      | No    | Max Detect <       |
|                   | <b>a</b> . b. | j          | ,<br>,<br>, |                            |             |           |       | Screening Criteria |
| 2-Hevanone        | mø/kø         | 1/6        | 0.0014      | 0.0014                     |             |           | No    | Max Detect <       |
|                   | 0             |            |             |                            |             |           |       | Screening Criteria |
| Aretone           | me/ko         | 2/9        | 0.0328      | 0.0762                     |             | 20,000    | No    | Max Detect <       |
|                   | 0- A-         |            |             |                            |             |           | *     | Screening Criteria |
| Ranzene           | ma/ko         | 1/9        | 0.012       | 0.012                      |             | 200       | No    | Max Detect <       |
|                   | 0 A           | 1          |             |                            |             |           |       | Screening Criteria |
| Tetrachlorcethene | ma/ko         | 1/9        | 0.0097      | 0.0097                     |             | 110       | N0    | Max Detect <       |
|                   | Q., Q.,       |            |             |                            |             |           |       | Screening Criteria |
| Toluene           | mo/ko         | 2/9        | 0.0058      | 0.0058                     |             | 41,000    | No    | Max Detect <       |
|                   | 0             | ì          |             |                            |             |           |       | Screening Criteria |
| Yulenes Total     | mo/ko         | 1/9        | 0.0055      | 0.0055                     |             | 100,000   | No    | Max Detect <       |
| time t termster   | 0<br>D        |            |             |                            |             |           |       | Screening Criteria |
|                   |               |            | 8           | Burn Pit D Groundwater     | oundwater   |           |       |                    |
| Arcenic           | 110/1.        | 1/5        | 4.2         | 4.2                        | 3.02        | 0.045     | Yes   | Max Detect >       |
|                   | b<br>L        |            |             |                            |             |           |       | Screening Criteria |
| Rarium            | 1/011         | 5/5        | 12          | 9.66                       | 71.72       | 260       | No    | Max Detect <       |
|                   | b             | 9          |             |                            |             |           |       | Screening Criteria |
| Silver            | . <u>I/an</u> | 5/5        | 0.09        | 3.6                        | 1.12        | 18        | No    | Max Detect <       |
|                   | l<br>b<br>L   |            |             |                            |             |           |       | Screening Criteria |
|                   |               |            |             |                            |             |           |       |                    |

COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency.

#### 12.5 ECOLOGICAL RISK EVALUATION OF BURN PIT D

The ecological risk evaluation of Burn Pit A is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are considered further in PRE Steps ii through v. The results of the five steps of the PRE are reported below.

51 h

#### 12.5.1 Ecological Screening Value Comparison (Step i)

Three RCRA metals were detected in groundwater at Burn Pit D at concentrations exceeding reference background criteria. The results of the ESV comparison for groundwater at Burn Pit D are presented in Table 12-7. At the upgradient monitoring well (MW-2), barium and silver concentrations exceed the surface water ESVs. Where detected in the remaining five monitoring wells (MW-1, and MW-3 through MW-6), barium and silver again exceed the ESVs for those analytes. The ecological COPCs identified by the ESV comparison for groundwater at Burn Pit D are barium and silver.

Although ecological COPCs are identified in groundwater at Burn Pit D when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit D. There are no surface water bodies in the vicinity of Burn Pit D. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit D soils.

Because there are no ESVs for soil, all analytes detected in soil at the Burn Pit D are evaluated further in PRE Steps ii through v.

| Analyte | ESV                | MW-2*  | MW-1   | MW-3        | MW-4   | MW-5   | MW-6   | Maximum |
|---------|--------------------|--------|--------|-------------|--------|--------|--------|---------|
|         |                    |        | RC     | RA Metals ( | μg/L)  |        |        |         |
| Arsenic | 190°               |        |        |             |        |        | 4.2 J  | 4.2 J   |
| Barium  | 3.9 <sup>b</sup>   | 58 J   | 99.6 J | 12 J        | 63.5 J | 51.7 J | 50.1 J | 99.6 J  |
| Silver  | 0.012 <sup>e</sup> | 0.37 J | 3.6 J  | 0.09 J      | 0.53 J | 0.23 J | 0.85 J | 3.6 J   |

# Table 12-7. EPA Region 4 Ecological Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit D, Fort Stewart

\*Upgradient monitoring well,

J = Estimated concentration.

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1996) and, where indicated, alternative values for analytes without Region 4 ESVs.

RCRA = Resource Conservation and Recovery Act.

"Arsenic III.

<sup>b</sup>Office of Solid Waste and Emergency Response Ambient Water Quality Criteria or Tier-II value (Suter and Tsao 1996). <sup>c</sup>Hardness dependent, assumes 50 mg/L CaCo<sub>3</sub>.

Bold font indicates detected concentration exceeds ESV.

#### 12.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit D comprises approximately 4 acres (Figure 12-1). The site is surrounded by woods and has become overgrown with grass, shrubs, and small pine trees. There are a few large trees in the center of the site. Drainage occurs through overland flow radially in all directions and is

collected in a series of man-made ditches that convey runoff to Mill Creek. The closest surface stream to the site is Mill Creek, a tributary of Taylors Creek, which lies approximately 2500 feet east-northeast of the site.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit D are described in Section 8.1.2.

#### 12.5.3 Preliminary Effects (Step iii)

In the PRE for Burn Pit D, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the site. The derivation of TRVs is discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8-1.

## 12.5.4 Preliminary Exposure (Step iv)

Ecological receptors at Burn Pit D are likely exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews and robins exposed to COPCs in soil are presented in Table 8-2.

## 12.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate a HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0, and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

Surface Soil. There are no constituents detected above background criteria in surface soil at Burn Pit D. Therefore, no preliminary risk calculation was performed.

#### 12.5.6 Uncertainties

There are no ecological COPCs identified in the PRE for Burn Pit D, so no supplemental evaluation of risk using published dietary fractions and LOAELs is necessary.

#### 12.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT D

The following are the conclusions of the Phase II RFI for Burn Pit D:

- In surface soils, no constituent was found above background. There are, therefore, no human health or ecological COPCs in surface soil.
- In subsurface soil, two metals (cadmium and mercury) were reported at concentrations above their reference background value. The metals are not present in a consistent pattern and are not considered to be due to releases that may have occurred from site-related activities. BTEX compounds were detected in three of the four Phase I boreholes around the perimeter

of the site, but were not detected in any of the four Phase II samples, including the three soil borings in the middle of the source area. Therefore, BTEX contamination is not considered to be present at Burn Pit D. PCE was identified in only one of the nine subsurface soil samples collected.

- PCE and benzene are potential contaminant migration COPCs in subsurface soil because these chemicals may leach to groundwater, resulting in groundwater concentrations exceeding their respective MCLs. However, they were not detected in groundwater and were detected in only one soil sample of the nine collected. Off-site migration of these contaminants would be very limited due to retardation and biodegradation and the slow movement of groundwater. Even if these constituents were to reach groundwater at their predicted maximum concentrations, they would degrade to concentrations less than their MCLs in less than 7.7 years, and would have traveled less than 40 feet from the Burn Pit D site. Therefore, migration of these constituents is considered unlikely, and there are no contaminant migration COPCs.
- In groundwater, three metals (arsenic, barium, and silver) were reported at concentrations above background in different wells and at concentrations only slightly above background. The metals do not indicate a pattern of distribution indicating a potential release from the former Burn Pit.
- In groundwater, arsenic is a potential COPC because it exceeded its risk-based screening value for residential use of groundwater as drinking water. However, arsenic did not exceed its MCL and was detected only slightly above its reference background concentration in a single well. In addition, use of the surficial groundwater at this site for drinking water is unlikely. Therefore, arsenic is not considered a potential threat to human health at Burn Pit D. No other constituent exceeded its risk-based screening value or its MCL in groundwater at Burn Pit D.
- Barium and silver are ecological COPCs in groundwater because they exceed the ESVs for surface water. However, there are no surface water bodies in the vicinity of Burn Pit D, and off-site migration would be limited due to retardation and the slow movement of groundwater. Therefore, exposure of ecological receptors to these metals in surface water bodies downgradient from Burn Pit D is not a complete pathway. Therefore, barium and silver are eliminated as ecological COPCs for Burn Pit D, and further investigation and/or evaluation of these constituents is not required.

The following are the recommendations for Burn Pit D:

- 1. No further action is recommended for Burn Pit D.
- 2. All potential human health COPCs were eliminated for Burn Pit D (Section 12.4), and a BRA is not recommended for the site.
- 3. Because there are no ecological COPCs identified in the PRE for Burn Pit D (Section 12.5), an ERA is not recommended for the site.

# **13.0 CHARACTERIZATION OF BURN PIT E (SWMU 4E)**

#### **13.1 HISTORY AND DESCRIPTION OF BURN PIT E**

Burn Pit E (SWMU 4E) comprises approximately 0.8 acres and is located on the north side of Fort Stewart Route 144, approximately 800 feet east from its junction with Fort Stewart Route 50 and approximately 1.1 miles north-northwest of State Highway 144 (Savannah Highway). Most of the site is grass covered. Isolated pine and hardwood trees are present, and the site is surrounded by woods to the north, east, and west. Some debris (e.g., a few pieces of plastic, such as a plastic knife) were observed during a Phase I site reconnaissance on November 29, 1993. A low, bermed and railroad cross-tie barricaded square excavation with a sign labeled "Misfire Pit 5" is located among trees in the eastern area of the site.

Surface elevations vary from approximately 74 to 75 feet amsl. The site slopes gently eastward with about 4 feet of relief from the western site boundary to MW-1, located about 300 feet east of the site (Figure 13-1). Drainage occurs through overland flow to the east, collecting in small unnamed streams and swampy areas that discharge directly to the Canoochee River. The closest surface stream to the site is the Canoochee River, which lies approximately 2500 feet north of the site. Swampy areas within the Canoochee River floodplain lie approximately 1000 feet east of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 13-1). Three RCRA metals, including barium, chromium, and lead, were detected above site-specific background levels in soils. RCRA metals were also detected above site-specific background levels in groundwater and included arsenic, cadmium, chromium, and lead. No VOCs were detected in either soil or groundwater samples in Phase I.

During the Phase II RFI activities, two new monitoring wells (MW-5 and MW-6) were installed and three soil borings (SB-1 through SB-3) were drilled at the site (Figure 13-1). During redevelopment of the Phase I wells, MW-2 was discovered to have a bent or separated pipe casing, which prevented the surge block from being inserted into the well. MW-2 was, therefore, abandoned by filling the well with grout, and well MW-6 was constructed to replace it. MW-1 could not be developed or sampled due to a depth of water in the well of less than 1 foot and an extremely slow recharge rate, which resulted in the well being purged dry. Therefore, Phase I wells MW-3 and MW-4, together with the Phase II wells and borings, were used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of slightly silty sands. Geologic cross-sections of the Burn Pit E site are presented in Figure 13-2. Geotechnical parameters were measured in a bulk soil sample taken from MW-6 at a depth of 15 to 17.5 feet and in a Shelby tube sample taken from MW-5 at a depth of 15 to 17 feet. The results of the geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic sands with less than 4 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from MW-5, is  $3.11 \times 10^{-4}$  cm/second, typical for a fine sand.

Water levels in the monitoring wells were measured during well development between July 13 and 27, 1997, and again during well sampling between July 29 and August 11, 1997. The water table is present at a depth 13 feet bgs. A water table contour map is provided as Figure 13-3.



Figure 13-1. Locations of

|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 60 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 120                                                                                                             | 180                          | 240                                                                                                              | 530                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|----|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|    | A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2                                                                                                               |                              |                                                                                                                  | <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 75 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | B<br>B                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| -  | SILTY.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 70 | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|    | SILTY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | SILTY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | STLTY<br>SAND                                                                                                   | STL TY<br>SAND               |                                                                                                                  | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
| 65 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <ol> <li>Control de autoritation de la control de la c</li></ol> |                                                                                                                 |                              |                                                                                                                  | SILTY<br>SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| Ā  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 60 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  | - <u>\$</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| -  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 50 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 45 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              | and an entropy of the state of the |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                 |                              |                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 40 | and an a specific and | And the second s       | in the second | and the second second second | a sum and a speed on the second second or a factor of                                                            | , in the second se |

Figure 13-2. Geologic Cross-Sec

.

()



(

(

Ć

Figure 13-3. Water Table Contou

and the second second second second

Groundwater contours measured during the Phase I RFI activities in August 1993 indicate similar groundwater flow direction, although water levels were about 2 feet higher in the August 1993 data. Groundwater flow is generally to the east, discharging ultimately to low-lying swampy areas within the Canoochee River floodplain, approximately 1000 feet east of the Burn Pit E site. The calculated horizontal hydraulic gradient in the water table across the site is 0.009 foot/foot. Based on a hydraulic conductivity of  $3.11 \times 10^{-4}$  cm/second and a porosity of 0.464, the groundwater flow rate is approximately 6.2 feet/year toward the east.

#### **13.2 NATURE AND EXTENT OF CONTAMINATION AT BURN PIT E**

#### **13.2.1 Surface Soil Contamination**

The nature and extent of surface soil contamination was evaluated using the results of analyses on surface soil samples taken from the two new monitoring wells (MW-5 and MW-6) and the three soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 13-1 summarizes the analytical results for surface soil samples, and Figure 13-5 shows their distribution. This assessment presents Phase II contaminant data only because no surface soil samples were collected during Phase I.

|                      |                         |                                          | P                                        | hase II San                              | nples                                    |                                          |  |  |  |  |  |
|----------------------|-------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|------------------------------------------|--|--|--|--|--|
| Parameter            | Reference<br>Background | SB-1<br>4E1111<br>07/13/97<br>(0-1 foot) | SB-2<br>4E1211<br>07/13/97<br>(0-1 foot) | SB-3<br>4E1311<br>07/13/97<br>(0-1 foot) | MW-5<br>4E1511<br>07/23/97<br>(0-1 foot) | MW-6<br>4E1611<br>07/24/97<br>(0-1 foot) |  |  |  |  |  |
|                      | Semivolatile            | Organic Co                               | ompounds (µ                              | g/kg)                                    |                                          |                                          |  |  |  |  |  |
| Benzo(a)pyrene       | 0.0                     |                                          |                                          |                                          |                                          | 21                                       |  |  |  |  |  |
| Benzo(b)fluoranthene | 0.0                     |                                          |                                          |                                          |                                          | 54.8                                     |  |  |  |  |  |
| Fluoranthene         | 0.0                     |                                          |                                          |                                          |                                          | 63                                       |  |  |  |  |  |
| RCRA Metals (mg/kg)  |                         |                                          |                                          |                                          |                                          |                                          |  |  |  |  |  |
| Arsenic              | 2.10                    |                                          |                                          |                                          | 0.17                                     | **************************************   |  |  |  |  |  |
| Barium               | 14.7                    | 15.4                                     | 11.4                                     | 15                                       | 18.3                                     | 17.2                                     |  |  |  |  |  |
| Chromium             | 6.21                    | 1.8                                      | 1.3                                      | 1.6                                      | 1.2                                      | 2.3                                      |  |  |  |  |  |
| Lead                 | 8.81                    | 3                                        | 4.4                                      | 6.4                                      | 8.5                                      | 5.4                                      |  |  |  |  |  |
| Mercury              | 0.03                    | 0.03                                     |                                          | 0.04                                     |                                          | 0.03                                     |  |  |  |  |  |
| Selenium             | 0.41                    |                                          |                                          |                                          | 0.39                                     |                                          |  |  |  |  |  |

# Table 13-1. Summary of Analytical Results for Surface Soils Burn Pit E, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

RCRA = Resource Conservation and Recovery Act.

**SVOCs.** Benzo(a)pyrene (21  $\mu$ g/kg), benzo(b)flouranthene (54.8  $\mu$ g/kg), and flouranthene (63  $\mu$ g/kg) were detected in the surface soil sample at MW-6 at the southeastern edge of the former burn area. SVOCs were not detected in other surface soil samples at the site.

RCRA metals. Barium and mercury were the only RCRA metals reported above reference background values. Barium was found at four of the five surface soil sampling locations with a

:


ſ

Ć

Figure 13-4. Results of Analys

maximum concentration reported at MW-6 (18.3 mg/kg). Mercury was reported only at SB-3 at a concentration (0.04 mg/kg) that is only slightly greater than background and is, therefore, not likely to be site related.

#### 13.2.2 Subsurface Soil Contamination

The subsurface soil samples were collected from the four Phase I monitoring well boreholes, the two Phase II well boreholes, and the three Phase II soil borings. The samples were analyzed for VOCs and RCRA metals. Table 13-2 summarizes analytical results for the subsurface soil samples, and Figure 13-5 shows their distribution. Both Phase I and Phase II data are shown.

VOCs. Toluene was identified in four of the Phase II subsurface soil samples, at SB-1, SB-3, MW-5, and MW-6. Maximum concentration of toluene (8.8  $\mu$ g/kg) was reported at MW-2. These concentrations are lower than the average detection limit for toluene and may, therefore, not be indicative of site-related contamination. Benzene (7.9  $\mu$ g/kg at MW-4) was the only BTEX compound found in any of the Phase I subsurface soil samples.

RCRA metals. Three metals were reported in subsurface soils at concentrations above their reference background value. Barium was detected at concentrations above background in one of the nine subsurface soil samples, with a maximum value (27 mg/kg) at MW-1, which is located approximately 300 feet from the burn area. Maximum concentrations of chromium (9.5 mg/kg at SB-2) and mercury (0.063 mg/kg at MW-3) were also reported. These metals were detected at concentrations only slightly above background and do not show a consistent pattern of distribution.

#### 13.2.3 Groundwater Contamination

Groundwater contamination was evaluated using the results from water samples taken from two monitoring wells installed during Phase I (MW-3 and MW-4) and two monitoring wells installed during Phase II (MW-5 and MW-6). The groundwater samples were analyzed for VOCs and RCRA metals (total metal analysis on unfiltered water samples). Table 13-3 summarizes the analytical results for groundwater samples, and Figure 13-6 shows their distribution. This assessment focuses on the Phase II contaminant data because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

VOCs. Low concentrations of BTEX compounds were found in groundwater at MW-4. Ethylbenzene  $(1.1 \ \mu g/L)$  and xylenes  $(6.3 \ \mu g/L)$  were reported at concentrations only slightly above their detection limits. No VOCs were reported above the detection limit in Phase I groundwater samples. These results suggest that the low concentrations of BTEX present in subsurface soils at the site may have also impacted groundwater, but not at significant concentrations.

PCE was detected in MW-4 at 1.1  $\mu$ g/L, but was not detected in other wells during Phase II. No PCE was reported in Phase I groundwater samples. Acetone was reported in MW-4 and MW-6 at a maximum concentration (53.4  $\mu$ g/L).

**RCRA metals**. Cadmium was the only RCRA metal detected above background in groundwater. Cadmium was found at a maximum concentration in MW-4 (4.3  $\mu$ g/L) and was also detected above background in MW-6 (1.2  $\mu$ g/L). Because both of these wells are downgradient of the site, cadmium is considered a potential SRC.

| Table 13-2. Summary of Analytical Results for Subsurface Soils           Documents | During 1 to 29 Junior at 1 |
|------------------------------------------------------------------------------------|----------------------------|

|                                              |                    |            | Phase I    | <b>Phase I Samples</b> |                                      |             | Ρh                     | <b>Phase II Samples</b> | S            |              |
|----------------------------------------------|--------------------|------------|------------|------------------------|--------------------------------------|-------------|------------------------|-------------------------|--------------|--------------|
|                                              |                    | 1WM        | MW2        | MW3                    | MW4                                  | MW-5        | 9-WM                   | SB-1                    | SB-2         | SB-3         |
|                                              |                    | SL1-6      | SL2-8      | <b>SL3-8</b>           | SL4-6                                | 4E1512      | 4E1612                 | 4E1112                  | 4E1212       | 4E1312       |
|                                              |                    | 06/26/93   | 06/29/93   | 06/30/93               | 06/29/93                             | 07/23/97    | 07/24/97               | 07/13/97                | 07/13/97     | 07/13/97     |
|                                              |                    | (4-6 feet) | (6-8 feet) | (6-8 feet)             | (4-6 feet)                           | (2.5-5 feet | (10-12.5 feet)         | (2.5-4.5 feet)          | (4.5-7 feet) | 9.5-12 feet) |
|                                              | Reference          | 0004.0     | 0000       | 0.000                  | 0004.0                               | 0002.5      | 0010.0                 | 0002.5                  | 0004.5       | 0009.5       |
| Parameter                                    | Background         | 000        | 0008.0     | 0008.0                 | 0006.0                               | 0005.0      | 0012.5 0004.5 0007.0 0 | 0004.5                  | 0007.0       | 0012.0       |
|                                              |                    |            | Vola       | tile Organi            | Volatile Organic Compounds ( µ g/kg) | ds (μg/kg)  |                        |                         |              |              |
| Benzene                                      | 0.0                |            |            |                        | 7.9                                  |             |                        |                         |              |              |
| Toluene                                      | 0.0                |            |            |                        |                                      | 2.7         | 8.8                    | 7.4                     |              | 5.5          |
|                                              |                    |            |            | RCRAI                  | RCRA Metals (mg/kg)                  | kg)         |                        |                         |              |              |
| Arsenic                                      | 8.04               |            |            |                        |                                      |             |                        |                         | 0.2          |              |
| Barium                                       | 17.0               | 27         | 14         | 8.5                    | 11                                   | 14.5        | 4.8                    | 11                      | 14.6         | 11.3         |
| Chromium                                     | 11.6               | 12         | 6,5        | 5.6                    | 6.9                                  | 1.8         | 5.7                    | 2.5                     | 8            | 7.9          |
| Lead                                         | 11.1               | 5.9        | 2.6        | 5.2                    | 4.8                                  | 2.9         | 2.2                    | 2.5                     | 4.7          | 5.2          |
| Mercury                                      | 0.05               | 0.026      | 0.039      | 0.062                  | 0.043                                |             | 0.03                   | 0.03                    | 0.08         | 0.04         |
| Silver                                       | 0.46               |            |            |                        |                                      | 0.19        |                        |                         |              |              |
|                                              |                    |            | -          | Other Inc              | Other Inorganics (mg/kg)             | g/kg)       |                        |                         |              |              |
| Total Organic Carbon                         | 2,200              |            |            |                        |                                      | 824         |                        |                         |              |              |
|                                              |                    | -          |            |                        |                                      |             |                        |                         |              |              |
| Note: A blank indicates analyte not detected | analyte not detect | ha         |            |                        |                                      |             |                        |                         |              |              |

Note: A blank indicates analyte not detected. Note: **Bold** font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

97-160P(EXCEL)/011499

A Control Constant ACC ......



(

(

(

Figure 13-5. Results of Analyse

------

- -

- 1964

## THIS PAGE INTENTIONALLY LEFT BLANK



Figure 13-6. Results of Analy

()

()

: )

## THIS PAGE INTENTIONALLY LEFT BLANK

|                   |                         |                | P                          | hase II Sample             | S                          |                            |
|-------------------|-------------------------|----------------|----------------------------|----------------------------|----------------------------|----------------------------|
| Parameter         | Reference<br>Background | MCL            | MW-3<br>07/29/97<br>4E4311 | MW-4<br>07/30/97<br>4E4411 | MW-5<br>08/11/97<br>4E4511 | MW-6<br>08/11/97<br>4E4611 |
|                   |                         | Volatile Organ | ic Compounds (             | ug/L)                      |                            |                            |
| Acetone           | 0.0                     | ·········      | Í                          | 53.4                       |                            | 14.5                       |
| Ethylbenzene      | 0.0                     | 700            |                            | 1.1                        |                            |                            |
| Tetrachloroethene | 0.0                     | 5              |                            | 1                          |                            |                            |
| Xylenes, Total    | 0.0                     | 10,000         |                            | 6.3                        |                            |                            |
|                   |                         | RCRA           | Metals (µg/L)              |                            | •                          |                            |
| Barium            | 71.72                   | 2000           | 6.9                        | 10.2                       | 17.7                       |                            |
| Cadmium           | 0.43                    | 5              |                            | 4.3                        | 0.23                       | 1.2                        |
| Chromium          | 3.56                    | 100            | 1.5                        | 1.3                        | 1.7                        | 1.7                        |
| Selenium          | 1.90                    | 50             |                            |                            | 0.43                       |                            |
| Silver            | 1.12                    |                |                            | 0.1                        | 0.16                       |                            |

# Table 13-3. Summary of Analytical Results for Groundwater Burn Pit E, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act.

#### 13.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 13-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

#### **13.3 FATE AND TRANSPORT CONSIDERATIONS OF BURN PIT E**

The results of contaminant migration soil screening (discussed in Section 6.0) for Burn Pit E are presented in Table 13-5. Only benzene is identified as a CMCOPC based on leaching to groundwater. However, benzene is not currently observed in groundwater, indicating that it may be degraded before reaching the water table. The predicted maximum concentration of benzene in groundwater, based on a maximum of 7.9  $\mu$ g/kg in soil and a DAF of 1, is 20  $\mu$ g/L. Through biodegradation, even if the benzene were to reach groundwater at such a concentration, it would degrade to its MCL value (5  $\mu$ g/L) in less than 2.0 times its half-life. Assuming a conservative half-life of 2.0 years, the benzene would degrade to its MCL value in 4.0 years.

The maximum concentrations of benzene were reported in Phase I soil samples only and were not detected in Phase I samples.

| Analyte              | Surface Soil | Subsurface Soil | Groundwater |
|----------------------|--------------|-----------------|-------------|
|                      | (μg/kg)      | (µg/kg)         | (µg/L)      |
| Acetone              | na           | nd              | 53.4        |
| Benzene              | na           | 7.9             | nd          |
| Ethylbenzene         | na           | nd              | 1.1         |
| Tetrachloroethene    | na           | nd              | 1           |
| Toluene              | na           | 8.8             | nd          |
| Xylenes, total       | na           | nd              | 6.3         |
|                      | (µg/kg)      | (µg/kg)         | (μg/L)      |
| Benzo(a)pyrene       | 21           | na              | na          |
| Benzo(b)flouranthene | 54.8         | na              | na          |
| Flouranthene         | 63           | na              | na          |
|                      | (mg/kg)      | (mg/kg)         | (μg/L)      |
| Arsenic              | brc          | brc             | nd          |
| Barium               | 18.3         | 27              | brc         |
| Cadmium              | nd           | nd              | 4.3         |
| Chromium             | brc          | 12              | brc         |
| Lead                 | brc          | brc             | nd          |
| Mercury              | 0.04         | 0.08            | nd          |
| Selenium             | brc          | nd              | brc         |
| Silver               | Nd           | brc             | brc         |

# Table 13-4. Summary of Site-Related Constituents Burn Pit E, Fort Stewart

brc = Below background reference criteria

na = Not analyzed in any sample in that medium

nd = Not detected in any sample in that medium

BOLD font indicates the analyte is an SRC in that medium

#### Table 13-5. CMCOPCs Based on Soil Screening for Burn Pit E, Fort Stewart, Georgia

| SRCs                 | Maximum<br>Concentration | GSSL              | Is Maximum<br>Concentration >GSSL? |
|----------------------|--------------------------|-------------------|------------------------------------|
|                      | RCRA Meta                | ls (µg/kg)        |                                    |
| Barium               | 27.00                    | 1600              | No                                 |
| Chromium             | 12.00                    | 38                | No                                 |
| Mercury              | 0.08                     | 0.4               | No                                 |
|                      | Volatile Organic Co      | mpounds (µg/kg)   | ·                                  |
| Benzene              | 7.90                     | 2                 | Yes                                |
| Toluene              | 7.00                     | 600               | No                                 |
|                      | Semivolatile Organic     | Compounds (µg/kg) | I                                  |
| Benzo(a)pyrene       | 21.00                    | 400               | No                                 |
| Benzo(b)fluoranthene | 54.80                    | 200               | No                                 |
| Fluoranthene         | 63.00                    | 210000            | No                                 |

CMCOPCs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

SRCs = Site-related Contaminants.

The most likely pathway of contaminant migration from Burn Pit E is via groundwater discharge to swampy areas along the Canoochee River, located approximately 1000 feet east of the site. Based on the calculated groundwater flow velocity of 6.2 feet/year, the estimated arrival time for the site groundwater to reach the Canoochee River is expected to be more than 161 years. Therefore, benzene would degrade to less than its MCL within 30 feet of the Burn Bit E Site, well before reaching the Canoochee River.

Although acetone, ethylbenzene, PCE, and xylene are not identified as CMCOPCs based on leaching to groundwater from Burn Pit E, these constituents are currently observed in the site groundwater. However, the maximum observed concentrations for these constituents are below their respective MCLs. Metals detected in the groundwater at concentrations above background include cadmium only. However, the maximum concentration of cadmium does not exceed its MCL and, based on fate and transport analysis, cadmium is not expected to increase in the future. Therefore, none of the constituents is expected to be of concern for contaminant migration from Burn Pit E.

#### 13.4 HUMAN HEALTH RISK CHARACTERIZATION OF BURN PIT E

SRCs were identified in Section 13.3 for surface soil, subsurface soil, and groundwater. The identification of risk-based action levels for all of the Burn Pits has been previously discussed in Section 4.7 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit E is given below.

#### 13.4.1 Exposure Evaluation

This site is completely covered by vegetation, and no VOCs were detected in surface soils. Therefore, exposure via inhalation based on current land use is not a viable exposure pathway. The site is infrequently used by the Army as a training area. Therefore, current on-site receptors include a juvenile trespasser and a hunter. Current off-site receptors would be represented by an off-site hunter.

Land use at this site is unlikely to change in the future. However, the Army may use the site as a training area in the future. Construction may occur on the site or the vegetative cover may be gone as a result of on-site activities. Future on-site receptors include military personnel and a construction worker. Future off-site receptors include hunters in the area.

A summary of the potential exposure pathways for each of the receptor populations is given below.

**Current juvenile trespasser.** The juvenile trespasser may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Hunter. The hunter is representative of a current on-site receptor and a current and future off-site receptor. The current on-site receptor is not likely to be exposed to chemicals given the absence of an air migration pathway, and clothing the hunter would be wearing would eliminate

potential dermal exposure. VOCs were only detected in subsurface soils. Significant exposure via volatilization is not likely.

The current off-site hunter may be exposed via ingestion of contaminants bioaccumulated in game animals. The future off-site hunter may be exposed via ingestion of game and inhalation of fugitive dust if the vegetative cover were to be removed.

**Future on-site military personnel**. The on-site military personnel may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

**Future construction worker.** The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

#### 13.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals will be evaluated further in a BRA.

None of the contaminants in surface soils or subsurface soils exceeded its respective risk-based screening value for direct exposure of a potential receptor population (Table 13-6).

The maximum concentration of cadmium in groundwater (4.3  $\mu$ g/L) exceeds the screening value of 1.8  $\mu$ g/L (Table 13-6). However, the maximum concentration of cadmium was below its MCL of 5  $\mu$ g/L. The remaining chemicals were below their respective screening values. In addition, the use of surficial groundwater at the sites for drinking water is unlikely. Therefore, cadmium is not considered a potential threat to human health.

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit E. A human health BRA is not warranted. No further action is required for protection of human health.

#### 13.5 ECOLOGICAL RISK EVALUATION AT BURN PIT E

The ecological risk evaluation of Burn Pit A is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are considered further in PRE Steps ii through v. The results of the five steps of the PRE are reported below.

#### 13.5.1 Ecological Screening Value Comparison (Step i)

One RCRA metal and four VOCs were detected in groundwater at the Burn Pit E at concentrations exceeding reference background criteria. The results of the ESV comparison for groundwater at the Burn Pit E are presented in Table 13-7. At the upgradient monitoring well

|                                  | 1                       |                                 | Т                               | 7                               | 1                               |                                 |                            | 1                               |                                 | T I                             |                                 | 1                               | T                      | <u> </u>                        | η                               | <b>T</b>                        |                                 | 1                               |
|----------------------------------|-------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|----------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|
| Turtificantion                   | VUSHACALIVI             | Max Detect < Screening Criteria |                            | Max Detect < Screening Criteria |                        | Max Detect > Screening Criteria | Max Detect < Screening Criteria |
| 6.JQUJ                           |                         | No                              | No                              | No                              | No                              | No                              |                            | No                              | No                              | No                              | No                              | No                              |                        | Yes                             | No                              | No                              | No                              | No                              |
| EPA III<br>Screening<br>Criteria |                         | 550                             | 0.08                            | 0.8                             | 310                             | 2.3                             | oil                        | 14,000                          | 1,000                           | 61.0                            | 200                             | 41,000                          | r                      | 1.8                             | 3700                            | 1300                            | 1.1                             | 12000                           |
| Background<br>Criteria           | Burn Pit E Surface Soil | 14.7                            |                                 |                                 |                                 | 0.03                            | Burn Pit E Subsurface Soil | 17.0                            | 11.6                            | 0.05                            |                                 |                                 | Burn Pit E Groundwater | 0.43                            |                                 |                                 |                                 |                                 |
| Maximum<br>Detect                | Burn F                  | 18.3                            | 0.02                            | 0.054                           | 0.06                            | 0.04                            | Burn Pit                   | 27                              | 12                              | 0.08                            | 0.0079                          | 0.0074                          | Burn Pi                | 4.3                             | 53.4                            | 1.1                             | 1                               | 6.3                             |
| Minimum<br>Detect                |                         | 11.4                            | 0.02                            | 0.054                           | 0.06                            | 0.03                            |                            | 11                              | 1.8                             | 0.026                           | 0.0079                          | 0.0027                          |                        | 0.23                            | 14.5                            | 1.1                             | -                               | 6.3                             |
| Frequency<br>of Detects          |                         | 5/5                             | 1/5                             | 1/5                             | 1/5                             | 3/5                             |                            | LIL                             | LIL                             | 6/7                             | 1/6                             | 3/6                             |                        | 3/3                             | 2/3                             | 1/3                             | 1/3                             | 1/3                             |
| Units                            |                         | mg/kg                           | mg/kg                           | mg/kg                           | mg/kg                           | mg/kg                           |                            | mg/kg                           | mg/kg                           | mg/kg                           | mg/kg                           | mg/kg                           |                        | μg/L                            | µg/L                            | цg/L                            | µg/L                            | μg/L                            |
| Analyte                          |                         | Barium                          | Benzo(a)pyrene                  | Benzo(b)fluoranthene            | Fluoranthene                    | Mercury                         |                            | Barium                          | Chromium                        | Mercury                         | Benzene                         | Toluene                         |                        | Cadmium                         | Acetone                         | Ethylbenzene                    | Tetrachloroethene               | Xylenes, Total                  |

Table 13-6. Comparison to Action Levels-Burn Pit E, Fort Stewart

(

(

COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency.

#### Table 13-7. EPA Region 4 Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit E, Fort Stewart

| Analyte           | ESV               | MW-3*       | MW-4        |       | MW-5     | MW-6   | Maxim | ım |
|-------------------|-------------------|-------------|-------------|-------|----------|--------|-------|----|
| <u></u>           | <u> </u>          | RC          | RA Metals ( | ug/L) |          |        |       |    |
| Cadmium           | 0.66 <sup>a</sup> |             | 4.3         | J     | 0.23 J   | 1.2    | 4.3   | J  |
|                   | ·                 | Volatile Or | ganic Comp  | ounds | s (μg/L) |        |       |    |
| Acetone           | 10000             |             | 53.4        |       |          | 14.5 J | 53.4  |    |
| Ethylbenzene      | 453               |             | 1.1         | J     |          |        | 1.1   | J  |
| Tetrachloroethene | 84                |             | 1           | J     |          |        | 1     | J  |
| Xylenes, Total    | 7100 <sup>c</sup> |             | 6.3         |       |          |        | 6.3   |    |

\*Upgradient monitoring well.

J = Estimated concentration.

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1996) and, where indicated, alternative values for analytes without Region 4 ESVs.

RCRA = Resource Conservation and Recovery Act.

"Hardness dependent, assumes 50 mg/L CaCO3.

<sup>b</sup>LC<sub>50</sub> daphnid (Verchueren 1977).

LCo salmonid (Clayton and Clayton 1981).

Bold font indicates detected concentrations exceeds ESV.

(MW-3), no constituent concentration exceeds the surface water ESVs. Where detected in the remaining three monitoring wells (MW-4 through MW-6), the maximum detected concentration of cadmium exceeds the ESV. The only ecological COPC identified by the ESV comparison for groundwater at Burn Pit E is cadmium.

Although ecological COPCs are identified in groundwater at Burn Pit E when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit E. There are no surface water bodies in the vicinity of Burn Pit E. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of the contaminants in Burn Pit E soils.

Because there are no ESVs for soil, all analytes detected in soil at the Burn Pit E are evaluated further in PRE Steps ii through v.

#### 13.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit E (SWMU E) comprises approximately 0.8 acres (Figure 13-1). Most of the site is grass covered. Isolated pine and hardwood trees are present, and the site is surrounded by woods to the north, east, and west. The site slopes gently eastward. Drainage occurs through overland flow to the east, collecting in small unnamed streams and swampy areas that discharge directly to the Canoochee River. The closest surface stream to the site is the Canoochee River, which lies approximately 2500 feet north of the site.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit E are described in Section 8.1.2.

#### 13.5.3 Preliminary Effects (Step iii)

In the PRE for the Burn Pit E, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the site. The derivation of TRVs is discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8-1.

#### 13.5.4 Preliminary Exposure (Step iv)

Ecological receptors at Burn Pit E are likely exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews, and robins exposed to COPCs in soil are presented in Table 8-2.

#### 13.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate a HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0, and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

**Surface Soil.** The preliminary risk calculation for shrews and robins exposed to ecological COPCs detected in soil at Burn Pit E is presented in Table 13-8. This table shows the maximum detected concentrations and the TRVs for shrews and robins. Concentrations resulting in HQs exceeding 1.0 are shown in **boldface** font.

There is no RCRA metal present in Burn Pit E surface soil at concentrations exceeding the TRV for the surrogate species. An HI is not calculated for the surrogate species exposed to RCRA metals detected in surface soil because they are assumed to have dissimilar mechanisms of toxicity.

The three organic ecological COPCs are not likely to pose a risk to ecological receptors. The maximum detected concentrations are several hundred times smaller than the conservative TRVs for small mammals. An HI can be calculated for the surrogate species exposed to SVOCs detected in surface soil, assuming that they have similar mechanisms of toxicity on small mammals. The HI for the shrew exposed to the three SVOCs in surface soil is 0.03, using the surrogate TRVs for benzo(b)fluoranthene and fluoranthene. Because the HI is less than 1.0, these three SVOCs in soil at Burn Pit E are not ecological COPCs for populations of small mammals ingesting earthworms and other soil-dwelling invertebrates. There are, therefore, no ecological COPCs in Burn Pit E surface soil.

#### Table 13-8. Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 ft) at Burn Pit E, Fort Stewart

|                      | TF    | <b>IV</b> |              |               |            |        |      |         |
|----------------------|-------|-----------|--------------|---------------|------------|--------|------|---------|
| Analyte              | Shrew | Robin     | MW-5         | MW-6          | SB-1       | SB-2   | SB-3 | Maximum |
|                      |       |           | RCRA M       | etals (mg/kg) |            |        |      |         |
| Barium               | 2627  | 2293      | 18.3         | 17.2          | 15.4 J     | 11.4 J | 15 J | 18.3    |
| Mercury              | 14.1  | 1.1       |              | 0.03          | 0.03       |        | 0.04 | 0.04    |
|                      | A =   | Semiv     | olatile Orga | iic Compound  | ls (mg/kg) |        |      |         |
| Benzo(a)pyrene       | 39.6  | -         |              | 0.021         |            |        |      | 0.021   |
| Benzo(b)fluoranthene | 3.96  | -         |              | 0.055         |            |        |      | 0.055   |
| Fluoranthene         | 3.96  | -         | ·            | 0.063         |            |        |      | 0.063   |

J = Estimated concentration.

- = No data to derive TRV.

RCRA = Resource Conservation and Recovery Act.

TRV = Toxicity Reference Value = (NAOEL × BW/Food Ingestion)/BAFearthworm (see Table 8-1).

Blank cells indicate analyte not detected at site.

Boldface indicates detected concentration exceeds TRB (HQ>1).

#### 13.5.6 Uncertainties

There are no ecological COPCs identified in the PRE for Burn Pit E, so no supplemental evaluation of risk using published dietary fractions and LOAELs is necessary.

#### 13.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PITE

The following are the conclusions of the Phase II RFI for Burn Pit E:

- In surface soils, barium and mercury were detected in four of the five samples. Barium contamination extends throughout the Burn Pit E site and is an SRC. Mercury was found at a concentration only slightly above background in a single sample and is not likely site related.
- SVOCs were detected in a single surface soil sample at Burn Pit E at the southeastern edge of the former burn area. The SVOCs were detected at concentrations only slightly above their detection limit and do not pose a human health or ecological risk.
- There are no human health COPCs in surface or subsurface soil because no constituent exceeded its risk-based screening value.
- There are no ecological COPCs in surface soils because no constituent was detected above background that also exceeded its TRVs for ecological receptors.
- In subsurface soil, three metals (barium, chromium, and mercury) were reported at concentrations above their reference background value. Barium was detected in one of the nine subsurface soil samples. Chromium and mercury were detected at concentrations only slightly above background, are not present in a consistent pattern, and are, therefore, not considered to be due to releases that may have occurred from site-related activities. BTEX compounds were detected in five of the nine subsurface soil samples, although at concentrations near their detection limits.

- Benzene is a potential contaminant migration COPC in subsurface soil because it may leach to groundwater, resulting in groundwater concentrations exceeding its MCL. However, benzene was not detected in groundwater and was detected in only one soil sample of the nine collected. Off-site migration of benzene would be very limited due to retardation and biodegradation and the slow movement of groundwater. Even if it were to reach groundwater at its predicted maximum concentration, it would degrade to a concentration less than MCL in less than 4.0 years, and would have traveled less than 30 feet from the Burn Pit E site. Therefore, migration of benzene is considered unlikely, and there are no contaminant migration COPCs.
- In groundwater, cadmium was the only RCRA metal detected above background. Cadmium was found in two of the three downgradient wells and is considered a potential SRC. BTEX compounds (ethylbenzene and xylenes) and PCE were also reported, although at concentrations only slightly above their detection limits. No VOCs were reported in Phase I groundwater sampling.
- Cadmium is a potential human health COPC at Burn Pit E, where the maximum concentration of cadmium in groundwater exceeded its risk-based screening value for residential use of groundwater. The maximum concentration of cadmium was less than its MCL of 5 µg/L at each site. In addition, use of the surficial groundwater at the sites for drinking water is unlikely. Therefore, cadmium is not considered a potential threat to human health at Burn Pit E.
- Cadmium is an ecological COPC in groundwater because it exceeds its ESV for surface water. However, there are no surface water bodies in the vicinity of Burn Pit E, and off-site migration would be limited due to retardation and the slow movement of groundwater. Therefore, exposure of ecological receptors to cadmium in surface water bodies downgradient from Burn Pit E is not a complete pathway. Therefore, cadmium is eliminated as an ecological COPC for Burn Pit E, and further investigation and/or evaluation of this constituent is not required.

The following are the recommendations for Burn Pit E:

- 1. No further action is recommended for Burn Pit E.
- 2. All potential human health COPCs were eliminated for Burn Pit E (Section 13.4), and a BRA is not recommended for the site.
- 3. Because there are no ecological COPCs identified in the PRE for Burn Pit E (Section 13.5), an ERA is not recommended for the site.

97-160P(DOC)/020899

## 14.0 CHARACTERIZATION OF BURN PIT F (SWMU 4F)

#### 14.1 HISTORY AND DESCRIPTION OF BURN PIT F

Burn Pit F (SWMU 4F) comprises approximately 3 acres and is located on the western side of Fort Stewart Route 51, approximately 3400 feet south from its junction with State Highway 144 (Savannah Highway). Clearing and grubbing debris are present in small mounded areas within the site. The debris consist of stumps, branches, and similar vegetative material. Two dump trucks were observed off-loading branches at the site during a November 16, 1993, reconnaissance of the site. Occasional stands of pine or small individual trees are present. Grass covers some of the debris piles and open areas. The site is surrounded by woods.

Surface elevations vary from approximately 33 to 39 feet amsl, sloping gently southeastward (Figure 14-1). Drainage occurs through overland flow to the southeast, collecting in swampy areas that comprise Big Swamp and Goshen Swamp. The closest surface stream to the site is an unnamed tributary of Goshen Swamp, which lies approximately 2000 feet west of the site. Swampy areas located within the floodplain of Goshen Swamp lie approximately 1000 feet south and west of the site.

During the Phase I RFI activities, four monitoring wells were installed at the site (Figure 14-1). Methylene chloride was detected in soil samples, and three RCRA metals, including barium, chromium, and mercury, were detected above site-specific background concentrations in soils. In groundwater, carbon disulfide and methylene chloride were detected, and BTEX compounds were present at levels exceeding MCLs. RCRA metals were also detected above site-specific background levels in groundwater and included chromium and lead.

During the Phase II RFI activities, three soil borings (SB-1 through SB-3) were drilled at the site (Figure 14-1). These borings, together with the four Phase I wells, were used to determine physical and chemical characteristics of the site.

Soils encountered in the soil borings and well boreholes consisted predominantly of slightly silty sands. Geologic cross-sections of the Burn Pit F site are presented in Figures 14-2 and 14-3. Geotechnical parameters were measured in a Shelby tube sample taken from SB-2 at a depth of 19.5 to 21.5 feet. The results of the Geotechnical testing, listed in Table 4-1, indicate that the soils are non-plastic sands with less than 9 percent by weight of fine-grained material. Permeability, as measured in the Shelby tube sample from SB-2, is  $5.81 \times 10^4$  cm/second, typical of a fine sand.

Water levels in the monitoring wells were measured during well development on July 14, 1997, and again during well sampling between July 30 and August 11, 1997. The water table is present at a depth 7 to 13 feet bgs. A water table contour map is provided as Figure 14-4. Groundwater contours measured during the Phase I RFI activities in August 1993 indicate water levels were about 4 feet lower in the August 1993 data. The groundwater flow direction was interpreted in the Phase I RFI to be strongly influenced by a potentiometric trough oriented toward the southeast. The current interpretation is based on water levels measured at the time of well development (instead of at the time of sampling) because the wells were not all sampled at the same time. The current interpretation suggests that groundwater may be slightly mounded beneath the site, highest at MW-4 and lowest at MW-1 and MW-3. This interpretation assumes that the water table is a subdued reflection of the overlying topography. Groundwater



Figure 14-1. Locations of S

, A

|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 120                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 180                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 240                                                                                                                                                                                                                                | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|      | a na panana ang kanang kan<br>Kanang kanang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | n de la carte caracter a caracter a la caracter de                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | ┉┿╍┥┉┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|      | and the state of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | a na tanàna amin'ny fisia amin'ny tanàna mandritra dia mandritra dia mandritra dia mandritra dia mandritra dia<br>Ny fisiana dia mandritra dia                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ana kumu dan selama ku mangan selama ku ma<br>Mangan ku mangan selama ku mangan ku mangan selama ku mangan selama ku mangan selama ku mangan selama ku mangan                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | t and and a star and a star for the star for the star of the st                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and an international states of the second state |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | in the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|      | The second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and a part course of a produce produced and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 4 m p 44, m p 44, m p 44, m p 4 m p 44, m p 44 | and and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 45   | - Marine Street                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an dia mandritra dia mandri<br>Mandridra dia mandritra dia                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and the second s |                                                                                                                                                                                                                                    | MM 2004 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n alama kanala na ana na ana na ana na ana na ana na                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | net a second and a second and a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a na ana ana ana ana ana ana ana ana an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | A                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | an al analysis and a surface share a first star and a surface of the surface of t | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and a second second of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | al and a second seco                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 10   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a final strategy and the second strategy and the s | امۇرىي بىرىيە br>بىرىيەر بىرىيە بىرىيەت بىرىيە بىرىيە بىرىيە بىرىيەت بىرىيە بىرىيەت بىرىيە بىرىيە بىرىيەت بىرىيەت بىرىيەت بىرىيەت                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | n de la companya de<br>La companya de la comp                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                    | ┉╢╌┟╌┟╴┇                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 40   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - <b>P</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | and and a second se                                                                                                                     | and a second particular second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|      | and a second a second and the second second and the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | مؤسية بالمستحم سأسافه                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 88<br>03                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | . Name and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and a start of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | na an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|      | SANDY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | i for her in the former in the second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | SILTY SAN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SB-02                                                                                                                                                                                                                              | and a standard and a standard and a standard                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|      | - Contraction of the second se | n ganta ang ang ang ang ang ang ang ang ang an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | W/ GLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 35   | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ***                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | X III                                                                                                                                                                                                                              | <u>ry</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 1 1 1 1 1 1 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ╈┲╌┝╌┝╌┝╌┝╌┼╸┼╸┥╸╡                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | SAN                                                                                                                                                                                                                                | <b>&gt;</b> +++                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Salar da se andre a far da se da s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ╺╾┥╾┥╾┥╼┥╼┥╌┥╴┥╌┥╴                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | +++++++++++++++++++++++++++++++++++++++                                                                                                                                                                                            | 411                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|      | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | en en felden fre einer bleren men er en feren forse med en einer en feren i se einer en einer en einer en einer                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ╧╧╧╧                                                                                                                                                                                                                               | ℽ┼┼┼                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
|      | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ┼╼┼╍┝╍┢╍┟╍┽╍┽╍┼╍╎╍┶╌╷                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                               | Y                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | CLAY                                                                                                                                                                                                                               | +++++++++++++++++++++++++++++++++++++++                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|      | and a second a second and a second a s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and group and personal surgery and an easy                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and the second sec                                                                                                                                                                                                                                             | in the first and a first second s                                                                                                                     | 1111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 30   | SAN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | W L. T. L.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ╺╍┝╾┾╼┿╾╽╼┿┥╽┉┷╍╽╺┵┿                                                                                                                                                                                                               | +                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|      | SIL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ry                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SILT                                                                                                                                                                                                                               | <u>r</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|      | CLA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | Contensis and a subserve and a subserve and a subserve a subserve a subserve a subserve a subserve a subserve a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a single state of a single state and a second state of the single state of the sin |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SAND.                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | 4-4-4-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|      | - mode and and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | monterior and the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and the second s | har fan an weren ar geregen fan fan geregen.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                    | 124                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
| 25   | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | าา จุ้งหาสำองสุของ การสู่เหาสู้แมงสิ่งสะสุของ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1114 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 1144 - 11 | an ar fart af new consider an and a serie and an and a series of the series of the series of the series of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | SAN                                                                                                                                                                                                                                | formation and the second secon |
|      | CLAYE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Y                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    | V                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a alian fan fan fan fan fan fan fan fan fan f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and a second and a second a second a second a second as a given of a second a second a second as a second a sec                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | er se anna an strange, car a costa an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ┿┿┥ <u>┶</u> ╠╌╌┾╴                                                                                                                                                                                                                 | forderstander                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|      | and a star for the second and and and a star for the second second second second second second second second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a provide a survey and a second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and have a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | nije v na konstruktion provi pose stanovni sa sa stanovni sa                                                                                                                                   | and and and and and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |
|      | a martine and a first a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | e vel versensjonen en staar beverse versensjonen.<br>In 2-maar en sember aanderer beverse staarderer beverse staar                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an faanaan ah                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | CLAY                                                                                                                                                                                                                               | ĘY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |
| 20   | and a second and a second a second a second and a second as a s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | monthe and an and a production of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | n version and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ere we beauty hits ere af eren andere er sammer i gestalligen                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | SANU                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a de la parte de la constant d'antifica persona                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - Constant of the second s                                                                                                                     | (and a second se |
|      | a na pana kang kana pana pana kana ng ana pang ng ang ang ang ang ang ang ang ang a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | er a franses i nem franses en a franses franses en andere same                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                    | <b> </b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|      | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a la construction de la construction<br>La construction de la construction d                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | nen de mensen en englennet, werden de englement en<br>mensen en en englement en en enderen terre                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Constant from the second s                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | o provinsi na segunda n                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a second and a second and a second and a second as                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | and the finance of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                    | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|      | e realized to describe a strange and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | na na maria na maria na matanga                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and an internal produced and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a set of the second s                                                                                                                                                                                                                                             | lan periodi an ang sa periodi ang s<br>Periodi ang sa periodi |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Alfonder at the end of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Contraction of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | en a ser a se and second and a second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | have a star and share a stranger and                                                                                                                                                                                               | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
| 15   | and the second and the second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a state of the | n ang series br>In ang series                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | a san di silana in si ana ana angan di sinanan angan si si                                                                                                                                                                         | and the base of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|      | an and an approximately an and a second and a second as a second second as the second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | ميسين ويتبع ستقمين والمستهد الوصم المشير والمحصورة فالمحا                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | a server serv                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a construction of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and the state of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ter in a second and in a second and in the                                                                                                                                                                                         | fer viewe i verseere<br>provinseere versigen                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|      | a de angenera agan panagan na angenera angenera angenera angenera angenera angenera angenera angenera angenera                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | n san an har la sha an an an an ar a ba                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | the property line of a line of party party party of the state of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | server and an analysis of a surface of the                                                                                                                                                                                         | iann an an Anna                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|      | entition was made after a single was a single was a way a manufacture of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | an internal and and and and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | n de la compañía de<br>La compañía de la comp                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | مىرى بىرەتچىيە چى ئېرىدە ، مەمامەمىر مىرە .<br>ئەمەر بىچە بېيرى بەرمەر - خىرا - مەمەر مىرە                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                    | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|      | a na se a companya na sense a se a se a se a se a se a se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | a barrela a constructiva a constructiva.<br>La constructiva a con                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | and the second provides a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an gran gran a star gran gran a star st                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                    | يه ينه م                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
| 10   | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | and a star of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a series and a series of the s | nenden finde en finde en de saderen en sig                                                                                                                                                                                         | ha ng suliva yan.<br>Hani ng kata a ta                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
| 10 1 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |

Figure 14-2. Geologic Cro

 $\langle \rangle$ 

ĺ

(

)

|                | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 0 120                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 180                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 240                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 300                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|----------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                | and a second and a second an analysis and an angle of the second and a second and a second and a second and a s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | สารสารสารสารสารสารสารสารสารสารสารสารสารส                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                | and for the second containing a shared and so                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | and and and a special company and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | - produce and send and send and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | (                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | a war an                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | and an advantage of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | andread and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | maging and a find and and and and and and and and and a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and an experiment of the providence of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | and a super second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - North Start and Strategy and Strategy and Strategy                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 45             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Sun and allow from the second se                                                                                                                                                                                                                                                   | an ferret an ferret ferret and market and ferret                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | al new first first and a second and first     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 45             | en men er sjærne sjærne for stære for med an sig som fra støre for som for som støre som som som som som som s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | profession of a family and a second and a longer                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and a star for the star of the | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | negera konsela milera ofice e instant oficer of market as well                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Sanda after free and in free for after a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and in the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | pro- hanife a sport and wife reduce have a specific gas                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | - for any of the first start and see and see                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 100 (218 gr 11-fran 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 1. 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                | B                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | for for for the stand of the st       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a fundamental of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | มากระบบครามสาราชาวารสาราชาวารสาราชาวาร<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | and a specific set                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | - manghar angkana kangkana funca far angkana panang nangkana diana di                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and a second and a second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | an fair an an the section of the sec | francista en francista ante ante ante ante ante forma de series de la companya de la companya de la companya d<br>La companya de la comp                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a state of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | žerski začeni kon<br>do nakonski se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 40             | ر این به مید با می است. می می می می می می می است از می است.<br>مربق است است است از می می می می می می می است است است است است.<br>مربق است                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | - andre takan de andre de andre andre de andre<br>In andre de a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ann ferner ferrer a search ann fer a star an ferrer a fair a star an ferrer ann a star an ferrer ann a star an<br>a star ferrer ferrer a star an ferrer a ferrer a ferrer a star a star a star a star a star a star a ferrer ferr                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 88-02<br>88-02                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | for the star and the former of the star and     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | +++++                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                | a ser and a second a second and a second as a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                | <b>N</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and a state of the       | North and a start of the start  | interior and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - frankrisk                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                | man from from from the information of any and a state of the state of  | rangere a fair takanang sanganang na ajan takan <u>ana s</u> ang sa sang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 144                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 35             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | and a second sec    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 11111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | - Constant and a state of the second se                                                                                                                                                                                                                                                   | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | CLAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SANDY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | under de la colorista esta esta esta esta esta esta esta e                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | and want was seen a provide a provide a set of the set  | 111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | with a specific strain the specific spe       | CLAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ┽┼┽┽┾┾                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 44                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                | a state of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | **************************************                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 111                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | And and and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | $\frac{1}{2}$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
| 30             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | and the second s | - frankreder<br>Andre frankreder                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SILTY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <u>this has a feature of the second sec</u> | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ****                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                | +¢LAYEY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | uterspecture<br>starspecture                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |
|                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | n alasta kana sa                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ╈╋╗                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | -la fa-la fa-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ╷┼╌┥╌┥╌┥╌┥╌┥╌┥╴┥                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | a state and the state of the st |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 25             | ten pro la state de la seconda d                                                                                                                                                                                                                                             | and a second sec       | SANDY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | Bankar and an and a start                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Action Incomentation                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a survey of a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | en in fan Y Lifer Arlen fan yn de geregen yn geregen yn geregen yn geregen yn geregen yn geregen yn geregen yn<br>Lifer yn geregen yn ger                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | . 9 m - 19 m - 1    | for the second s | haifa ha ha ha                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | HHHM                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | faither the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | 1 - 1979 19 - 19 - 19 - 19 - 19 - 19 - 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | - And        | CLAYEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | and the second s    | for the standard and a straight of the standard and the s |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and and the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and an interest and an interest of an interest of the second second second second second second second second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SAND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| 20             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | aleman br>In analogo aleman alem                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | بيدون م من ما ومخطعة أ<br>مساوحه الد مود فا وحجا و                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
| _20            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | we have a set on the set of the s | and the standard from from from from the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | n a ginn fan Galerstan (gerender oan gerender oan gerender oan gerender oan gerender oan gerender oan gerender                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ากกระบบการสารกระบบสู่สารกระบบสายสารกระบบสายการสารกระบบสายการสารกระบบสายการสารกระบบสายการสารกระบบสายการสารกระบบ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | hand and a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | a generalization de la comme de la construction de la construction de la construction de la construction de la<br>La construction de la construction d                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ne providente de la resta de la construcción de                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | anna tar igu un aslandan isan isan isang unitar ing br>unitar ing unitar ing uni | y an only the first activity of a standard and a second standard activity of the second standard activity of the                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | and a second and a second a s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| -              | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | the standard of the standard o       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | - man man non non and a second and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | nie nam Martinie jest Wingeleicht albeide auf zu eine Angeleichte an mat eine artebe auf die Jermage                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | an a fan ste fan in de ste ander en al ste fan de ste ste ste ste s<br>Na ste                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | and and a second se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | and a far far far far far far far far far f                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | andre franken<br>Sendenden                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |
|                | man in the second water and a second with a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | pargundangen fan fan de niger fan                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | no per estar de la companya de la co                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | provide and the second second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | م يشم أسبح أسبخ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 1.5            | nande strander generalise former is former and and and a second second second second second second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | for have a start of the start o       | and the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | Andrew Street and stre    | en and former (an angles a single and a single and a single a single a single and a single a single a single a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and the second state of th | a may a sumption and a market of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | a - june for strange internet and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | an or only an effect of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | a for the map of the set of the first of the set of the |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | In the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ana ngana nga maliputan menghani ng mangka serifatan faran s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | i-and and indus                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and a standard the second and a standard the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | to any new own were water the same of a large station                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | and a stranger of a stranger of a stranger of and a stranger of a stranger                                                                                                                                                                                                                                             | for the first state of the product of the state of the st    | manger og vig vi græfinge for og er og er færdere<br>- Toger og er seger af er seger seger og er og er                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | na in an an anna agus ina na an a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | - martin the property of the second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | the star from the production of                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                | we have an end of the second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | manifestine of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | and in the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ng manandar a siyang mananan sa ma                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | an managan a sa ang pangangan ang mgangan pangangan pangan pangangan pangangan pangangan pangangan pangangan p<br>Pangangang sa pangangangan pangangan pangangangangang pangangang pangangang pangang pangang pangang pangang pang                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | na je na jene<br>na se na se na se na se                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |
| 10             | المراقع الميس مرد مريون من الأرب المعالم من المار.<br>الإمراقي مع مسالية الإعماق مراقعهم هريون المحافظ ا                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | and the second sec                                                                                                                                                                                                                                                   | e og e la man andere ander<br>In the spectra and an andere and and andere and andere and andere and                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | الي المراجعة علي المراجع br>المراجع المراجع                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | and the second sec                                                                                                                                                                                                                                             | and the second sec |
| • <del>-</del> |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | the state of the s | a and a ser and a product of the second s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | and and a second second second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | and a second                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |

Γ

(

 $\langle \rangle$ 

Figure 14-3. Geologic Cro



Figure 14-4. Water Tabl

flows from a high in the north and west portion of the site to the south, southeast, and east before joining the regional shallow groundwater flow. Regional groundwater flow is expected to be to the south or southwest, discharging ultimately to low-lying swampy areas within the Big Swamp and Goshen Swamp, approximately 1000 feet south and west of the Burn Pit F site. The calculated horizontal hydraulic gradient in the water table across the site is 0.0023 foot/foot. Based on a hydraulic conductivity of  $5.81 \times 10^{-4}$  cm/second and a porosity of 0.408, the groundwater flow rate is approximately 3.4 feet/year toward the south, southeast, and east.

260

Following receipt of GEPD comments dated October 28, 1999, on the Phase II RFI for Burn Pit F, Fort Stewart conducted a supplemental investigation of groundwater. The scope of that supplemental investigation included the installation of temporary piezometers to verify groundwater flow directions, sampling of water from piezometers to verify whether a source of BTEX contamination is present, and sampling and analysis of the four existing wells to verify that natural attenuation of BTEX is occurring. Results of the supplemental investigation are given in Appendix H.

#### 14.2 NATURE AND EXTENT OF CONTAMINATION AT BURN PIT F

#### 14.2.1 Surface Soil Contamination

The nature and extent of surface soil contamination was evaluated using the results of analyses on surface soil samples taken from the three Phase II soil borings (SB-1 through SB-3) at the site. These samples were analyzed for SVOCs and RCRA metals. Table 14-1 summarizes the analytical results for surface soil samples, and Figure 14-5 shows their distribution. This assessment presents Phase II contaminant data only because no surface soil samples were collected during Phase I.

|          |           |                            | Phase II Samples           |                            |
|----------|-----------|----------------------------|----------------------------|----------------------------|
|          | Reference | SB-1<br>4F1111<br>07/13/97 | SB-2<br>4F1211<br>07/14/97 | SB-3<br>4F1311<br>07/14/97 |
|          |           | RCRA Metals (mg/kg)        |                            |                            |
| Arsenic  | 2.10      | 2.7                        | 0.88                       | 0.91                       |
| Barium   | 14.7      | 13.6                       | 10.7                       | 11.1                       |
| Chromium | 6.21      | 13.5                       | 8.1                        | 6.3                        |
| Lead     | 8.81      | 6.9                        | 7.3                        | 5.7                        |
| Mercury  | 0.03      | 0.03                       | 0.03                       | 0.04                       |

#### Table 14-1. Summary of Analytical Results for Surface Soils Burn Pit F, Fort Stewart, Georgia

Note: **Bold** font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.

ðr.



Figure 14-5. Results of Analy

.

SVOCs. No SVOCs were detected in surface soil samples at the site.

**RCRA metals.** Maximum concentrations of arsenic (2.7 mg/kg) and chromium (13.5 mg/kg) were reported at SB-1 at concentrations exceeding background. Chromium also exceeded reference background concentrations at the other two boring locations (SB-2 and SB-3). Mercury was reported only at SB-3 at a concentration (0.04 mg/kg), which is only slightly greater than background, and is, therefore, not likely to be site related.

#### 14.2.2 Subsurface Soil Contamination

The subsurface soil samples were collected from the four Phase I monitoring well boreholes and the three Phase II soil borings. The samples were analyzed for VOCs and RCRA metals. Table 14-2 summarizes analytical results for the subsurface soil samples, and Figure 14-6 shows their distribution. Both Phase I and Phase II data are shown.

VOCs. Toluene was detected in two subsurface soil samples at SB-1 and MW-4, with the maximum concentration (5.5  $\mu$ g/kg) reported at MW-4. These concentrations are lower than the average detection limit for toluene and may, therefore, not be indicative of site-related contamination. Methylene chloride was detected in MW-1 and MW-3 at concentrations up to 7.4  $\mu$ g/kg.

**RCRA metals.** Five metals were reported in subsurface soils at concentrations above their reference background value. Barium was detected at concentrations above background in three of the seven subsurface soil samples, with a maximum value (125 mg/kg) at SB-3. Chromium was also found in three of the seven samples, with a maximum concentration of (65.8 mg/kg) also at SB-3. Other metals detected above background include lead (26.9 mg/kg at SB-3), mercury (0.051 mg/kg at MW-4), and silver (2.6 mg/kg at SB-1).

Table 14-2. Summary of Analytical Results for Subsurface Soils Burn Pit F, Fort Stewart

ſ

(

(

|                      |            |            | Phase I      | <b>Phase I Samples</b>                      |              | H              | Phase II Samples             | 50            |
|----------------------|------------|------------|--------------|---------------------------------------------|--------------|----------------|------------------------------|---------------|
|                      |            | IWM        | MW2          | EWM                                         | MW4          | SB-1           | SB-2                         | SB-3          |
|                      |            | SL1-8      | SL2-8        | SL3-8                                       | <b>SL4-4</b> | 4F1112         | 4F1212                       | 4F1312        |
|                      | Reference  | 06/29/93   | 06/29/93     | 06/30/93                                    | 06/30/93     | 07/13/97       | 07/14/97                     | 07/14/97      |
| Parameter            | Background | (6-8 feet) | (6-8 feet)   | (6-8 feet) (6-8 feet) (6-8 feet) (2-4 feet) | (2-4 feet)   | (10-12.5 feet) | (12-14.5 feet) (9.5-12 feet) | (9.5-12 feet) |
|                      |            | Vola       | tile Organic | Volatile Organic Compounds (ug/kg)          | i (ug/kg)    |                |                              |               |
| Methylene Chloride   | 0.0        | 9          |              | 7.4                                         |              |                |                              |               |
| Toluene              | 0.0        |            |              |                                             | 5.5          | 2.9            |                              |               |
|                      |            |            | RCRA M       | RCRA Metals (mg/kg)                         | -            |                |                              |               |
| Arsenic              | 8.04       |            |              |                                             |              | 6.0            |                              | 3.3           |
| Barium               | 17.0       | 6.2        | 17           | 17                                          | 18           | 16             | 110                          | 125           |
| Chromium             | 11.6       | 5.1        | 4.6          | 15                                          | 8.8          | 4.3            | 13                           | 65.8          |
| Lead                 | 11.1       | 16         | 8.3          | 7.7                                         | 4.8          | e              | 6.4                          | 26.9          |
| Mercury              | 0.05       |            | 0.023        |                                             | 0.051        | 0.02           |                              | 0.04          |
| Silver               | 0.46       |            |              |                                             |              | 2.6            |                              |               |
|                      |            |            | Other Inory  | Other Inorganics (mg/kg)                    | <u>(</u> 2)  |                |                              |               |
| Total Organic Carbon | 2,200      |            |              |                                             |              |                | 279                          |               |
|                      |            |            |              |                                             |              |                |                              |               |

Note: A blank indicates analyte not detected. Note: **Bold** font indicates value exceeds reference background criterion. RCRA = Resource Conservation and Recovery Act.



Figure 14-6. Results of Analy

e z

()
#### 14.2.3 Groundwater Contamination

Groundwater contamination was evaluated using the results from water samples taken from the four monitoring wells installed during Phase I (MW-1 through MW-4). The groundwater amples were analyzed for VOCs and RCRA metals (total metal analysis on unfiltered water samples). Table 14-3 summarizes the analytical results for groundwater samples and Figure 14-7 shows their distribution. This assessment focuses on the Phase II contaminant data because the Phase I data are considered suspect due to high metals content as a result of turbidity. Phase I VOCs data are discussed qualitatively with respect to trends observed.

|                |                         |              | P                          | hase II Sampl                         | es                         |                            |
|----------------|-------------------------|--------------|----------------------------|---------------------------------------|----------------------------|----------------------------|
| Parameter      | Reference<br>Background | MCL          | MW-1<br>07/30/97<br>4F4111 | MW-2<br>07/30/97<br>4F4211            | MW-3<br>08/11/97<br>4F4311 | MW-4<br>07/30/97<br>4F4411 |
|                |                         | Volatile Org | anic Compound              | s (μg/L)                              |                            |                            |
| Ethylbenzene   | 0.0                     | 700          |                            |                                       |                            | 637                        |
| Styrene        | 0.0                     | 100          |                            |                                       |                            | 33.5                       |
| Xylenes, Total | 0.0                     | 10,000       |                            |                                       |                            | 3170                       |
|                |                         | RCR          | A Metals (µg/L)            |                                       | Ann                        |                            |
| Barium         | 71.72                   | 2,000        | 22.8                       | 49.1                                  | 26.3                       | 24.6                       |
| Cadmium        | 0.43                    | 5            | 1.1                        | · · · · · · · · · · · · · · · · · · · | 0.65                       |                            |
| Chromium       | 3.56                    | 100          | 1                          | 0.7                                   | 2.8                        | 1.6                        |
| Lead           | 4.69                    | 15           | 0.08                       |                                       |                            |                            |
| Silver         | 1.12                    |              | 1                          |                                       |                            | 0.1                        |

# Table 14-3. Summary of Analytical Results for Groundwater Burn Pit F, Fort Stewart

Note: A blank indicates analyte not detected.

Note: Bold font indicates value exceeds reference background criterion.

MCL = Maximum Contaminant Level.

RCRA = Resource Conservation and Recovery Act.

VOCs. Two BTEX compounds (ethylbenzene at 637  $\mu$ g/L and total xylenes at 3170  $\mu$ g/L) were found in groundwater at MW-4. These concentrations are less than the respective MCLs for ethylbenzene (700  $\mu$ g/L) and total xylenes (10,000  $\mu$ g/L). In the Phase I sample results, both ethylbenzene and total xylenes were reported at much higher concentrations (2,800 and 15,800  $\mu$ g/L, respectively), and benzene (22,700  $\mu$ g/L) and toluene (41,100  $\mu$ g/L) were also reported above their MCLs. These results suggest that the high concentrations of BTEX present in groundwater at the site may be attenuating naturally, either through dispersion, volatilization, or biodegradation.

Styrene was also observed in MW-4 at 33.5  $\mu$ g/L and is considered a secondary contaminant in the primary BTEX plume. The source and extent of the BTEX contamination is unknown. MW-4 is located hydraulically higher than other wells at the Burn Pit F site and is considered side-gradient to flow from the site. BTEX compounds were not detected at significant concentrations in the soil borings within the probable source area and were not detected in the remaining (downgradient) monitoring wells.



4 m - 141

Figure 14-7. Results of Analys

J.

**RCRA metals.** Cadmium was the only RCRA metal detected above its background concentration in groundwater and was detected at such concentrations in MW-1 (1.1  $\mu$ g/L) and MW-3 (0.65  $\mu$ g/L). MW-1 is considered upgradient of the site.

2/2

#### 14.2.4 Summary of Site-related Contaminants

A summary of the SRCs by medium and their maximum concentrations is presented in Table 14-4. SRCs include all organics that are detected, and inorganics detected above reference background criteria. These SRCs are carried forward for evaluation under fate and transport, human health PRE, and ecological PRE.

| Analyte            | Surface Soil | Subsurface Soil | Groundwater |
|--------------------|--------------|-----------------|-------------|
|                    | (µg/kg)      | (µg/kg)         | (µg/L)      |
| Ethylbenzene       | na           | nd              | 637         |
| Methylene Chloride | na           | 7.4             | nd          |
| Styrene            | na           | nd              | 33.5        |
| Toluene            | na           | 5.5             | nđ          |
| Xylenes, total     | na           | nd              | 3170        |
|                    | (mg/kg)      | (mg/kg)         | (µg/L)      |
| Arsenic            | 2.7          | brc             | nd          |
| Barium             | brc          | 125             | brc         |
| Cadmium            | nd           | nd              | 1.1         |
| Chromium           | 13.5         | 65.8            | brc         |
| Lead               | brc          | 26.9            | brc         |
| Mercury            | 0.04         | 0.051           | nd          |
| Selenium           | nd           | nd              | nd          |
| Silver             | nd           | 2.6             | brc         |

# Table 14-4. Summary of Site-Related Constituents Burn Pit F, Fort Stewart

brc = Below background reference criteria.

na = Not analyzed in any sample in that medium.

nd = Not detected in any sample in that medium.

BOLD font indicates the analyte is an SRC in that medium.

#### 14.3 FATE AND TRANSPORT CONSIDERATIONS AT BURN PIT F

The results of contaminant migration soil screening (discussed in Section 6.0) for Burn Pit F are presented in Table 14-5. Chromium and methylene chloride are identified as CMCOPCs based on leaching to groundwater from this site. Neither of these two constituents are currently observed in groundwater. However, there is a potential for chromium to be leaching from the soil and migrate to the water table at concentration exceeding its MCL. Methylene chloride, on the other hand, may be degraded before reaching the water table, or it may be a laboratory artifact.

The predicted maximum concentration of methylene chloride in groundwater, based on a maximum of 7.4  $\mu$ g/kg in soil and a DAF of 1, is 37  $\mu$ g/L. Through biodegradation, even if the methylene chloride were to reach groundwater at such a concentration, it would degrade to its

MCL value (5  $\mu$ g/L) in less than 2.9 times its half-life. Assuming a conservative half-life of 0.3 years, the methylene chloride would degrade to its MCL value in 0.9 years. The methylene chloride was reported in Phase I soil samples only and was not detected in Phase II samples.

| SRCs               | Maximum<br>Concentration | GSSL            | Is Maximum<br>Concentration > GSSL?    |
|--------------------|--------------------------|-----------------|----------------------------------------|
|                    | RCRA Meta                | l (mg/kg)       |                                        |
| Arsenic            | 3.30                     | 29              | No                                     |
| Barium             | 125.00                   | 1600            | No                                     |
| Chromium           | 65.80                    | 38              | Yes                                    |
| Lead               | 26.90                    | 400             | No                                     |
| Mercury            | 0.05                     | 0.4             | No                                     |
| Sliver             | 2.6                      | 34              | No                                     |
|                    | Volatile Organic C       | ompound (µg/kg) | ······································ |
| Methylene Chloride | 7.40                     | 1               | Yes                                    |
| Toluene            | 5.50                     | 600             | No                                     |

| Table 14-5. | . CMCOPCs Based on Soil Screening | for Burn Pit F |
|-------------|-----------------------------------|----------------|
|-------------|-----------------------------------|----------------|

CMCOPCs = Contaminant Migration Constituents of Potential Concern.

GSSL = Generic Soil Screening Level.

RCRA = Resource Conservation and Recovery Act.

The most likely pathway of contaminant migration from Burn Pit A is via groundwater flow toward Goshen Swamp, located approximately 1000 feet south and west of the site. Based on the calculated groundwater flow velocity of 3.4 feet/year, the estimated arrival time for the site groundwater to reach Goshen Swamp is expected to be more than 294 years. Methylene chloride would degrade to less than its MCL within a few feet of the Burn Pit F site, well before reaching Goshen Swamp.

Although cadmium, ethylbenzene, styrene, and total xylenes are not identified as CMCOPCs based on leaching to groundwater from Burn Pit F, they are currently observed in the site groundwater. However, the maximum observed concentrations for these constituents are below their respective MCLs and, based on fate and transport analysis, they are not expected to increase in the future. Therefore, it can be concluded that chromium is the only constituent that is expected to be of potential concern for contaminant migration from Burn Pit F.

# 14.4 HUMAN HEALTH RISK EVALUATION OF BURN PIT F

SRCs were identified in Section 14.2 for surface soil, subsurface soil, and groundwater. The identification of risk-based action levels for all of the Burn Pits has been previously discussed in

Section 7.3 and will not be discussed in detail here. The site-specific risk evaluation for Burn Pit F is given below.

## 14.4.1 Exposure Evaluation

This site is completely covered by vegetation, and no VOCs were detected in surface soils. Therefore, exposure via inhalation based on current land use is not a viable exposure pathway. The site is not currently used by the Army. Therefore, current on-site receptors include a juvenile trespasser and a hunter. Current off-site receptors would be represented by an off-site hunter.

Land use at this site is unlikely to change in the future. However, the Army may use the site as a training area in the future. Construction may occur on the site or the vegetative cover may be gone as a result of on-site activities. Future on-site receptors include military personnel and a construction worker. Future off-site receptors include hunters in the area.

A summary of the potential exposure pathways for each of the receptor populations is given below.

**Current juvenile trespasser.** The juvenile trespasser may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for the juvenile trespasser. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Hunter. The hunter is representative of a current on-site receptor and a current and future off-site receptor. The current on-site receptor is not likely to be exposed to chemicals given the absence of an air migration pathway, and clothing the hunter would be wearing would eliminate potential dermal exposure. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

The current off-site hunter may be exposed via ingestion of contaminants bioaccumulated in game animals. The future off-site hunter may be exposed via ingestion of game and inhalation of fugitive dust if the vegetative cover were to be removed.

Future on-site military personnel. The on-site military personnel may be exposed to surface soil. Incidental ingestion and dermal absorption are complete exposure pathways for this receptor. Volatile organics were only detected in subsurface soils. Significant exposure via volatilization is not likely.

Future construction worker. The construction worker may be exposed to surface and subsurface soils. Complete exposure pathways include incidental ingestion, inhalation of volatiles and fugitive dust, and dermal absorption.

#### 14.4.2 Risk Evaluation

The purpose of the risk evaluation is to determine what contaminants present a potential threat to human health. These chemicals will be evaluated further in a BRA.

Arsenic in surface soil is the only contaminant that exceeds its risk-based screening concentration for direct exposure of a receptor population (Table 14-6). Arsenic in surface soil is a potential COPC at this site because it exceeds the risk-based screening value for exposure of a residential receptor. It should be noted that the maximum concentration of arsenic (2.7 mg/kg) is below the average concentration for soils in the eastern United States and only slightly exceeds reference background criteria (2.1 mg/kg) at one location, SB-1. Arsenic was not found above background in either subsurface soil or groundwater. In addition, a residential screening value was used although residential land use at this site is unlikely. Therefore, arsenic is not considered a potential threat to human health at Burn Pit F.

Table 14-6. Comparison to Action Levels-Burn Pit F, Fort Stewart

,

|                    |           |                         |                   |                            |                        | EPA III               |       |                                    |
|--------------------|-----------|-------------------------|-------------------|----------------------------|------------------------|-----------------------|-------|------------------------------------|
| Amolista           | Tinite    | Frequency<br>of Detects | Minimum<br>Detect | Maximum<br>Detect          | Background<br>Criteria | Screening<br>Criteria | COPC? | Justification                      |
|                    |           |                         |                   | Rurn Pit F. Surface Soi    | face Soil              |                       |       |                                    |
|                    | 2         | 2 r                     | 000               | 170                        | 110                    | 0.43                  | Yes   | Max Detect >                       |
| Arsenic            | mg/Kg     | c/c                     | 0.00              |                            |                        |                       |       | Screening Criteria                 |
| Chromium           | mg/kg     | 3/3                     | 6.3               | 13.5                       | 6.21                   | 39                    | No    | Max Detect <                       |
|                    |           |                         |                   |                            |                        |                       |       |                                    |
| Mercury            | mg/kg     | 3/3                     | 0.03              | 0.04                       | 0.03                   | 2.3                   | No    | Max Detect <<br>Screening Criteria |
|                    |           |                         | B                 | Burn Pit F Subsurface Soil | urface Soil            |                       |       |                                    |
| Barium             | mg/kg     | 9/9                     | 16                | 125                        | 17.0                   | 14,000                | No    | Max Detect < Screening Criteria    |
|                    | - And And | 6/6                     | 43                | 65.8                       | 11.6                   | 1,000                 | No    | Max Detect <                       |
| Chromann           | SA YIII   | 5                       | ]                 |                            |                        |                       |       | Screening Criteria                 |
| Lead               | mg/kg     | 9/9                     | 3                 | 26.9                       | 11.1                   | 400                   | No    | Max Detect <<br>Screening Criteria |
|                    |           |                         |                   | 190.0                      | 0.05                   | 61.0                  | No    | Max Defect <                       |
| Mercury            | mg/kg     | 4/0                     | 70.0              | 100.0                      | 000                    |                       |       | Screening Criteria                 |
| Ciluar             | mo/ko     | 1/6                     | 2.6               | 2.6                        | 0.46                   | 1,000                 | No    | Max Detect <                       |
|                    | a., A.,   | •                       |                   |                            |                        |                       |       | Screening Criteria                 |
| Methylene chloride | mg/kg     | 1/6                     | 0.0074            | 0.0074                     |                        | 16                    | No    | Max Detect <                       |
|                    | )         |                         |                   |                            |                        |                       |       | Screening Uniteria                 |
| Toluene            | mg/kg     | 2/6                     | 0.0029            | 0.0055                     |                        | 41,000                | No    | Max Detect <<br>Screening Criteria |
|                    |           |                         |                   | Burn Pit F Groundwater     | oundwater              |                       |       |                                    |
| Cadmium            | J/au      | 1/3                     | 0.65              | 0.65                       | 0.43                   | 1.8                   | No    | Max Detect <                       |
|                    | 1         |                         |                   |                            |                        | 1200                  | AL.   | May Datast /                       |
| Ethylbenzene       | hg/L      | 1/3                     | 637               | 637                        |                        | 1300                  | ON    | Screening Criteria                 |
| Styrene            | µg/L      | 1/3                     | 33.5              | 33.5                       |                        | 1600                  | No    | Max Detect <                       |
|                    | )<br>-    | _                       |                   |                            |                        |                       |       | Screening Uniteria                 |
| Xylenes, Total     | μg/L      | 1/3                     | 3170              | 3170                       |                        | 12000                 | No    | Max Detect <<br>Screening Criteria |
|                    | _         |                         |                   |                            |                        |                       |       |                                    |

COPC = Chemical/Contaminant of Potential Concern. EPA = U.S. Environmental Protection Agency. All of the contaminants in subsurface soils and groundwater were below their respective risk-based screening concentration for direct exposure of a receptor population (Table 14-6).

In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at Burn Pit F. A human health BRA is not warranted. No further action is required for protection of human health.

### 14.5 ECOLOGICAL RISK EVALUATION OF BURN PIT F

The ecological risk evaluation of Burn Pit A is a PRE conducted according to GEPD (1996) guidance (see Section 8.0). The PRE compares the maximum detected concentrations of analytes directly to conservative screening values for those substances. If no ecological COPCs are identified based on the ESV comparison (Step i), then no further evaluation is required. If ecological COPCs are identified based on the Step i screening, then those ecological COPCs are considered further in PRE Steps ii through v. The results of the five steps of the PRE are reported below.

## 14.5.1 Ecological Screening Value Comparison (Step i)

One RCRA metal and three VOCs were detected in groundwater at Burn Pit F at concentrations exceeding background criteria. The results of the ESV comparison for groundwater at Burn Pit F are presented in Table 14-7. Because the detected concentrations of cadmium in the downgradient wells MW-2 through MW-4 did not exceed the ESV, cadmium is not considered an ecological COPC. The maximum detected concentration of ethylbenzene exceeded its surface water ESV; therefore, the ecological COPC identified by the ESV comparison for groundwater at Burn Pit F is ethylbenzene.

Although ecological COPCs are identified in groundwater at Burn Pit F when it is screened according to GEPD (1996) and EPA Region 4 (1995) guidance, groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at Burn Pit F. There are no surface water bodies in the vicinity of Burn Pit F. Also, groundwater contaminants are not expected to migrate from the site because the movement of groundwater is slow relative to the high adsorption and biodegradation of contaminants in Burn Pit F soils.

Because there are no ESVs for soil, all analytes detected in soil at Burn Pit F are evaluated further in PRE Steps ii through v.

#### 14.5.2 Preliminary Problem Formulation (Step ii)

Burn Pit F comprises approximately 3 acres (Figure 14-1). Clearing and grubbing debris are present in small mounded areas within the site. The debris consist of stumps, branches, and similar vegetative material. Occasional stands of pine or small individual trees are present. Grass covers some of the debris piles and open areas. The site is surrounded by woods. Drainage occurs through overland flow to the southeast, collecting in swampy areas that comprise Big Swamp and Goshen Swamp. The closest surface stream to the site is an unnamed tributary of Goshen Swamp, which lies approximately 1000 feet west of the site.

| Analyte        | ESV                                    | MW-1*        | MW-2           | MW-3      | MW-4   | Maximun |
|----------------|----------------------------------------|--------------|----------------|-----------|--------|---------|
|                | ······································ | RCR          | A Metals (µg/L | )         |        |         |
| Cadmium        | 0.66ª                                  | 1.1 J        |                | 0.65 J    |        | 0.65    |
| t              |                                        | Volatile Org | ganic Compour  | id (μg/L) |        |         |
| Ethylbenzene   | 453                                    |              |                |           | 637    | 637     |
| Styrene        | 10,000 <sup>b</sup>                    |              |                |           | 33.5 J | 33.5 J  |
| Xylenes, Total | 7,100 <sup>e</sup>                     |              |                |           | 3170   | 3170    |

# Table 14-7. EPA Region 4 Ecological Screening Value Comparison for Analytes Detected in Groundwater at Burn Pit F, Fort Stewart

\*Upgradient monitoring well.

J = Estimated concentration.

ESV = U.S. Environmental Protection Agency Region 4 Ecological Screening Values (EPA 1996) and, where indicated, alternative values for analytes without Region 4 ESVs.

RCRA = Resource Conservation and Recovery Act.

"Hardness dependent; assumes 50 mg/L CaCO3.

<sup>b</sup>LC<sub>50</sub> fish (Clayton and Clayton 1981).

<sup>c</sup>LC<sub>0</sub> salmonid (Clayton and Clayton 1981).

Bold font indicates detected concentration exceeds ESV.

The preliminary assessment endpoints, ecological receptors, and surrogate species representative of those receptors selected for evaluation in the preliminary risk calculation for Burn Pit F are described in Section 8.1.2.

#### 14.5.3 Preliminary Effects (Step iii)

In the PRE for Burn Pit F, TRVs are required for shrews and robins ingesting contaminated biota exposed to soils near the site. The derivation of TRVs is discussed in Section 8.1.3. The TRVs derived for shrews and robins for ecological COPCs in soil are presented in Table 8-1.

#### 14.5.4 Preliminary Exposure (Step iv)

Ecological receptors at Burn Pit F are likely exposed by ingestion of contaminated soil or biota exposed to contaminated soil. The exposure parameters for the surrogate species, shrews and robins exposed to COPCs in soil are presented in Table 8-2.

# 14.5.5 Preliminary Risk Calculation (Step v)

The preliminary risk calculation (Step v) uses HQs, the ratio of the measured maximum concentration and the TRV, to evaluate the potential for risk. The HQs of ecological COPCs with consistent modes of toxicity and effects endpoints are added to calculate an HI. Metals are assumed to have distinct modes of toxicity and effects endpoints. Therefore, HIs are only calculated for VOCs and SVOCs when no individual ecological COPC has an HQ greater than 1.0, and HQs are calculated for more than one chemical. Ecological COPCs with HQs and HIs less than 1.0 indicate little to no likelihood of risk to the ecological receptors. An ERA using site-specific data is indicated for those ecological COPCs with calculated HQs or HIs exceeding 1.0 (GEPD 1996).

Surface Soil. The preliminary risk calculation for shrews and robins exposed to ecological COPCs detected in soil at Burn Pit F is presented in Table 14-8. This table shows the maximum detected concentrations and the TRVs for shrews and robins. Concentrations resulting in HQs exceeding 1.0 are shown in boldface font.

The only ecological COPC present in Burn Pits F surface soil at concentrations exceeding the TRV for the surrogate species is chromium. Only the robin is potentially at risk from chromium in surface soil. An HI is not calculated for the surrogate species exposed to RCRA metals detected in surface soil because they are assumed to have dissimilar mechanisms of toxicity.

|          | T     | RV    |                 |        | · · · · · · · · · · · · · · · · · · · | [       |
|----------|-------|-------|-----------------|--------|---------------------------------------|---------|
| Analyte  | Shrew | Robin | SB-1            | SB-2   | SB-3                                  | Maximum |
|          |       | RC    | RA Metals (mg/k | g)     |                                       |         |
| Arsenic  | 37.8  | 651   | 2.7 J           | 0.88 J | 0.91 J                                | 2.7 J   |
| Chromium | 62662 | 5.18  | 13.5            | 8.1    | 6.3                                   | 13.5    |
| Mercury  | 14.1  | 1.1   | 0.03            | 0.03   | 0.04                                  | 0.04    |

#### Table 14-8. Preliminary Risk Evaluation of Ecological COPCs in Surface Soil (0 to 1 foot) at Burn Pit F, Fort Stewart

J = Estimated concentration. -= No data to derive TRV.

RCRA = Resource Conservation and Recovery Act.

TRV = Toxicity Reference Value = (NOAEL × BW/Food ingestion)/BAFeanhworm (see Table 8-1). Boldface indicates detected concentration exceeds TRV (HQ>1).

#### 14.5.6 Uncertainties

The risk to ecological receptors from chromium in surface soil at Burn Pit F is overestimated by the preliminary risk calculations. Chromium was detected in the three surface soil samples at concentrations ranging from 6.3 to 13.5 mg/kg. Chromium was detected in only one surface soil sample at a concentration barely exceeding the NOAEL-based TRV for the robin with a realistic diet (12.44 mg/kg). However, chromium concentrations in surface soil at Burn Pit F are many times less than the LOAEL-based TRV for the robin calculated using published dietary fractions (62.1 mg/kg). Therefore, chromium in surface soil at Burn Pit F is unlikely to pose a risk to ecological receptors.

## 14.6 CONCLUSIONS AND RECOMMENDATIONS FOR BURN PIT F

The following are the conclusions of the Phase II RFI for Burn Pit F:

- In surface soils, arsenic, chromium, and mercury were reported above background concentrations and are considered possible SRCs. Chromium was found in all three surface soil samples and, therefore, extends throughout the Burn Pit F site.
- Arsenic is a potential human health COPC in surface soil at Burn Pit F because it exceeds the risk-based screening value for exposure of a residential receptor. However, the maximum concentration of arsenic is below the average concentration for regional soils, and residential land use is unlikely. Therefore, arsenic is eliminated as a human health COPC.

- Chromium and mercury are not human health COPCs in surface soil because the maximum concentration of each constituent did not exceed its respective screening criteria.
- Chromium is a potential ecological COPC in surface soils because it exceeds the TRV for one of the surrogate species (robin only). Chromium is considered to be site related and was found in all three of the surface soil samples within the source area at concentrations exceeding its TRV. However, the maximum concentration of chromium (13.5 mg/kg) is within the USGS range of background concentrations for soils in the Eastern United States (up to 1000 mg/kg). In addition, chromium concentrations in surface soil are many times less than the LOAEL-based TRV for the robin and are unlikely to pose a risk to ecological receptors. The risk to ecological receptors from chromium in surface soil at Burn Pit F is overestimated by the preliminary risk calculations. As discussed in the uncertainties section (14.5.6), the maximum detected concentration of chromium in surface soil is expected to result in exposures to the surrogate ecological receptors (shrews or robins) that are less than the lowest exposures associated with adverse effects on them (LOAEL-based TRV). Therefore, chromium is eliminated as an ecological COPC for Burn Pit F, and further investigation and/or evaluation is not required.
- In subsurface soil, five metals (barium, chromium, lead, mercury, and silver) were reported at concentrations above their reference background value. Barium and chromium were detected in three of the seven samples. Toluene and methylene chloride were each detected in two subsurface soil samples.
- There are no human health COPCs in subsurface soil because no constituent exceeded its risk-based screening value.
- Chromium and methylene chloride are potential contaminant migration COPCs in subsurface soil because they may leach to groundwater, resulting in groundwater concentrations exceeding their respective MCLs. However, chromium was not detected in groundwater at concentrations above its reference background value. Off-site migration of chromium would be limited due to retardation and the slow movement of groundwater. Similarly, methylene chloride was not detected in groundwater, and its migration would be limited due to retardation and biodegradation as well as the slow movement of groundwater. Therefore, migration of either chromium or methylene chloride is considered unlikely, and there are no contaminant migration COPCs.
- In groundwater, cadmium was the only RCRA metal detected above its reference background value. Cadmium was detected at its highest concentration in well MW-1, which is considered upgradient of the site.
- BTEX compounds (ethylbenzene and total xylenes) were reported in groundwater at MW-4, although they were present at concentrations less than their respective MCLs and less than the concentrations reported during the Phase I RFI, suggesting that they may be attenuating naturally, either through dispersion, volatilization, or biodegradation.
- There are no human health COPCs in groundwater because no constituent exceeded its risk-based screening value or its MCL in groundwater.

• Ethylbenzene is an ecological COPC in groundwater because it exceeds the ESV for surface water and may present a potential risk to amphibians. However, there are no surface water bodies in the vicinity of Burn Pit F, and off-site migration would be limited due to retardation and/or biodegradation and the slow movement of groundwater. Therefore, exposure of ecological receptors to constituents in surface water bodies downgradient from Burn Pit F is not a complete pathway. Therefore, ethylbenzene is eliminated as an ecological COPC for Burn Pit F, and further investigation and/or evaluation of this constituent is not required.

The following are recommendations for Burn Pit F:

- 1. No further action is recommended for Burn Pit F.
- 2. All potential human health COPCs were eliminated for Burn Pit F (Section 14.4), and a BRA is not recommended for the site.
- 3. All potential ecological COPCs were eliminated for Burn Pit F (Section 14.5), and an ERA is not recommended for the site.
- 4. As discussed in Section 14.1, a supplemental groundwater characterization was performed in November 1998 at Burn Pit F (Appendix H). Based upon the results of that supplemental characterization, BTEX contamination is restricted to the immediate area around well MW-4. Concentrations of ethylbenzene and total xylenes are less than their respective MCLs and continue to decline, indicating that they are attenuating naturally. The groundwater flow direction at the Burn Pit F site has been better defined (as indicated on Figure H-2). These results fully support Fort Stewart's request for "No Further Action" for this site.

فاقره ممتا فكففت فاخترا والرومون

and a support of the state of the

# **15.0 CONCLUSIONS AND RECOMMENDATIONS**

#### 15.1 SUMMARY OF FINDINGS

The Phase II RFI presented in this report was conducted to collect additional analytical data for determining the nature and extent of contamination in environmental media in the vicinity of the former Burn Pits (SWMUs 4A through 4F). The data were derived from analyzing samples of surface soils, subsurface soils, and groundwater at each of the six Burn Pit sites. Surface soil samples were analyzed for SVOCs and RCRA metals; subsurface soil and groundwater samples were analyzed for VOCs and RCRA metals.

Results of these analyses indicated that soils and groundwater at the various sites contain both organic and metal contaminants at concentrations greater than their reference background concentrations. The following summarizes the significant findings of the Phase II RFI sampling and analysis:

- SVOCs were found only in a single surface soil sample at low concentrations at Burn Pit E.
- Metals found above background in surface soils include arsenic, barium, cadmium, lead, and mercury in the northern and eastern portions of Burn Pit A; lead and mercury at Burn Pit B; barium, cadmium, lead, and mercury in the center of Burn Pit C; barium and mercury at Burn Pit E; and arsenic, chromium, and mercury in the northern and northeastern portion of Burn Pit F.
- BTEX compounds were found sporadically in subsurface soil at low concentrations at all six Burn Pit sites. A consistent pattern of distribution is evident only at Burn Pit B and Burn Pit C.
- Metals in subsurface soils were generally found sporadically at all six Burn Pit sites at concentrations less than twice background and in no consistent pattern of distribution. Notable presence of metals found in subsurface soils includes lead and mercury in the central and northeastern portion of Burn Pit C, and five metals (predominantly barium, lead, and chromium) throughout the Burn Pit F site.
- In groundwater, the highest levels of BTEX contamination were found in a single well located west (side-gradient) of the Burn Pit F site. Concentrations of BTEX were less than their respective MCLs and less than reported Phase I concentrations, suggesting that natural attenuation may be occurring. The source of the BTEX contamination is unknown; BTEX compounds were not found at high concentrations in soils at Burn Pit F.
- Metals in groundwater were found at all six sites at concentrations less than their respective MCLs and much less than reported Phase I concentrations, suggesting that redevelopment of the wells and use of low-flow sampling equipment to reduce turbidity in the samples were successful.
- Metals found in groundwater include barium, mercury, and silver at Burn Pit A; barium, cadmium, chromium, selenium, and silver at Burn Pit B; barium, cadmium, and mercury at Burn Pit C; arsenic, barium, and silver at Burn Pit D; and cadmium at Burn Pits E and F.

## **15.2 CONCLUSIONS**

Several assessments were conducted to determine the significance of the contaminant concentrations found at the six Burn Pit sites with respect to their impact on human health and the environment. The assessments included the following:

- A contaminant fate and transport analysis (Section 6.0), which provided an assessment of the potential migration pathways and transport mechanisms affecting the chemicals at the sites.
- An HHRA (Section 7.0), which conducted a Step 1 risk evaluation to determine potential human health risks associated with the contaminants.
- An ERA (Section 8.0), which provided a Phase 1 PRE for potential terrestrial and aquatic receptors at the sites.

The following summarizes the conclusions regarding contaminant fate and transport:

- Among the metal constituents, only chromium from Burn Pit F is considered a contaminant migration COPC based on contaminant transport analysis for leaching from soil to groundwater. At Burn Pit F, chromium could reach the water table at a predicted concentration exceeding its MCL. However, off-site migration of chromium would be very limited due to its high K<sub>d</sub> (i.e., high retardation factor) as well as the slow movement of groundwater toward Goshen Swamp, located approximately 1000 feet west of the site.
- Benzene, methylene chloride, and PCE could migrate from soils to the water table at concentrations exceeding their respective MCLs. However, off-site migration of these constituents would be very limited due to retardation and biodegradation as well as the slow movement of groundwater. Predicted maximum concentrations of these constituents at a distance of 300 feet from the source are not expected to exceed their MCLs even with the most conservative assumption for their biodegradation half-lives. None of the organic compounds that is currently observed in groundwater exceeds its respective MCL in groundwater.

The following summarizes the conclusions of the HHRA:

- In surface soil, arsenic is a potential COPC at Burn Pits A and F, where arsenic exceeds its risk-based screening value for exposure of a residential receptor. It should be noted that the maximum concentration of arsenic at these sites (2.7 µg/kg) is below the average concentration of arsenic for soils regionally (7.4 µg/kg). In addition, a residential screening value was used, even though residential land use at these sites is unlikely. Therefore, arsenic in surface soil is not considered a potential threat to human health at the Burn Pits.
- In subsurface soil, no constituent exceeded its respective risk-based screening value for exposure to an industrial receptor.
- In groundwater, arsenic is a potential COPC only at Burn Pit D, where the maximum concentration of arsenic in groundwater exceeded its risk-based screening value for residential use of groundwater as drinking water. However, the maximum concentration of arsenic (4.2 µg/L) was well below its MCL of 50 µg/L, and was only slightly above its reference background concentration of 3.4 µg/L. Arsenic exceeded background in only a single well at Burn Pit D (MW-6). In addition, use of the surficial groundwater at this site for

drinking water is unlikely. Therefore, arsenic in groundwater is not considered a potential threat to human health at the Burn Pits.

- Similarly, cadmium is a potential COPC at Burn Pits C and E, where the maximum concentrations of cadmium in groundwater exceeded its risk-based screening value for residential use of groundwater. The maximum concentrations of cadmium were less than its MCL of 5 µg/L at each site. In addition, use of the surficial groundwater at the sites for drinking water is unlikely. Therefore, cadmium is not considered a potential threat to human health at the Burn Pits.
- In conclusion, there are no constituents in either surface soil, subsurface soil, or groundwater that are considered a potential threat to human health at the Burn Pits. A human health BRA is not warranted. No further action is required for protection of human health.

The following summarizes the conclusions of the ERA:

- According to GEPD and EPA Region 4 guidance, groundwater was screened as surface water in the ecological PRE. Barium, cadmium, lead, mercury, silver, and ethylbenzene are present in groundwater at one or more of the former Burn Pits at concentrations that exceed EPA Region 4 ESVs for surface water.
- Groundwater is not evaluated further in the PRE because ecological receptors are not likely to be exposed to groundwater at the Burn Pits. Treating groundwater as surface water is unrealistic because there are no surface water bodies in the vicinity of the Burn Pit sites. As concluded in the fate and transport evaluation (Section 6.0), off-site migration of these contaminants would be limited due to retardation or biodegradation as well as the slow movement of groundwater. Therefore, migration of contaminants to surface waters via groundwater discharge is not a complete exposure pathway for ecological receptors.
- Cadmium, chromium, and lead are present in surface soil at one or more of the former Burn Pits at concentrations that exceed the TRVs for the robin and, in some cases, the shrew. Cadmium, chromium, and lead in surface soil at these two sites are, therefore, identified as potential ecological COPCs in the PRE. Based on the results of additional evaluations using published dietary fractions and LOAEL-based TRVs and the uncertainties in the preliminary risk calculations, cadmium, chromium, and lead in surface soil at the Burn Pits are unlikely to pose a risk to ecological receptors.

#### **15.3 SUPPLEMENTAL PHASE II GROUNDWATER CHARACTERIZATION**

**VOCs.** Two BTEX compounds (ethylbenzene at 289  $\mu$ g/L and total xylenes at 1420  $\mu$ g/L) were detected in groundwater at MW-4 during the supplemental groundwater sampling at Burn Pit F conducted in November 1998 (Appendix H). The concentrations are less than the respective MCLs for ethylbenzene (700  $\mu$ g/L) and total xylenes (10,000  $\mu$ g/L). In Phase I sample results, both ethylbenzene and total xylenes were reported at much higher concentrations (2,800 and 15,800  $\mu$ g/L, respectively). In the Phase II RFI, both ethylbenzene and total xylenes were also reported at higher concentrations of 637 and 3170  $\mu$ g/L, respectively. These results indicate that the concentrations of BTEX present in groundwater at the site are attenuating naturally, either through dispersion, volatilization, or biodegradation.

Low concentrations of total xylenes (3.8  $\mu$ g/L) were also found in MW-1. Total xylenes were found at a concentration only slightly above their detection limit and significantly less than their MCL of 10,000  $\mu$ g/L. No VOCs were reported above the detection limit in MW-1 during the Phase II RFI groundwater sampling.

Chloroform and styrene (2.8  $\mu$ g/L and 12.2  $\mu$ g/L, respectively) were observed in MW-4. Styrene is considered a secondary contaminate in the primary BTEX plume. Chloroform is a common laboratory contaminant, was not detected in the Phase II RFI sampling event, and is substantially less than its MCL of 100  $\mu$ g/L.

The following conclusions and recommendations have been made based on the results of the Phase II RFI and the supplemental groundwater investigation (Appendix H):

- 1. The exact source of the BTEX contamination at Burn Pit F is unknown. MW-4 is located hydraulically higher than the other wells at the Burn Pit F site and is considered side-gradient to the existing groundwater flow. BTEX compounds were not detected at significant concentrations in the soil borings within the potable source area, were not detected in the remaining downgradient wells, and were not detected in the eight additional piezometers installed around the perimeter of MW-4. Therefore, the source and extent of contamination is expected to be localized and confined to the immediate vicinity of MW-4.
- 2. BTEX compounds (ethylbenzene and total xylenes) continue to be present in groundwater at MW-4. However, their concentrations are less than their respective MCLs and less than the concentrations reported during the Phase I and II RFIs, indicating they are attenuating naturally, either through dispersion, volatilization, or biodegradation.
- 3. No further investigation or action is recommended for Burn Pit F. The existing wells should be retained in the event that future site monitoring is required to confirm that concentrations of ethylbenzene and total xylenes continue to decrease.

### **15.4 RECOMMENDATIONS**

The results of the RFI and supplemental groundwater sampling, conclusions regarding nature and extent of contamination, fate and transport, human health risk, and ecological risk, result in the following recommendations:

- 1. No further action is recommended for all six Burn Pits (i.e., SWMUs 4A, 4B, 4C, 4D, 4E, and 4F). If approved by GEPD, Fort Stewart respectfully requests that the Installation's Subpart B permit be amended to reflect this change in investigative status.
- 2. Due to the fact that all human health COPCs have been eliminated in site-specific evaluations, none of the Burn Pits will require a BRA.
- 3. Due to the fact that all ecological COPCs have been eliminated in site-specific evaluations, none of the Burn Pits will require an ERA.

4. If GEPD approves Fort Stewart's recommendation for "No Further Action" at SWMUs 4A, 4B, 4C, 4D, 4E, and 4F, the existing site monitoring wells will be properly abandoned, with the exception of the wells located at SWMU 4F. Fort Stewart proposes to retain the wells at SWMU 4F in the event that future site monitoring is required to confirm that concentrations of ethylbenzene and total xylenes continue to decrease.

N. Aleron

.....

# **16. REFERENCES**

- American Society for Testing and Materials. 1995. <u>Standard Guide for Risk-Based Corrective</u> <u>Action Applied at Petroleum Release Sites. ASTM Designation E:1739-95</u>, Annual Book of ASTM Standards, West Conshohocken, Pa.
- Arora, R. 1984. <u>Hydrologic Evaluation for Underground Injection Control in the Coastal Plain of Georgia</u>, Department of Natural Resources, Environmental Protection Division, Georgia Geological Survey.
- Baes, C. F., R. D. Sharp, A. L. Sjoreen and R. W. Shore. 1986. A Review and Analysis of Parameters for Assessing Transport of Environmentally Released radionuclides through Agriculture. ORNL-5786. Martin Marietta Energy Systems, Oak Ridge, Tenn.
- Clayton, G. D., and F. E. Clayton (eds). 1981. <u>Patty's Industrial Hygiene and Toxicology</u>, <u>Vol. 2A: Toxicology, 3rd Revised Edition</u>. John Wiley & Sons, New York.
- Eisler (U.S. Fish and Wildlife Service). 1987. <u>Mercury Hazards to Fish, Wildlife, and</u> <u>Invertebrates: A Synoptic Review</u>. Biological Report 85(1.10), Fish and Wildlife Service, U.S. Department of the Interior, Patuxent, Md.
- Eisler. 1988. <u>Lead Hazards to Fish, Wildlife, and Invertebrates: A Synoptic Review</u>. Biological Report 85(1.14), Fish and wildlife Service, U.S. Department of the Interior, Patuxent, Md.
- Environmental Science and Engineering. 1982. Fort Stewart Military Reservation RCRA Studies: Final Engineering Report.
- EPA (U.S. Environmental Protection Agency). 1987. <u>Hazardous Waste Treatment, Storage, and</u> <u>Disposal Facilities (TSDF) - Air Emission Models</u>, Office of Air and Radiation, Research Triangle Park, N.C.
- EPA. 1991. <u>Risk Assessment Guidance for Superfund, Volume I: Human Health Evaluation</u> <u>Manual, Part B, Development of Risk-Based Preliminary Remediation Goals, Interim,</u> Publication 9285.7-01B, EPA, Office of Emergency and Remedial Response, Washington, D.C.
- EPA. 1991. Maximum Contaminant Level Goals and National Primary Drinking Water Regulations for Lead and Copper, Final Rule 40 CFR Parts 141 and 142, <u>Federal Register</u>, Vol. 56, No. 110, pp. 26460-26564.
- EPA. 1991. Soil Transport and Fate (STF) Database and Model Management System, Version 2.0, R. S. Kerr Environmental Research Laboratory, Ada, Okla.
- EPA. 1993. <u>Risk Reduction Engineering Laboratory (RREL) Treatability Database</u>, Version 4.0, Office of Research and Development, Cincinnati, Ohio.
- EPA. 1993a. <u>Guidelines for Deriving Site-specific Sediment Quality Criteria for the Protection of</u> <u>Benthic Organisms</u>, EPA-822-R-93-017.

- EPA. 1993b. Wildlife Exposure Factors Handbook, Vol. 1 EPA/600/R-93/187a.
- EPA. 1994. <u>Technical Background Document for Soils Screening Guidance</u>, EPA/540/R-94/106, NTIS PB95-963532.
- EPA 1994. <u>The Hydrologic Evaluation of Landfill Performance (HELP) Model</u>, Version 3, EPA/600/R-94/168b, Office of Research and Development, Washington, D.C.
- EPA. 1996a. <u>Supplemental Guidance to RAGS, Region IV Bulletins</u>, Ecological Risk Assessment, Nos. 1-5, USEPA Region 4, Office of Health Assessment, October 1996 (Draft).
- EPA 1996. Soil Screening Guidance: Technical Background Document, EPA/540/R-95/128, Office of Solid Waste and Emergency Response, May.
- EPA. 1996b. Proposed Guidelines for Ecological Risk Assessment, EPA/630/R-95/002B, Washington, D.C.
- EPA. 1997. Ecological Risk Assessment Guidance for Superfund, Process for Designing and Conducting Ecological Risk Assessments, Environmental Response Team, Edison, N.J., Interim Final, June 1997.
- Georgia Environmental Protection Division (GEPD). 1996. <u>Guidance for Selecting Media</u> <u>Remediation Levels at RCRA Solid Waste Management Units</u>, Georgia Environmental Protection Division, Atlanta, Ga., November.
- GEPD. 1997. Personal communication from Rod Stafford, GEPD, Department of Natural Resources, Atlanta, Ga., October 6.
- Geraghty and Miller. 1992. <u>RCRA Facility Investigation Final Work Plan, Fort Stewart, Georgia</u>, June.

Krause and Randolph. 1989.

- HAZWRAP (Hazardous Waste Remedial Actions Program). 1994. Loring Air Force Base Ecological Risk Methodology, Martin Marietta Energy Systems, Inc. Draft.
- Howard, P. H., R. S. Boethling, W. F. Jarvis, W. M. Meylan, and E. M. Michalenko. 1991. Environmental Degradation Rates, Lewis Publishers, Inc., Chelsea, Mich.
- Looper, E. E. 1982. Soil Survey of Liberty and Long Counties, Georgia, U.S. Department of Agriculture, Soil Conservation.
- Metcalf and Eddy. 1996. Final Work Plan for RCRA Facility Investigation at Bulk Fuel Storage System, Wright Army Airfield, Fort Stewart, Georgia.
- Miller, J. A. 1990. <u>Groundwater Atlas of the United States, Segment 6</u>, U.S. Department of the Interior, U.S. Geological Survey, Hydrologic Inventory Atlas 730G.

- Mills, W. B., D. B. Porcella, M. J. Ungs, S. A. Gherini, K. V. Summers, G. L. Rupp, and G. L. Bowie. 1985. <u>Water Quality Assessment: A Screening Procedure for Toxic and Conventional Pollutants: Parts 1,2, and 3</u>, EPA/600/6-85/002, EPA Environmental Research Laboratory, Office of Research and Development, Athens, Ga.
- Perry, R. H., Chilton, C. H., and Kirkpatrick, S. D. 1969. <u>Chemical Engineers Handbook</u>, McGraw Hill Book Company, New York.
- Rust (Rust Environmental and Infrastructure, Inc.). 1996. <u>Phase I RCRA Facility Investigation</u> <u>Report for 24 Solid Waste Management Units at Fort Stewart, Georgia, Vols. I-III, May.</u>
- Sample, B. E., Opresko, D. M., and G. W. Suter II. 1996. <u>Toxicological Benchmarks for</u> <u>Wildlife: 1996 Revision, ES/ER/TM-86/R3.</u>
- SAIC (Science Applications International Corporation). 1997. <u>Sampling and Analysis Plan for</u> <u>the Phase II RCRA Facility Investigation at the Former 724th Tanker Purging Station.</u> <u>SWMU 26. Fort Stewart, Georgia</u>, May.
- Shen, T. T., Schmidt, C. E., and Card, T. R. 1993. <u>Assessment and Control of VOC Emissions</u> from Waste Treatment and Disposal Facilities, Van Nostrand Reinhold, New York.
- Shuman, L. M. 1991. "Chemical forms of micronutrients in soils," in J. J. Montvedt (ed.), <u>Micronutrients in Agriculture</u>, Soil Sci. Soc. Am. Book Series No. 4, Soil Sci. Soc. Amer., Inc., Madison, Wis.
- Suter, G. W. II, and C. L. Tsao. 1996. <u>Toxicological Benchmarks for Screening Potential</u> <u>Contaminants of Concern for Effects on Aquatic Biota: 1996 Revision</u>. ES/ER/TM-96/R2. Lockheed Martin Energy Systems, Oak Ridge National Laboratory, Oak Ridge, Tenn.
- Tracer Research Corporation. 1994. Fort Stewart, Savannah, Georgia, 1993 Annual Sampling Event of 135 Underground Storage Tanks. January.
- U.S. Department of Commerce, 1990. <u>1990 Census</u>, Bureau of the Census.
- USGS (U.S. Department of Interior, Geological Survey). 1984. <u>Element Concentrations in Soils</u> and Other Surficial Materials of the Conterminous United States. Open-File Report 81-197.